MIGRATION OF

OLFACTORY ENSHEATHING CELLS GRAFTED INTO ADULT RAT SPINAL CORD

A Thesis Submitted to the College of

Graduate Studies and Research

In Partial Fulfillment of the Requirements

for the Degree of Doctor of Philosophy

in the Department of Anatomy and Cell Biology

University of Saskatchewan

Saskatoon

By

Viktor Mykolayovyuch Skihar

© Copyright Viktor Mykolayovyuch Skihar, November 2004. All rights reserved.

PERMISSION TO USE

In presenting this thesis in partial fulfillment of the requirements for a Doctor of Philosophy degree from the University of Saskatchewan, I agree that the Libraries of this university may make it freely available for inspection. I further agree that permission for copying of this thesis in any manner, in whole or in part, for scholarly purposes may be granted by the professor who supervised my thesis work or, in his absence, by the Head of the Department or the Dean of the College in which my thesis work was done. It is understood that any copying or publication or use of this thesis or parts thereof for financial gain shall not be allowed without my written permission. It is also understood that due recognition shall be given to me and the University of Saskatchewan in any scholarly use, which may be made of any materials in my thesis.

Requests for permission to copy or to make other use of material in this thesis in whole or in part should be addressed to:

Head of the Department of Anatomy and Cell Biology 107 Wiggins Road University of Saskatchewan Saskatoon, SK, S7N 5E5 CANADA

ABSTRACT

Olfactory ensheathing cells (OECs) are non-myelinating glial cells that provide ensheathment for axons of the olfactory nerve *in vivo*. OECs have been shown to facilitate the regeneration of CNS axons, to assemble a myelin sheath around demyelinated axons, and it has been suggested OECs migrate very well within the microenvironment of the injured CNS. However, there has been no direct test of their migratory ability *in vivo*. The aims of this study were to determine whether: 1) OECs can be induced to migrate towards an ethidium bromide (EtBr)-induced focal (~1 mm long) demyelination of the spinal cord white matter; 2) OECs migrate away from a focal demyelination either into normal CNS tissue or towards a second demyelinated lesion; 3) microglial reactivity is required for the generation of the migratory signal(s) inducing OECs to migrate towards a focal demyelination; 4) OECs grafted into the subarachnoid space surrounding the spinal cord will migrate into the neuropil in the absence of demyelination.

To achieve these aims, we developed an *in vivo* model for studying the migratory ability of OECs within the adult rat spinal cord. A small focal EtBr-induced demyelination of the dorsal funiculus (unilaterally) of the spinal cord was made at variable distances from the site of a DiI-labelled OEC graft. The major findings were: i) the strength of the migratory signal(s) inducing OECs to migrate increased as the demyelinated lesion was located closer to the grafting site; ii) the OEC migration towards a distal demyelinated lesion was greatly enhanced when the cells were grafted directly into a second demyelinated lesion; iii) the cell migration occurred along a migratory path containing many reactive astrocytes and microglia; iv) the migration of OECs was significantly reduced when the microglial reactivity was dampened using minocycline; and v) OECs survived grafting into cerebrospinal fluid (i.e. subarachnoid space) and migrated into the neuropil of the brain and spinal cord. The major conclusions are that OECs can respond to migratory signal(s) arising as a result of a focal EtBr-induced demyelination and that microglia are one potential source of these migratory signal(s).

ii

ACKNOWLEDGEMENTS

First of all, I would like to thank Dr. Ronald Doucette for his courage accepting me as his graduate student and giving me a chance in my life, which I thought I lost back in Ukraine, to do what I had dreamed before coming to Canada. I really appreciate his help, financial support for my research and for my living expenses. It has been an honor to work with such a prominent neuroscientist as Dr. Doucette.

I would like to thank the members of my graduate committee, Drs. Patrick Krone, Guilian Muir, Andrew Gloster, Richard Devon and David Schreyer for finding time in their busy schedules to advise me during my research project.

I would like to thank Dr. Jeffery Kocsis for coming all the way to Saskatoon for my thesis defense. I am grateful to have such a knowledgeable scientist evaluate one of my achievements, the completion of my Ph.D. thesis, as one step in my becoming a neuroscientist.

I would like to thank Cindy Farrar, Anita Givens-Amilio and Darren Nesbitt for teaching me various research techniques that I used to conduct the experiments in my research, for their assistance and help in facilitating my work and for their being with me in the time of sadness and success. In addition, thanks to Karen Yuen and Tonya McGowan for sharing their knowledge in the areas of their expertise.

I would like to acknowledge my relative, Metro Dmetrichuk, who lent me his financial support in order to improve my English before starting my graduate studies.

I would like show my appreciation to Ronald and Daphne Doucette, Shirley and George West, Irene Partridge, Karen and Chris Yuen, Darren Nesbitt, Dixie Swanson, and Tonya McGowan for their contribution and financial support to myself and the life of my family in Ukraine by making me feel like I was a part of their family.

DEDICATION

To my parents,

Hlykeria Mykolaivna Skihar and Mykola Kostyantynovych Skihar.

To all the patients who battle multiple sclerosis.

TABLE OF CONTENTS

PERMISSION TO USE i			i
ABSTRACT			ii
ACKN	ACKNOWLEDGEMENT		
DEDI	CATIO	N	iv
TABL	E OF C	CONTENTS	V
LIST	OF TAE	BLES	Х
LIST	LIST OF FIGURES		
LIST	OF ABI	BREVIATIONS	xiii
1.	INTR	ODUCTION	1
1.1	The P	rimary Olfactory Pathway	1
	1.1.1	The Olfactory Mucosa	1
	1.1.2	The Olfactory Nerves	4
	1.1.3	The Olfactory Bulb	5
	1.1.4	Embryonic and Fetal Development	6
	1.1.5	Neuronal Regeneration	8
1.2	Olfact	ory Ensheathing Cells	9
	1.2.1	Ensheathment of Olfactory Axons	12
	1.2.2	Embryonic and Fetal Development	13
	1.2.3	PNS-CNS Transitional Zone	14
	1.2.4	Tissue Culture Protocols	15
	1.2.5	Therapeutic Potential	18
1.3	Myelin	nating Glia	20
	1.3.1	Myelin	20
	1.3.2	Oligodendrocytes	26
	1.3.3	Schwann Cells	28
	1.3.4	Olfactory Ensheathing Cells	30

1.4	Multi	ple Sclerosis	34
	1.4.1	Neuropathology and Clinical Course	34
	1.4.2	Theories on Etiology of MS	37
	1.4.3	Host Glial Cell Response	39
	1.4.4	Current Treatment	40
1.5	Glial	Cell Reaction to Injury	42
	1.5.1	Astrocytes	42
	1.5.2	Microglia	44
	1.5.3	Inflammation	46
	1.5.4	Degeneration	48
1.6	Anima	al Models for Studying Demyelination and Remyelination	49
	1.6.1	Ethidium Bromide	49
	1.6.2	Lysolecithin	50
	1.6.3	Cuprizone	50
	1.6.4	Experimental Autoimmune Encephalomyelitis	51
	1.6.5	Theiler's Virus	53
1.7	Cell T	Transplantation: A Potential Therapeutic Approach to Treating	
	Multi	ple Sclerosis	54
1.8	Migra	tion of Myelinating Glia Within Central Nervous System	60
	1.8.1	Oligodendrocyte Migration	61
	1.8.2	Schwann Cell Migration	62
	1.8.3	Olfactory Ensheathing Cell Migration	63
2.	MAT	ERIALS AND METHODS	65
2.1	Olfact	tory Ensheathing Cell Cultures	65
	2.1.1	Dissection of Embryos	65
	2.1.2	Disaggregation of Tissue and Setting Up Cell Culture	65
	2.1.3	Purification and Expansion of Cell Culture	66
	2.1.4	Harvesting Cells and Long-Term Storage of Cells	67
	2.1.5	Prelabelling Cells with Dil	67

68 69
69
70
71
71
72
73
73
74
74
74
75
76
76
78
78
79
80
80
80
81
81
82
83
84
85
87

4.1	Olfactory Ensheathing Cell Cultures	87
	\mathcal{O}	

4.2	The A	nimal Model For Studying OEC Migration	90
	4.2.1	The Ethidium Bromide Lesion	90
	4.2.2	The Cell Graft	93
	4.2.3	Using Dil to Identify Grafted Cells	96
4.3	OEC	Migration in the Absence of Demyelination	96
	4.3.1	The Cell Graft	100
	4.3.2	Host Glial Cell Reactivity	100
	4.3.3	DiI+ve Cells Rostral to Cell Graft	100
4.4	Induci	ing OECs to Migrate Towards a Focal Demyelination	103
	4.4.1	The Cell Graft	103
	4.4.2	The Ethidium Bromide Lesion	114
	4.4.3	Host Glial Cell Reactivity	114
	4.4.4	DiI+ve Cells Rostral to Cell Graft	117
4.5	Induci	ing OECs to Migrate Away From a Focal Demyelination	120
	4.5.1	The Ethidium Bromide Lesions	120
	4.5.2	The Cell Graft	120
	4.5.3	Host Glial Cell Reactivity	121
	4.5.4	DiI+ve Cells Rostral to Cell Graft	121
4.6	Micro	glial Reactivity Contributes to the Migratory Signal	137
	4.6.1	Ethidium Bromide Lesion at T10 (Group 6 rats)	137
	4.6.2	Ethidium Bromide Lesions at T10 and T12 (Group 7 rats)	140
4.7	Grafti	ng Retroviral-Infected OECs	145
	4.7.1	Retroviral-Infected OECs in Vitro	145
	4.7.2	OECs Grafts into Dorsal Funiculus	150
	4.7.3	OEC Grafts Onto Pial Surface	153
5.	DISC	USSION	156
5.1	Anima	al Models of CNS Demyelination for Studying the In Vivo	
	Migra	tory Ability of Myelinating Glia	156
5.2	Migra	tion of Myelinating Glial Cells in CNS	159
	5.2.1	Host Myelinating Glia	160

	5.2.2	Donor-Derived Myelinating Glia	163
5.3	Contr	ibution of Reactive Microglial Cells to Migration of	
	Myeli	nating Glia	165
5.4	Relev	ance of Migratory Ability of OECs to Neural Repair	170
5.5	Prelab	belling Cells Prior to Grafting	178
	5.5.1	Fluorescent dyes	180
	5.5.2	Viral Infection	182
5.6	Metho	ods of Delivery of Cells to Injured CNS	184
	5.6.1	Grafting Cells Directly into CNS Tissue	185
	5.6.2	The Vascular Route	186
	5.6.3	Injection of Cells Into CSF	188
5.7	Gener	al Conclusions	190
5.8	Future	e Directions	191

6. **REFERENCES**

LIST OF TABLES:

TABLE 2.1	ABLE 2.1 Cell counts of viable OECs before and after being frozen in	
	liquid nitrogen	77
TABLE 4.1	The experimental groups used to address each Specific Aim	97
TABLE 4.2	Number of β -Gal-expressing OECs found in the T12 and T8	
	spinal cord segments after a pial surface graft	155

LIST OF FIGURES

FIGURE 4.1:	Morphology and Phenotype of OECs in vitro	89
FIGURE 4.2:	A small, focal EtBr-induced demyelinated lesion	92
FIGURE 4.3:	DiI+ve OECs grafted into the dorsal funiculus of T12	95
FIGURE 4.4:	Schematic diagrams summarizing the experimental groups	99
FIGURE 4.5:	Microglial and astrocytic reactivity in the dorsal funiculus	102
	of T11 in Group 1 rats	
FIGURE 4.6:	Number of DiI+ve OECs in the dorsal funiculus of T11 in	105
	Groups 1-3 rats.	
FIGURE 4.7:	Number of DiI+ve OECs in the dorsal funiculus of T10 in	107
	Groups 1-3 rats.	
FIGURE 4.8:	Number of DiI+ve OECs in the dorsal funiculus of T9 in	109
	Groups 1-3 rats.	
FIGURE 4.9:	Number of DiI+ve OECs in the dorsal funiculus of T8 in	111
	Groups 1-3 rats.	
FIGURE 4.10:	DiI+ve cells in the dorsal funiculus of T11 and T9 in	113
	Groups 1-3 rats	
FIGURE 4.11:	Microglial and astrocytic reactivity in the dorsal funiculus	116
	of T11 in Group 2 rats	
FIGURE 4.12:	Microglial and astrocytic reactivity in the dorsal funiculus	119
	of T11 in Group 3 rats	
FIGURE 4.13:	Microglial reactivity in the dorsal funiculus of T11 in Group	123
	4 and 5 rats	
FIGURE 4.14:	Astrocytic reactivity in the dorsal funiculus of T11 in Group	125
	4 and 5 rats	
FIGURE 4.15:	DiI+ve cells in the dorsal funiculus of T11 in Group 4 and 5	127
	rats.	
FIGURE 4.16:	Number of DiI+ve OECs in the dorsal funiculus of T11 in	130
	Groups 1, 4 and 5 rats.	

FIGURE 4.17:	Number of DiI+ve OECs in the dorsal funiculus of T10 in	132
	Groups 1, 4 and 5 rats.	
FIGURE 4.18:	Number of DiI+ve OECs in the dorsal funiculus of T9 in	134
	Groups 1, 4 and 5 rats.	
FIGURE 4.19:	Number of DiI+ve OECs in the dorsal funiculus of T8 in	136
	Groups 1, 4 and 5 rats.	
FIGURE 4.20:	Microglial and astrocytic reactivity in the dorsal funiculus	139
	of T11 in Group 6 rats	
FIGURE 4.21:	DiI+ve OECs in the dorsal funiculus of T11 in Group 6 rats	142
FIGURE 4.22:	Microglial and astrocytic reactivity in the dorsal funiculus	144
	of T11 in Group 7 rats	
FIGURE 4.23:	DiI+ve OECs in the dorsal funiculus of T11 in Group 7 rats	147
FIGURE 4.24:	Schematic description of retroviral infection of OEC	149
	cultures	
FIGURE 4.25:	β -Gal+ve OECs in the spinal cord, brainstem and	152
	cerebellum	

LIST OF ABBREVIATIONS:

AMPA	α -amino-3hydroxy-5-methyl-4-isoxazoleppropionic acid
APP	amyloid precursor protein
ATP	adenosine triphosphate
bFGF	basic fibroblast growth factor
β-Gal	β-Galactosidase
BCS	bovine calf serum
BMCs	Bone marrow cells
BSA	bovine serum albumen
CG4	oligodendrocyte progenitor cell line
CNPase	2',3'-Cyclic nucleotide-3'phosphohydrolase
CNS	central nervous system
CSF	cerebrospinal fluid
Cx32	connexin 32
DAB	diaminobenzidine tetrahydrochloride
DiI	1,1'- dioctadecyl - 3,3,3',3' - tetramethylindocarbocyanine perchlorate
DIV	days in vitro
DMEM	Dulbecco's Modified Eagle's Medium
DMSO	dimethyl sulfoxide
DPBS	Dulbecco's phosphate buffered saline
DREZ	dorsal root entry zone
DRG	dorsal root ganglion
E1	embryonic day 1
E18	18 day embryo
EAE	experimental autoimmune encephalomyelitis
EDTA	ethylenediaminetetraacetic
EG	Enteric glia
EGF	epidermal growth factor
EtBr	ethidium bromide
FBS	fetal bovine serum
FGF	fibroblast growth factor
Gal-C	galactocerebroside
GAP-43	43 Kd growth-associated protein
GFAP	glial fibrillary acidic protein
GFP	green fluorescent protein
H_2O_2	hydrogen peroxide
HBSS	Hank's balanced salt solution
HEPES	N-2-Hydroxyethylpiperazine-Ní-2-ethanesulfonic acid
HLA	human leukocyte antigen
HS	horse serum
Ig	immunoglobulin
IGF	insulin-like growth factor
IL	interleukin
INF	interferon
i.p.	intraperitoneally

MAG	myelin-associated glycoprotein
MAPs	microtubule associated proteins
MBP	myelin basic protein
mG5	modified G5 medium
MHC	major histocompatibility complex
MMPs	matrix metalloproteases
MOG	myelin/oligodendrocyte glycoprotein
MRI	magnetic resonance imaging
MS	multiple sclerosis
MSC80	immortalized Schwann cell line
N-CAM	neural cell adhesion molecule
NGF	nerve growth factor
NSCs	neuronal stem cells
Oct-6 /SCIP/ tst-1	POU domain factor Oct-6
OECs	olfactory ensheathing cells
OMP	olfactory marker protein
ONL	olfactory nerve fiber layer
OX-18	antibody, which recognizes the MHC class 1 antigen
OX-42	antibody, which recognizes the complement type 3 receptor
p75NGFR	low-affinity neurotrophin receptor
PBS	phosphate buffered solution
PBSS	Puck's balanced salt solution
PDGF	platelet derived growth factor
PF	paraformaldehyde
PKH26	lipophilic fluorescent dye
PLP	proteolipid protein
PMP22	peripheral myelin protein 22
PNS	peripheral nervous system
PPG	picric acid/paraformaldehyde/glutaraldehyde
PSA-N-CAM	embryonic polysialic acid containing N-CAM
RV	retrovirus
SMP	skim milk powder
SV 40	simian virus 40
TGF-β	transforming growth factor-β
Th	helper T cells
TMEV	Theiler's murine encephalomyelitis virus
TNF-α	tumor necrosis factor-a
TX-100	Triton X-100
V-CAM	vascular cell adhesion molecule
VLA	very late antigen
	-

1 INTRODUCTION

1.1 The Primary Olfactory Pathway

The primary olfactory pathway consists of the olfactory epithelium, the olfactory nerve, and the olfactory bulb. Primary olfactory neurons are located in the olfactory epithelium, which is the epithelial component of the olfactory mucosa and lines part of the nasal cavity, and project their axons via the first cranial nerve, the olfactory nerve, to the olfactory bulb. In the bulb, the olfactory axons innervate a discrete lamina near the pial surface of the ipsilateral olfactory bulb where they form synaptic connections with neurons of the bulb in specialized areas of neuropil called glomeruli. The olfactory epithelium is the part of the adult mammalian nervous system in which it was first discovered that neurogenesis continued past the early postnatal period of development and into adulthood (Barber, 1982; Calof *et al.*, 1998; Graziadei and Monti Graziadei, 1979). New olfactory receptor neurons continue to be generated throughout the life of mammals, a process that provides new neurons to replace those lost from injury or disease (Graziadei and Monti Graziadei, 1978; Moulton *et al.*, 1970; Simmons *et al.*, 1981).

1.1.1 The olfactory mucosa

The olfactory mucosa lines the superior part of the nasal cavity and consists of two layers, a superficial olfactory epithelium and a deeper lamina propria (Morrisson and Costanzo, 1992). The former is an avascular pseudostratified columnar epithelium containing three principal cell types: olfactory receptor neurons, sustentacular (supporting) cells, and horizontal basal cells. The olfactory epithelium is mainly sensory in function and is specialized for the detection of odorants. The lamina propria contains numerous fascicles of olfactory axons, which are all unmyelinated, multicellular Bowman's glands and blood vessels (Getchell *et al.*, 1984); in addition to these elements being embedded within the loose connective tissue of the lamina propria, there is also an occasional small fascicle of myelinated and unmyelinated axons originating from the trigeminal nerve. Sustentacular cells of the olfactory epithelium and

Bowman's glands, the ducts of which course through the epithelium enroute to the apical surface of the epithelium, are believed to be responsible for producing the layer of mucus that covers the free surface of the olfactory mucosa (Getchell *et al.*, 1984).

Three main neuronal populations have been identified in the olfactory epithelium of adult mammals (Graziadei and Monti Graziadei, 1978, 1979), with each population representing a different stage in the maturation of newly formed olfactory receptor neurons. Globose basal cells comprise the first population and are believed to be the youngest neurons in the olfactory epithelium, with the most immature of these just beginning to emit an axon from the basal surface of the cell (Graziadei and Monti Graziadei, 1978, 1979). These cells are situated near the base of the epithelium, immediately superficial to a single layer of horizontal basal cells. The second neuronal population is made up of immature olfactory receptor neurons that have extended their axons into the lamina propria of the olfactory mucosa, with some of the axons reaching as far as the olfactory bulb. Although the neurons of this second population may respond to odor molecules, their activation does not contribute to the perception of smell because the axons of these neurons have not vet formed synaptic contacts in the olfactory bulb. Neurons of the first two neuronal populations share several phenotypic features with immature neurons from other regions of the developing nervous system, expressing for example vimentin (Schwob et al., 1986), microtubule associated proteins (MAPs) characteristic of immature neurons (i.e. MAP5 and MAP1B; Schoenfeld et al., 1989; Tucker et al., 1989; Viereck et al., 1989), the embryonic form of the neural cell adhesion molecule (N-CAM; Miragall et al., 1988, 1989), and the 43 Kd growth-associated protein that is better known as GAP-43 (Verhagen et al., 1989, 1990). In addition, the second progenitor cell in the continuum of globose basal cell population expresses MASH-1 (Calof et al., 1996, 2002; Schwob, 2002), which is required for commitment to the neuronal lineage (Murray et al., 2003). In MASH-/- mice, most of the progenitors develop along the sustentacular cell pathway instead (Murray et al., 2003). Later, the third distinct subclass of globose basal cells, an immediate neuronal precursor expresses neurogenin-1 (Calof et al., 1996, 2002; Schwob, 2002).

Mature olfactory receptor neurons comprise the third neuronal population and are bipolar cells that possess a fusiform cell body. A single unmyelinated axon arises from the basal pole of each neuron and projects without axon collaterals to the olfactory bulb where it makes synaptic contacts with neurons in the olfactory bulb. Most of the olfactory receptor

neurons are believed to never fully mature because they retain the expression of MAP5, MAP1B (Schoenfeld *et al.*, 1989; Tucker *et al.*, 1989; Viereck *et al.*, 1989), and vimentin (Schwob *et al.*, 1986); they do, however, stop expressing GAP-43 (Verhaagen *et al.*, 1989, 1990) and the embryonic form of N-CAM (Miragall *et al.*, 1988, 1989). Mature olfactory receptor neurons also exclusively express olfactory marker protein (OMP; Margolis, 1985). OMP-deficient mice showed decreased ability to respond to odorants compared with wild type mice, thus OMP was demonstrated to modulate the odor detection and signal transduction cascade in these cells (Buiakova *et al.*, 1996).

A single dendrite arises from the apical pole of the cell body of each olfactory receptor neuron and projects towards the free surface of the olfactory epithelium where it terminates in an apical knob from which extend several cilia. These olfactory receptor neurons express chemoreceptors on the surface of their dendrites, which enable the neurons to detect the presence of volatile stimuli in the surrounding environment (Getchell *et al.*, 1984). When odorants reach the superior aspect of the nasal cavity they must traverse a layer of mucus before they can bind to odorant receptors in the chemosensitive plasma membrane of the olfactory receptor neurons (Getchell *et al.*, 1984). This binding of odorant molecules by the receptors initiates a series of membrane events leading to sensory transduction and initiation of an action potential. Olfactory receptor neurons encode information about the intensity, duration and quality of the odorant and transmit this information to the brain.

At the base of the olfactory epithelium there are horizontal basal cells that lie directly against the basal lamina separating the epithelium from the lamina propria. The horizontal basal cells express the epidermal growth factor (EGF) receptor, keratin and a surface glycoprotein called alpha-N-acetyl galactosamine that binds I β -4 isolectin from Banderaea simplicifolia-1 (Holbrook *et al.*, 1995). These cells are attached by desmosomes to sustentacular cells and globose basal cells, and by hemidesmosomes to the basal lamina. The horizontal basal cells enfold virtually all bundles of olfactory axons within tunnels formed when horizontal basal cells arch over the basement membrane (DeLorenzo, 1957; Frisch, 1967; Graziadei and Monti Graziadei, 1978, 1979; Holbrook *et al.*, 1995). The horizontal basal cells are suggested to be the stem cell for olfactory neurogenesis (Calof and Chikaraishi, 1989; Satoh and Takeuchi, 1995).

The sustentacular cells extend from the basement membrane to the epithelial surface and are described by several authors (Getchell et al., 1984; Rafols and Getchell, 1983) as modified ependymal cells. Rafols and Getchell (1983) characterized these cells in the olfactory mucosa of the salamander as two morphological types with type I resembling ependymal tanycytes and type II resembling protoplasmic astrocytes. The apical portions of the sustentacular cells are bound together with adjacent sustentacular cells or with the olfactory receptor neuron dendrites by tight junctions at the mucosal surface, where each dendrite is surrounded by sustentacular cells and thus is isolated from adjacent dendrites (DeLorenzo, 1957; Frish, 1967; Graziadei and Monti Graziadei, 1979). At the basement membrane, the basal portion of sustentacular cells lies very close to capillaries and to Bowman's glands that lie immediately on the lamina propria side of the basement membrane. Sustentacular cells are believed to (1) isolate olfactory receptor neurons from one another for electrical purposes (Graziadei and Metcalf, 1971); (2) secrete mucopolysaccharides from the apical pole (Getchell et al., 1984); (3) transport molecules from the lamina propria through the olfactory epithelium to its free surface and vice versa (Rafols and Getchell, 1983), and (4) guide the migration of postmitotic neurons towards the apical surface of the epithelium (Simmons et al., 1981; Rafols and Getchell, 1983).

1.1.2 The olfactory nerves

The axons, arising from the basal pole of olfactory receptor neurons, course through the olfactory epithelium en route to the lamina propria where they join with other axons to form the peripheral olfactory fascicles (fila olfactoria) (Graziadei and Monti Graziadei, 1978). In the lamina propria, the unmyelinated olfactory axons are ensheathed by glial cells referred to as olfactory ensheathing cells (OECs; see Section 1.2). The olfactory fascicles eventually coalesce in the submucosa to form the olfactory nerve bundles that penetrate the cribriform plate of the ethmoid bone, enter the cranial cavity, travel within the nerve fiber layer (i.e. the most superficial layer; ONL) of the olfactory bulb, and synapse in the olfactory glomerulus. Newly formed olfactory bulb (Costanzo, 1985; Oley *et al.*, 1975; Simmons and Getchell, 1981). Connective tissue sheaths surround each olfactory fascicle within the lamina propria. Several of these olfactory fascicles and their connective tissue sheaths fuse together to form much larger fascicles, which in turn enter and usually almost completely fill each

foramina of the cribriform plate (Doucette, 1991). The olfactory fascicles then travel within the subarachnoid space bridging the gap between the foramina of the cribriform plate and the ONL of the olfactory bulb. The connective tissue sheaths of the intracranial portion of these fascicles fuse with both the dural and leptomeningeal linings of the cranial cavity (Doucette, 1991).

Olfactory axons are small, being only 0.1 to 0.5 µm in cross-sectional diameter, and never branch until they approach their target in the olfactory bulb (Graziadei and Monti Graziadei, 1978). Numerous axons are grouped together in one common mesaxon formed by the OEC (DeLorenzo, 1957). OECs are the only glial cells present in the peripheral olfactory fascicles (DeLorenzo, 1957; Doucette, 1990; Field *et al.*, 2003). The cell membrane of an OEC forms one mesaxon for a large number of axons rather than for each individual axon and separates fascicles from each other by OEC cytoplasm (DeLorenzo, 1957). The OECs of each peripheral olfactory fascicle are enclosed within a common basal lamina, which surrounds the entire fascicle (DeLorenzo, 1957; Frisch, 1967). Adjacent fascicles are separated from one another by fibroblasts and collagen (Doucette, 1993). In contrast, OECs within the ONL of the olfactory bulb are apposed to a basal lamina only where these cells contribute to the formation of the glia limitans (Barber and Lindsay, 1982; Doucette, 1984, 1986). Astrocytes, but not fibroblasts and collagen fibers, occupy the spaces between adjacent olfactory fascicles in the central nervous system (CNS; Doucette, 1984, 1986).

1.1.3 The olfactory bulb

The axons of olfactory receptor neurons enter the olfactory bulb through its ventral surface, which lies on the intracranial side of the cribriform plate immediately ventral to the surface of the frontal lobe. The olfactory axons grow through this part of the CNS to synapse with their targets (Barber and Linsday, 1982; Graziadei and Monti Graziadei, 1978). The olfactory bulb is a bilateral oblong structure attached to the rest of the brain by means of the olfactory tract. The olfactory bulb has a laminated structure and arises embryologically as an evagination of the rostral wall of the cerebral vesicle.

Proceeding from the pial surface inward, the laminar pattern of the olfactory bulb comprises the following layers: a) ONL, b) glomerular layer, c) external plexiform layer, d) mitral cell layer, e) internal plexiform layer, and f) granule cell layer. The ONL forms a superficial lamina, which extends around the entire outer circumference of the bulb and contains glial cells and fascicles of densely packed unmyelinated olfactory axons (Barber and Linsday, 1982; Doucette, 1984, 1986, 1990). Astrocytes and OECs are the glial cells of the ONL. The synaptic terminals of the axons cluster in spherical structures know as glomeruli, and each glomerulus receives input primarily from different olfactory receptor neurons. In the glomerulus, olfactory neuron axons synapse on dendrites of the mitral, tufted and periglomerular cells. Periglomerular cells are interneurons, the axons of which synapse on the dendrites of mitral and tufted cells as well as on the cell bodies and dendrites of other periglomerular cells (Pinching and Powell, 1971). Thus, the periglomerular cells influence the transmission of the olfactory information between neighboring glomeruli. The largest (20 - 30)um in diameter) neurons in the olfactory bulb are the mitral cells, which are arranged in a layer of one to two cell bodies thick and which are separated from the glomerular layer by their own dendrites forming the external plexiform layer. Tufted cells, also known as displaced mitral cells, can be found sparsely scattered throughout the external plexiform layer and even in parts of the periglomerular region. Dendrites and axons of the tufted cells make connections similar to those of mitral cells, respectively. However, collaterals of tufted cell axons ramify around several adjacent glomeruli, where they terminate on the cell bodies and dendrites of periglomerular cells (Pinching and Powell, 1971). The axons of mitral and tufted cells leave the olfactory bulb through the granule cell layer to enter the olfactory tract (Haberly and Price, 1977) and innervate the anterior olfactory nucleus, olfactory tubercle, amygdaloid nucleus, piriform cortex (White, 1965, Heimer, 1968; Price and Powell, 1971) and entorhinal cortex (White, 1965, Heimer, 1968). The granule cell layer forms the innermost lamina of the olfactory bulb. Granule cells extend their dendrites into the external plexiform layer where they make reciprocal dendro-dendritic contacts with mitral and tufted cells.

1.1.4 Embryonic and fetal development

The development of the olfactory system has been extensively studied in a variety of animals including mice (Hinds, 1972), frog (Byrd and Burd, 1991), ducks (Rebiere and Dainat, 1981), lungfish (Derivot, 1984), and snakes (Holtzman and Halpern, 1990). During the second week of gestation two olfactory placodes, the thick areas of cranial ectoderm situated rostrolateral to the developing cerebral vesicles, invaginate to form the olfactory pits, which will give rise to the olfactory epithelium (Graziadei, 1973; Hinds, 1972). At the base of the epithelium, processes of the epithelial cells penetrate the basal lamina and, surrounded by glial progenitor cells, extend into the mesenchyme adjacent to the olfactory epithelium (Cuschieri

and Bannister, 1975). At the end of the second week of gestation of mouse embryos, OMP can be detected using immunohistological methods in cell bodies near the surface of the olfactory epithelium (Farbman and Margolis, 1980; Graziadei *et al.*, 1980a). Since only mature olfactory receptor neurons express OMP in the adult (Monti-Graziadei *et al.*, 1977), the presence of OMP in cells of developing olfactory epithelium indicates that these cells have formed synaptic contacts and have reached maturity. In the third week of gestation OMP is found in an increased number of cells at the epithelial surface, in the olfactory nerve fascicles of the lamina propria, in the ONL and by the end of the third week in the glomeruli of the bulb (Graziadei *et al.*, 1980a).

During embryogenesis, olfactory receptor neurons influence the development of the olfactory bulb, the initial formation of which from the rostral wall of the cerebral vesicle occurs near the end of the second week of gestation (Hinds, 1972). At this time the number of olfactory axons and mitral/tufted cells in the olfactory bulb increases in developing *Xenopus* larvae (Byrd and Burd, 1991). During the early developmental stages of the mouse embryos, the olfactory axons make their initial contact with the rostral wall of the cerebral vesicle, referred to as the olfactory bulb primordium (Doucette, 1989). Once the olfactory axons and peripheral glial progenitor cells of the olfactory fascicle cover the external exposed surface of the bulb primordium, they form the presumptive ONL. At this time point, which corresponds to Theiler stages 19 and 21 (Theiler, 1972), the cellular constituents of the ONL are separated from those of the mouse bulb primordium by a relatively intact glia limitans. This glia limitans eventually disappears thus leaving no physical barrier between the cellular constituents of the definitive ONL and the marginal zone of the developing olfactory bulb (Doucette, 1989).

Present at the early developmental stages, olfactory axons are found to penetrate the bulb reaching its deep intermediate zone. Here they make contact with neuronal precursor cells of the olfactory bulb and possibly influence their development (Stout and Graziadei, 1980). Later the olfactory axons retract to their final location in the olfactory nerve and glomerular layers. Olfactory afferent innervation is critical for the differentiation and maintenance of the dopamine phenotype of interneurons in the olfactory bulb (Baker and Farbman, 1993). Decreasing the numbers of the olfactory axons by partial deafferentation of the olfactory bulb results in a reduction in the number of mitral neurons in the olfactory bulb of either *Xenopus* embryos (Byrd and Burd, 1993) or the moth, *Manduca sexta* (Hildebrand *et al.*, 1997).

Complete removal of the olfactory placode in amphibians resulted in formation of a small or no olfactory bulb (Stout and Graziadei, 1980). Vasculature plays an important role in neuronal development and responds to alterations in sensory stimulations early in life.

Doucette (1989) described the development of the ONL in the olfactory bulb of mouse embryos, when Marin-Padilla and Amieva (1989) characterized the early neurogenesis of the mouse olfactory nerve. The olfactory axons are identified in the mesenchyme growing dorsocaudally toward the cerebral vesicle. Peripheral glial progenitor cells, which ensheath adjacent axons, accompany them. Olfactory axons and OECs from the same regions of the nasal mucosa establish an early association, which is maintained up to their glomerular terminal. Marin-Padilla and Amieva (1989) suggested that simple anatomical arrangement between olfactory axons and OECs allows establishing a topographic map of the sensory olfactory epithelium in the developing olfactory bulb. Thus, the axons of newly formed olfactory neurons from different nasal regions could be guided by their OEC conduits toward their target glomeruli; hence, the olfactory map may be maintained undisturbed through the animal's life span. Axons, but not the glial progenitors, penetrate into the marginal zone through small breaks in the glia limitans of the bulb primordium. Doucette (1989) suggested that the ONL of the olfactory bulb forms superficial to, rather than within, the marginal zone of the rostral wall of the cerebral vesicle. The peripheral glial progenitor cells contributed to the formation of the ONL and presumably differentiated into OECs. Arising from the peripheral nervous system (PNS), OECs are one of the two glial cell types in the ONL, with astrocytes being the second cell type and having a CNS origin. However, by Theiler stage 24 (embryonic day 18) of developing rat embryos and stage 16 of mouse embryos the OECs are the only glial cells present in the ONL (Doucette, 1989). Therefore, OECs are the only glial cells present in the OEC cultures initiated using the ONL of E18 rat olfactory bulbs.

1.1.5 Neuronal regeneration

Generation of new sensory neurons in the olfactory epithelium is an ongoing process, which continuous throughout life (Barber, 1982; Calof *et al.*, 2002; Graziadei and Monti Graziadei, 1978, 1979; Schwob, 2002). The neuronal turnover occurs both spontaneously and in response to cellular and tissue trauma (Schwob, 2002). New neurons arise as a result of the proliferation and differentiation of stem/progenitor cells located near the base of the epithelium (Graziadei and Monti Graziadei, 1978, 1978, 1983; Graziadei *et al.*, 1980b). Neurogenesis in the

olfactory epithelium is in a state of dynamic equilibrium between positive and negative regulators. Indirectly controlling the generation of new neurons, the mature neurons of the olfactory epithelium appear to be a source of negative feedback signals, which regulate the proliferation of the precursor cells. A presumptive stem cell divides asymmetrically approximately every 50 days producing another stem cell, which stays close to the basal lamina, and a neuronal precursor cell. The latter cell further divides rapidly several times producing many immature neurons, each of which migrate away from the basal lamina (Mackay-Sim and Kittel, 1991). Because in the olfactory epithelium neurogenesis is a continuous process, there are neurons at varying stages of differentiation, reflecting different stages of maturation. Therefore, there are olfactory receptor neurons with different morphological (Graziadei and Monti Graziadei, 1978), biochemical (Harding et al., 1977), and physiological (Simmons and Getchell, 1981) characteristics residing within the same part of the olfactory epithelium. Caggiano et al. (1994) and Schwob et al. (1994) used a retroviral vector to label proliferating cells in the olfactory epithelium and obtained data consistent with the neurons arising from horizontal basal cells. An additional in vivo study suggests that a stem cell resides in the olfactory epithelium very close to the basal lamina, which is where the horizontal basal cells are localized (Mackay-Sim and Kittel, 1991). Olfactory receptor neurons can also be generated *in vitro* with the evidence suggesting they may arise from horizontal basal cells (Mahanthappa and Schwarting, 1993; Satoh and Takeuchi, 1995). However, the actual identity of the stem and progenitor cells in the olfactory epithelium is yet to be resolved.

1.2 Olfactory Ensheathing Cells

OECs are the non-myelinating glial cells that ensheath olfactory axons within both the PNS and CNS portions of the primary olfactory pathway (DeLorenzo, 1957; Doucette, 1990; Field *et al.*, 2003; Frisch, 1967). OECs are located in a region of the mammalian nervous system where axonal growth occurs throughout the lifetime of the organism. They are distributed within the lamina propria of the olfactory mucosa, the olfactory nerve, and the first two layers of the olfactory bulb (Cancalon, 1985; Doucette, 1990; Gong *et al.*, 1994). Rafols and Getchell (1983) examined the morphological relationship between the receptor neurons and nonneuronal cells, i.e. glial fibrillary acidic protein (GFAP) expressing OECs, which surround the unmyelinated olfactory axons and suggested that these OECs contribute to the guidance of

the olfactory axons to their synaptic targets in the olfactory bulb. Although Rafols and Getchell (1983) described these cells as astrocytes they were later named as OECs (Doucette, 1984).

In both the olfactory nerves and the olfactory bulbs the OECs display a very characteristic morphology (Bailey and Shipley, 1993; Doucette, 1984, 1986, 1990, 1991, 1993). Golgi in 1875 and Blanes in 1898 (as described in Ramon-Cueto and Avila, 1998) were the first to describe the glial population of the olfactory bulb of adult mammals. They observed two types of macroglia in the olfactory bulb, one stellate in morphology and the other fusiform. Stellate cells were distributed throughout the entire olfactory bulb, whereas the fusiform cells were located only in the first two layers. More recent studies have revealed the stellate cells are astrocytes and the fusiform cells are the OECs (Barber and Lindsay, 1982; Doucette, 1984, 1986, 1991, 1993). More recently, Bailey and Shipley (1993) described unipolar and irregular subtypes of astrocytes that were exclusively located in the ONL of the olfactory bulb. These astrocytes separated olfactory nerve fascicles, and had long, unbranched cytoplasmic processes (irregular subtype). The unipolar or unbranched astrocytes were different from most white matter fibrous astrocytes that elongate in a bipolar arrangement along axons in other white matter tracts.

The OECs reside within olfactory fascicles in both the PNS and CNS portions of the primary olfactory pathway, aligning their perikarya along the longitudinal axis of the fascicles (Barber and Lindsay, 1982). On the basis of ultrastructural criteria, OECs have a characteristic indented nucleus with uniformly distributed heterochromatin. Free ribosomes and large inclusion bodies are abundant in the cytoplasm as well. Compared to astrocytes, the cytoplasm of OECs is more electron dense, which may be due to the presence of more water-soluble proteins in the cytoplasm of the latter cells (Doucette, 1984). The intermediate filaments of OECs are fewer in number and are scattered throughout the cytoplasm instead of being grouped into closely packed bundles the way they are in astrocytes (Barber and Lindsay, 1982; Doucette, 1984, 1986, 1990, 1991). At the glia limitans, the plasma membranes of end-feet of OECs lack the orthogonal arrays of intramembranous particles (Barber and Lindsay, 1982; Doucette, 1991, 1993) that are so characteristic of astrocytic end-feet forming the glia limitans in other parts of the CNS (Peters *et al.*, 1976).

OECs are a potential source of growth-promoting and neurite-promoting factors during development and in the postnatal period (reviewed in Ramon-Cueto and Avila, 1998). These

cells express platelet-derived growth factor (Kott et al., 1994), neuropeptide Y (Ubink et al., 1994), and the calcium-binding protein S100 (Cummings and Brunjes, 1995), which is suggested to facilitate the growth and survival of neurons (Bhattacharyya et al., 1992; Van Eldik *et al.*, 1991). A neurite-promoting factor referred to as glia-derived nexin is also synthesized by both the astrocytes of the nerve fiber layer of the bulb (Scotti et al., 1994) and by OECs (Reinhard, 1988; Scotti et al., 1994). The low-affinity neurotrophin receptor (p75NGFR) is also expressed by OECs, with the expression of this receptor being under strong developmental regulation (Barnett et al., 1993; Franceschini and Barnett, 1996; Gong et al., 1994; Miwa et al., 1993; Turner and Perez-Polo, 1993, 1994; Vickland et al., 1991). At all embryonic and fetal stages of development in rodents, p75NGFR expression is associated with the OECs ensheathing olfactory axons. In neonatal rodents, OECs show an intense expression of p75NGFR at the light and electron microscope levels that decreases to almost undetectable levels in the adult. Ramon-Cueto and Nieto-Sampedro (1992) identify four subpopulations of OECs in cultures obtained from adult rat olfactory nerve and glomerular layers of the olfactory bulb. One subpopulation expresses both p75NGFR and fibronectin (5% of the total population), while a second subpopulation is p75NGFR positive and fibronectin negative (26%). The remaining two subpopulations do not express p75NGFR, although one is fibronectin positive (50%) and the other is fibronectin negative (19%).

At all developmental stages, OECs express in their plasma membranes the L1 cell adhesion molecule (Miragall *et al.*, 1988), the embryonic polysialic acid containing N-CAM (PSA-N-CAM), as well as adult forms of N-CAM (Franceschini and Barnett, 1996; Miragall *et al.*, 1988); they also synthesize and secrete laminin into the extracellular matrix (Doucette, 1996; Kafitz and Greer, 1997; Liesi, 1985). All of these molecules are known to be involved in cell adhesion and axonal growth (Bixby *et al.*, 1988). The CAMs presumably mediate olfactory axon elongation on OECs (Barnett *et al.*, 1993; Chuah and Au, 1994; Franceschini and Barnett, 1996; Key and Akeson, 1990; Miragall *et al.*, 1988; Ramón-Cueto and Nieto-Sampedro, 1992) since N-CAM and L1 are only found in that part of the OEC plasma membrane in direct contact with the axons (Miragall *et al.*, 1988). Only those OECs forming part of the glia limitans express PSA-N-CAM, with this molecule being present in the part of their plasma membrane that is not apposed to the basal lamina (Miragall *et al.*, 1988). On the other hand, type IV collagen and fibronectin are expressed in the region of the OEC plasma membrane that is in contact with the basal lamina (Doucette, 1995).

Vimentin is the major constituent of the OEC intermediate filaments during development and in the adult (Barnett *et al.*, 1993; Franceschini and Barnett, 1996; Ramón-Cueto and Nieto-Sampedro, 1992; Schwob *et al.*, 1986). Besides vimentin, OECs also express nestin in the embryo and neonate (Doucette, 1993; Valverde *et al.*, 1992), and GFAP in the neonate and adult (Barber, 1982; Barnett *et al.*, 1993; Franceschini and Barnett, 1996; Pixley, 1992).

1.2.1 Ensheathment of olfactory axons

OECs envelop multiple axons in a common mesaxon with their inner process-bearing (adaxonal) surface, which has no basal lamina (DeLorenzo, 1957; Doucette, 1990; Field *et al.*, 2003; Frisch, 1967). Winding, sheet-like inner OEC cytoplasmic processes separate the unmyelinated axons into smaller groups within the inner compartment of each fascicle (Valverde and Lopez-Mascaraque, 1991). In cross section, each peripheral olfactory fascicle is seen to contain several glial cells, with the cytoplasmic processes of these cells extending towards the center of the fascicle to enclose variable numbers of olfactory axons within common mesaxons (Doucette, 1990). The outer (abaxonal) surface of OECs is smooth and faces the connective tissue, which in the PNS portion of the pathway contains fibroblasts and collagen fibers and from which they are separated by a basal lamina. This basal lamina is continuous with the epithelial basal lamina at the numerous points where small bundles of olfactory axons leave the olfactory epithelium and are first ensheathed by OECs (DeLorenzo, 1957; Doucette, 1984, 1990). Raisman (1985) suggested that the peculiar multiaxonal sheathing arrangement allows the newly growing axons to slide in along existing axonal surfaces without requiring them to make direct contact with the glial cell membrane.

Throughout their length in both the PNS (fila olfactoria) and CNS (ONL and glomerular layers) portions of the primary olfactory pathway, the olfactory axons are ensheathed by OECs, such that the axons do not contact any other glial cell type (Barber and Lindsay, 1982; Carr and Farbman, 1993; Doucette, 1984, 1986, 1990, 1991, 1993; Pixley, 1992; Raisman, 1985). Unlike Schwann cells, OECs do not ensheath individual axons; instead, the mesaxon of each OEC encloses densely packed bundles of unmyelinated axons (Barber and Lindsay, 1982; Doucette, 1993; Farbman and Squinto, 1985; Peters, 1976). Similarly, at variance with the

manner in which Schwann cells ensheath unmyelinated non-olfactory axons, OECs within the ONL of the olfactory bulb are only apposed to a basal lamina in those regions where they contribute to the formation of the external glial limiting membrane. However, similar to other peripheral nerves (Bunge *et al.*, 1986), the olfactory fascicles in the lamina propria are enclosed within a basal lamina that separates the OECs from the cells and matrix of the surrounding connective tissue.

The ability of the olfactory nerves to re-establish synaptic connection with central adult neurons of the olfactory bulb in adult mammals has been attributed to OECs and to the glial arrangements at the olfactory glomerulus (Barber, 1981; Barber and Raisman, 1978; Graziadei *et al.*, 1980b; Graziadei and Monti Graziadei, 1978, 1979; Wilson and Raisman, 1981). Unmyelinated axons of the spinal nerves and cranial nerves 3 to 12 cross from the PNS into the CNS and are "handed" directly from a Schwann cell to an astrocyte (Berthold and Carlstedt, 1977). In contrast, olfactory axons are never "handed" from OECs to another glial cell as the axons enter the CNS. Olfactory axons do, however, lose their glial ensheathment as they enter a glomerulus where the cytoplasmic processes of astrocytes may come into close apposition to the entering axon fascicles although the axons are never actually ensheathed by these glial cells (Doucette, 1984; Raisman, 1985). OECs extend their cytoplasmic processes in the shape of "prongs" that penetrate deeply along the periphery of each glomerulus. However, OEC processes do not enter the neuropil of the glomerulus (Raisman, 1985).

1.2.2 Embryonic and fetal development

The differentiation of OECs during embryonic and fetal development is intimately related to the development of the first cranial nerve. Progenitor cells originating from the olfactory placode give rise to OECs (Doucette, 1989, 1990; Farbman and Squinto, 1985; Marin-Padilla and Amieva, 1989). During the development of the embryo and later of the fetus, the OEC progenitors migrate away from the olfactory placodes toward the rostral wall of the telencephalic vesicle where they eventually contribute to the formation of the ONL of the presumptive olfactory bulb (Doucette, 1989; Marin-Padilla and Amieva, 1989). As these immature OECs enter the telencephalic vesicle during early embryonic development, they begin to express the N-CAM and L1 cell adhesion molecules (Miragall *et al.*, 1989). Doucette (1993) suggested that these CAMs facilitate the adhesion and growth of olfactory axons from the olfactory epithelium into the developing olfactory bulb with the growing axons being able

to use the cell surface of OECs as a substrate. On the other hand, OECs ensheath multiple axons, most of which do not come in a close contact with OEC plasma membrane (DeLorenzo, 1957; Doucette, 1990; Field *et al.*, 2003; Frisch, 1967). Such arrangement of olfactory axons within an OEC mesaxon allows newly formed axons to extend to the olfactory bulb using other axons for guidance (Doucette, 1984, 1990; Raisman, 1985). These OECs also begin to express the p75NGFR (Gong *et al.*, 1994) and the L14 lectin (Mahanthappa *et al.*, 1994). Although the function of p75NGFR remains to be determined, it is thought to provide a chemotactic guidance to growing olfactory axons (Doucette, 1993; Li *et al.*, 1997). Mahanthappa *et al.* (1994) showed that the L14 lectin also promotes the adhesion of primary olfactory neurons to laminin *in vitro* in an integrin-independent manner.

During the development of the rodent olfactory system, peripheral glial progenitor cells accompany and ensheath olfactory axons growing dorsocaudally toward the cerebral vesicle (Doucette, 1989). In Theiler stage 19-21 embryos, the peripheral glial progenitor cells and olfactory axons are located under the pia matter along the external surface of the bulb primordium forming a presumptive ONL immediately superficial to the bulb's external glial limiting membrane. By the first part of stage 21, the peripheral progenitor cells are identified within the presumptive ONL with an electron-dense cytoplasmic matrix filled with expanded rough endoplasmic reticulum, mitochondria, numerous free ribosoms and polyribosomes, and a well-developed Golgi apparatus. Some cells acquire bipolar shape with their long axis oriented parallel to the direction of axonal growth. Olfactory axons but not peripheral glial progenitors penetrate the marginal zone of the bulb primordium. The only glial cells found within the ONL by Theiler stage 24 of developing rat and mouse embryos are the peripheral glial progenitor cells.

1.2.3 PNS-CNS transitional zone

OECs are the only glial cells known to ensheath axons on both sides of the PNS-CNS transitional zone (Doucette, 1984, 1986, 1990, 1991; Fraher, 1999). In contrast, when myelinated primary sensory axons enter the CNS, their ensheathment abruptly changes from Schwann cell ensheathment to that of oligodendrocytes or astrocytes (Berthold and Carlstedt, 1977; Maxwell *et al.*, 1969; Steer, 1971). The relationship of unmyelinated non-olfactory primary sensory axons to surrounding glial cells is similar to that of myelinated fibers at the PNS-CNS transitional zone with the axons being ensheathed in pockets of the Schwann cells in

the PNS and those of astrocytes in the CNS. The axons of spinal cord and cranial nerve rootlets pass through pores or openings in the glia limitans at the transitional zone between PNS and CNS. Each Schwann cell within the PNS compartment is surrounded by basal lamina and collagen fibers. The basal lamina of each Schwann cell at the PNS-CNS junction is continuous with that of the glia limitans of the subpial astrocytes (reviewed in Fraher, 1999).

Each olfactory nerve rootlet travels within the subarachnoid space after leaving foramina of the cribriform plate before entering the ONL of the olfactory bulb. The basal lamina covering the external surface of each of these rootlets is continuous with that covering the external surface of the glia limitans of the ONL. Within the PNS, the OECs within each olfactory fascicle are enclosed within a common basal lamina, which surrounds the entire fascicle (DeLorenzo, 1957; Frisch, 1967). Fibroblasts and collagen fibers separate adjacent peripheral olfactory fascicles from one another (Doucette, 1993). In contrast, the OECs within the CNS portion of the primary olfactory pathway are opposed to a basal lamina only where these glial cells contribute to the formation of the glia limitans (Barger and Lindsay, 1982; Doucette, 1984, 1986). Furthermore, in the ONL astrocytes are found between adjacent olfactory fascicles instead of fibroblasts and collagen fibers (Doucette, 1984, 1986, 1993). In addition, astrocytes are excluded from the inside of these fascicles and from the PNS-CNS transitional zone of the first cranial nerve, which is where OECs have the exclusive role of forming the glia limitans (Doucette, 1991), even though elsewhere this functions is the exclusive domain of astrocytes. At the surface of the olfactory bulb, the cytoplasmic processes and plasma membrane of OECs mingle and come into close contact, respectively, with those of astrocytes, and form part of the glial limiting membrane with no intervening basal lamina between the two types of glial cell (Doucette, 1984, 1990).

1.2.4 Tissue culture protocols

OECs have attracted the attention of researchers as potential candidates for the treatment of different neurological diseases such as multiple sclerosis and spinal cord injury. When grafted into demyelinated lesions, OECs were found to form myelin around naked axons (Franklin *et al.*, 1996; Imaizumi *et al.*, 1998). Transplantation of these cells into a mechanically damaged spinal cord resulted in axonal regeneration, remyelination of axons in the damaged spinal cord tracts and functional recovery of locomotion in the animals (reviewed

by Boyd *et al.*, 2003; Franklin and Barnett, 2000; Plant *et al.*, 2003; Raisman, 2001; Ramon-Cueto *et al.*, 2000).

OECs can be isolated from 18 day (E18) rat embryos (Boyd *et al.*, 2004; Devon and Doucette, 1992, 1995; Doucette, 1993; Doucette and Devon, 1994, 1995), newborn rats (Barnett *et al.*, 1993; Chuah and Au, 1993, 1994; Franceschini and Barnett, 1996; Pixley and Pun, 1990) and from adult rats (Gudino-Cabrera and Nieto-Samperdro, 1999; Nash *et al.*, 2002; Ramon-Cueto and Nieto-Sampedro, 1992) and humans (Barnett *et al.*, 2000; Kato *et al.*, 2000). The main advantage of using embryonic tissue is that at day 18 of embryonic development all of the glial cells in the ONL of the olfactory bulb of rats are OECs (Doucette, 1989). On the other hand, cell cultures obtained from the olfactory bulbs of newborn and adult animals require additional purification procedures, such as immunopanning (Ramon-Cueto and Nieto-Sampedro, 1994) or using a cell sorter (Pascual *et al.*, 2002), to eliminate contaminating cells.

Primary cultures of OECs have been obtained from different parts of the olfactory system including the olfactory mucosa, olfactory nerve, and olfactory bulb. Pixley and Pun (1990) harvested olfactory mucosa from newborn Sprague Dawley rat pups, whereas Lu et al. (2002) used the lamina propria of olfactory mucosa they obtained from adult rats. Devon and Doucette (1992, 1995), Boyd et al. (2004), Chuah and Au (1993), Barnett et al. (1993), Ramon-Cueto and Nieto Sampedro (1992) used the ONL of the olfactory bulb as the source of tissue to initiate OEC cultures. Another way to acquire OECs for transplantation is a creation of a cell line. Boruch et al. (2001) generated a cell line of OECs producing nerve growth factor (NGF) whereas Goodman et al. (1993) generated a p75NGFR-positive OEC clonal cell line from neonatal Sprague-Dawley rats using the SV40 large T antigen. In order to break intercellular connection, Ramon-Cueto and Nieto Sampedro (1992) treated their cell cultures with trypsin containing medium, whereas others utilized a combination of trypsin and collagenase (Chuah and Au, 1991; Pixley and Pun, 1990). There are advantages and disadvantages to using primary cultures or cell lines. Engineering the cell line prevents contamination present in primary cultures and provides a long-term stable supply of a purified cell population. On the other hand, OEC cell lines are unlikely to be useful in future treatments of patients with demyelinating diseases and CNS injury unless we know that grafted cells will not form tumors or cause any other disorders as a consequence of uncontrolled cellular growth.

Most studies have used pure population of OECs, whereas Li *et al.* (1997, 1998) utilized unpurified OECs and Lakatos *et al.* (2003) implemented a mixture of OECs with other cells, i.e. meningeal cells, in their experiments. Unsatisfactory results in OEC harvesting procedures or OEC transplantation studies have prompted researchers to implement additional techniques (i.e. immunoadsorbtion, enzymatic cell separation) or chemicals (i.e. growth factors). Chuah and Teague (1999) used bovine pituitary extract in order to increase the number of OECs.

A marker specific to OECs is still unavailable for identifying these cells in both in vivo and in vitro experiments. This has prompted researchers to use various cell labelling techniques. Prelabelling OECs with a nuclear fluorochrome bisbenzimide (Hoechst 33342 dye; Imaizumi et al., 2000; Lu et al., 2002; Pascual et al., 2002; Ramon-Cueto et al., 1998, 2000), a fluorescent dye 1,1'- dioctadecyl - 3,3,3',3' - tetramethylindocarbocyanine perchlorate (DiI) (Skihar and Doucette, 2001, 2003), and Cell-Tracker probes (Li et al., 1997, 1998) were used before transplanting cells into the CNS of animals. Even though researchers did not find evidence that these dyes can be transferred directly to surrounding cells within the experimental time frame; macrophages could phagocytose dying OECs at the implantation site. For example, Hoechst prelabelling of cells prior to transplantation can result in false positive signals (Ruitenberg et al., 2002); therefore, the survival and migration of Hoechst-prelabelled transplanted cells may be overestimated. Adenoviral-mediated (Ruitenberg et al., 2002, 2003) and retroviral-transfer (Boyd et al., 2004; Lakatos et al., 2003) of LacZ or adenoviral transfer of green florescent protein (Li et al., 2003) were used to identify cells with fluorescent, light, or electron microscopy. The microscopy observations revealed that adult (Li et al., 2003; Ruitenberg et al., 2002, 2003) and neonatal (Boyd et al., 2004; Lakatos et al., 2003) rat OECs survived the implantation. Retroviral transfection of OECs with LacZ containing viral vector also allowed in-depth characterization of these cells at both the light and electron microscopic level. These observations provided unequivocal ultrastructural identification that LacZexpressing OECs neither associated directly with central axons nor did they form myelin sheaths or basal lamina after transplantation into the damaged spinal cord of adult rats, even when regeneration and remyelination of lesioned axons were evident (Boyd et al., 2004).

1.2.5 Therapeutic potential

Several approaches to enhance the remyelination of demyelinated axons in the adult mammalian CNS have been considered, with one approach being the transplantation of myelinforming cells (Akiyama et al., 2001, 2002a, 2002b; Barnett et al., 2000; Franklin et al., 1996; Imaizumi et al., 1998, 2000; Smith et al., 2002). One myelinating cell that is attracting increasing attention is the OEC. These glial cells have been transplanted into the electrolytically lesioned rat corticospinal tract (Li et al., 1997, 1998), into proximal and distal stumps of the transected rat spinal cord with a guidance channel containing Schwann cells embedded in Matrigel replacing a transected thoracic segment (Ramón-Cueto et al., 1998), into stumps of the completely transected adult rat spinal cord (Ramón-Cueto et al., 2000), and into the spinal cord dorsal column that had been demyelinated by ethidium bromide (EtBr) and xirradiated to prevent host cells from contributing to its repair (Imaizumi et al., 1998). The grafted OECs adopted a myelinating phenotype that persisted throughout the lesion site (Franklin *et al.*, 1996; Imaizumi *et al.*, 1998). They assemble a myelin sheath around single regenerating or remyelinated axons. Examinations of the morphology of the remyelinating glial cells indicated substantial structural differences in comparison with that of the normal spinal cord white matter. The OECs established an end-to-end PNS type myelinating relationship with individual axonal segments. An outer sheath was formed around groups of myelinated axons; it was suggested that the outer sheath arose from the grafted OECs (Imaizumi et al., 1998; Li et al., 1997, 1998; Ramón-Cueto et al., 1998). However, authors failed to provide evidence that transplanted OECs remyelinated axons and formed an outer sheath for these regenerating axons.

In the experiments of Ramon-Cueto *et al.* (1998, 2000), transplanted OECs were found to facilitate the regeneration of axons in descending motor pathways (i.e. corticospinal, raphespinal, and coeruleospinal). Regenerating motor fibers surrounded by OECs not only crossed the gliotic tissue formed at the proximal cut surface of the spinal cord but also grew across the scar formed between the proximal and distal stumps; some of these regenerating axons (noradrenergic and serotonergic) grew for up to 3 cm (approximately 6 spinal cord segments) distal to the lesion.

Kato *et al.* (2000) demonstrated that transplanted OECs derived from adult human olfactory nerves were capable of extensive remyelination of demyelinated axons after being

grafted into the adult rat spinal cord. Human OECs remyelinated axons in the x-irradiated and EtBr-induced demyelinated lesion (see Section 1.6.1). As had been reported previously for grafted rat OECs, the remyelinating human cells assembled a PNS-type myelin sheath around the axons (see Section 1.3.1).

OECs have features of both Schwann cells, a glial cell of the PNS, and astrocytes, one of the macroglia of the CNS (Doucette, 1990, 1995; Franklin and Barnett, 1997; Franklin, 2002b; Ramon-Cueto and Avila, 1998; Ramon-Cueto and Valverde, 1995). Similar to Schwann cells, which secrete growth-promoting and neurite-promoting factors that support the regeneration of both PNS and CNS axons (Aguayo, 1985), OECs facilitate the growth of the olfactory axons (Doucette, 1990). On the other hand, CNS glial cells and the microenvironment they create are believed to be primarily responsible for the failure of axonal regrowth in the adult mammalian CNS (Reier *et al.*, 1989; Schwab, 1990).

OECs express growth- and neurite-promoting factors such as platelet derived growth factor (PDGF; Kott *et al.*, 1994) neuropeptide Y (Ubink *et al.*, 1994; Ubink and Hokfelt, 2000), protein S-100 (Cummings and Brunjes, 1995; Franceschini and Barnett, 1996; Gong *et al.*, 1996), p75NGFR (Gong *et al.*, 1994; Treloar *et al.*, 1999; Vickland *et al.*, 1991) and gliaderived netrin (Reinhard *et al.*, 1988; Treloar *et al.*, 1999). Extracellular matrix and cell adhesion molecules such as N-CAM (Franceschini and Barnett, 1996), laminin (Doucette, 1996), and tenascin (Kafitz and Greer, 1998) are suggested to facilitate axonal growth and guidance in the olfactory nerve and in the ONL of the olfactory bulb, where the OECs are found.

A great amount of research data have been collected over the past decade on clinical relevance of using OEC for promoting neural repair in the CNS of adult mammals (Doucette, 1990, 1995; Franklin and Barnett, 1997; Franklin *et al.*, 2002a, 2002b). OECs have been shown to promote long-distance regeneration of axons in the adult mammalian spinal cord and facilitate remyelination of CNS axons such that conduction velocities of previously demyelinated axons are restored to normal levels. OECs can be easily obtained from extracranial sources, such as from olfactory epithelium of human (Feron *et al.*, 1998; Jafek *et al.*, 2002; Lanza *et al.*, 1993, 1994) or from similar tissue in the nasal cavity of rodents (Franklin *et al.*, 2002a; Lu *et al.*, 2002). Thus, patients can provide their own olfactory tissue for autologous transplantations, which will avoid unnecessary immunosuppression.

Furthermore, OECs can more easily integrate into CNS tissue after grafting into the injured or demyelinated spinal cord (Lakatos *et al.*, 2000; Li *et al.*, 1998) than Schwann cells (Baron-Van Evercooren *et al.*, 1992; Blakemore, 1992; Duncan and Milward, 1995).

1.3 Myelinating Glia

In addition to OECs there are two additional cell types, namely oligodendrocytes (Baumann and Pham-Dinh, 2001) and Schwann cells (Bunge *et al.*, 1968), possessing the ability to myelinate axons. Oligodendrocytes are myelinating glia of the CNS whereas Schwann cells perform the same role in the PNS. All three types of myelinating glia could potentially contribute to neural repair in areas where CNS myelin is damaged or absent (see Section 1.8).

1.3.1 Myelin

Myelin, which was named so by Virchow (reviewed in Baumann and Pham-Dinh, 2001), is a very important constituent of both CNS and PNS axons providing electrical insulation for these axons. Rich in lipids and low in water content, the myelin sheath has a unique segmental structure, which permits the saltatory conduction of nerve impulses, thus supporting fast nerve conduction over long distances. Myelin is very important for the development and function of the human CNS and PNS. Myelin structure and/or function are perturbed in a number of different neurological disorders of the CNS and PNS such as in multiple sclerosis and in peripheral neuropathies, respectively (Fazakerley and Buchmeier, 1993; Wekerle, 1993).

Both oligodendrocytes and Schwann cells extend cytoplasmic processes, to ensheath a segment of axon in a "jelly-roll" form (Morell *et al.*, 1994; Peters *et al.*, 1991). The cytoplasmic leaflets of each concentric layer of myelin membrane becomes closely apposed as the myelin is compacted, thus forming the major dense line. The extracellular leaflets of adjacent concentric layers of the myelin membrane fuse together to form the intraperiodic line (or minor dense lines). A narrow extracellular cleft, the periaxonal space, separates the myelin sheath from the axonal membrane. Myelin sheaths in the PNS have cytoplasmic pockets, called Schmidt-Lantermann clefts or incisures, containing cytoplasm between myelin membrane; such cytoplasmic pockets are rarely found in CNS myelin. At the inner (i.e. adaxonal) and outer (i.e. abaxonal) end of the spiral, myelin is less compact and forms inner

and outer loops, which retain small amounts of either oligodendrocyte or Schwann cell cytoplasm. Each myelin sheath, or internode, for example in rat optic nerve is about 150-200 μ m in length (Butt and Ransom, 1989) and is separated by a short length of axon where there is no myelin, the node of Ranvier, from an adjacent internode (Bunge, 1968). The myelin and nodes of Ranvier ensure fast saltatory conduction allowing the nerve impulses to jump from node to node, rather than more slowly progressing along the axon as they do in unmyelinated or demyelinated fibers.

The insulating properties of the myelin sheath that facilitate the saltatory conduction of action potentials are largely due to its structure, its thickness, its low water content, and its richness in lipids (Ritchie, 1984). Oligodendrocytes and Schwann cells produce the specific constituents of myelin, such as myelin lipids and myelin proteins. Myelin lipids are represented by galactolipids (i.e. cholesterol, cerebroside, sulfatide) and phospholipids (ethanolamine phosphatides, lecithin, sphingomyelin, phosphatidylserine, phosphatidylinositol, plasmalogens). Both CNS and PNS myelin lipids comprise up to 70 % and protein molecules approximately 30% of myelin dry weight, whereas other cellular membranes have a higher ratio of proteins to lipids (Morell et al., 1994, Norton, 1981). The PNS myelin has many of the same lipids as myelin of the CNS, although myelin in the PNS contains less cerebroside and sulfatide but more sphingomyelin than that in CNS (Smith et al., 1983). In oligodendrocytes, myelin lipids are especially rich in glycosphingolipids, galactosylceramides, in particular galactocerebroside (e.g. GAL-C) and their sulfated derivatives, sulfatides (i.e. sulfogalactosylceramides) (Raff et al., 1978, Zalc et al., 1981). Deficiencies in some lysosomal enzymes result in an abnormal degradation of sulfatides, galactocerebrosides, or proteins and accumulation of these undegraded fatty acids. These disturbances of myelin metabolism result in leukodystrophies (Kolodny, 1993).

Myelin proteins are specific membrane components of the myelin sheath in both the PNS and CNS. The PNS and CNS myelin proteins include myelin basic protein (MBP) (Kies *et al.*, 1972) and proteolipid protein (PLP) (Folch and Lees, 1951). 2',3'-Cyclic nucleotide-3'phosphohydrolase (CNPase; Trapp *et al.*, 1988) and several glycoproteins including myelin-associated glycoprotein (MAG; Quarles, 1997) and connexin 32 (Cx32) are also present in peripheral and central myelin (Scherer *et al.*, 1995). However, the expression of peripheral myelin protein 22 (PMP22) and P0 is found only in PNS myelin, whereas
myelin/oligodendrocyte glycoprotein (MOG; Brunner *et al.*, 1989) can only be detected in the CNS.

MBP is a family of protein isoforms with different molecular masses and plays a major role in myelin compaction in the CNS (Wood and Moscarello, 1989). The MBP gene has been identified on chromosome 18 and assigned to 18q22-qter in the human genome (Saxe *et al.*, 1985; Sparkes *et al.*, 1987). Posttranslational modifications of MBP including in particular acetylation, phosphorylation, and methylation, are important for compaction of the myelin membrane (Wood and Moscarello, 1989). In the shiverer mutant mouse, which presents with a deletion of a large part (five out of six exons) of the MBP gene (Roach *et al.*, 1985), CNS myelin lacks a major dense line, whereas PNS myelin appears normal (Privat *et al.*, 1979).

PLPs are lipid-protein complexes, so they were given the generic name of proteolipids to distinguish them from water-soluble lipoproteins. In addition to the main 25kDa isoform of PLP, there is a second PLP isoform called DM-20 (i.e. 20kDa). Both the PLP and the DM-20 isoforms are expressed in the CNS and PNS, with the former being predominant over DM-20 in CNS myelin and DM-20, on the other hand, being the dominant form in PNS myelin (Pham-Dinh et al., 1991). In contrast to PLP/DM-20 expression in oligodendrocytes, neither of these lipid proteins is incorporated into Schwann cell myelin of wild type mice (Anderson et al., 1997). The cohort study on patients with peripheral neuropathy who were diagnosed with Pelizaeus-Merzbacher disease, a leukodystrophy, demonstrated that expression of PLP but not DM-20 was necessary to prevent peripheral neuropathy (Shy et al., 2003). DM-20 expression appears earlier than PLP during development and it is thought that it might have a role in oligodendrocyte differentiation in addition to a structural role in the formation of a myelin sheath (Ikenaka et al., 1992; Peyron et al., 1997; Timsit et al., 1992). The gene coding for PLP is located on the X chromosome in humans (Willard and Riordan, 1985) at position Xq22 (Mattei et al., 1986) and in mice in the H2C area (Dautigny et al., 1986). Jimpy mice and myelin deficient rats have sex-linked recessive mutations located in the PLP gene. In humans such spontaneous mutation occurs in Pelizaeus-Merzbacher disease, also known as an Xchomosome-linked dysmyelinating disorder, which is characterized by abnormal formation and maintenance of myelin (Nave and Boespflug-Tanguy, 1996; Seitelberger, 1995). In PLP knockout mice, oligodendrocytes can still assemble a myelin sheath around large diameter axons, but the concentric layers of myelin membrane are loosely wrapped with wide

extracellular spaces between the layers (Boison *et al.*, 1995). Therefore, PLP is suggested to be responsible for the tight adhesion of the outer leaflets of the myelin membrane surfaces leading to the formation of the minor dense line (Boison *et al.*, 1995; Duncan *et al.*, 1987, 1989). In addition, axonal swelling followed later by axonal degeneration associated with motor impairment was observed in 16 month old PLP/DM-20 null mice (Griffiths *et al.*, 1998).

CNPase represents 4% of total CNS myelin protein. Although the biological role of CNPase remains obscure, this protein is present in the periaxonal membrane, inner mesaxon and in the paranodal loops of oligodendrocyte myelin (Trapp *et al.*, 1988). CNPase is also found in PNS myelin being localized similar to that of CNS myelin. CNPase is present within the oligodendrocyte plasma membrane during the earliest phase of axonal ensheathment prior to the appearance of PLP and MBP. Gravel *et al.* (1996) suggested that CNPase might play a role in the extension of oligodendrocyte cytoplasmic processes around the axons during the early stages of myelinogenesis. Over expression of CNPase in transgenic mice perturbed myelin formation, resulted in a formation of intramyelenic vacuoles, aberrant expansion of oligodendrocyte cytoplasmic processes and precocious oligodendrocyte maturation (Gravel *et al.*, 1996).

MAG, as a quantitatively minor constituent of myelin, represents only 1% of the total protein found in CNS myelin and only 0.1% in PNS myelin. MAGs are transmembrane proteins with extracellular regions that are homologous to N-CAM and other members of the immunoglobulin gene superfamily (Salzer *et al.*, 1987). Recently, MAGs have been identified as members of sialic acid binding lectins, the sialoadhesins, which preferentially bind to neuron cell surface sialoglycolipids (Yang *et al.*, 1996). In the adult rat, MAG is largely distributed across different regions of the PNS myelin internode (Meyer-Franke and Barres, 1994; Trapp and Quarles, 1982), whereas it is confined to the periaxonal collar of the myelin sheath in the CNS (Trapp *et al.*, 1989). MAG is involved in the activation of a protein tyrosine kinase Fyn and a member of the src family, during the initial stages of myelination (Umemori *et al.*, 1999) and morphological differentiation of oligodendrocytes (Osterhout *et al.*, 1999). Although MAG-deficient mice develop almost normal CNS myelin, the myelin that is formed contains cytoplasmic organelles between noncompact lamellae, no periaxonal cytoplasmic periaxonal collar in most of the internodes, and an abnormal collar in the remainder (Li *et al.*, 1994, Montag *et al.*, 1994). Montag *et al.* (1994) suggested that MAG might participate in the

formation of the periaxonal cytoplasmic collar of oligodendrocytes and be responsible for the adhesion of oligodendrocyte processes to axons in the CNS. Two MAG proteins have been identified, large MAG and small MAG (Frail *et al.*, 1985; Salzer *et al.*, 1987), which differ not only in terms of their molecular weight but also in regard to their function. Pathological abnormalities of CNS myelin, but no PNS axon or myelin degeneration, are reported in large MAG mutant mice (Carenini *et al.*, 1997), although full null MAG mutants (i.e. both large and small MAGs) develop a peripheral neuropathy in addition to CNS myelin pathology (Fruttiger *et al.*, 1995). These data indicate that large MAG is the critical MAG isoform in CNS myelin, whereas small MAG is important in the PNS for the maintenance and integrity of the myelin sheath (Fujita *et al.*, 1998). Consistent with this suggestion, large MAG is expressed earlier than small MAG during CNS myelination and is the predominant isoform in adult CNS myelin, whereas small MAG is the predominant isoform in PNS myelin in both rodents (Frail *et al.*, 1987; Salzer *et al.*, 1987) and humans (Miescher *et al.*, 1997).

MOG was first identified as the antigen responsible for the demyelination observed in animals injected with whole CNS homogenates (Lebar *et al.*, 1976, 1986). Present only in mammalian species, the MOG gene is co-localized to the human major histocompatibility complex (MHC) on chromosome 6 in the human genome, and on chromosome 17 in mice (Pham-Dinh *et al.*, 1993). MOG is located in the plasma membrane of oligodendrocytes, especially on the cytoplasmic processes, and on the outermost lamellae of the myelin sheaths (Brunner *et al.*, 1989). MOG is also a surface marker of mature (Birling *et al.*, 1993; Coffey and McDermott, 1997; Pham-Dinh, 1993) and myelinating (Solly *et al.*, 1996) oligodendrocytes. The extracellular domain of MOG, which is a member of the immunoglobulin superfamily, is believed to induce both a T cell-mediated inflammatory immune reaction and a demyelinating antibody-mediated response in the experimental autoimmune encephalomyelitis (EAE) animal model (Lassmann *et al.*, 1988).

PMP22 is a small, hydrophobic glycoprotein most prominently expressed by Schwann cells as a component of compact myelin of the PNS (Naef and Suter, 1998; Snipes *et al.*, 1992). Functionally, PMP22 is involved in myelination during the development of peripheral nerves and in the maintenance of peripheral myelin structure (Uyemura, 1993). Mutations affecting the PMP22 gene include duplications, deletions, and point mutations. Point mutations are responsible for the most common forms of hereditary peripheral neuropathies including

Charcot-Marie-Tooth disease type 1A (Patel *et al.*, 1992; Timmerman *et al.*, 1992), hereditary neuropathy with liability to pressure palsies and a subtype of Dejerine-Sottas syndrome (Roa *et al.*, 1993).

P0 glycoprotein is the major transmembrane protein of PNS myelin of mammals as well as of fish and amphibia (Uyemura et al., 1995). In contrast to what is seen for mammalian oligodendrocytes, the oligodendrocytes of fish express P0 (Jeserich and Waehneldt, 1986). Lemke and Axel (1985), Lemke et al. (1988) and Lemke (1993) suggest that P0 mediates both the extracellular and the cytoplasmic compaction of myelin. The extracellular domain of P0 interconnects the turning loops of Schwann cell processes to form the intraperiod line (D'Urso et al., 1990; Filbin et al., 1990; Schneider-Schaulies et al., 1990). PMP22 interacts with P0 to form complexes, which probably hold adjacent Schwann cell membranes together (D'Urso et al., 1999). The intracellular part of P0 contains positively charged basic residues and is believed to interact with the phospholipids and MBP in the cytoplasmic face of apposing Schwann cell membranes, thus forming the major dense line (Kirschner and Ganser, 1980; Ding and Brunden, 1994; Martini et al., 1995). P0-deficient mice develop severely compromised myelin compaction, primarily at the site of the intraperiod line, whereas the major dense line continues to be formed by Schwann cells (Giese et al., 1992; Martini et al., 1995). In the absence of MBP, Schwann cells expressing P0 are still able to form the major dense line in shiverer mice (Kirschner and Ganser, 1980; Inouye et al., 1985). The lack of both MBP and P0 in the double mutants results in an almost complete loss of major dense lines in PNS myelin (Martini et al., 1995). Therefore, the formation of the major dense line in peripheral nerve myelin is formed by both P0 and MBP (Martini et al., 1995).

In multiple sclerosis (MS), an immune response is directed against components of CNS myelin, including MBP (Zamvil and Steinman, 1990; Steinman, 1993, 1995) and MOG (Kerlero de Rosbo *et al.*, 1993). B-cell and T-cell reactivity against PLP, CNPase and MAG have also been observed in MS patients (Oksenberg *et al.*, 1993; Steinman, 1995). The MOG gene has been identified within the major histocompatibility complex 1 region on the chromosome 6 (Daubas *et al.*, 1994; Pham-Dinh *et al.*, 1993). Steinman (1993) suggested that MOG might display some immune function in the CNS. MOG can be an autoantigen as it is located on the surface of the outermost lamellae of the myelin sheath, where MOG is more exposed to antibodies than are other myelin proteins like MBP or PLP. Oligodendrocytes are

found to be susceptible to lysis by complement (Compston *et al.*, 1989). Piddlesden and Morgan (1993) suggested that the absence of the complement-inhibitory protein CD59 from the oligodendrocyte membrane contributes to this susceptibility to lysis by oligodendrocytes.

There are morphological and molecular differences between CNS and PNS myelin. Although MBP, PLP, MAG and Cx32 are expressed in both oligodendrocyte and Schwann cell myelin, oligodendrocytes also produce MOG and CNPase, whereas Schwann cells make PMP22 and P0 myelin proteins. In contrast to oligodendrocytes, which typically ensheath and myelinate multiple CNS axons, myelinating Schwann cells always have a 1:1 relationship with PNS axons. A myelinated axon is always associated with an adjacent Schwann cell, the nucleus of which gives the appearance of a "signet ring". The cytoplasmic processes of adjacent myelinating Schwann cells interdigitate at the node of Ranvier, in marked contrast to the exposure of CNS axons to extracellular space at each node (Bunge, 1968). The soma of a Schwann cell unlike that of an oligodendrocyte is circumferentially covered by a basal lamina, which is continuous with that of the adjacent internode, regardless of whether the Schwann cell is myelinating or not. This type of myelination is called peripheral or PNS-type. Schwann cell myelinated fibers are separated from each other by extracellular compartment, the endoneurium, whereas the oligodendrocyte myelinated fibers are in close contact.

1.3.2 Oligodendrocytes

Rio Hortega first introduced the term oligodendroglia in 1921 describing neuroglial cells with few processes in metallic impregnation stained brain tissue (reviewed in Baumann and Pham-Dinh, 2001). Although a major role of oligodendrocytes is the myelination of CNS axons, there are also satellite oligodendrocytes, which are found perineuronally and which are believed to regulate the microenvironment around neurons (Ludwin, 1997). A myelinating oligodendrocyte extends several cytoplasmic processes, each of which contacts and repeatedly envelops a stretch of axon with subsequent condensation of multispiral myelin (Bunge *et al*, 1962; Bunge, 1968). Adjacent myelin segments (i.e. internodes) on the same axon belong to different oligodendrocytes, with each segment being separated by a node of Ranvier. In contrast to the PNS, the axolemma is exposed to the extracelluar milieu at the nodes of Ranvier in the CNS (Bunge, 1968) and is in close proximity to astrocytic processes (Black and Waxman, 1988) and to NG2 positive glial precursor cells (Butt *et al.*, 1999). The number of axons that are myelinated by each oligodendrocyte varies according to the area of the CNS and

possibly also the species, from 40 in the optic nerve of the rat (Peters *et al.*, 1991) to 1 in the spinal cord of the cat (Bunge *et al.*, 1961).

Butt *et al.* (1995) classified myelinating oligodendrocytes into four types based on their morphology and on the size and thickness of the myelin sheath they formed. Type I oligodendrocytes were small cells supporting short, thin myelin sheaths for 15-30 small diameter axons. Intermediate types of myelinating oligodendrocytes were classified as types II and III. The type IV oligodendrocytes were the largest cells, forming long, thick myelin sheaths around 1-3 large diameter axons (Bunge, 1968).

Originating from migratory and mitotic precursors, oligodendrocyte progenitors differentiate progressively into postmitotic myelin-producing cells as they sequentially express developmental and then myelinating markers (Hardy and Reynolds, 1993; Pfeiffer et al., 1993). At the early developmental stages, oligodendrocyte precursors have been characterized by their bipolar morphology and by the expression of nestin (Gallo and Armstrong, 1995), the DM-20 isoform of PLP (Ikenaka et al., 1992; Peyron et al., 1997; Timsit et al., 1992, 1995), the PDGFβ receptor (reviewed in Richardson et al., 2000; Spassky et al., 2000), PSA-NCAM (Grinspan and Franceschini, 1995; Hardy and Reynolds, 1991), and CNPase. Later, PSA-NCAM and nestin are no longer expressed. As the oligodendrocyte progenitors are actively proliferating and migrating within the CNS during embryonic and fetal development, they maintain production of PDGF-β receptor, DM-20 and CNPase and begin synthesizing the GD3 ganglioside (Hardy and Reynolds, 1991), the gangliosides recognized by the monoclonal antibody A2B5 (Fredman et al., 1984), and the rat NG2 proteoglycan (Nishiyama et al., 1996, 1999). These migrating oligodendrocyte progenitors eventually settle along fiber tracts of the presumptive white matter as pro-oligodendrocytes, which are cells of the oligodendrocyte lineage that express sulfatides and unidentified glycolipids that bind the O4 monoclonal antibody (Bansal et al., 1989).

As the pro-oligodendrocytes differentiate into oligodendrocytes they become less motile (Orentas and Miller, 1996), lose their mitogenic response to PDGF-β receptor (Gao *et al.*, 1998; Hart *et al.*, 1989; Pringle and Richardson, 1993), and they no longer express antigenic epitopes recognized by the A2B5, GD3 and NG2 antibodies (reviewed in Baumann and Pham-Dinh, 2001). These immature oligodendrocytes are identified by virtue of their expression of Gal-C (Pfeiffer *et al.*, 1993; Raff *et al.*, 1978; Zalc *et al.*, 1981), and by the expression of the

RIP antigenic epitope (Butt et al., 1995; Friedman et al., 1989). As the cells differentiate into oligodendrocytes they begin to express MBP (Butt et al., 1995). Even in the absence of neurons, oligodendrocyte progenitors can differentiate into cells that extend myelin-like membranes (Sarlieve et al., 1983) and express the myelin proteins PLP, MBP and MAG. Thus, the capacity of oligodendrocyte progenitors to differentiate into cells expressing a myelinating phenotype is specific to the lineage (Temple and Raff, 1986). Nevertheless, co-culture of oligodendrocyte progenitors with neurons increases gene expression of these myelin proteins in the oligodendrocytes (Macklin et al., 1986; Matsuda et al., 1997), similar to that observed in vivo (Kidd et al., 1990). When axonal contact occurs, oligodendrocytes dramatically modify their morphology, retract cytoplasmic processes that have not made contact with an axon (Hardy and Friedrich, 1996), downregulate DM-20 expression, and begin to express the fulllength isoform of the PLP gene (Trapp et al., 1997). Those oligodendrocytes in white matter that do not make axonal contact undergo programmed cell death and die, thus eliminating unneeded cells (Barres et al., 1992). This switch from DM-20 expression to full-length PLP expression is believed to signal the beginning of the process of myelination (Trapp et al., 1997).

1.3.3 Schwann cells

Professor Theodor Schwann in 1839 published his treatise on the cell theory, where he showed that in spite of a fibrous nature peripheral nerves are made up of cells, which were later named after him (reviewed in Gould, 1990). Schwann cells are the myelinating glia of the PNS.

Schwann cells develop from the neural crest (Le Douarin *et al.*, 1991) and evolve through three steps (Mirsky *et al.*, 2001). First, after the transition from the neural crest Schwann cell precursors are typically found in rat and mouse nerves of 14-15 and 12-13 day old embryos, respectively. Second, precursors become immature Schwann cells that are present in 17 day old rat embryo and 15 day old mouse embryo to around birth. Subsequently, the immature Schwann cells differentiate along the peripheral nerves into two mature types with myelinating Schwann cells appearing first and mature non-myelinating Schwann cells shortly after (Mirsky and Jessen, 1996; Jessen and Mirsky, 1997, 1998). It is not known how glial cells originate from the neural crest, which also gives rise to other lineages including neurons and melanocytes. Henion and Weston (1997) suggest that while migrating to their

final destination cells of the neural crest enter distinct lineages under the influence of instructive signals. For example, β -neuregulins appear to bias neural crest cell differentiation towards the glial lineage by blocking entry of the cells into the neuronal lineage (Shah *et al.*, 1994), whereas transforming growth factor β (TGF- β) promotes the generation of smooth muscle cells and bone morphogenic protein 2 stimulates development of autonomic neurons (Shah *et al.*, 1996).

In postnatal rodents, immature myelinating Schwann cells ensheath the large axons and the non-myelinating cells accommodate small diameter axons in their membrane invaginations. Driven by axonal signals, myelinating Schwann cells carry out a large amount of membrane synthesis and ensheathment needed to form the myelin sheath (Jessen and Mirsky, 1992). Extensive changes in gene expression occur in immature Schwann cells when they are induced to myelinate an axon including up-regulation of myelin proteins such as P0, MBP, PMP22 and down-regulation of NCAM, p75NGFR and GFAP (Jessen et al., 1990). Three transcriptional factors that are known to play an important role in Schwann cell myelination include Sox-10 a member of the SRY-like high motility group domain family of transcription factors (Kuhlbrodt et al., 1998; Southard-Smith et al., 1998), the POU domain factor Oct-6/SCIP/tst-1 (Arroyo et al., 1998) and the zinc finger protein Krox-20 (Topilko et al., 1997). In mice with transiently arrested Oct-6/SCIP/tst-1 expression or permanently blocked Krox-20 transcription factors, promyelinating Schwann cells appear normal; they express a basal lamina and normal levels of MAG, MBP and P0 mRNAs. However, Schwann cells in these transgenic animals do not form a concentric compacted myelin sheath (Bermingham et al., 1996; Topilko et al., 1994). Recently, the mutation of Egr-1 (Krox-20) has been identified in patients with Charcot-Marie-Tooth disease, an inherited demyelinating neuropathy, (Warner et al., 1998, 1999; Timmerman et al., 1999).

During embryonic development, the survival of Schwann cell precursors depends on axonal signals (i.e. β -neuregulins), whereas mature Schwann cells can survive in the absence of axons (Grinspan *et al.*, 1996; Li *et al.*, 1998; Mirskiy *et al.*, 2001; Trachtenberg and Thompson, 1996). This is due primarily to the presence of an autocrine growth factor loop in Schwann cells starting in rats from embryonic day 18 onwards (Meier *et al.*, 1999). The most important growth factor components of autocrine survival include insulin-like growth factor (IGF)–2, PDGF-BB and neurotrophin-3 (Meier *et al.*, 1999). This ability of Schwann cells to survive in

the absence of axons is critical in the context of peripheral nerve regeneration after injury, when the axons must enter the distal stump and regrow into their target tissue. Living Schwann cells are necessary for successful peripheral nerve repair, providing essential adhesion molecules and trophic factors (Fawcett and Keynes, 1990; Nadim *et al.*, 1990).

Two distinct forms of Schwann cells exist in peripheral nerves in adult mammals: myelinating and nonmyelinating Schwann cells. Schwann cells form contacts with axons, other Schwann cells and with the extracellular matrix. During differentiation, myelinating Schwann cells lose contact with all but a single axon segment, which they subsequently ensheath with cytoplasmic processes and myelinate (Zorick and Lemke, 1996). Axonal contacts are essential for initiation of myelination, which induce a Schwann cell to express myelin genes, the expression of which is influenced by extracellular matrix (Bunge *et al.*, 1989, Scherer *et al.*, 1994; Scherer, 1997). Schwann cells are known to produce basement membrane components including laminin, type IV collagen and fibronectin, to which they adhere via integrin receptors, dystroglycans and *N*-syndecan (Bunge *et al.*, 1989; Mirsky and Jessen, 1996; Scherer, 1997). Interactions with laminin are suggested to be important for Schwann cell migration (Milner *et al.*, 1997) and myelination (Chen *et al.*, 2000; Fernandez-Valle *et al.*, 1994). In addition, morphological arrangements of actin cytoskeleton are also critical for Schwann cell differentiation and myelination (Fernandez-Valle *et al.*, 1997).

1.3.4 Olfactory ensheathing cells

In vivo, OECs ensheath but do not myelinate small C fibers (i.e. the olfactory axons; Doucette, 1984, 1990; Raisman, 1985). Barnett *et al.* (1993) and Devon and Doucette (1994, 1995) have demonstrated, OECs also express a nonmyelinating phenotype when grown in neuron-free cell cultures. OECs express a mixture of astrocyte-specific and Schwann-cell specific phenotypic features (Doucette, 1990, 1995; Franklin, 2002b; Franklin and Barnett, 1997; Ramon-Cueto and Avila, 1998; Ramon-Cueto and Valverde, 1995) and have been reported to perform the roles of both astrocytes and Schwann cells (Doucette, 1990, 1993; Doucette and Devon, 1995). Given the fact that OECs only ensheath unmyelinated axons *in vivo* (Barnett *et al.*, 1993; Devon and Doucette, 1992, 1995), they would be expected to express a nonmyelinating phenotype when grown *in vitro*. The media shown to be effective *in vitro* in promoting the growth and differentiation of neopallial and olfactory bulb oligodendrocyte progenitor cells failed to induce OECs to express a myelinating phenotype (Barnett *et al.*, 1993; Doucette and Devon, 1995). Moreover, OECs in neuron-free cultures treated with dBcAMP can be induced to express GAL-C but not induced to express MBP (Doucette and Devon, 1994, 1995). In contrast, Schwann cells re-express myelin proteins (MBP and P0) when cAMP had been added to the culture medium (Lemke and Chao, 1988; Morgan *et al.*, 1991; Pleasure *et al.*, 1985). The results of these studies indicated that the molecular mechanisms regulating the expression of a myelinating phenotype by OECs are different from those that function in Schwann cells.

The first evidence that OECs could assemble a myelin sheath was reported by Devon and Doucette (1992), who co-cultured OECs obtained from the olfactory bulbs of E18 rat embryos with dorsal root ganglion (DRG) neurons. Co-culturing of DRG neurons with OECs resulted in the expression of a myelinating phenotype and the assembly of a PNS-type myelin sheath around the DRG neurites. Devon and Doucette (1992) used fluoredeoxyuridine to eliminate non-neuronal cells from the DRG cultures and confirmed the purity of these cultures by microscopy before adding OECs prelabelled with a lipophilic fluorescent dye (PKH26). In these co-cultures, each OEC either myelinated only one neurite or ensheathed several unmyelinated neurites at the same time. Each OEC, whether it was myelinating single neurites or ensheathing multiple unmyelinated axons, was also enclosed within its own basal lamina, a situation that does not exist in vivo in the primary olfactory pathway (Doucette, 1984, 1990; see Section 1.2.1). Only occasionally, two or more myelinated neurites were ensheathed by the same glial cell and were enclosed within a common basal lamina. Immunocytochemical stainings revealed that S100-positive OECs ensheathing single DRG neurites were also MBPpositive. These MBP-positive OECs assumed a bipolar morphology and formed a contiguous series of cells along the length of each DRG neurite (Devon and Doucette, 1992, 1995). Axonal contact is one essential prerequisite for Schwann cells to express a myelinating phenotype (Brunden and Brown, 1990; Brunden et al., 1990), a requirement that also applies to OECs (Barnett et al., 1993; Devon and Doucette, 1992, 1995; Doucette and Devon, 1994, 1995). In contrast to Schwann cells, however, OECs do not require addition of L-ascorbic acid to the growth medium for them to assemble either a basal lamina or a myelin sheath in vitro (Doucette and Devon, 1995).

Franklin *et al.* (1996) used an O4-positive (+ve) clonal OEC cell line they had generated using tissue from the newborn rodent olfactory bulb and grafted these cells into an

EtBr/x-irradiated (see Section 1.7.1) demyelinated area of the adult rat spinal cord. Their study provided the first *in vivo* evidence of the ability of OECs to remyelinate demyelinated CNS fibers; similar to what had been previously described *in vitro* (Devon and Doucette, 1992, 1995). This clonal cell line myelinated the axons in a Schwann cell-like fashion. Subsequent studies by Imaizumi *et al.* (1998) using primary rat cell cultures of OECs not only replicated these *in vivo* experiments but, most importantly, also showed that the OEC remyelinated area. The grafting of OECs obtained from human OB (Bartnett *et al.*, 2000), human olfactory nerve (Kato *et al.*, 2000) or canine OB (Smith *et al.*, 2002) into x-irradiated/EtBr-induced demyelinated rat spinal cord resulted in Schwann-like remyelination of CNS axons within the demyelinated lesion. Radtke *et al.* (2004) utilized OECs harvested from an adult H-transferase transgenic pig for transplantation into the nonhuman primate spinal cord and observed remyelination of EtBr-induced demyelinated CNS axons.

Thus, OEC remyelination was demonstrated to be functionally significant. OECs were also found capable of remyelinating electrolytically lesioned corticospinal axons that regenerated into and through the site of the lesion after grafting OECs into the lesion site (Li et al., 1998). However, the remyelinating ability of OECs was questioned when the *in vitro* study of Plant et al. (2002) failed to identify any myelination of DRG neurites by co-cultured OECs, which were obtained using the technique of Ramon-Cueto and Nieto-Sampedro (1992). Schwann cells form more myelin sheaths in co-culture with DRG neurons than OECs in these experiments. Such lack of myelination in vitro by OECs, perhaps, could have been determined by the extent of the purification procedure used by Plant et al. (2002). p75NGFR-expressing OECs used in the study by Plant et al. might not be a subset of OECs responsible for the myelinating potential. Another possible explanation is that Plant et al. use adult OECs that may not have the same myelinating potential as embryonic OECs (Devon and Doucette, 1992, 1995). Takami et al. (2002) compared regenerative abilities of OECs and Schwann cells in vivo and found that Schwann cells but not OECs improved the functional recovery of adult rats after contusive spinal cord injury. Furthermore, OECs myelinated fewer axons than Schwann cells. However, the authors reported that only a few OECs could be identified within spinal cord tissue. Either susceptible to cytotoxicity or other unidentified causes, OECs failed to stay

around and therefore neither supported axon regeneration nor contributed to the remyelination of the regenerated axons in experiments performed by Takami *et al.* (2002).

The molecular mechanisms controlling the expression of a myelinating phenotype by OECs remain to be elucidated. Smith *et al.* (2001b) initiated a study to identify which transcription factors were expressed by OECs engaged in remyelinating spinal cord axons. Similar to Schwann cells (Blanchard *et al.*, 1996; Zorick *et al.*, 1999), the expression of Oct-6 /SCIP/ tst-1, Krox-20 and desert hedgehog transcription factors was detected in an EtBr/x-irradiated rat spinal cord lesion after OEC transplantation (Smith *et al.*, 2001b). However, the authors did not present direct evidence that these transcription factors were expressed by grafted OECs, as opposed, for example, to invading host Schwann cells. Other studies (Barnett *et al.*, 1993; Devon and Doucette, 1995; Doucette and Devon, 1994, 1995) indicate that OECs have similar, but not identical, requirements to Schwann cells for expression of a myelinating phenotype. So methodological differences, such as the age of animals from which the OECs were obtained, the method of purification, etc. might have had an impact on the results rather than the actual myelinating ability of the OECs. Moreover, there is little known yet about factors regulating the myelinating phenotype in OECs and whether all OECs even possess the ability to myelinate.

Oligodendrocytes are known to synthesize significant levels of myelin-associated molecules (i.e. Gal-C and MBP) *in vitro* even in the absence of axonal contact, provided the growth medium contains insulin, hydrocortisone, and triiodothyronine (Lopes-Cardozo *et al.*, 1989; McMorris *et al.*, 1986; Mozell and McMorris, 1991; Poduslo *et al.*, 1990; Ved *et al.*, 1989). When similar growth medium is used to feed neuron-free cultures of OECs, these glial cells are not induced to express either Gal-C or MBP (Barnett *et al.*, 1993; Doucette and Devon 1994, 1995). These data suggest that the promoter region of the MBP gene is under different regulatory control in oligodendrocytes and OECs (Doucette and Devon, 1994, 1995). The MBP promoter is also believed to be controlled differently in oligodendrocytes and Schwann cells (Foran and Peterson, 1992; Gow *et al.*, 1992). Gal-C expression by OECs in neuron-free cultures can, however, be induced by increasing the intracellular level of cAMP (Doucette and Devon, 1994, 1995), which at the same time fails to induce these cells to express MBP (Doucette and Devon, 1995).

1.4 Multiple Sclerosis

MS is a relatively common and always disabling disease of the CNS primarily affecting young adults (20 to 40 years of age) that exhibits extraordinary clinical, radiological, and pathological heterogeneity. The MS Society of Canada estimates that 50,000 Canadians have MS, with this disease affecting 1-2 people per 1000 population (MS Society of Canada website; October, 2004; http://www.mssociety.ca).

1.4.1 Neuropathology and clinical course

MS is characterized by ongoing lesion formation in the CNS and increasing cumulative damage and is now also recognized as a disease with on-going pathological processes in most patients most of the time. Autopsy studies indicate that there are individuals who never develop clinical symptoms, in spite of having neuropathologic changes typical of MS. Although no accurate data have been collected yet to estimate subclinical disease, it has been suggested that asymptomatic MS might account for up to 25% of all cases (Joy and Johnston, 2001). The majority of MS patients (~85%) develop neurological signs and symptoms over a period of several days, reach a plateau, and then show some improvement followed by a period of clinical quiescence (i.e. remission) that is variable in length but is rarely permanent (Wingerchuk et al., 2001). New attacks (i.e. relapses) typically consist of one or a combination of symptoms including optic neuritis, axial or limb parasthesias with neck flexion (Lhermitte's sign), weakness, gait ataxia, diplopia, and symptomatic worsening with increases in body temperature (Uhthoff symptom). The relapses of disabling symptoms repeat themselves over and over, interspersed with periods of remission; these patients are referred to as having "relapsing-remitting MS". As the disease progresses, the frequency and duration of the relapses often increase while that of the remissions usually decreases until they no longer experience any periods of remission; at this point in the progression of the disease these patients are considered to have switched from a "relapsing-remitting" to a "secondary progressive" form of MS (Prineas et al., 2001).

For the remaining 15% of MS patients, the disease presents first and foremost with a gradually progressive loss of neurological function that presents with no intervening periods of remission. This form of MS is referred to as "primary progressive". The appearance of, and

worsening of, the clinical exacerbations in all MS patients are usually accompanied by inflammatory infiltrates and demyelination in the brain and spinal cord white matter giving rise to the pathological hallmark of the disease, the MS plaque. Patients with primary progressive MS tend to be older. Although women are twice as likely as men to develop relapsing-remitting MS, men are as likely to develop primary progressive MS as are women.

An MS plaque is a circumscribed area of demyelination within the brain or spinal cord within which there is a relative preservation of axons. Although plaques may occur anywhere within the white matter, the periventricular area of the cerebral hemispheres, optic nerve, brain stem, cerebellum and spinal cord are among the more frequent sites (Wingerchuk et al., 2001). Varying proportions of immune cells and immunoreactive substances (i.e. immunoglobulins) populate plaques, which can be classified as being active, inactive or chronic. Active plaques contain leukocytes that have invaded the CNS from the vascular system, resident activated macrophages/microglia, and reactive astrocytes located at the periphery of the plaque (Gutowski et al., 1999). Schwann cells may be present inside plaques, but if present are only found inside the astrocyte-poor central area of the lesion (Itoyama et al., 1983). The number of plaques gradually increases over time as the disease progresses resulting in a gradual increase in the number of demyelinated axons. Demyelination and axonal pathology are important contributors to the neurological signs and symptoms of MS (Bjartmar et al., 2001; Ferguson et al., 1997; Nowacki et al, 2000; Peterson et al., 2001). Although it has been assumed axons and gray matter are spared in MS, recent observations indicate that MS is a destructive process leading to axonal damage (Trapp et al., 1998) and brain atrophy (Jeffery et al., 2000; Sailer et al., 2003), even in spite of current anti-inflammatory or immunosuppressive therapies. Progressive neurological deterioration in MS patients has also been correlated with atrophy in the spinal cord (Bot et al., 2004; Losseff et al., 1996), cerebellum (Davie et al., 1995), and cerebral cortex (Losseff et al., 1996). The onset of the symptoms of MS is often associated with breakdown of the blood-brain barrier, which can be visualized by gadolinium-enhanced lesions on magnetic resonance imaging (McFarland et al., 1992; Rudick et al., 1996; Werring et al., 2000).

Depending on cellular composition and on the relative balance between demyelination and remyelination, MS plaques can be divided into several different types (Lassmann *et al.*, 2001; Lucchinetti *et al.*, 1996, 1999; 2000). Type I and II lesions are characterized by the

presence of a prominent inflammatory reaction, which is likely mediated in part by tumor necrosis factor- α (TNF- α) and by the release of reactive oxygen species by T lymphocytes and macrophages (Lucchinetti et al., 1996, 1999; 2000). The deposition of antibodies and complement is seen in Type II lesions but these are absent from lesions of Type I. Antibodies against various myelin proteins have been identified in cerebrospinal fluid (CSF) and in sera obtained from MS patients, although their role in the pathogenesis is not clear. For example, anti-MOG antibodies have been identified both in serum and in CSF of some MS patients (Reindl et al., 1999). Hypoxic damage in oligodendrocytes is predominant in type III lesions (Aboul-Enein et al., 2003; Lassmann, 2003), which are characterized by ill-defined borders, rims of spared myelin surrounding inflamed vessels, and preferential loss of MAG. Other myelin proteins (PLP, MBP, CNP) are still present within active lesions. The Type III lesion is considered to be active because activated macrophages filled with myelin degeneration products are present. Marked degeneration of oligodendrocytes, a minor inflammatory reaction and extensive tissue destruction are present at the Type IV lesion. Similar to that of Type I and II lesions, the perivenous and radial plaque growth and simultaneous loss of all myelin protein are also present in Type IV lesions. Various degrees of remyelination are seen in both Type I and II lesions, but not in either Type III or Type IV lesions.

Depending on a number of factors such as location of lesions, disease activity, medication and genetic susceptibility, axonal loss reaches a certain threshold and becomes clinically obvious at various time-points. In longstanding lesions, an increased number of inflammatory attacks axonal injury can be substantial and results in considerable irreversible neurological impairment and disability of MS patients.

This irreversible disability that MS patients exhibit has been attributed to axonal dysfunction and loss (Davie *et al.*, 1995; De Stefano *et al.*, 1998). Acute axonal injury has been demonstrated in early lesion formation and is most pronounced in actively demyelinating plaques in part due to the inflammatory edema (Bitsch *et al.*, 2000; Ferguson *et al.*, 1997; Kornek *et al.*, 2000; Trapp *et al.*, 1998). Axonal injury begins at disease onset and remains clinically silent for many years. Irreversible neurological disability develops when axonal loss reaches its threshold and compensatory resources of the CNS tissue are exhausted. Accumulation of axonal amyloid precursor protein (APP), a marker of axonal injury or

dysfunction, was found in active plaques and at the border of the chronic MS lesions (Ferguson *et al.*, 1997).

1.4.2 Theories on etiology of MS

The primary cause of MS remains unknown. Many investigators generally accept the concept that the immune system contributes to the destruction in multiple sclerosis and CD4 positive T cells initiate an inflammatory cascade that leads to demyelination. However, it has not yet been determined what causes MS. Therefore, researchers investigate different hypotheses such as environmental (Agranoff and Goldberg, 1974; Bach, 2002; Kurtzke, 1980), viral (Challoner *et al.*, 1995; Friedman *et al.*, 1999; Gilden, 1999; Kazmierski *et al.*, 2004; Lucchinetti *et al.*, 2000), genetic (Ebers *et al.*, 1995; Sadovnick *et al.*, 1993; Weinshenker *et al.*, 1998), vascular (Aboul-Enei *et al.*, 2003; Lassman *et al.*, 2001) and inflammatory (Bo *et al.*, 1994; Bonetti and Raine, 1997; Dowling *et al.*, 1996; Hofman *et al.*, 1989; Merrill *et al.*, 1993; Trapp *et al.*, 1999). A majority of researchers believe that a viral agent directly or indirectly triggers inflammatory and autoimmune cascades which initiate MS pathogenesis.

Epidemiological findings support an environmental hypothesis since the prevalence of MS generally increases with distance from the equator, and thus the disease occurs preferentially in areas that receive a relatively low amount of ultraviolet radiation from sunlight (Agranoff and Goldberg, 1974; Kurtzke, 1980). As a consequence, low vitamin D production is detected in the population of these areas and especially in MS patients (Bach, 2002). Other environmental factors suspected in predisposing people to the development of MS include infections (Challoner et al., 1995; Friedman et al., 1999; Gilden, 1999; Kazmierski, 2004; Lucchinetti et al., 2000) and ecological factors (Lauer, 1997, reviewed in Wingerchuk et al., 2001), such as climate, soil conditions, and diet. It has been hypothesized that some infections including human herpes virus 6 (Challoner et al., 1995; Friedman et al., 1999, Lucchinetti et al., 2000; Knox, 2000; Goodman, 2003), Chlamydia pneumonia (Gilden, 1999; Sriram et al., 1999), and influenza virus (Kazmierski, 2004) are causative agents of MS. The genetic hypothesis is based on the evidence of a higher prevalence of MS in monozygotic twins as opposed to heterozygotic siblings (Sadovnick et al., 1993). Many ethnic groups, including North American Indians, showed reduced incidence of MSdespite them living in regions with a high prevalence for this disease. Many patients with MS were tested for human leukocyte antigen (HLA) DR2, as a more susceptible locus on chromosome 6 (Ebers *et al.*, 1995).

Although HLA polymorphism is not considered to be a significant contributor, genetic factors may still be important determinants of the disease course (Weinshenker et al., 1998). Polymorphisms in the interleukin (IL)-1β and IL-1β receptor antagonist genes (Schrijver et al., 1999), the apolipoprotein E gene (Evangelou et al., 1999), and immunoglobulin (Ig) G Fc receptor genes (Myhr et al., 1999) have been also associated with MS course. MS patients with a combination of IL-1ß and IL-1ß receptor antagonist polymorphisms are found to have a higher rate of progression on the Expanded Disability Status Scale when compared with other possible combinations (Schrijver et al., 1999). At the apolipoprotein E gene, the ɛ4 allelle is found to be more common in MS patients with more rapid progression of disability in MS (Evangelou et al., 1999). Authors suggest that apolipoprotein E is not implicated in the early pathogenesis of MS; however, patients possessing the ɛ4 allelle have a reduced capacity for neuronal remodeling after relapses. Although the frequencies of two IgG Fc receptors IIA and IIIB do not differ between the MS patients and the controls, MS patients homozygous for the IgG Fc receptor IIIB neutrophil antigen 1 allele have more benign course of disease than that of patients heterozygous or homozygous for the IgG Fc receptor IIIB neutrophil antigen 2 allele (Myhr et al., 1999).

There are an increasing number of studies reporting the role of an immunomediated vasculitis leading to MS plaque development (Aboul-Enei *et al.*, 2003; Lassman *et al.*, 2001). Activated T-cells and macrophages damage endothelial cells causing microvessel thrombosis with secondary ischemic damage of white matter. Activated microglia secrete nitric oxide (Bo *et al.*, 1994; Merrill *et al.*, 1993) and cytokines TNF- α (Hofman *et al.*, 1989) and lymph toxin (Selmaj *et al.*, 1991), which were found to selectively damage oligodendrocytes *in vitro* and in some MS lesions (Bonetti and Raine, 1997; Dowling *et al.*, 1996; Trapp *et al.*, 1999). Matrix metalloproteases (MMPs), a family of proteolytic enzymes including collagenasese, gelatinases, stromelysins, membrane-type metalloproteinases, matrilysin and metalloelastase, are known to be involved in remodeling of extracellular matrix of many tissues (Birkedal-Hansen *et al.*, 1993). They have been implicated in degradation of vascular basement membrane thereby facilitating leukocyte migration into inflammatory sites (Yong *et al.*, 1998). The authors focused their attention on MMP-9 (gelatinase B) because it is known to degrade type IV collagen and to disrupt cell-matrix contacts. The subsequent loss of integrin signaling

leads to oligodendrocyte apoptosis (Chandler *et al.*, 1997). MMP-9 has been shown to degrade myelin basic protein of the myelin sheath (Chandler *et al.*, 1995; Gijbels *et al.*, 1993).

1.4.3 Host glial cell response

The MS lesions have been characterized by the composition of demyelination, oligodendrocyte pathology, the inflammatory infiltrate, astrogliosis and axonal damage (Kornek and Lassmann, 2003; Lassmann *et al.*, 1998). However, it remains unclear, which of the processes develop primarily in response to still unknown cause and which develop secondarily as a consequence of, or as a reaction to, an initiating event. The pathology of MS is associated with immunological and inflammatory processes that result in complex changes in gene expression by astrocytes and microglia in the vicinity of the lesion, in the recruitment of a number of cell types (e.g. leucocytes) into the lesion (Adamson *et al.*, 1999; Al-Omaishi *et al.*, 1999; Asensio *et al.*, 1999), and in the alteration of the extracellular matrix of the plaque (Gutowski *et al.*, 1999; Opdenakker, 1997). Chemokines, which are expressed in response to CNS inflammation, are known to induce a chemotactic migration of microglial cells (Badie *et al.*, 1999; Cross and Woodroofe, 1999; Yamagami *et al.*, 1999). INF- γ and TNF- α are the most studied inflammatory cytokines and both molecules are expressed in MS lesions (Wingerchuk *et al.*, 2001).

Heterogeneity of pathological changes in MS plaques has been previously documented (Lucchinetti *et al.*, 2000). However, one of the constant features of all plaques is the presence of microglial and astrocytic reactivity. A very recent study of patients with relapsing and remitting MS, who died during or shortly after the onset of a relapse, showed extensive oligodendrocyte apoptosis, as well as microglial and astroglial activation, in myelinated tissue in which there were few or no lymphocytes or myelin phagocytes (Barnett, 2004).

Microglia along with T cells infiltrate injured CNS tissue and play an important role in the development and progression of active MS lesions. Sriram and Rodrigez (1997) implicated macrophage/microglia as principal players in the demyelination process due to activation of these cells and release of the factors that are toxic to the oligodendrocytes and myelin. This model was supported by the findings of Chiang *et al.* (1996), who described macrophage/microglia-mediated demyelination induced by low-level chronic secretion of IL-3. The macrophages/microglia within active plaques are filled with myelin degradation products, whereas inactive plaques contain far fewer cells and the macrophages/macroglia that are

present do not normally contain myelin debris. Demyelination is believed to be occurring at the site of active plaques, whereas inactive plaques are most likely areas where some degree of tissue repair either has occurred or is actively taking place. The demyelinated lesions in patients with chronic MS have well-demarcated areas with very few cells visualized with Luxol Fast Blue staining, varying degrees of axonal loss, and persistent but minor inflammation; these lesions are referred to as chronic plaques. In recent years a harmful role of macrophage/microglia has been questioned. Diemel et al., (1998) and Loughlin et al., (1997) suggested that macrophages can be beneficial for MS lesion and that they also may promote recovery from demyelination by the phagocytosis of myelin debris and the release of trophic factors. Macrophages were found to stimulate myelinogenesis (Hamilton and Rome, 1994) as well as remyelination (Cuzner et al., 1994; Loughlin et al., 1997) in cell cultures. There are still questions also to whether macrophages found in MS lesions are agents of destruction or contributors to recovery. Bartnik et al. (2000) showed that macrophages promoted myelination in embryonic spinal cord aggregate cultures at normal and low levels of oxidative stress and on the other hand were part of the cause of neuronal damage when the levels of oxidative stress were higher.

A network of reactive astrocytes can be found in the majority of MS lesions. It is believed that astrocytes are activated by pro-inflammatory cytokines (Selmaj *et al.*, 1990) and such an astrocytic reaction is not specific for MS. Although the role of astrocytic activation is not yet clear, Rosen *et al.* (1989) suggested that astrocyte sclerosis might serve as a barrier for regeneration. Other studies showed that astrocytes may be a source of growth factors, including vascular endothelial growth factor and fibroblast growth factor (FGF) 2, and could therefore support regenerative processes (Horner and Palmer, 2003; Ridet *et al.*, 1997). Reactive astrocytes present inside chronic MS plaques produce molecules such as tenascin-R and –C, which are believed to impede remyelination and axonal repair by contributing to the formation of a glial scar (Gutowski *et al.*, 1999).

1.4.4 Current treatment

The treatment of MS patients is very complex and best accomplished with a multidisciplinary approach involving a neurologist, ophtalmologist, psychiatrist, urologist, gastroenterologist, nurse, physical therapist, rehabilitation specialist, and social worker. There is no cure for MS at the present time, because its cause remains

unknown. Therefore, the treatment of patients with MS is very challenging and complex. The patients with MS often develop complications such as fatigue, depression, sleep disorders, spasticity, pain, decubitus ulcers, speech and swallowing disorders, mood and cognition disorders, and others. Such complications usually require attention of a psychiatrist, physiotherapist and rehabilitation specialist.

Clinically significant acute relapses are currently being treated with the use of corticosteroids and plasma exchange to hasten recovery. Although no consensus about the optimal form, dose, route and duration of corticosteroid therapy has been accepted, corticosteroids such as methylprednisolone and prednisone have been used to reduce the rate of clinical relapses, delay progression of disability and reduce CNS lesion activity as evidenced by magnetic resonance imaging (MRI; Beck, 1992, 1995). Plasma exchange is found to enhance recovery of relapse-related neurological deficits in MS patients who show no response to high-dose corticosteroids (Weinshenker *et al.*, 1999).

Phase 3 trials indicate that treatment with interferon derivatives, such as INF β -1a (Rebif[®]) may delay the development of clinical and MRI evidence of a second diagnosis-defining bout after a first episode of acute neurological syndromes suggestive of MS (Jacobs et al., 2000). The rate of clinical relapses, the development of new lesions and increase in volume of existing lesions on MRI scans are also reduced by approximately 30 percent with administration of INF β -1b (Betaseron[®], Berlex Laboratories) (Paty and Li, 1993; The IFNB Multiple Sclerosis Study Group, 1995) and INF β-1a (Avonex[®], Biogen; Rudick *et al.*, 1997). INF β-1a and glatiramer acetate, a mixture of synthetic polypeptides containing glutamic acid, alanin, lysine, and tyrosine, (Copaxone[®], Teva Pharmaceutical Industries) were found to reduce the frequency of relapse (Johnson *et al.*, 1995; Johnson *et al.*, 1998). In addition, INF β -1a may also delay the progression of disability in MS patients with minor disability and a relapsing form of MS (Simon et al., 1998; Jacobs et al., 1996; Rudick et al., 1997). The specific mechanisms of action of interferons in MS are not completely understood. These agents are believed to exert their immunomodulating activities by reducing both the proliferation of T cells and the production of TNF- α , by decreasing antigen presentation, by altering cytokine production and favoring ones governed by type 2 helper T (Th) cells, by increasing the secretion of IL-10, and by reducing the passage of

immune cells across the blood-brain barrier by means of their effects on adhesion molecules, chemokines, and proteases (Rudick *et al.*, 1997). Glatiramer acetate is found to promote the proliferation of Th2 expressed cytokines, compete with MBP for presentation on MHC class II molecules, thereby inhibiting antigen-specific T-cell activation, altering the function of macrophages, and inducing antigen-specific suppressor T-cells (Bornstein *et al.*, 1987). An elevation of serum IL-10 levels, suppression of the pro-inflammatory cytokine TNF α mRNA, and elevation of the antiinflammatory cytokines TGF- β and IL-4 mRNAs in leukocytes were observed in a 12 month study conducted by Miller *et al.* (1998). They suggested that the beneficial clinical effects of glatiramer acetate in MS patients may be attributed to changes in activation of T cells and a shift from a Th1 to a Th2/Th3 cytokine profile, probably leading to glatiramer acetate-driven mechanisms of bystander suppression.

1.5 Glial Cell Reaction to Injury

Following an acute insult to the CNS, a complex of morphological and metabolic changes takes place within the nervous system with the goal to repair the damaged tissue (Deckert-Schluter *et al.*, 1995; Drescher *et al.*, 1997; Griffin *et al.*, 1992; Kreutzberg, 1996; Raivich *et al.*, 1996). The injured neurons undergo crucial steps in a fight for their survival and stimulate nearby neuroglia including microglia and astrocytes for support (Kreutzberg, 1996; Raivich *et al.*, 1995). The activation of microglia (Kreutzberg, 1996) and astrocytes (Eddleston and Mucke, 1993) has been described as a graded, stereotypic response, which is readily observed in stroke and ischemia, in neurodegenerative diseases, after direct or indirect axonal injury (Norenberg, 1994) or during inflammation due to infections or autoimmune disease (Perry *et al.*, 1995). The production of proinflammatory cytokines, functional changes in brain vascular endothelia and a recruitment of cells of the immune system into the damage tissue has been also identified after acute nervous system injury.

1.5.1 Astrocytes

The astrocytes are considered to be the predominant neuroglial cell of the CNS and exist in two typical forms stellar-fibrillary astrocytes and protoplasmatic astrocytes. The former cells are normally located in the white matter and bear long slender processes, which stain with GFAP, a typical astrocyte cytoskeletal protein (Bignami *et al.*, 1972). The

protoplasmic astrocytes, on the other hand, are residents of grey matter, exhibit numerous short and highly branched processes and only a few of these astrocytes express GFAP. Astrocytes provide metabolically optimized nutrients to the highly energy-consuming neurons and maintain ionic balance in the extracellular space (reviewed in Tsacopoulos and Magistretti, 1996). In addition, astrocytes induce tight junctions on endothelial cells (Janzer and Raff, 1987; Pekny *et al.*, 1998; Tao-Cheng *et al.*, 1987), induce the ramified phenotype in microglia and blood-derived monocytes (Kloss *et al.*, 1997; Schmidtmayer *et al.*, 1994) and provide numerous trophic factors including PDGF (Hermanson *et al.*, 1995), NGF, IGF-1, TGF- α for adjacent neurons and oligodendrocytes, particularly following injury (Eddleston and Mucke, 1993; Ridet *et al.*, 1997).

In response to an injury that occurred some distance away, astrocytes became activated and increase of their synthesis of GFAP within 24 hours for example after facial nerve axotomy (Tetzlaff et al., 1988). Reactive astrocytes in the adult brain appear to recover the capacity to express vimentin (Pixley and De Vellis, 1984; Schiffer et al., 1988; Takamiya et al., 1988), which normally is expressed only during embryogenesis and during the first two postnatal weeks in rodents (Pixley and De Vellis, 1984; Schnitzer et al., 1981; Voigt, 1989). Reactive astrocytes create a physical barrier between damaged and healthy cells. Beginning 2 weeks after axonal injury, the reactive astrocytes gradually replace microglia from the neuronal surface and begin to enwrap the perikarya of injured neurons with thin, flat cytoplasmic processes (Graeber and Kreutzberg, 1988; Guntinas-Lichius et al., 1997). Processes of adjacent astrocytes adhere to each other and form a multilayered stack of astrocytic lamellae, which could act as a small glia scar surrounding the axotomized neuron. Physical damage, ischemia or inflammation will lead to disruption of the blood brain barrier, collateral activation of adjacent microglia and a much stronger response of astrocytes, which will proliferate and form a massive scar (Balasingam et al., 1994; Giulian and Lachman, 1985). A dense network of hypertrophic astrocytes with thick, interdigitating processes and associated extracellular matrix compose the glial scars (Brodkey et al., 1995). It is commonly believed that the cell surface of astrocytes and the extracellular matrices in which they are embedded are more likely to hinder rather than facilitate the growth of axons in vivo in the adult mammalian CNS (Fraher, 1999; Reier et al., 1986; Reier and Houle, 1988; Ridet et al., 1997). The components of the extracellular matrix such as chondroitin sulfate proteoglycans (Giftochristos and David,

1988; Lips *et al.*, 1995; McKeon *et al.*, 1991), tenascin (Laywell *et al.*, 1992; Lips *et al.*, 1995; McKeon *et al.*, 1991) and collagen IV (Stichel *et al.*, 1999) have been found to inhibit neurite outgrowth.

However, astrocytes are not always nonpermissive for axonal growth. In fact, there appear to be differences in the ability of astrocytes to support axonal growth. This can depend on the degree of differentiation of the cells (Baehr and Bunge, 1989; Hatten *et al.*, 1991; Lucius *et al.*, 1996), on whether the astrocytes have formed a two- or three-dimensional substratum for the growing axons (Fawcett *et al.*, 1989), on the region of the CNS that is injured (Alonso and Privat, 1993a, 1993b; Chauvet *et al.*, 1998; Prieto *et al.*, 2000), and on the presence or absence of macrophages (David *et al.*, 1990), Schwann cells (Berry *et al.*, 1992; Dezawa *et al.*, 1999) or OECs (Li *et al.*, 1997, 1998; Ramon-Cueto *et al.*, 1998, 2000).

1.5.2 Microglia

Microglia are thought to be derived from infiltrating hematopoietic or mesodermal cells during the early development of the CNS (Eglitis and Mezey, 1997; Hickey and Kimura, 1988; Hickey *et al.*, 1991, 1992; Kurz and Christ, 1998; Ling and Wong, 1993). The normal, resting microglia have long, ramified processes oriented parallel to the nerve fibers in the white matter, and display a stellate morphology in the gray matter (Compston *et al.*, 1997). Microglia appear to participate in the immune surveillance of the nervous system. During embryonic and early postnatal (up to postnatal day 21) rat development, microglia express the complement type 3 receptor and the MHC class 1 antigen, which can be recognized by monoclonal antibodies OX-42 and OX-18, respectively (Domaradzka-Pytel *et al.*, 1999). After P21, microglia within the normal CNS no longer express the MHC class 1 antigen but remain OX-42 positive. However, microglia will re-express complement type 3 receptor in response to neural damage (Graeber *et al.*, 1988).

Microglia respond to neural damage by undergoing characteristic changes, which involve cellular and morphological events, such as increased cell size and number of activated cells, as well as the expression of cell surface molecules (i.e. MHC class 1 antigen; Graeber *et al.*, 1988), cytokines (i.e. IL-1 β , TNF- α ; Bartholdi and Schwab, 1997) and growth factors (i.e. NGF, neurotrophin-3, ciliary neurotrophic factor; Eddleston and Mucke, 1993; Elkabes *et al.*, 1996; Mallat *et al.*, 1989). Following axotomy of the facial nerve, resting microglia become activated and then proliferate and tightly ensheath injured motoneurons with enlarged cytoplasmic processes (Blinzinger and Kreutzberg, 1968). If an injured neuron survives and regrows its axon, reactive microglia may revert back to the resting form. If neurons die, microglia change their form from process bearing to rounded (ameboid) macrophages, which remove the dead cells (Streit and Kreutzberg, 1988). Activated microglia develop enlarged cell processes, which give the cells a bushy appearance. Cellular hypertrophy is concomitant with upregulation of complement type 3 receptor expression (Graeber *et al.*, 1988) and both are apparent by 24 hours after injury. Microglia begin to proliferate 2-3 days after injury and their numbers reach maximum levels after 4-7 days. Following an epidural application of kainic acid over the rat cerebral hemispheres, Kaur and Ling (1992) observed extensive neuronal degeneration, which elicited an increased expression of the complement type 3 receptor (marked by OX-42) and induced MHC 1 expression (marked by OX-18) in microglia. The elevated expression of the former receptor may be related to the phagocytic activity of these reactive microglia, whereas re-expression of the latter antigen was suggested to facilitate the activated microglial cells interaction with T lymphocytes as part of an immune response.

Streit *et al.* (1999) suggest that activation of microglia shortly after injury may serve to limit secondary damage to neurons, whereas chronically activated microglia may take an active role in neuron destruction. Microglial activation after acute CNS injury is described as a transient and self-limited event and it usually subsides within about one month after either traumatic or ischemic insults (Gehrmann et al., 1991; Morioka et al., 1991; Koshinaga and Whittemore, 1995). An increasing number of studies have focused on the expression of microglia-derived cytokines and growth factors following CNS injury. The mRNA levels of microglia-derived cytokines and growth factors fluctuate depending on the injury type. The expression of proinflammatory cytokines, such as IL-1 β and TNF- α , is minimally upregulated following axotomy of the facial nerve (Streit et al., 1998), whereas a sharp increase in the expression of these cytokines is seen immediately after spinal cord injury (Bartholdi and Schwab, 1997). The upregulation of proinflammatory cytokine mRNAs is observed for a very short duration, with the levels returning to baseline levels one or two days after injury (Bartholdi and Schwab, 1997; Buttini et al., 1994; Streit et al., 1998; Taupin et al., 1993; Woodroofe et al., 1991). Streit et al. (1999) suggested that this transient increase in IL-1ß and TNF- α expression is tightly controlled and these cytokines may function in triggering secondary events, including the production of neurotrophins, which may be beneficial to the

subsequent repair and regeneration process (Fagan and Gage, 1990; Lindholm *et al.*, 1987). In contrast, prolonged increases in the production of proinflammatory cytokines (i.e. TNF- α ; Probert and Akassoglou, 2001) such as occurs in chronic conditions such as MS may be detrimental to neural repair and cause secondary, irreversible damage to nervous tissue.

1.5.3 Inflammation

Reactive gliosis is a universal event after CNS injury and is a prominent pathological feature in MS where reactive astrocytes constitute the major cell type in demyelinating plaques (Raine, 1984). After injury and/or during inflammation, TNF- α and IL-6 (Selmaj *et al.*, 1990) induce rapid proliferation and hypertrophy of astrocytes, which in turn secrete numerous humeral factors (e.g. TNF- α , prostaglandins, and TGF- β ; Hatten *et al.*, 1991; Nieto-Sampedro *et al.*, 1983; Reier and Houle, 1988). Therefore, secreted products of activated inflammatory cells contribute to the reactive gliosis found in white matter diseases of the CNS.

MS lesions are commonly characterized by the presence of inflammatory infiltrates consisting of lymphocytes and macrophages/microglia. Oligodendrocytes and the myelin sheaths they produce can be destroyed by different immunological mechanisms, such as direct interaction with cytotoxic T-cells and T-cell cytokines (Selmaj et al., 1990), demyelinating antibodies (Linington *et al.*, 1988; Raine, 1984), and cytotoxic cytokines (i.e. TNF- α ; Probert and Akassoglou, 2001). It has been suggested that inflammation in MS is induced by the migration of activated T cells through the blood-brain barrier (Conlon et al., 1999). INF-y and TNF- α , which are released as part of the inflammatory response, induce the capillary endothelial cells to express vascular cell adhesion molecule (V-CAM) and class II molecules of the MHC (Cannella and Raine, 1995). Activated T cells express integrins, particularly alpha-4 integrin, and CD4, a member of the immunoglobulin supergene family, which bind to V-CAM and MHC class II molecules, respectively. T cells express very late antigen (VLA)-4, which enables them to penetrate the endothelium. VLA-4 was found on T cells that collected around veins and capillaries and formed perivascular lymphocyte cuffs in acute MS lesions. The blockade of VLA-4 with cyclic peptides resulted in a reversion of clinical paralysis in EAE animals (Dutta *et al.*, 2000). Once outside the blood vessels, the activated lymphocytes expressed alpha 1 integrin, which binds to collagen type IV in the extracellular matrix. T cells here began to secrete enzymes such as MMPs, some of which, MMP-2 and MMP-9, also called gelatinase A and B, specifically degrade collagen type IV (Brosnan and Raine, 1996; Conlon et

al., 1999; Gijbels *et al.*, 1994). In this way activated lymphocytes gain access to the CNS where they then induce an inflammatory reaction in the white matter, creating a situation that is potentially very detrimental to CNS myelin and oligodendrocytes.

In the neuropil, activated Th1 cells may stimulate local microglia and hematopoetic macrophages, through the secretion of cytokines such as TNF- α , INF- γ , lymphotoxin- α (also known as TNF- β) and IL-12. These in turn damage myelin and cause a further release of potential antigens. These cytokines are expressed in actively demyelinating MS lesions and elevated levels of TNF- α correlate with disease activity and blood-brain barrier damage (Bitsch et al., 2000). However, TNF- α was recently shown to also protect neurons (Akassoglou et al., 1997; Liu et al., 1998; Probert et al., 1995), to promote proliferation of oligodendrocyte progenitors and to facilitate their remyelination of CNS axons (Arnett et al., 2001). Lymphotoxin- α and TNF- α then stimulate macrophages, microglia and astrocytes to produce nitric oxide and osteopontin (Steinman, 2001). Merrill et al. (1993) suggested that nitric oxide, one of the free radicals that in combination with antibodies, complement and TNF- α (Selmaj et al., 1991) mediate damage to oligodendrocytes by macroglia, plays a role in lesion formation in MS. Osteopontin, an extracellular glycosylated bone phosphoprotein, further induces Th1 cytokine production, including INF- γ and IL-12, while downregulating Th2 cytokines such as IL-10. Th1 cytokines are found during MS exacerbation whereas Th2 cytokines may reduce the size of MS lesions (Bielekova et al., 2000; Chabas et al., 2001; Kappos et al, 2000; Steinman, 2001).

CD4+ helper cells are present mainly in perivascular areas of the CNS of MS patients, whereas CD8+ cells are more prominent within the MS plaques and, therefore, are thought to be more directly involved in the tissue destruction than are the CD4+ cell population (Babbe *et al.*, 2000; McCallum *et al.*, 1987; Steinman, 2001; Woodroofe *et al.*, 1986). The number of CD8+ cells also correlates better with the extent of acute axonal injury than does the number of CD4+ cells (Bitsch *et al.*, 2000).

A number of pro- and anti-inflammatory cytokines have been identified in the serum and CSF of MS patients. The expression of some of these markers, such as TNF- α (Bertolotto *et al.*, 1999; Rieckmann *et al.*, 1998), as well as molecules like adhesion molecules (Kraus *et al.*, 2002; Rieckmann *et al.*, 1998), MMPs (Waubant *et al.*, 1999), chemokine receptors (Strunk *et al.*, 2000) and macrophage-related protein (e.g. MRP8/14; Bogumil *et al.*, 1998), was shown

to correlate with clinical course and lesion activity identified with MRI. However, none of these markers are specific for MS and the results of different studies remain controversial and have not been adopted for use yet in clinical practice (Bertolotto *et al.*, 1999, Bitsch *et al.*, 1998, Carrieri *et al.*, 1998; Strunk *et al.*, 2000; van Oosten *et al.*, 1998).

1.5.4 Degeneration

The primary injury caused by direct mechanical trauma to the brain or spinal cord encompasses the focal destruction of neural tissue. The initial insult usually also results in a progressive wave of secondary injury, which via a complex of harmful pathophysiological mechanisms exacerbates the injury to CNS tissue and causes both necrosis and apoptosis. Following either brain or spinal cord injury, the extracellular glutamate concentration rises continuously leading, especially in an hypoxic environment (Choi, 1993), to overstimulation of ionotropic α -amino-3hydroxy-5-methyl-4-isoxazoleppropionic acid (AMPA)/kainate receptors and triggering excitotoxic cell death (Doble, 1999). Precisely what factors expressed by dying neurons trigger activation of microglia remains poorly characterized. Damaged axons may release adenosine triphophate (ATP), which could activate microglial cells via purinoreceptors. The increase of ATP in culture medium was found to induce microglia membrane currents and transmembrane Ca²⁺ fluxes (Haas et al., 1996; Illes et al., 1996; Kettenmann et al., 1993; Walz et al., 1993). High levels of extracellular potassium could also be responsible for mediating certain aspects of glial cell activation. Gehrmann et al. (1993) described a transient potassium chloride-induced microglial activation, which was accompanied by short-lasting hypertrophy and increased OX-42 expression, but not proliferation of microglia.

In both the acute and the chronic phases of MS, demyelination and axonal degeneration have both been correlated with the severity of disabilities such as paralysis and the inability to walk (Losseff *et al.*, 1996; Trapp *et al.*, 1998). During the inflammatory reaction in both MS and in EAE, lymphocytes, microglia and macrophages produce excessive amounts of glutamate, which in turn activates AMPA/kainate receptors present on oligodendrocytes and neurons (reviewed in Steinman *et al.*, 2002). Glutamate via its receptors induces calcium influx and causes necrotic damage to oligodendrocytes and axons. In the EAE animal model, blocking the AMPA/kainate with quinoixalin derivatives (NBQX and MBQX) provided some protection to the oligodendrocytes and axons from immune-mediated damage, although the immune response to myelin antigens was not altered (Pitt *et al.*, 2000; Smith *et al.*, 2000). It

has been suggested that cytotoxic CD8+ T cells might mediate axonal damage in inflammatory lesions of MS tissue (Babbe *et al.*, 2000), in EAE mice (Huseby *et al.*, 2001), and *in vitro* (Medana *et al.*, 2001). Within MS plaques, axons are particularly vulnerable to degeneration when exposed to nitric oxide (Smith *et al.*, 2001a). In addition, inflammatory edema increases extracellular pressure contributing to axonal damage especially in CNS anatomical locations such as the spinal cord where room for tissue expansion is limited (Shi and Blight, 1996). Wujek *et al.*, (2002) noted there was a 9% increase in the volume of the spinal cord during the first attack in mice with relapse-remitting EAE.

1.6 Animal Models for Studying Demyelination and Remyelination

Many animal models have been developed to study the biology and pathology of demyelination and remyelination in the CNS (Steinman, 1999), since it is not feasible to perform experimental manipulations in humans. These animal experiments provide investigators with the opportunity to study the morphological, cellular and molecular mechanisms involved in CNS demyelination and remyelination and are designed to mimic some, but not all, of the pathological and neurological sequelae of MS. The available animal models fall into the following categories: (1) neurotoxicant-induced demyelination, including EtBr (Yajima and Suzuki, 1979), lysolecithin (Hall, 1972) and cuprizone; (Blakemore, 1973), (2) autoimmune-induced demyelination, including EAE (Raine and Traugott, 1984); and (3) viral-induced demyelination, including corona-virus (Herndon *et al.*, 1977) and Theiler's murine encephalomyelitis virus (TMEV) (Rodriguez *et al.*, 1988).

1.6.1 Ethidium bromide

EtBr is a nucleic acid-binding neurotoxicant that has often been used to induce demyelination because of its effectiveness in killing CNS glial cells, but not axons, when injected into white matter. This nucleic acid chelating agent reacts with the nuclei of those cells residing in the vicinity of the injection (Blakemore, 1984; Yajima and Suzuki, 1979). When injected in the dorsal column of the spinal cord, EtBr demyelinates axons of the dorsal column sensory pathways by killing all the glial cells, including the oligodendrocytes, astrocytes and microglia, in the vicinity of the injection. However, axons are spared from the toxic effects because their cell bodies lie elsewhere, specifically within the dorsal root ganglia situated some distance from the site of EtBr injection. Within three days, the EtBr-induced

demyelinating lesion is composed of naked, demyelinated axons, and a total absence of glial cells. During lesion repair, invading macrophages scatter among the demyelinated axons and around blood vessels, and cluster around disintegrating myelin sheaths to clean up the debris. In addition, oligodendrocyte progenitor cells, astrocytes and Schwann cells all migrate into the lesion site from the surrounding tissue to assist in neural repair. The oligodendrocytes contribute to the remyelination of the demyelinated axons in the central part of such EtBr lesions, while Schwann cells remyelinate axons in the subpial areas and astrocyte-free regions (Felts and Smith, 1996; Yajima and Suzuki, 1979). Blakemore and Patterson (1978) were the first to expose the animal's spinal cord to x-irradiation prior to injection of EtBr, using 40 Grays of x-irradiation to suppress the spontaneous repair of the lesion by host glial progenitor cells. Such a lesion provides an excellent environment in which the myelinogenic properties of grafted myelin-forming cells can be tested (Akiyama *et al.*, 2001, 2002a, 2002b; Blakemore and Crang, 1985; Franceschini and Barnett, 1996; Franklin *et al.*, 1996; Groves *et al.*, 1993).

1.6.2 Lysolecithin

Another neurotoxicant-induced animal model of demyelination is one using lysolecithin (lysophosphatidyl choline), which was first used by Hall and Gregson (1971) for tissue in the PNS. Numerous investigators have subsequently injected lysolecithin into the CNS to permeabilize the plasma membranes of cells (Blakemore *et al.*, 1977; Blakemore, 1978; Waxman *et al.*, 1979). Due to its permeabilizing property lysolecithin affects oligodendrocytes, thus resulting in demyelination in the immediate vicinity of the injection site (Blakemore, 1978).

Woodruff and Franklin (1999) compared lesions induced by either EtBr or lysolecithin and found that a significant proportion of axons remained demyelinated for a longer time after an EtBr injection than after a lysolecithin-induced demyelination. Although, EtBr-induced demyelinated lesions remyelinated more slowly than after lysolecithin injection, exposure to the latter toxin did cause axonal damage.

1.6.3 Cuprizone

The neurotoxicant cuprizone, bis–cyclohexanone-oxaldihydrazone, is a copper chelator, causes copper deficiency resulting in liver and brain tissue damage (Blakemore, 1972, 1973; Ludwin, 1978; Suzuki and Kikkawa, 1969). Rats, guinea pigs, and hamsters acquired a severe spongiform encephalopathy after being on an 8 week cuprizone (0.2%) diet (Carey and

Freeman, 1983). In Wister male rats, only oligodendrocyte perturbation and intramyelinic edema were observed (Love, 1988). However, after a short 4 or 6 week exposure to a diet containing cuprizone C57BL mice developed acute demyelination. This occurred as a result of the apoptotic death and depletion of mature oligodendrocytes within the corpus callosum (Armstrong *et al.*, 2002; Arnett *et al.*, 2001; Hiremath *et al.*, 1998; Mason *et al.*, 2000, 2001; Morell *et al.*, 1998) and the superior cerebellar peduncle (Komoly *et al.*, 1987; Stidworthy *et al.*, 2003). Cuprizone treatment has been also used as a model for investigating the damage to oligodendrocytes in Swiss, CD1, and ICI mice (Bakker and Ludwin, 1987; Blakemore, 1974, Ludwin, 1978). Oligodendrocytes are considered to be the main CNS target of cuprizone because copper is required in high levels for proper mitochondrial function and oligodendrocytes are much more metabolically active (Hiremath *et al.*, 1998; Morell *et al.*, 1998) than other cell types in the central nervous system (Blakemore, 1973; Cammer and Zhang, 1993). When cuprizone was used in SJL mice before inducing EAE by PLP peptide administration, it was found to inhibit both T-cell activation and function and to attenuate the severity of the EAE (Emerson *et al.*, 2001).

The cuprizone-induced demyelination is presumably mediated by microglia/macrophages (Hiremath *et al.*, 1998) since the blood-brain barrier remains intact and there is no lymphocyte-induced inflammatory reaction (Bakker and Ludwin, 1987; Kondo *et al.*, 1987; McMahon *et al.*, 2002). After cuprizone removal from the diet, oligodendrocyte progenitors start to proliferate and migrate from the subventricular zone. They accumulate within the demyelinating lesions and differentiate into mature oligodendrocytes, which in turn remyelinate the demyelinated axons (Mason *et al.*, 2000). However, prolonged maintenance of C57BL mice on the cuprizone diet resulted in the failure of remyelination (Mason *et al.*, 2004). Newly regenerated mature oligodendrocytes along with the resident progenitors become progressively depleted within the chronically demyelinated corpus callosum.

1.6.4 Experimental autoimmune encephalomyelitis

The basis for the development of models of acute EAE emerged from "paralytic accidents", which were noted by Stuart and Krikorian (1928), Rivers *et al.* (1935), Kabat *et al.* (1947) after rabies immunizations in which spinal cord inflammation led to clinical paralysis (reviewed by Steinman, 1999). An animal model of acute disseminated EAE by Rivers (1935) was developed in monkeys once immunization against brain tissue was identified as the cause

of the demyelination and paralysis (reviewed by Steinman, 1999). A mixture of brain material with complete Freund's adjuvant, an emulsion made from killed mycobacteria tuberculosis, was used by Kabat *et al.* (1947) to produce highly reproducible white matter inflammation, demyelination and acute paralysis (reviewed by Steinman, 1999).

Although the actual mechanisms responsible for the extensive CNS demyelination in MS remain to be elucidated, it is believed the pathology is related to autoimmunological phenomena induced by viral infection or by exposure to an unknown allergen (see Section 1.5.2). Once sensitized to an antigen present on CNS (but not PNS) myelin, lymphocytes enter the neuropil and induce macrophages to engulf CNS myelin. Myelin breakdown products further stimulate the immune response in the form of a positive feedback loop.

The presence of autoimmune antibodies against myelin components in the sera of MS patients lends support to the use of EAE as an animal model for MS (Raine *et al.*, 1974, 1977; Steinman, 1999). Experimental animals can be immunized with MBP (Raine *et al.*, 1974, 1977), or with a synthetic peptide analog of MBP that contains only the essential amino acids (Fallis *et al.*, 1989), with PLP, or with MOG (Steinman, 1999). Then, adoptive transfer of T-cells sensitized to myelin antigens is required to induce demyelination in the CNS. When induced in the guinea pig (Raine *et al.*, 1974, 1977; Stone and Lerner, 1965; Traugott *et al.*, 1982) or mouse (Brown and McFarlin, 1981; Lublin *et al.*, 1981), EAE presents the neurological features of chronic CNS demyelination. Experimental animals with EAE display neurological signs such as weight loss, mouth wetting, some urinary and fecal incontinence, and mild hindlimb paraparesis. Incomplete recovery and progressive worsening follow after each relapse of demyelination until the animals often eventually develop quadriparesis or spastic paraparesis (Raine and Traugott, 1983).

These clinical features are the result of plaques of chronic demyelination scattered throughout the white matter. T cells, immunoglobulin, and complement play a role in the demyelinating pathogenesis in EAE. T cells reactive to myelin and antibodies to MBP, MOG, and PLP can be detected in the CSF and in demyelinated areas from brains of animals with EAE (Steinman, 1996). Adhesion molecules, cytokines, chemokines, HLA molecules, and metalloproteases are also critical participants in the development of inflammatory lesions (Steinman, 1999). Treatment with antibodies to α -integrin, CD4+ve T cells reactive to myelin proteins, and to the cytokines IL-6 and TNF- α have been found to reverse paralysis in EAE

(Conlon *et al.*, 1999). Targeting of these molecules is now in use in clinical trials on MS patients (Steinman, 1999). As a consequence of these studies, Copaxone, a synthetic polymer analog of MBP (Arnon, 1996) and INF- β (Ruuls *et al.*, 1996) were found to be effective in reducing the relapse rate in MS. On the other hand, INF- γ , also a critical cytokine tested in clinical trials, was found to exacerbate disease, and a monoclonal antibody to TNF- α and a soluble TNF- α receptor also the worsened clinical symptoms in MS patients (Steinman, 1999).

Although the EAE animal model gives rise to clinical signs comparable to those seen in MS, such as demyelination and a sparing of astrocytes and microglia, EAE-induced lesions are numerous and scattered all over the CNS with no experimental control over their location and size. In order to control where and how big the demyelinated lesions are, it is preferable to inject a single dose of EtBr, which results in the demyelination of the axons without inducing Wallerian degeneration.

1.6.5 Theiler's virus

Theiler's murine encephalomyelitis virus is a single-stranded RNA virus that belongs to the *Piconaviride* family and was first described by Theiler in 1937 (reviewed in Oleszak *et al.*, 1995). This virus has been reported to cause neurological and enteric diseases in susceptible strains of mice, such as SJL (Dal Canto *et al.*, 1996; Lipton and Jelachich, 1997; Monteyne *et al.*, 1997; Oleszak *et al.*, 1995; Tsunoda and Fujinami, 1996). After intracranial inoculation with the Theiler's virus DA strain, these mice develop a biphasic disease. Acute disease occurs within 3 to 12 days postinfection followed by late chronic demyelinating disease developing at 30 to 40 days postinfection. Later, the animals die being unable to completely clear the virus from their bodies (reviewed by Dal Canto *et al.*, 1996; Monteyne *et al.*, 1997; Oleszak *et al.*, 1995; Tsunoda and Fujinami, 1996). In contrast, in resistant strains of mice, such as C57BL/6, the animals develop only the early acute phase of the disease without progression into late chronic demyelinating disease and these mice also clear the virus completely from their bodies by about 3 weeks postinfection (Lipton, 1975; Lorch *et al.*, 1981).

Both susceptible and resistant strains of mice develop acute polioencephalitis with destruction of neurons to a variable degree, extensive mononuclear cell and lymphocyte infiltration in the CNS (reviewed by Begolka *et al.*, 1998; Drescher *et al.*, 1997; Oleszak *et al.*, 1995; Rodriguez *et al.*, 1988). Variably intense and multifocal demyelinating lesions of the cerebral hemisphere and spinal cord white/grey matter, mononuclear cell infiltrates in the

spinal cord, progressive spinal cord atrophy and axonal loss were observed in DA-induced susceptible mice 30 to 40 days after inoculation. The inflammatory areas were concentrated in the regions of the diencephalon, the hippocampus, the basal ganglia, the anterior horn of the spinal cord and leptomeninges with sparing of the white matter during the early acute phase of the disease. The monocytes and T lymphocytes infiltrated small and medium-sized parenchymal blood vessels forming perivascular cuffings and foci of ischemic-type coagulative necrosis. During the late chronic stage, myelin loss, myelin-laden phagocytic macrophages, axonal swellings, perivascular and leptomeningeal infiltrates were detected within the demyelinating lesions, which involved the spinal cord anterior, lateral, and dorsal columns with preferential susceptibility of the thoracic segments.

1.7 Cell Transplantation: A Potential Therapeutic Approach to Treating Multiple Sclerosis

Remyelination by transplanted myelin-forming cells has been explored extensively in animal experiments as a possible approach for achieving repair in demyelinating diseases. There are several different cell types for such transplantation, including oligodendrocytes and their progenitors, Schwann cells, OECs, bone marrow cells, neuronal stem cells, tanycytes and enteric glia. Franklin and Barnett (2000) reviewed the properties of the ideal cell for transplantation including an ability to remyelinate CNS axons, availability of cell source, migratory and survival capabilities, compatibilities with astrocytes, and potential vulnerability of the myelin produced by the cell candidate.

Oligodendrocytes are normally responsible for CNS myelination and are the cells that are typically targeted by the different approaches being used to induce demyelination in animal models. Therefore, the ability of oligodendrocytes in these animal models to sufficiently respond to demyelination is severely compromised. This leaves oligodendrocyte progenitor cells and stem cells as the main glial contributors of the CNS to the remyelination of the axons. There is a wealth of experimental evidence demonstrating the production of new myelin in animals after transplantation of oligodendrocyte lineage cells (Barnett *et al.*, 1993; Duncan *et al.*, 1992; Franklin and Blakemore, 1997; Groves *et al.*, 1993; Warrington *et al.*, 1993). Human oligodendrocyte progenitors have also been shown to myelinate congenitally dysmyelinated axons in the mouse brain (Windrem *et al.*, 2004). Remyelination by transplanted cells was also

accompanied by both improved conduction (Utzschneider *et al.*, 1994) and functional recovery (Jeffery *et al.*, 1999). Therefore, oligodendrocyte progenitors would be the most suitable for cell transplantation in MS. However, the isolation of oligodendrocytes from neural tissue is a complex process, which not only yields low cell numbers but also lacks broad ethical approval. Although Penderis *et al.* (2003) found no depletion of oligodendrocyte progenitors after repeated episodes of focal EtBr-induced demyelination in the rat caudal cerebellar peduncle, several authors have suggested that repeated demyelination in MS results in the depletion of oligodendrocyte progenitors (Niehaus *el al.*, 2000; Wolswijk, 2002). Furthermore, the ongoing immunological process directed against myelin and oligodendrocytes might attack the transplanted cells.

Schwann cells normally myelinate only PNS axons; however, it has been known for many years that in some conditions these cells can and will remyelinate axons within the confines of the CNS (Blakemore and Crang, 1985, 1989; Blakemore et al., 1995). In addition to experimental data, Schwann cells contribute to the spontaneous remyelination in MS (Ghatak et al., 1973; Itoyama et al., 1983, 1985). Although Schwann cells myelinate only a single axon in contrast to oligodendrocytes, it has been shown that transplantation of Schwann cells into demyelinated spinal cord can restore normal conduction properties of CNS axons (Honmou et al., 1996). In order for these cells to contribute to remyelination, unlike mature oligodendrocytes, the Schwann cells must migrate from the PNS towards the demyelinated area or be grafted into the lesion. The Schwann cell migration towards the demyelinated area was observed only when astrocytes were absent from a lysolecithin-induced lesion in irradiated rat spinal cord (Harrison, 1985). Similar to what is found in animal models, Schwann cells can sometimes be identified inside the astrocyte-poor central area of MS plaques (Itoyama et al., 1983). Unlike oligodendrocytes, human Schwann cells are readily accessible from the sciatic nerve, where peripheral myelin is intact in MS patients. Schwann cells can be expanded in vitro (Rutkowski et al., 1995; Van den Berg et al., 1995) and used after cryopreservation (Kohama et al., 2001). Therefore autologous Schwann cells could be harvested from an MS patient, expanded under appropriate conditions in vitro and injected directly into demyelinated lesions. The advantages of using Schwann cells are the accessibility of these myelinating cells, the possibility of autologous transplantation without the need of subsequent immunosupression, and the potential for avoidance of immune attacks against CNS myelin in MS. However,

Schwann cells are not a normal cellular component of the undamaged CNS. These cells migrate into areas of the damaged CNS, where they remyelinate axons only in astrocyte depleted areas of the CNS (Blakemore, 1975). Another potential problem for the use of Schwann cells is the inhibitory effect of astrocytes on Schwann cell remyelination of CNS axons (Franklin and Blakemore, 1993; Harrison, 1985; Woodruff and Franklin, 1999). In addition, MS plaques are disseminated throughout the CNS neuropil, thus Schwann cells would be required to migrate to many lesions. These cells have shown limited migratory ability when injected into an animal's spinal cord in the presence of a toxin-induced demyelinated lesion (see Section 1.8.2).

Bone marrow cells (BMCs) have recently also been considered as a potential candidate for transplantation as part of a therapeutic approach in the treatment of demyelinating diseases. BMCs are known to be a source of circulating erythrocytes, platelets, monocytes, granulocytes, and lymphocytes (Phinney et al., 1999). BMCs can be obtained from an individual's bone marrow and, therefore, they have been under the scope of researchers as a potential candidate for autologous cell therapies. Several studies have used BMCs isolated from rat femoral bones for transplantation into the demyelinated (x-irradiation/EtBr) spinal cord of adult rats (Akiyama et al., 2002a; Sasaki et al., 2001). Sasaki et al. (2001) observed predominantly a peripheral pattern of myelination, reminiscent of that of Schwann cells, after BMC transplantation, whereas Akiyama et al. (2002a) reported both peripheral and central types of remyelination of CNS axons, with a concomitant improvement in conduction velocity. Although xirradiation/EtBr-induced demyelination has been reported to lead to delayed endogenous remyelination (Blakemore and Crang, 1985; Honmou et al., 1996), there is a possibility that host Schwann cells were recruited from outside the lesion area as a result of performing the injection procedure of BMCs. Thus, host Schwann cells could have produced the peripheral type of remyelination seen in rats that received BMC transplant. This interpretation would be consistent with BMCs secreting various growth factors that could have created an environment suitable for Schwann cell migration and remyelination. Admittedly, BMCs have been also reported to differentiate into non-hematopoietic and non-lymphopoietic cells (Caplan, 1991; Owen, 1988; Prockop, 1997) and recently also to produce neurons and astrocytes when grown in cell culture (Woodbury et al., 2000). In vivo experiments have shown that BMCs possess the capacity to differentiate into astrocytes when transplanted into normal (Azizi et al., 1998) and

ischemic (Eglitis et al., 1999) brain or lead to neuronal differentiation when injected intravenously (Brazelton *et al.*, 2000; Mezey and Chandross, 2000). Therefore, the fate of transplanted BMCs requires further study to investigate further possible lineages these cells might take in their response to a new environment.

Tanycytes present in the mediobasal hypothalamus and pituicytes from the neurohypophysis were found to express a distinctive set of immunological markers in common with OECs and nonmyelinating Schwann cells, namely p75NTR, O4 antigen, estrogen receptor-α type, and IGF-1 (Gudino-Cabrera and Nieto-Sampedro, 2000). Gudino-Cabrera and Nieto-Sampedro (2000) suggested that OECs, tanycytes, and pituicytes share a common phenotype and are three members of a family of Schwann-like macroglia of the CNS. They referred to these three glial cells as aldynoglia, a term derived from a Greek word meaning "to make grow". It appears that tanycytes and pituicytes, like OECs, provide a more hospitable microenvironment for growing axons in the CNS, thus making them ideal candidates for glial cell transplantation into areas of CNS injury. Tanycytes have been shown to myelinate DRG neurites *in vitro* (Gudino-Cabrera and Nieto-Sampedro, 2000), and also promote the growth of CNS axons after grafting into the CNS of adult rats (Prieto *et al.*, 2000). Although tanycytes and pituicytes have been shown to promote axonal regeneration and myelination of DRG neurites, the harvesting procedure of these cells in humans is complicated and would require a challenging neurosurgical manipulation.

Dogiel (1899) was the first to describe the glial cells of the enteric nervous system (Brehmer *et al.*, 1999). Enteric glia (EG) have been shown to facilitate the regeneration of dorsal root axons through the normally non-permissive environment of the dorsal root entry zone (DREZ) into the spinal cord and as well as an incomplete functional recovery of the cutaneous trunci muscle reflex after injury (Jiang *et al.*, 2003). However, EG can only be obtained by means of an abdominal surgical procedure, which can lead to unnecessary postsurgical complications.

Neuronal stem cells (NSCs) are multipotent cells, which have been identified in the developing and adult CNS (Gage *et al.*, 1995; Johe *et al.*, 1996; Lois and Alvarez-Buylla, 1993; Palmer *et al.*, 1997; Reynolds and Weiss, 1992). Grafts of normal mouse clonal NSCs injected into the brain ventricles on newborn shiverer mice have been shown to effectively reverse the pathology and improve some symptoms of the mutant strain (Yandava *et al.*, 1999). The
researchers found that most of the transplanted cells migrated through the brain and differentiated into oligodendrocytes. The injection of NSCs derived from wild type mouse either intravenously or into the cerebral ventricles of C57/BL mice with MOG-induced EAE resulted in a marked increase in the numbers of oligodendrocyte progenitors within demyelinated areas, a decreased extent of demyelination and axonal loss, and functional recovery of the mice receiving such transplants (Pluchino *et al.*, 2003). Transplantation of human NSCs into demyelinated rat spinal cord resulted in axonal remyelination similar to that of Schwann cells (i.e. PNS-type myelin) and improvement of conduction velocities of the remyelinated axons (Akiyama *et al.*, 2001).

NSCs can be isolated from the adult human brain only during a neurosurgical procedure, such as collection of a biopsy (Kukekov *et al.*, 1999). Addition of growth factors such as EGF and basic fibroblast growth factor (bFGF) to culture medium is required to expand these cells *in vitro* (Cattaneo and McKay, 1990; Gensburger *et al.*, 1987; Hammang *et al.*, 1997; Kalyani *et al.*, 1997; Svendsen *et al.*, 1996). Mouse NSCs have been shown to differentiate into both neurons (Brustle *et al.*, 1995; Stemple and Anderson, 1992; Svendsen *et al.*, 1996) and oligodendrocytes (Hammang *et al.*, 1997; Milward *et al.*, 1997) when transplanted into the embryonic and neonatal CNS but only into astrocytes when grafted into the adult CNS (Lundberg and Bjorklund, 1996). Therefore, these cells differentiate along the different cell lineage depending on environmental signals. Fetal tissue is considered to be the best current tissue source for human neural stem cells; however, unresolved ethical issues are a major concern in future clinical applications. The other important issue is that stem cells are highly proliferative and thus also bear the risk of tumor formation (Bjorklund *et al.*, 2002). Although stem cell research has a lot of questions to be answered, the potential of such an approach should not be neglected.

OECs are the glial cells that ensheath the olfactory axons of the first cranial nerve (Doucette, 1984, 1990, 1995; Ramon-Cueto and Avila, 1998; Ramon-Cueto and Valverde, 1995). Although *in vivo* OECs express only a nonmyelinating phenotype, Devon and Doucette (1992, 1995) demonstrated their myelinating ability *in vitro*, which was independently confirmed *in vivo* by several other research groups (Franklin *et al.*, 1996; Imaizumi *et al.*, 1998; Li et al., 1997, 1998; Lakatos *et al.*, 2003). The regenerative effort of most types of axons within the CNS fails to support more than a minimal amount of growth (Kiernan, 1979; Reier

58

and Houle, 1988; Fawcett, 1998), but olfactory axons are usually, in the absence of experimental intervention, successful at reinnervating the olfactory bulb (Graziadei and Monti Graziadei, 1979; Barber and Lindsay, 1982; Doucette *et al.*, 1983; Doucette, 1990). It appears the growth of olfactory and non-olfactory axons is favored by a permissive OEC environment (Doucette, 1990, 1995; Ramon-Cueto and Valverde, 1995; Franklin and Barnett, 1997; Fawcett, 1998; Ramon-Cueto and Avila, 1998).

A number of studies over the past several years have demonstrated the ability of OECs to support the growth of non-olfactory CNS axons *in vivo* (Lustgarten *et al.*, 1991; Ramon-Cueto and Nieto-Sampedro, 1994; Smale *et al.*, 1996; Li *et al.*, 1997, 1998; Perez-Bouza *et al.*, 1998; Ramon-Cueto *et al.*, 1998; Navarro *et al.*, 1999). OECs were shown to facilitate the regeneration of primary afferent axons after rhizotomy across the DREZ of the spinal cord (Ramon-Cueto and Nieto-Sampedro, 1994; Navaro *et al.*, 1999), cholinergic fibers of the fimbria/fornix pathway across a cavity formed after aspiration of a large portion of the fiber pathway (Smale *et al.*, 1996), lesioned corticospinal tract axons (Li *et al.*, 1997, 1998; Nash *et al.*, 2002) and after spinal cord transection injury (Ramon-Cueto *et al.*, 1998, 2000). Therefore, OECs have been receiving increased attention as possible candidates for neural repair through the grafting of myelinating glia.

Cellular interactions within demyelinating CNS lesions are complex and ultimately determine whether oligodendrocytes or Schwann cells will assume primary responsibility for remyelinating the axons. Astrocytes are one of the key players in determining which glial cell type will remyelinate axons in each demyelinated area of the CNS (Blakemore *et al.*, 2003). Cells of the oligodendrocyte lineage are given such preference in the presence of astrocytes. However, it is unlikely that any one glial cell type can correct all of the problems in injured CNS tissue. A cell that can adopt several different roles is needed. OECs were demonstrated to possess this feature, being able to switch their phenotype from that resembling an astrocyte to one more like that of a myelinating Schwann cell (Devon and Doucette, 1992, 1995; Doucette, 1995). Due to their highly malleable phenotype, OECs will very likely be able to combine the roles of astrocytes and Schwann cells when transplanted into a lesion cavity (Doucette, 1995). Even if OECs may not promote axonal growth better than Schwann cells, or any other potential candidate, OECs might differ from Schwann cells in restoring the glia limitans and blood brain barrier and allowing maximal re-entry of axons into the host tissue.

59

Schwann cells fail to integrate into the CNS, which in part is due to astrocytic perturbation of the neuron-Schwann cell interaction and a consequent lack of ensheathment and myelination (Blakemore and Crang, 1989; Franklin *et al.*, 1992; Guenard *et al.*, 1994). The migratory ability of Schwann cells within the CNS parenchyma is also limited, being mostly confined to the vicinity of the graft when they are transplanted into the cerebral hemisphere even of newborn shiverer mice (Baron-Van Evercooren *et al.*, 1992). Eventually, grafted Schwann cells are usually extruded from the CNS, even after remyelinating axons in the spinal cord of mice are exposed to an intracisternal injection of cholera toxin B (Jasmin *et al.*, 2000). Several authors suggested that astrocytes inhibit the migration of Schwann cells within the parenchyma of the CNS (Fawcett and Asher, 1999; Franklin and Blakemore, 1993). Schwann cells and astrocytes have been shown to avoid one another when they are co-cultured in the same culture dish. Astrocytes either grown together with Schwann cells *in vitro* (Lakatos *et al.*, 2000) or after transplantation of the latter cells into the adult rat spinal cord were induced to increase the expression of axon growth inhibitory chondroitin sulfate proteoglycans (Lakatos *et al.*, 2003).

The repair of the CNS damage in patients with demyelinating disorders remains a major challenge for physicians and neuroscientists. Therapeutic cell transplantation has shown potential in promoting CNS repair. However, what cell or combination of different cells is the best for transplantation? The cell candidate should be of a nature that avoids the need for immunosupression and this can be achieved if the cells were obtained from the patient. Other important issues are the cell source, the approach for cell harvesting with minimal damage to the patient, the efficiency of both the harvesting technique and the tissue processing to yield a large enough number of cells. The chosen cell has to have the ability to form a myelin sheath around demyelinated axons, even in the presence of activated host cells (i.e. astrocytes and microglia) and even within an inflammatory environment. In addition, the transplanted cells will be required to migrate to multiple demyelinated areas as seen in MS and remyelinate axons at least in those located in the vitally important parts of the CNS like brainstem.

1.8 Migration of Myelinating Glia Within Central Nervous System

If glial cell transplantation is to be considered a realistic therapy for the repair of demyelinated CNS lesions in disorders such as MS, a number of important questions have to be

resolved, among which is the ability of each glial candidate to migrate following implantation into the CNS. This is a very important issue in MS since transplanted cells will be required not only to grow and myelinate within the extent of a single demyelinated area but also to migrate between at least several plaques so that a minimal number of cell injections will be needed to repair multiple lesions.

1.8.1 Oligodendrocyte Migration

The migratory abilities of oligodendrocytes and their progenitors have been studied extensively both *in vitro* (Armstrong *et al.*, 1990; Milner *et al.*, 1997) and *in vivo* (Fruttiger *et al.*, 1999; Milner and Ffrench -Constant, 1994; Milner, 1997; Niehaus *et al.*, 1999; Osterhout *et al.*, 1999). The oligodendrocyte progenitors have been shown to be highly motile (Gansmuller *et al.*, 1991; Small *et al.*, 1987). Growth factors such as PDGF (Armstrong *et al.*, 1990; Fruttiger *et al.*, 1999; Osterhout *et al.*, 1999) and bFGF (Milner *et al.*, 1997) promote oligodendrocyte progenitor migration either *in vitro* (Armstrong *et al.*, 1990; Milner *et al.*, 1997) or *in vivo* (Fruttiger *et al.*, 1999; Osterhout *et al.*, 1999). In contrast to oligodendrocyte progenitors, more mature oligodendrocytes bear multiple processes, which probably impede the migratory capabilities of these cells. Oligodendrocyte maturation correlates with downregulation of integrin expression (i.e. $\alpha \vee \beta 1$, $\alpha \vee \beta 3$, $\alpha \vee \beta 5$ and $\alpha \vee \beta 8$) by oligodendrocytes, receptors known to play an important role in cell migration (Milner and ffrench-Constant, 1994; Milner, 1997). Endogenous oligodendrocytes have been observed migrating from the subventricular zone toward the lysolecithin-induced lesion in the corpus callosum of adult rat (Nait-Oumesmar *et al.*, 1999).

Transplantation of cells of the oligodendrocyte lineage has also shown that progenitor cells migrate further *in vivo* than do the more mature cells (Warrington *et al.*, 1993; Archer *et al.*, 1997). When grafted into uninjured CNS tissue, oligodendrocyte progenitor cells migrated for up to 3-4 mm in the rostral-caudal direction from the injection site in the spinal cord of the myelin deficient newborn rat (Tontsch *et al.*, 1994) and for 20 mm along the dorsal columns of Shiverer rat pups (Archer *et al.*, 1997). When CG4 cells (oligodendrocyte progenitor cell line) were transplanted into the normal rat spinal cord, their migration and survival was found to be limited to areas of the injection site (Franklin *et al.*, 1996). In contrast, following transplantation into the x-irradiated spinal cord of adult rats, CG4 cells migrated from the grafting site (i.e. ventral funiculi) through both white and grey matter towards the x-

irradiation/EtBr-induced demyelinated lesion in the dorsal funiculi (Franklin *et al.*, 1996). Baron-Van Evercooren *et al.* (1996) observed migration of transplanted oligodendrocyte progenitors from the grafting site towards a lysolecithin-induced demyelinated lesion in adult wild type and Shiverer mouse spinal cords. Migrating cells were identified along the meninges, perivascular space, as well as the outer surface of myelin sheaths in CNS wild type and Shiverer mouse white matter. In summary, it has been shown that the more immature cells of the oligodendrocyte lineage migrate the best within the CNS are.

1.8.2 Schwann Cell Migration

Schwann cells have been reported to migrate into the CNS and myelinate CNS axons in a number of developmental and pathological studies (reviewed in Franklin and Blakemore, 1993). Blakemore (1975) observed Schwann cells migrating along perivascular spaces and reaching demyelinated axons after an injection of lysolecithin into the spinal cord of adult rats and after the consequential destruction of the glia limitans in this area of the CNS. The migratory ability of Schwann cells grafted into the cerebral hemisphere of newborn Shiverer mice is mostly confined to the vicinity of the graft (Baron-Van Evercooren et al., 1992). Schwann cells from an immortalized cell line (MSC80) but not Schwann cells that were obtained from the sciatic nerves of newborn rats migrated towards a focal lysolecithin-induced demyelinated lesion of the Shiverer mouse spinal cord (Baron-Van Evercooren et al., 1992, 1993). The migrating MSC80 cells used the ependyma, meninges, and blood vessels as the main migratory routes rather than white matter tracts. Limited migration of Schwann cells along perivascular routes and intrafascicularly within the fimbria was also noted after grafting primary Schwann cells into the hippocampus (Brook et al., 1993; Raisman et al., 1993). These Schwann cells extended long fiber-like cytoplasmic processes that extended singly amongst and parallel to the orientation of the host glial cells and the axons in the fimbria (Raisman et al., 1993). Although the transplanted Schwann cells induced a transient but marked hypertrophy of host fimbrial astrocytes, the migration of the donor Schwann cells did not appear to be impeded. In other studies describing Schwann cell migration, blood vessels (and the glia limitans) were also found to be favored routes of migration (Blakemore, 1984; Blakemore et al., 1986; Langford and Owens, 1990). Laminin, a glycoprotein constituent of blood vessel basal lamina, has been reported to facilitate migration of Schwann cells in vitro (McCarthy et al., 1983). Iwashita and Blakemore (2000) created areas of persistent demyelination in the

dorsal funiculus of the rat thoracolumbar spinal cord by injecting EtBr into white matter exposed to x-irradiation. Their results indicated that significant numbers of transplanted Schwann cells were not attracted to migrate through normal tissue towards the demyelinated area (Iwashita and Blakemore, 2000). Schwann cells also show poor long-term survival after transplantation into either normal or x-irradiated adult white matter (Iwashita *et al.*, 2000). Fawcett and Asher (1999) suggest that astrocytes inhibit the migration of Schwann cells via the secretion of chondroitin sulphate proteoglycans.

1.8.3 Olfactory ensheathing cell migration

Several studies have reported that OECs have the ability to migrate within the microenvironment of the injured spinal cord when Hoechst prelabelled cells were grafted very close to the site of injury (Navarro et al., 1999; Ramon-Cueto and Nieto-Sampedro, 1994; Ramon-Cueto *et al.*, 1998, 2000). Transplanted neonatal OECs expressing β -Galactosidase (β -Gal) were found to migrate from the DREZ into the dorsal horn and adjacent dorsal funiculus (Riddell *et al.*, 2004). However, other reports presented evidence for little migration of virallyinfected OECs that were grafted directly into the cystic cavity of a spinal cord contusion injury (Boyd et al., 2004; Plant et al., 2003). These data may indicate that OECs migrate best within the injured CNS when grafted close to, but not within, the area of tissue injury. OECs may also prefer to stay within the lesion site when grafted directly into it and help the host tissue facilitate a more hospitable environment for axon regeneration. Takami et al. (2002) grafted virally infected OECs into the cystic cavity of the compression injury of the spinal cord in the adult rat but after 12 weeks found no β -Gal+ve OECs within or around the damaged area; therefore, this study raises the question as to whether the OECs actually survived being grafted into the lesion cavity. When mouse OECs derived from the lamina propria of the olfactory mucosa were grafted into the DRG of adult rats they migrated towards the DREZ of the spinal cord but did not cross into the CNS and also failed to promote the regeneration of dorsal root axons across the DREZ (Ramer et al., 2004).

In contrast to the limited migration of OECs observed in spinal cord injury, these cells have been reported to migrate within the large demyelinated (x-irradiation/EtBr) area induced in the dorsal funiculi of the adult rat spinal cord (Imaizumi *et al.*, 1998). OECs grafted into a focal electrolytically induced lesion of the corticospinal tract at the upper cervical segment of the rat spinal cord not only infiltrated the lesion area but also extended beyond it for at least 2-3

63

mm in the caudal direction (Li *et al.*, 1997, 1998). Interestingly, when cat OECs were transplanted into the medial longitudinal fasciculus of adult cats immediately caudal to a transection, Gudino-Cabrera *et al.* (2000) found the cells migrated retrogradely towards the abducens nucleus but failed to find any anatomical or functional evidence for the regeneration of the axonal projection from the abducens nucleus towards the oculomotor nucleus.

2. MATERIALS AND METHODS

2.1 Olfactory Ensheathing Cell Cultures

2.1.1 Dissection of embryos

Breeding colonies of adult Wistar rats were kept over one night in an animal room with a 12/12 hour light/dark cycle (Doucette, 1993). The next morning the females were checked for the presence of a vaginal plug; if present, this day was designated embryonic day 1 (E1). The pregnant rats were placed in a chamber filled with 5% CO₂ for 5 -10 min and then cervically dislocated. Using sterile techniques, the uterine horns of pregnant rats (E18) were exposed by transabdominal cesarean section, fetuses dissected from the surrounding tissue, and immersed in a room temperature bath of Hank's balanced salt solution (HBSS). To reduce developmental variation, only embryos at Theiler stage 23 (Theiler, 1972) of development were used for these experiments. Immediately after dissection, embryos were observed under a dissecting microscope in order to determine their developmental stage and separate the embryos at Theiler stage 23 from those of younger or older stages (1-2 embryos per litter). Four phenotypic features were used to identify embryos at Theiler stage 23 of development. The first feature was the presence of hair follicles over all the skin of the embryos and a second feature was the complete separation and divergence of the digits in both the forelimbs and the hind limbs. A third feature was an ear pinna that covered more than half of the external auditory meatus and the fourth feature was the presence of an umbilical hernia. The Theiler stage 23 embryos were decapitated, their heads transferred to a 1:1 mixture Dulbecco's Modified Eagle's Medium (DMEM; 4.5 g/L of glucose; Sigma-Aldrich Canada Ltd.) and F12 containing 10% fetal bovine serum (FBS; Hyclone, Logan, UT) and then split in half along the sagittal midline.

2.1.2 Disaggregation of tissue and setting up cell culture

The olfactory bulbs were dissected from embryo heads as described in Doucette (1993, 2001). Each head was transferred into a 60 mm Petri dish containing growth medium (DMEM/F12/10% FBS). Under a dissecting microscope, the heads were fixed upright with a pair of Dumont Forceps inserted into the tissue lying to either side of the nose. The skin,

subcutaneous tissue, and fibrous tissue of the cranium were cut between the presumptive cerebral hemispheres and cerebellum and then between the right and left cerebral hemispheres along the midline, beginning posteriorly and continuing as far forward as the presumptive nasal bones of the skull. The prongs of the Dumont Forceps were placed between the cerebral hemispheres and the cerebellum to fix the embryo's head while cutting it with the scalpel in the sagittal plane, thus completely separating each head into a right and left half. Using a pair of spring scissors, the olfactory nerves were cut along the ventral surface of the bulb. The exposed olfactory bulbs were dissected out of their position in the anterior cranial fossa by cutting the olfactory peduncle. The bulbs were collected in a 60 mm Petri dish, while the rest of the embryo's head was discarded. This dissection procedure for obtaining the olfactory bulbs was repeated for each fetus.

The ONL was dissected from each olfactory bulb as follows. The prongs of one pair of forceps were placed inside the bulb's cut end that previously had been attached to the olfactory peduncle. The free edge of the outer ONL was grasped with a second pair of forceps and gently peeled off the deeper layers of the bulb primordium. The ONLs were collected in a growth medium (DMEM/F12/10% FBS). This peeling procedure removed almost all of the ONL from the entire circumference of the bulb, with the exception of an occasional small tag of olfactory bulb tissue. The dissected ONLs were gently forced through a Nitex mesh (75 μ m pore size) to disaggregate the cells mechanically. The resulting cell suspension was diluted to a final volume of 40 ml in DMEM/F12/10% FBS. Five milliliters of the cell suspension were plated inside each of eight 25 cm² Falcon flasks. For immunostaining, 2 ml of the cell suspension were placed into each of ten 35 mm Falcon dishes with glass coverslips inside and 5 ml of the remaining cell suspension were plated inside each of four 25 cm² Falcon flasks. All cultures were incubated at 37°C in a humidified atmosphere of 5% CO₂ in air.

2.1.3 Purification and expansion of cell culture

Beginning with the first feed after 3-4 days *in vitro* (DIV), twice a week (for 2 weeks) the cultures were fed with a modification of the G5 medium (mG5) of Bottenstein (1985). The mG5 medium consisted of DMEM (4.5 g/l of glucose) as the basal medium, to which insulin (5 mg/ml), transferrin (50 mg/ml), selenium (5.2 ng/ml), hydrocortisone (3.63 ng/ml), and biotin (10 ng/ml) were added. Any meningeal fibroblasts contaminating the sparse primary cell cultures did not grow very well in this serum-free condition. In comparison to the G5 medium,

the mG5 medium contained neither epidermal growth factor nor fibroblast growth factor, which were found not to be mitogenic for fetal OECs (Doucette and Devon, 1995). After 10-14 days in mG5, the medium was changed to DMEM/F12/10% FBS and the cultures were fed with this medium until they approached confluency.

2.1.4 Harvesting cells and long-term storage of cells

Cell cultures were rinsed three times with Puck's balanced salt solution (PBSS) to remove all traces of serum. The cultures were then incubated in 0.025% trypsin/0.01% ethylenediaminetetraacetic acid (EDTA)/PBSS for 5 min at room temperature. Using the phase contrast microscope to monitor the progress of the trypsinization, the flasks were lightly tapped on the countertop several times. The trypsin solution was then triturated to assist in dislodging the cells. The resulting cell suspension was centrifuged at the relative centrifugal force of 160 G (800 RPM, for 10 min) and then resuspended in DMEM/F12/10% FBS. In order to determine the number of viable cells, 400 μ l of the cell suspension was stained with 0.3% Nigrosin dye. Dead or dying cells were stained purple-black by the Nigrosin dye, whereas living (i.e. viable) cells remained unstained. The number of viable (i.e. Nigrosin-excluding) cells were counted in four corners of an Improved Neubauer Levy Hemacytometer (Hausser Scientific Partnership, Horsham, PA, 19044). The number of viable cells per ml was calculated using the following formula:

 $\frac{\text{Total count}}{4} \ge 10 \ge 1000 \ge \frac{5}{4} = \text{number of viable cells/ml.}$

Using this count, the cell suspension was diluted in DMEM/F12/10% FBS to the desired cell concentration and 1 ml of the resulting cell suspension was placed into each of several Nalgene cryovials ($\approx 1 \times 10^6$ cells per 1 ml of medium). To each vial, 110 µl of dimethyl sulfoxide (DMSO) was added prior to sealing the cryovials and placing them into a NALGENE Cryo 1°C Freezing Container (Nalge Company, Rochester, N. Y., USA) containing isopropyl alcohol. The NALGENE Cryo Container was stored overnight in a -70°C freezer. The following morning, the cryovials were transferred to the vapor phase of liquid nitrogen in a Locator (JR Plus, Cryo Biological Storage System) for long-term storage.

2.1.5 Prelabelling cells with Dil

When the cell cultures approached confluency the cells were ready to be labelled by DiI. The cell cultures were first trypsinized, as described in Section 2.1.4, after which the resulting cell suspension was incubated in DMEM/F12/10% FBS containing 40 µg/ml of DiI for 1 hour at 37° C in a 5% CO₂ atmosphere. At the end of the 1 hour incubation the cell suspension was centrifuged at 160 G (for 10 min) and resuspended in DMEM/F12/10% FBS. Following a second centrifugation and resuspension in DMEM/F12/10% FBS, a cell count was performed to determine the number of viable cells (see Section 2.1.4) and the DiI-prelabelled OECs were frozen in aliquots of ~1x10⁶ cells per 1 ml, as described in Section 2.1.4.

2.2 Immunofluorescent Staining of Cell Cultures

Some OEC cultures initiated at the time of the initial dissection were plated into 35 mm dishes containing uncoated glass coverslips to assess the phenotype of the cells in the primary cultures. In addition, when some of the OEC primary cell cultures were close to being confluent (e.g. ~80%) they were trypsinized (see Section 2.1.4) and a cell suspension of OECs was plated into 35 mm dishes at a density of 5×10^4 cells per ml to examine the immunocytochemical phenotype of the cells. All cultures were incubated at 37°C in a humidified atmosphere of 5% CO₂ in air. The primary cultures were fed with mG5 medium for 14 days followed by DMEM/F12/10% FBS for an additional 14-21 DIV and harvested after a total of 28-35 DIV. The secondary cultures were fed with DMEM/F12/10% FBS for the entire 10-14 DIV prior to harvesting the cultures for staining.

2.2.1 Intracellular antigens

Immediately after the coverslips were removed from the dishes, the cells were fixed in either Zamboni's fix, consisting of 0.2% picric acid and 4% paraformaldehyde in 0.2 M phosphate buffer (pH 7.4), or in a mixture of 0.2% picric acid/4% paraformaldehyde/0.05% glutaraldehyde (PPG) in 0.2 M phosphate buffer solution (PBS, pH 7.4) for 15 min at room temperature. The Zamboni's fixative was used when the cell culture was to be stained with mouse anti-nestin; the PPG fixative was used when staining the cultures with the rabbit anti-S100β antibody. After fixation, the cell cultures were rinsed twice in 0.03M PBS (30 sec each at room temperature), followed by a third PBS wash for 30 min at 4°C and a fourth PBS rinse for 1 min at room temperature. The cells were then incubated for 5 min in a solution of 0.2% Triton X-100 (TX-100) in order to permeabilize the cell membrane and were then washed in PBS for 10 min. A 10 min blocking in 2% skim milk powder (SMP) was followed by a 30 min incubation at room temperature in a 1:1 mixture of the following antibodies: mouse anti-nestin (1:100) and

rabbit anti-S100β (1:200). These antibodies were diluted in 1% SMP/0.2% TX-100 in 0.03 M PBS. The control sections were incubated in diluent only.

The cell cultures were rinsed in 0.03 M PBS for 10 min. Appropriate FITC and Texas Red-conjugated secondary antisera were then applied in a 1:1 mixture on the cell cultures for a 30 min incubation at room temperature. The secondary antibodies included: swine anti-rabbit IgG FITC conjugated (1:100, Dimention Labs, Dako A/S, Denmark) and goat anti-mouse IgG Texas Red conjugated (1:100, Bio/Can Scientific, Mississauga, ON, Canada).

The cells were washed in PBS for 5 min before incubation in a Hoechst solution (diluted 1:50) for 15 min at room temperature. The cells were then given a final rinse in PBS for 5 min. The coverslips were mounted onto slides using Citifluor (Marivac Ltd, Halifax, NS, Canada) medium (a Glycerol/PBS solution). After excess Citifluor was removed from around the edges, the coverslips were sealed with clear nail polish.

2.2.2 Live method

The coverslips were removed from the culture dishes and incubated in DMEM medium containing 5% horse serum (HS) and 15 mM N-2-Hydroxyethylpiperazine-Ní-2-ethanesulfonic acid (HEPES) for 5 min at room temperature. The cells were then incubated for one hour in rabbit anti-mouse p75NGFR (1:100) diluted in DMEM/5% HS/15 mM HEPES. The control sections were incubated in diluent only.

The coverslips were washed four times (5 min each at 4°C) with Dulbecco's phosphate buffered saline (DPBS). This solution contained 2.68 mM potassium chloride, 1.47 mM potassium dihydrophosphate, 0.14 M sodium chloride, 8.1 mM di-sodium hydrophosphate, 0.49 mM magnesium chloride and 1.19 mM calcium chloride dissolved in distilled water. Then the cells were fixed by immersion in Zamboni's fixative (pH 7.4) for 15 min at room temperature. The cells were washed twice in DPBS (30 sec each at room temperature), followed by a third wash in DPBS for 30 min at 4°C and a fourth DPBS rinse for 1 min at room temperature. The swine anti-rabbit FITC-conjugated secondary antiserum (1:100) was then applied to the cell cultures for a 30 min incubation at room temperature. The cell cultures were then washed with four changes of DPBS for 5 min each at 4°C before incubation in a Hoechst solution (see Section 2.2.1). The cells were then given 4 final rinses in DPBS for 5 min each, after which the coverslips were mounted onto slides using Citifluor medium (see Section 2.2.1).

2.2.3 Live method in combination with intracellular antigens

After being removed from the culture dishes, the coverslips were incubated in DMEM medium containing 5% HS and 15 mM HEPES for 5 min at room temperature. The cells were then incubated for one hour in rabbit anti-mouse p75NGFR (1:100) diluted in DMEM/5% HS/15 mM HEPES. The control sections were incubated in diluent only.

The coverslips were washed with four changes in DPBS for 5 min each at 4°C and then fixed by immersion in Zamboni's fixative (pH 7.4) for 15 min at room temperature. The cells were washed twice in DPBS (30 sec each at room temperature), followed by a third wash in DPBS for 30 min at 4°C and a fourth DPBS rinse for 1 min at room temperature. The cultures were then incubated for 15 min in a solution of 0.2% TX-100, 5% HS and 0.1% BSA in 0.03 M PBS (pH 7.4).

These cell cultures were fixed with Zamboni's fixative for immunostaining with mouse anti-nestin. Following a 15 min fixation at room temperature, the cells were washed twice in DPBS (30 sec each at room temperature), followed by a third wash in DPBS for 30 min at 4°C and a fourth DPBS rinse for 1 min at room temperature. The coverslips were incubated for 15 min in a solution of 0.2% TX-100, 5% HS and 0.1% bovine serum albumen (BSA) in 0.03 M PBS (pH 7.4). Preincubation in TX-100 was required to permeabilize the cell membranes. A 30 min incubation in the mouse anti-nestin (1:100) antibody followed immediately after the TX-100 incubation.

The cell cultures were rinsed in DPBS four times for 5 min each. Appropriate FITC and Texas Red-conjugated secondary antisera (see Section 2.2.1) were then applied to the cell cultures for a 30 min incubation at room temperature. Swine anti-rabbit IgG FITC conjugated secondary antiserum (1:100) was used in a 1:1 mixture with a goat anti-mouse IgG Texas Red conjugated secondary antiserum (1:100).

The cell cultures were washed with four changes of DPBS for 5 min each at 4°C before incubation in a Hoechst solution (see Section 2.2.1). The cells were then given 4 final rinses in DPBS for 5 min each. The coverslips were mounted onto slides using Citifluor medium (see Section 2.2.1).

2.3 Retroviral Infection

All procedures involving handling of the retroviral supernatant and cells infected with the retrovirus (RV) were performed in accordance with requirements and standards of the Biosafety Code of the University of Saskatchewan (endorsed by the Board of Governors on March 29, 2001), the Laboratory Biosafety Guidelines (Laboratory Center for Disease Control, Health Protection Branch, Health Canada, Second Edition, 1996), as well as the Policies of the Department of Health, Safety and Environment and the Biosafety Advisory Committe of the University of Saskatchewan. Use of the RV in these experiments was covered under Biosafety Permit #ANA-02. All of the *in vitro* work done with the RV was done in a Biosafety Level 2 hood in a designated Biohazard room.

2.3.1 Production and collection of retrovirus

The BAG replication-deficient RV was constructed by Price et al. (1987). They cloned the Escherichia coli Lac-Z gene, which produces an indelible and innocuous marker enzyme called β -galactosidase, into a vector (pDOL) that was derived from the Moloney murine leukemia virus (Mo-MuLV). In the pDOL vector, the wildtype Mo-MuLV LTR provides the promoter to drive stable expression of the Lac-Z gene in infected cells. Downstream from the Lac-Z gene they cloned the simian virus 40 (SV 40) early promoter and the Tn5 neo gene to provide a selectable marker for removing uninfected cells from cell cultures infected with the BAG replication-deficient RV.

The ψ (PSI) 2 BAG alpha cells, the packaging cell line for this replication-deficient RV, were purchased from the American Type Culture Collection company (Catalog # CRL 9560, Lot # 206280, ATCC, Manassas, VA, USA) and were used to produce supernatant containing a high titer of RV. The ψ 2 BAG alpha packaging system, which was developed from the NIH/3T3 cell line, provides viral envelopes to package the BAG replication-deficient RV so it can be used to infect other cells (e.g., the OECs). Once the BAG RV has infected an OEC it is unable to spread to adjacent non-infected cells due to the absence of the genetic machinery in the OEC genome that is required to make a viral envelope.

Vials of frozen ψ 2 BAG alpha cells were thawed in a 37°C water bath. The cells were transferred into a DMEM medium containing 10% bovine calf serum (BCS) and 1% antibiotic-antimycotic (Sigma-Aldrich Canada Ltd, Oakville, ON). After the cell suspension was centrifuged at 160 G for 10 min, the cell pellet was resuspended in a desired volume of

DMEM/10% BCS/1% antibiotic-antimycotic. The resulting cell suspension was plated into 25 cm² flasks, which were gassed with 5% CO₂ for approximately 1 min and then tightly closed with sealed caps. Cultures of ψ 2 BAG alpha cells were fed with DMEM/10% BCS/1% antibiotic-antimycotic medium twice a week. When the cell cultures became confluent, the growth medium was removed and discarded and fresh DMEM/10% BCS/1% antibiotic-antimycotic medium was added to each flask. Within 2-3 days the medium in the flasks had turned yellow reflecting the high density and metabolic activity of the cells, which in turn indicated a high titer of RV was present in the medium. This yellow-colored medium was collected from the cell cultures every 2-3 days, stored in sterile 50 ml plastic centrifuge tubes at 4°C, and filtered through a 0.45 µm low protein binding filter. Aliquots of filtered RV-containing medium were either used right away or frozen and stored at -80°C until needed.

2.3.2 Titration of retrovirus

NIH 3T3 cells were used to determine the viral titer in the ψ 2 BAG α conditioned medium. The NIH 3T3 cells were purchased from the American Type Culture Collection (ATCC Catalog # CRL-1658). Vials of frozen NIH 3T3 cells were thawed in a 37°C water bath and then the cells were transferred into a DMEM/10% BCS/1% antibiotic-antimycotic medium. After the cell suspension was centrifuged at 800 RPM for 10 min, the cell pellet was resuspended in the desired volume of DMEM/10% BCS/1% antibiotic-antimycotic medium. The resulting cell suspension was plated into 60 mm dishes at a density of 5 x 10^4 cells/ml. The day after the cells were plated, the growth medium was replaced with DMEM/10% BCS/1% antibioticantimycotic medium to which was added one of several different volumes of RV supernatant and 8 µg/ml polybrene; the cell cultures were placed in the incubator at 37°C for 2.5 hours. The NIH 3T3 cells were infected with logarithmic dilutions $(10^{-1} \text{ through } 10^{-5})$ of retroviral supernatant. In order to ensure an even distribution of the virus over the surface of the cell culture, the dishes were agitated every 30 min throughout the 2.5 hour incubation period. After 2.5 hours of incubation, a sufficient volume of DMEM/10% BCS was added to each dish to dilute the concentration of polybrene in the growth medium to 2µg/ml. The infection of NIH 3T3 cells with RV containing supernatant was repeated daily for an additional two days. These NIH 3T3 cell cultures were trypsinized (see Section 2.1.4) immediately upon finishing the infection. Cell suspensions were collected separately from each dish, centrifuged and resuspended in DMEM/10% BCS. The resulting cell suspensions, derived from infections with logarithmic

dilutions of RV supernatant, were diluted 1:20 in DMEM/10%BCS containing 400 μ g/ml geneticin (G 418 sulfate, Life Technologies, Gibco BRL); each cell suspension was plated into three 60 mm dishes. The cell cultures were fed with fresh DMEM/10% BCS containing 400 μ g/ml geneticin twice a week for the next 10-14 days. At the end of the culture period, the growth medium was removed from the dishes and the cells were fixed in methanol for 15 min at room temperature. The fixed cells were allowed to air dry, after which they were stained with 0.45% Coomassie brilliant blue (color index = 42660) dissolved in Methanol/acetic acid for 1 min at RT and rinsed with distilled water. The average number of virus-expressing Geneticin-resistant colonies was determined using the following formula:

<u>number of colonies x dilution factor</u> = number of colony forming units/ml volume of RV supernatant

2.3.3 Infection of olfactory ensheathing cells

The OEC cultures were initiated as described in Section 2.1 and grown in 25 cm² Falcon flasks. The cell cultures were fed with mG5 medium as described in Section 2.1.3. Infection with RV began on the day the cell cultures were switched from mG5 medium to serum-containing medium. One control flask from each set of cell cultures was fed with the DMEM/F12/10% FBS medium containing no RV. One ml of the retroviral supernatant in DMEM/10% BCS medium containing 8 μ g/ml polybrene was added to each 25 cm² Falcon flask designated for RV infection and the cell cultures were incubated for 2.5 hours at 37°C. At the end of this time 3 ml of DMEM/F12/10% FBS were added to each flask after which the flasks were returned to the incubator for 21.5 hours. Fresh RV/polybrene solution was added daily, as just described, for 2 weeks. The control flask was treated identically on a daily basis, but received no exposure to RV.

2.3.4 Geneticin selection of infected cells

After the 2 week period of daily exposure to RV containing medium, the RV-infected and uninfected cell cultures were fed twice a week with DMEM/F12/10% FBS medium containing geneticin (400 μ g/ml); control (uninfected) cell cultures were fed in an identical manner. Close monitoring of control cultures determined when all uninfected cells were killed, which usually occurred within 4-5 days of being first exposed to the geneticin. The RV-infected OEC cultures were then expanded with DMEM/F12/10%FBS medium that contained no geneticin, until they

approached confluence. Prior to harvesting, $400 \ \mu g/ml$ of geneticin was added to the growth medium for the last 3-4 days.

2.3.5 x-Gal staining of OEC cultures

Some RV-infected OEC cultures were stained with x-Gal solution to confirm that the cells were expressing β -galactosidase. The following steps were performed in the Biohazard hood. The growth medium was aspirated from the dishes prior to rinsing them 3 times with 0.03M PBS at room temperature. The cell cultures were then fixed for 5 min at room temperature with a 0.2% glutaraldehyde solution (in 0.03M PBS) and washed 3 more times with 0.03M PBS (room temperature). Once the cells were fixed, the following steps could be carried out in the general lab. The OEC cultures were incubated with a solution containing 0.1% x-Gal (5-bromo-4-chloro-3-indolyl-β-d-galactoside; Gibco BRL, Grand Island, N.Y. 14072 USA) in 20 mM potassium ferricyanide, 20 mM potassium ferrocyanide, and 2 mM MgCl dissolved in 0.03M PBS for 4 hours at 37°C. These cultures were then rinsed 3 times with 0.03M PBS and stained for 2 min at room temperature with a solution containing 3.7 mM Nuclear Fast Red (5-chloro-2-methoxybenzenediazonium chloride hemi (zinc chloride) salt; Sigma-Aldrich, Oakville, ON) and 75 mM aluminum sulfate dissolved in distilled water. After a final wash with distilled water, the OEC cultures were dehydrated in an ascending series of ethanols (70%, 95%, 100%), cleared in xylene for 2 min and then mounted on slides using Shandon xylene substitute mountant.

2.4 Surgery

All surgical procedures were performed under sterile conditions. Animal care was provided in accordance with the requirements of the Canadian Council on Animal Care, and the experimental protocols were evaluated and approved by the Protocol Review Committee, which is a Subcommittee of the University Committee on Animal Care and Supply (University of Saskatchewan). Animals had free access to food and water throughout the period of study.

2.4.1 Anesthesia and analgesia

Thirty six adult (3-4 months old) female Wistar rats weighing 210-280 grams (U of S Laboratory Animal Service Unit) were anesthetized intraperitoneally (i.p.) with sodium pentobarbital (Somnotol; MTC Pharmaceuticals, Cambridge, Ontario, Canada) at a dose of 55 mg per kg body weight (Groups 1-4; see Table 4-1). In addition, Buprenex (Buprenorphine

74

hydrochloride, Reckitt & Colman Products Limited, Hull, England), an opioid analgesic, was administered intramuscularly before surgery was started at a dose of 0.05 mg per kg body weight.

An additional 21 rats were anaesthetized by inhalation of Halothane (MTC Pharmaceuticals, Ontario, Canada) (Groups 5-8; see Table 1). The induction of anesthesia was achieved by a mixture of 3% Halothane in 100% oxygen using a Boyle anaesthetic machine (Ohmeda, Mississauga, ON). After 5-10 min, the animals were immobile and unresponsive to a pinch test on their hind paws (pedal reflex). A single intramuscular injection of Buprenex (0.05 mg per kg body weight) was then administered prior to making the skin incision. The concentration of 3% Halothane was maintained during the initial parts of the surgery until it was time to begin the laminectomy. Beginning with the laminectomy, and continuing through the EtBr and cell grafting, the Halothane concentration was kept at 2-2.5%; closure of the wound was done using 1-1.5% Halothane.

During the course of the entire surgery, each animal was closely monitored (every 3-5 min) for their respiration, heart rate, selected reflexes, and the color of their paws. For all of the rats in Groups 1-8 they continued to breath at a normal rate, their heart rate seemed appropriate (as judged by palpation), pedal and tail pinch reflexes were absent, and the paws remained pink in color throughout the duration of each operation.

2.4.2 Laminectomy

An operation field was designated on the midline of the back. The rats were placed onto a reusable heating pad (Safe and Warm Incorporated, Seattle WA, USA) to maintain normal body temperature. The shaved operating field was washed with a 1:1 mixture of Betadine antiseptic solution (10% povidone-iodine topical solution UPS) and 70% ethanol. Using a #23 Almedic stainless steel scalpel blade, a 3-5 cm longitudinal incision was made through the skin and the superficial fascia along the posterior midline, immediately superficial to the spines of the lower thoracic vertebrae. In the surgical area, the muscles located on the dorsal surface of the spine were cut and moved laterally. A laminectomy was performed of the designated thoracic vertebra, after the appropriate spinous process had been removed using bone rongeurs. The dura mater was cut longitudinally within the hole created by the laminectomy.

2.4.3 Injection of ethidium bromide

A 32 Gauge Hamilton needle was connected to a 5 μ l Hamilton syringe, which was then mounted onto a rodent stereotaxic apparatus using the syringe holder of a KD Scientific Syringe Pump (KDS 310). A previously prepared stock solution of 0.03% EtBr dissolved in distilled water was loaded into the 5 μ l Hamilton syringe. The 32 Gauge needle was lowered so it just touched the spinal cord surface, as seen through a Nikon SMZ-1B stereoscopic dissecting microscope. After zeroing the XYZ coordinates of the stereotaxic apparatus, the needle was slowly lowered into the right dorsal funiculus to a depth of 1 mm, approximately 1 mm to the right of the dorsal spinal artery. The needle was left in place for 1 to 2 min after being inserted into the cord, prior to initiating the injection of EtBr. Using the KDS 310 syringe pump, 1 μ l of 0.03% EtBr was injected over 10 min using a flowrate of 0.10 μ l/min. After completion of the injection, the needle was left in the spinal cord for 1-2 min prior to being slowly withdrawn. The syringe with attached needle was immediately rinsed with dH₂O, followed by a couple of rinses with acetone and then a few final rinses with dH₂O.

2.4.4 Grafting Dil-labelled olfactory ensheathing cells

Immediately before surgery, cryovials were withdrawn from the locator and thawed in a bath of lukewarm water for approximately 1 min. The cell suspension was then slowly added to medium in a Falcon centrifuge tube to dilute the DMSO, followed by centrifugation at 160 G (for 10 min) and resuspension in medium (x2). After the last wash in medium, the total number of viable cells was determined as described in Section 2.1.4. The cell viability averaged 73% (see Table 2.1).

The cell pellet was resuspended in a grafting solution containing 1 μ g/ml of CaCl₂, 1 μ g/ml of MgCl₂, 0.1% Glucose in 0.1M PBS; the final cell concentration was 50,000 cells per μ l. This cell suspension was stored at 4°C for no longer than 1 hour prior to grafting. This cell suspension was loaded into the 5 μ l Hamilton syringe, as just described for the EtBr. The injection of cells into the dorsal funiculus was done using the same flowrate setting, as previously described for EtBr (see Section 2.4.3). A 1 μ l volume of cell suspension, containing 50,000 DiI-labelled OECs, was injected over a period of 10 min. After completion of the injection, the needle was left in the spinal cord for 1-2 min prior to being slowly withdrawn. The syringe with attached needle was immediately rinsed with dH₂O, followed by a couple of rinses with acetone and then a few final rinses with dH₂O.

The cell culture	Total number of	Total number of	% Viability
number	viable cells frozen	viable cells when	
		thawed	
K 427 ONL	1,152,600	703,125	61%
K 428 ONL	881,400	687,500	78%
K 429 ONL	863,000	725,000	84%
K431 ONL	1,001,600	781,250	78%
K 447 ONL	1,218,750	775,000	71%
K 462 ONL	1,127,300	721,500	64%
Mean number:		739,375	73%

Table 2.1Cell counts of viable OECs before and after being frozen in liquid
nitrogen

2.4.5 Grafting retroviral-infected olfactory ensheathing cells

The procedure for grafting of RV-infected OECs into the dorsal funiculus of the rat spinal cord (n=1) was identical to that described for DiI-labelled OECs in Section 2.4.4. In order to graft RV-infected OECs onto the pial surface of the rat spinal cord (n=3), the Anesthesia and Analgesia (Section 2.4.1) and Laminectomy (Section 2.4.2) parts of the surgical procedure were the same as previously described. The cell suspension was prepared for transplantation and loaded into the 5 µl Hamilton syringe similar to that described in Section 2.4.4. The 30 Gauge 90° needle was lowered so it just touched the spinal cord surface, as seen through a Nikon SMZ-1B stereoscopic dissecting microscope. The stereotaxic apparatus was zeroed, so the needle tip was inserted into the slit in the dura and touched the dorsal surface of the spinal cord under visual observation. The needle was slowly moved 1 mm to the right of the dorsal spinal artery so the right lip of slit in the dura surrounded the needle's tip without the needle being injected into the right dorsal funiculus. A 1 µl volume of cell suspension, containing 50,000 RV-infected OECs, was injected over a period of 10 min at the same flowrate setting described for EtBr (see Section 2.4.3). After completion of the grafting, the needle was left in position adjacent to the dorsal surface of the spinal cord for 1-2 min prior to being slowly withdrawn.

2.4.6 Wound closure and postoperative care

The needle was slowly withdrawn 1 to 2 min after either the cell grafting or EtBr injection had been completed. The muscles along the spine were sutured with Ethicon 4-0 surgical silk, and washed with a 3% solution of hydrogen peroxide (diluted in dH₂O) for antiseptic purposes. The skin wound was closed with 3-4 silk sutures, washed with Betadine antiseptic solution diluted 1:1 with 70% ethanol, and painted with a 2% Xylocaine local anaesthetic gel (Lidocaine Hydrochloride Jelly, USP Sterile Topical Anesthetic, Astra Pharma Inc., ON, Canada).

Each animal was monitored every 5 to 10 min during the first hour of the postoperative period for their respiration rate and for the color of their paws. The Group 1-4 rats were kept on a heating pad for 2-3 hours either until the pad had reached room temperature or until the rat had started to move around in the cage, whichever came first. Any bedding was kept away from the animal's face to avoid inhalation of small particles. The rats were observed each hour postsurgically for the next three hours and also the following morning to ensure they were

78

drinking water and eating food. On the first postoperative day the rats were transferred back to the animal facility room. Postoperative monitoring was done once a day until the rats were killed by an overdose of Somnotol and perfused for necropsy (see Section 2.5). During this monitoring, particular attention was directed towards the sutures and the healing of the wound. No problems were observed with the sutures on any of the rats. Postoperative wounds healed with no complications and were replaced by soft white linear scars after 2-3 weeks. By this time, the sutures were no longer observed and the new fur grew enough to cover the postoperative scar. On a daily basis, the rats were also checked for their mobility, alertness, food and water consumption, and the condition of their fur. The surgical manipulations did not cause noticeable mobility problems in any of the rats included in the Groups listed in Table 4-1.

With the exception of several animals, all rats were drinking water, eating food, and actively moving around in the cage during the postoperative period. Their activity, general appearance and behavior appeared to be normal. There were only three rats after somnotol anesthesia noticed lying in the cage for more than 30 min after surgery; these rats were barely moving but were breathing normally. These animals received 5 ml injections of saline i.p. one time each and their condition improved over the next two hours. Another three rats suddenly stopped breathing either during the last steps of the surgery or shortly after the operation, which might have been due to Halothane side effects, myocardial and respiratory depression, or the animal's individual sensitivity to Halothane. These animals were cervically dislocated and are not included in the animal numbers listed in Table 4-1. There were two additional rats whose entire body jerked when the needle was inserted into the dorsal funiculus of the spinal cord. Such a whole body reaction was likely due to the needle reaching the corticospinal tract located at the depths of the dorsal funiculus. During the postoperative period, it was noted that the right hind limb of these rats was paralyzed. Following necropsy, the lesion occupied the entire dorsal funiculus including the corticospinal tract. The tissue sections of these rats were excluded from analysis and the rats are not included in the animal numbers listed in Table 4-1.

2.4.7 Minocycline treatment

Six rats injected with EtBr at T10 and OECs at T12 (Group 6; see Table 4-1) and six additional rats that were injected with EtBr at both T10 and T12 and OECs at T12 (Group 7; see Table 4-1) received a 0.5 ml i.p. injection of either minocycline (45 mg/Kg in 0.9% saline; n=3 for each of Groups 6 and 7; Sigma-Aldrich, Oakville, ON) or vehicle (i.e. 0.9% saline; n=3 for

each of Groups 6 and 7) daily at approximately 24 hour intervals for the first 4 weeks after EtBr injection. The abdomen of all rats was examined each day prior to giving these daily injections for evidence of tenderness or other adverse reactions to either the minocycline or saline injections. To minimize the discomfort of daily injections, a new sterile needle was used for each injection. The minocycline and vehicle treated rats (Groups 6 and 7) were kept in the animal facility for an additional four weeks after the last minocycline or vehicle injection. A 4 week treatment time frame was chosen to ensure the anti-inflammatory property of minocycline was active for a sufficiently long period of time.

2.5 Necropsy of Tissue

2.5.1 Perfusion of the rats

Between 2 to 8 weeks after the last operation (i.e. Grafting DiI-labelled OECs) the rats were deeply anaesthetized with an i.p. injection of 0.35 c.c. Somnotol (~90 mg/kg body weight). The rat's chest cavity was opened transabdominally to expose the heart, after which the apex of the left ventricle was cut off with scissors. A needle, which was connected to the tubing of a Variable Flow Tubing Pump (VWP Scientific Products, VWR Canlab), was inserted into the left ventricle and pushed up into the ascending aorta. Blood exited the cardiovascular system via an additional incision in the wall of the right atrium; this incision was made immediately after the perfusion had begun. Approximately 300 ml of a 1% solution of sodium nitrite dissolved in 0.03 M PBS (pH 7.4) was passed through the vascular system to flush out the blood. The rat was then perfused with approximately 250 ml of a freshly made cold solution of 4% paraformaldehyde (PF) in 0.03 M PBS (pH 7.4) over a period of 30-40 min.

2.5.2 Dissection of spinal cord

When the perfusion was done, the skin and muscles were cut longitudinally along the posterior midline from the skull to the tail base. The exposed spinous processes and vertebral laminae were removed using bone rongeurs through the whole length of the spinal column. The spinal cord with its short rootlets were gently dissected out, immersed in fresh fixative (4% PF) and stored overnight at 4°C. After approximately 18-20 hours fixation, the spinal cords were immersed in a 30% solution of sucrose dissolved in 0.03 M PBS (pH 7.4) for 2-3 days at 4°C. The spinal cords were then ready for sectioning on a sliding microtome.

2.5.3 Cutting and storage of tissue sections

The rat thoracic spinal cord segments (T8-T12) were identified by counting the short rootlets left after dissection, beginning in the cervical area and proceeding caudally. These segments were individually separated using a scalpel blade. The specimen holder of a sliding microtome (American Optical Company, Scientific Instrument Division, Buffalo, New York, USA) was frozen in dry ice prior to application of Tissue-Tek O.C.T. 4583 compound (Sacura Finetek U.S.A., Inc., Torrance, CA 90504 USA) onto its surface. One spinal cord segment at a time was placed onto the prepared specimen holder, covered with O.C.T. medium, and frozen by means of dry ice. The sliding microtome was used to cut coronal sections of 40 µm thickness. Serial coronal sections were collected and placed in a 96 well storage plate filled with cryoprotectant (a solution of 0.11 M sodium phosphate monobasic and 0.38 M sodium phosphate dibasic in distilled water, containing 3% Glycerin and 3% Ethylene Glycol). Free floating sections were stored in this cryoprotectant at -20°C until they were mounted onto subbed slides for cell counts or for immunohistochemical staining.

2.5.4 Immunohistochemical staining

The 40 µm thick sections of spinal cord were removed from cryoprotectant and washed three times in 0.03 M PBS (30 sec each at room temperature), followed by a fourth PBS rinse for 30 min at 4°C. The sections were then incubated for 30 min in a solution of 3% SMP and 0.1% TX-100 in 0.03 M PBS (pH 7.4); preincubation in SMP/TX-100 was required to block non-specific binding sites and to permeabilize the cell membranes.

Incubation in primary antibodies was done overnight at 4°C. The following antibodies were used: rabbit anti-mouse p75NGFR (1:9000, Chemicon International Inc., Temecula, CA 92590, USA), rabbit anti-cow GFAP (1:11000, Dimension Labs, Dako A/S Denmark), mouse anti-rat MHC Class I (Clone OX-18, 1:9600, Secotec Product Datasheet, Kidlington, Oxford, England), mouse anti-CD11b (Clone MRC OX-42, 1:3200, Serotec Product Datasheet, Kidlington, Oxford, England), and mouse anti-Vimentin (Clone V9, 1:1000, Sigma Immuno Chemicals, St. Louis, MO 63178, USA). The primary antibodies were diluted in 0.03 M PBS. The control sections were incubated in diluent only.

The following day, the sections were washed with two changes of 0.03 M PBS and then blocked again in a solution of 3% SMP/0.1% TX-100 in 0.03M PBS for 30 min. Appropriate biotinylated secondary antisera were then applied to the sections for a 30 min incubation at room

temperature. The biotinylated secondary antibodies included: goat anti-rabbit IgG (1:200, Vectors Labs, Burlingame, CA 94010 USA) and horse anti-mouse IgG (1:150, Vectors Labs, Burlingame, CA 94010 USA).

The sections were then incubated for 15 min in a 1% solution of hydrogen peroxide (H_2O_2) in 30% methanol to quench the enzymatic activity of endogenous peroxidase. An additional wash in two changes of 0.03 M PBS (pH 7.4) was followed by incubation in a 3% solution of SMP in 0.03 M PBS (pH 7.4) (30 min). The Vectastain Elite ABC kit (Vector Laboratories, Inc., Burlingame, CA 94010 USA) was used to make the Avidin DH (reagent A)/ biotinylated horseradish peroxidase H (reagent B) cocktail; 8µl of each reagent per ml were diluted in 0.03 M PBS (pH 7.4) 30 min prior to being used. The sections were incubated in this solution for one hour at room temperature.

After two washes in 0.03 M PBS (pH 7.4), the sections were rinsed in 0.175 M sodium acetate (pH 6.8) for 10 min. A solution of 0.0185 M diaminobenzidine tetrahydrochloride (DAB) and 0.1586 M nickel sulfate (NiSO₄) in 0.175 M sodium acetate (pH 6.8) was made one hour prior to use and kept in the dark. Immediately after adding 85 μ l of a 30% solution of hydrogen peroxide to each 100ml of DAB/NiSO₄ solution, the sections were immersed in DAB/NiSO₄/H₂O₂ for 5 min in the dark.

The sections were then rinsed once (for 10 min) in 0.175 M sodium acetate (pH 6.8), given a final rinse in 0.03 M PBS (pH 7.4) for 10 min, and mounted onto subbed (1% gelatine, 0.1% chrome alum) slides. The sections were left to air dry for approximately 18-20 hours, after which they were dehydrated in an ascending series of ethanols (70%, 95%, 100%). The sections were cleared in xylene for 2 min and then coverslipped using Shandon xylene substitute mountant (Shandon, Pittsburgh, PA 15275, USA).

2.5.5 Data collection and statistical analysis

Four sections were randomly chosen from each of the caudal, middle, and rostral portions of each spinal cord segment (i.e T8-T12), mounted onto a microscope slide, and coverslipped using Citifluor antiquenching mounting medium. All tissue sections to be used for cell counting were stained with Hoescht dye (1:50 dilution) prior to mounting to ensure that the only DiI+ve profiles counted were those associated with a nucleus. Each tissue section was examined to determine whether it contained any DiI+ve cells, to identify the specific location in which the cells were found, and to count the total number

of labelled cells within the dorsal funiculus. For statistical analysis the mean cell count for each group of 4 sections was computed. The initial data analysis showed no differences between the data collected from the rostral, middle and caudal portions of each spinal cord segment for each Group. Therefore, these data were combined and plotted as one bar for each spinal cord segment and for the statistical analysis of the data. The data collected from Groups 1-5 (see Table 4.1) were statistically compared using a one-way analysis of variance (Prism program). The independent variable was the presence and location of one or two EtBr-induced demyelinated lesions (i.e. the experimental groups). The data on cell counts obtained from each spinal cord segment (i.e. T8-T11) and for each survival time (2, 4 or 8 weeks) were compared separately for statistical significance. The significance level was set at p < 0.05. Multiple post hoc comparisons between the different Groups (see Table 4.1) were performed with Bonferonni's multiple comparison post-test (Prism program) with the significance level being set at p < 0.05. The data collected from Groups 6 and 7 (see Table 4.1) were statistically compared using the Mann-Whitney U test (Prism program) and a significance level set at p < 0.05.

2.5.6 Luxol fast blue staining of tissue sections

The 40 μm thick spinal cord sections were removed from cryoprotectant and washed in 0.03M PBS for 10-20 min. The sections were then mounted onto gelatine-chrome alum subbed coverslips and were left to air dry for approximately two hours. Tissue sections were washed in 0.03M PBS for 10 min and then in distilled water for 10 min. The sections were dehydrated in 70% and 95% ethanol for 2 min each. Tissue sections were then incubated in a solution containing 1.05% Luxol Fast Blue (sulfonated copper phthalcyanine; Sigma-Aldrich, Oakville, ON), 90% ethanol and 10% acetic acid for 16 hours at 37°C. After washing in distilled water for 20 sec, followed by 70% ethanol and another rinse in distilled water. This was then followed by quick rinses in 70% ethanol and then 90% ethanol, followed by two changes of isopropyl alcohol for 3 and 2 min, respectively. The tissue sections were quickly rinsed in a 1:1 mixture of xylene and absolute alcohol. The sections were cleared with three changes in xylene and then mounted on slides using Shandon xylene substitute mountant.

2.5.7 x-Gal staining of tissue sections

The 40 μ m thick sections of the brain, olfactory bulbs, cerebellum, brain stem, cervical and thoracic spinal cord of Group 8 rats were stained with x-Gal solution to determine the β -galactosidase expression in OECs after either intraspinal implantation or pial surface grafting. Tissue sections were removed from cryoprotectant and washed in 0.03M PBS for 10-20 min. The sections were then mounted onto gelatine-chrome alum subbed coverslips and were left to air dry for approximately two hours. Tissue sections were incubated in a 0.1% x-Gal solution for 48 hours at 37°C. The sections were rinsed 3 times with 0.03M PBS and then stained for 2 min at room temperature with a 0.1% Nuclear Fast Red solution in distilled water also containing 5% ammonium sulfate. After a final wash with distilled water, tissue sections were cleared in xylene for 2 min and then mounted on slides using Shandon xylene substitute mountant.

3. SPECIFIC AIMS

Multifocal demyelinated lesions in white matter are the main characteristic pathologic features of MS, which is the most common CNS demyelinating disease in humans. Although the pathology of MS is believed to be related to immunological phenomena induced by viral infection or exposure to unknown allergens (Agranoff and Goldberg, 1974; Bach, 2002; Challoner et al, 1995; Friedman et al, 1999; Gilden, 1999; Kazmierski et al., 2004; Kurtzke, 1980; Lucchinetti et al., 2000), the mechanisms responsible for the extensive CNS demyelination and controlling whether remyelination occurs remain to be elucidated. In order to enhance remyelination in the adult mammalian CNS, several approaches involving the transplantation of myelinating glia have been considered (Barnett et al., 1993; Blakemore and Crang, 1985, 1989; Blakemore et al, 1995; Duncan et al., 1992; Franklin and Blakemore, 1997; Groves et al., 1993; Warrington et al., 1993). The cells of choice have been oligodendrocytes, Schwann cells and OECs (see Section 1.7). Although in vivo OECs provide ensheathment for the unmyelinated axons of the first cranial nerve, these glial cells can be induced to express a myelinating phenotype both in vitro (Devon and Doucette, 1992, 1995) and after grafting into demyelinated areas of the adult mammalian spinal cord (Franklin et al., 1996; Imaizumi et al., 1998; Li et al., 1997, 1998; Lakatos et al., 2003). It was the neurite-promoting and remyelinating abilities of the OECs that first attracted investigators to graft these cells into damaged CNS tissue (Lustgarten et al., 1991; Ramon-Cueto and Nieto-Sampedro, 1994; Smale et al., 1996; Li et al., 1997, 1998; Perez-Bouza et al., 1998; Ramon-Cueto et al., 1998; Navarro et al., 1999).

The demyelinated lesions in MS are scattered throughout the white matter of the CNS, making it impractical to graft myelin-forming cells into each demyelinated area as part of a therapeutic approach to remyelinating the axons. A more sensible approach would entail grafting the cells into a smaller number of sites and having the glial cells migrate towards each of several demyelinated areas. The purpose of my Ph.D. research project was to examine the

migratory ability of one type of myelin-forming cell, namely the OECs, after they were grafted into the spinal cord of adult rats and to identify the conditions under which they would be induced to migrate towards a focal area of demyelination. For these experiments the ensheathing cells were prelabelled with a fluorescent dye (i.e. DiI) prior to grafting so the cells could be identified for cell counting after they had mingled with host cells in spinal cord segments rostral to the graft. The specific aims of my research project were:

- Specific Aim # 1 To determine whether OECs will be induced to migrate towards a focal area of demyelination in the spinal cord of adult rats when the cells are grafted into normal CNS white matter.
- Specific Aim # 2 To determine whether the grafted OECs migrate away from a demyelinated area of spinal cord into normal CNS tissue.
- Specific Aim # 3 To determine whether the grafted OECs migrate away from a demyelinated area of spinal cord towards a second demyelinated area.
- Specific Aim # 4 To determine whether microglial reactivity contributes to the migratory signal inducing OECs to migrate towards a focal demyelination of CNS white matter.
- Specific Aim # 5 To determine whether OECs grafted onto the pial surface of the spinal cord will migrate into the neuropil in the absence of demyelination.

4. **RESULTS**

4.1 Olfactory Ensheathing Cell Cultures

At 3 DIV sparsely distributed flat and fusiform cells, either singly or in groups of a few cells, comprised the cell cultures initiated from the tissue obtained from the ONL of the E18 rat olfactory bulb. Over the next 10 DIV, as the cell cultures were fed twice a week with the mG5 serum-free medium, the morphology of the cells comprising the culture changed to predominantly a fusiform, bipolar or tripolar morphology. The bipolar and tripolar cells were small phase bright cells, whereas the somata of the fusiform cells were flatter (Fig. 4.1a). A very small number of large, flat cells, which were most likely meningeal fibroblasts, were present in the cell cultures when the growth medium was first changed from one containing serum to the mG5 medium; the mG5 medium supported an increase in the number of fusiform, bipolar and tripolar cells at the expense of the larger, flatter cells, which were usually almost completely absent from the cell cultures after 2 weeks (i.e. after ~ 10 days in mG5). The growth medium for the cell cultures was then switched back to DMEM/F12/10% FBS, which was used to expand the cell cultures over the next 20 to 30 DIV until they approached confluency. In the serum-containing medium the cells acquired a flatter multipolar and fusiform morphology (Fig. 4.1b).

To assess the purity of the cell cultures they were double immunostained with a monoclonal antibody to nestin in combination with either polyclonal antisera to p75 or to S100. Almost all of the cells in these cultures were both S100+ve (Fig. 4.1e) and nestin+ve (Fig. 4.1f). Likewise, the majority of the cells in these cultures co-expressed the p75 NGF receptor on their cell surface, including the somata as well as the cytoplasmic processes (Fig. 4.1c), and the intermediate filament protein nestin (Fig. 4.1d). There were only a small number of cells not expressing any of these markers (indicated by arrows in Fig. 4.1c-f).

FIGURE 4.1: Morphology and Phenotype of OECs in vitro.

Phase contrast micrographs showing OECs in serum-free medium (a) and in serum containing medium (b). Double immunostaining of OECs showing their co-expression of p75NGFR (c) and nestin (d), and of S-100 (e) and nestin (f). Cells that expressed none of these markers (the white arrows) were rare in any of the cell cultures used for this thesis. Bar = $100 \mu m$



4.2 The Animal Model For Studying OEC Migration

4.2.1 The Ethidium Bromide Lesion

The 0.03% EtBr solution used in this research project created a demyelinated lesion that remained confined to the right dorsal funiculus of the spinal cord with no detectable involvement of the contralateral funiculus. Practically the entire right dorsal funiculus was demyelinated at the epicenter of the EtBr injection, with the rostral-caudal extent of the damage extending for approximately 1 mm; this length corresponded to approximately 20% of the length of a spinal cord segment. Previous studies have used x-irradiation to impede host glial cells in their repair of demyelinated area of white matter (see Section 1.6.1) but in the model used here the cell migration of OECs was allowed to occur in conjunction with the host glial cells being allowed to contribute to the neural repair. With this concentration of EtBr, demyelination was still evident at 2 weeks after injury (Fig. 4.2e) but by 8 weeks this region of white matter had been almost entirely remyelinated (Fig. 4.2f).

The monoclonal antibody OX42 recognizes the CR3 complement receptor on the surface of resting and phagocytic microglia while the monoclonal antibody OX18 recognizes the rat class 1 major histocompatibility complex antigen on the surface of phagocytic microglia. Both of these antibodies were used in this research project to reveal the extent of the microglial reactivity to the cell graft. The spinal cord tissue was also stained with a monoclonal antibody to vimentin because astrocytes in the adult mammalian CNS do not normally express detectable levels of vimentin, unless they are induced to do so in response to brain or spinal cord injury. At 2 weeks after injury in a cross-section through the segment of injury, OX-18+ve reactive microglia (Fig. 4.2g) and vimentin+ve reactive astrocytes (Fig. 4.2h) were confined primarily to the demyelinated lesion within the right dorsal funiculus, with only an occasional reactive glial cell being found in the left dorsal funiculus or in the medial part of the right dorsal horn. Since this 0.03% EtBr concentration created an acceptably small, demyelinated lesion that was confined to one-half of the dorsal funiculus, this concentration was used in all subsequent experiments.

FIGURE 4.2: A small, focal EtBr-induced demyelinated lesion.

A schematic diagram (a) and a Luxol Fast Blue stained coronal section (b) of the normal adult rat spinal cord. c) A schematic diagram of an EtBr-induced demyelinated lesion at T10 (yellow circle). d) A micrograph showing the needle for injecting EtBr inserted into the right dorsal funiculus at T10. Luxol Fast Blue stained coronal sections of the T10 spinal cord segment two (e) and eight (f) weeks after injecting EtBr into this white matter. The OX-18+ve microglial (g) and vimentin+ve astrocytic (h) reactivity at the lesion site at two weeks after EtBr injection. Asterisks in c-h) denote the location of the EtBr-induced lesion. The white arrows indicate the astrocytic (g) and microglial (h) reactivity along the medial border of the EtBr-induced lesion. Bar = 400 μ m applies to all figures except a) and c).



4.2.2 The Cell Graft

One ul of cell suspension containing 50.000 DiI-labelled OECs was stereotactically injected into the right dorsal funiculus of the T12 spinal cord segment (Fig. 4.3a & b). The needle was inserted immediately lateral to the dorsal spinal artery and was lowered into the right dorsal funiculus to a depth of 1 mm. Macroscopically, the grafting site could be identified on the pial surface of the cord by a pink discoloration of approximately 1-1.5 mm in diameter that covered the surface of the right dorsal funiculus of the T12 spinal cord segment. Once the T12 spinal cord segment was cut in the coronal plane into 40 µm thick sections, a similar pink discoloration was also seen to be present throughout much of the depth of the right dorsal funiculus. When these coronal sections were examined with a dissecting microscope ($\sim 10x$ magnification) twenty to twenty-five of the sections (i.e. a length of $800-1000 \ \mu\text{m}$) through the midportion of T12 were seen to contain this pink discoloration of the right dorsal funiculus. The remainder of the coronal sections cut from the T12 spinal cord segment, both rostrally and caudally to the grafting site, did not contain visible discoloration of the dorsal funiculi. Microscopically, a compact mass of DiI+ve OECs could be seen with the fluorescent microscope; this DiI+ve OEC cell mass was seen within the right dorsal funiculus of each coronal section through the midportion of the graft (Fig. 4.3b). In addition to this compact mass, single as well as small groups of DiI+ve OECs were found within both dorsal funiculi and both dorsal horns of T12 (Fig. 4.3c).

Two weeks after grafting DiI+ve OECs into the right dorsal funiculus at the midpoint of the T12 spinal cord segment, OX-18+ve reactive microglia (Fig. 4.3e) and vimentin+ve reactive astrocytes (Fig. 4.3f) were detected within and around the location of the cell graft. These reactive microglia and astrocytes were localized predominantly within the right dorsal funiculus between the midline septum and the right dorsal horn and extending from the pial surface down to about the level of the corticospinal tract. No OX-18+ve microglial cells were seen in the dorsal funiculus of the T12 spinal cord segment in control animals, which received neither EtBr injection nor OEC graft
FIGURE 4.3:DiI+ve OECs grafted into the dorsal funiculus of T12.

a) A schematic diagram representing a graft of OECs (purple oval) in the dorsal funiculus of the T12 spinal cord segment. b-d) DiI+ve OECs in the right dorsal funiculus of T12 showing the graft at low power (b) and single cells at high power (c, d). For quantitation of cell migration, these sections were stained with Hoechst dye and each DiI-labelled OEC with a red fluorescent cell body and cytoplasmic processes (white arrowhead) and a blue Hoechst-stained nucleus (white arrow) was counted. Most of the cells with blue Hoechst-stained nuclei (grey arrow) were DiI-ve and were considered to be host cells. OX-18+ve reactive microglia (e) and vimentin+ve reactive astrocytes (f) at the site of an OEC graft at the level of T12 two weeks after grafting. OX-18 (g) and vimentin (h) immunostaining of the T12 spinal cord segment of an adult rat that received no cell graft or EtBr injection. Bar in 'd' = 100 μ m and applies to figures b) and e-h).



(Fig. 4.3g). Vimentin staining revealed occasional activated astrocytes in the neuropil only adjacent to pia matter (Fig. 4.3h).

4.2.3 Using Dil to Identify Grafted Cells

After grafting, the DiI-prelabelled OECs were identified for counting in the tissue sections of the spinal cord by virtue of the red fluorescence of their cell bodies and cytoplasmic processes (Fig. 4.3c & 4.3d). All of the spinal cord tissue sections that were to be used for generating cell counts of DiI+ve cells were stained with Hoechst dye, which stains all of the nuclei in the sections and can be visualized using a DAPI filter which shows the blue fluorescent nuclei. For the red fluorescence to be counted as a DiI+ve cell it had to be associated with a Hoechst+ve nucleus, as illustrated by the arrows in Fig. 4.3c and 4.3d. For the purposes of the cell counts, Dilnegative/Hoechst+ve nuclei were assumed to belong to cells of the host animal. The majority of the DiI fluorescence was confined to fusiform- and spindle-shaped cells (see Fig. 4.3c & 4.3d), and these were the ones that were counted in the experiments of Sections 4.3 to 4.6. Occasionally, red fluorescent particles were found scattered throughout the dorsal funiculi of the tissue sections, being present for the most part within and around the cell graft. These fluorescent particles might have been the remains of dead or dying cells that had been grafted into the spinal cord or the phagocytosed remnants of these cells within the microglia or macrophages that responded to the insertion of the needle into the spinal cord. There was no indication of any other possible leakage of DiI into the intercellular space; for example, there was no Dil staining of the axons of DRG neurons or the corticospinal tract, both of which are located within the dorsal funiculi at the grafting site.

4.3 OEC Migration In The Absence Of Demyelination

OECs were grafted into the right dorsal funiculus of the T12 segment of a control group of rats (Group 1; see Table 4.1), which received only an OEC graft (Table 4.1; Fig. 4.4a & 4.4.b). The rats were killed 2, 4, or 8 weeks after grafting, the spinal cords were sectioned in the coronal plane (40 μ m thick), and the tissue sections stained with Hoechst dye to label the nuclei, as described in Section 2.5.5 and 2.5.6.

Experimental Groups:	Number	Thesis	Type of
	of	section where	Anesthesia
	Animals	results can be	
	Used	found	
Group 1 rats			
(OEC at T12)	9	4.3	Somnotol
Group 2 rats			
(EtBr at T10 – OECs at T12)	9	4.4	Somnotol
Group 3 rats	2		a
(EtBr at T8 – OECs at T12)	9	4.4	Somnotol
Group 4 rats $(EtDr at T12)$	0	15	Samuatal
(ELDI at 112 - OECS at 112)	9	4.5	Sommotor
Group 5 rats			
(EtBr at $T10+12 - OECs$ at $T12$)	9	4 5	Halothane
	,	1.0	11410 thaile
Group 6 rats			
(EtBr at T10 – OECs at T12)			
- minocycline treatment	3	4.6	Halothane
- saline treatment	3		
Group 7 rats			
(EtBr at $T10+12 - OECs$ at $T12$)			
 minocycline treatment 	3	4.6	Halothane
- saline treatment	3		
Group 8 rats			
RV-labelled OECs at T12 pial surface	3	4.7	Halothane

Table 4.1 The experimental groups used to address each Specific Aim

FIGURE 4.4: Schematic diagrams summarizing the experimental groups.

a) The brain and spinal cord of an adult female Wistar rat; b) A schematic diagram representing a graft of OECs (purple oval) in the dorsal funiculus of the T12 spinal cord segment (OECs at T12; see Table 4.1). c) and d) Schematic diagrams representing an EtBr injection (yellow circle) at either T10 (c; EtBr at T10/OECs atT12; see Table 4.1) or T8 (d; EtBr at T8/OECs atT12; see Table 4.1) and the grafting of OECs into the T12 spinal cord segment (purple oval). e) and f) Schematic diagrams representing an OEC graft (purple oval) directly into an EtBr-induced demyelination at T12 (yellow oval) in the absence (e; EtBr at T12/OECs atT12; see Table 4.1) or the presence (f; EtBr at T10, T12/OECs atT12; see Table 4.1) of a second EtBr lesion located at T10 (yellow circle). g) and h) Schematic diagrams representing the experimental design for EtBr injection (yellow circle and yellow oval) and OEC cell graft (purple oval) used to test the effect of minocycline treatment on OEC migration.



4.3.1 The Cell Graft

The cell grafts were easily identified by virtue of their blue, Hoechst-stained nuclei and red, DiI+ve cell bodies and cytoplasmic processes (Fig. 4.3c). More than 200 DiI+ve OECs were present in the dorsal funiculus of each coronal spinal cord section through the midpoint of the rostral-caudal extent of the graft. These DiI+ve cells were so closely packed together that it was impossible to obtain an accurate count of the number of cells. At all survival times a very small number of DiI+ve cells were also found scattered throughout the dorsal funiculus of the T12 spinal cord segment.

4.3.2 Host Glial Cell Reactivity

Both reactive microglia and astrocytes were present in the dorsal funiculus of the T12 segment of the spinal cord, presumably in response to the placement of the cell graft (Fig. 4.3e & 4.3f). The vimentin+ve reactive astrocytes possessed an intensely stained cell body and several long cytoplasmic processes, whereas the OX-18+ve cells had small round somata and a few short, branched cytoplasmic processes. In more rostral segments of the spinal cord in these control rats, the microglial and astrocytic reactivity was less and less apparent in segments further away from the graft and with increasing survival times, but was always confined to the ipsilateral dorsal funiculus. Thus, OX-18+ve reactive microglia and vimentin+ve reactive astrocytes were found within the right dorsal funiculus of T12 at two weeks after grafting (Fig. 4.3e & f). Four (Fig. 4.5a, c, e &g) and eight (Fig. 4.5 b, d, f & h) weeks after cell grafting at T12, OX-42- (Fig. 4.5 e & f) and OX-18-expressing (Fig. 4.5 g & h) astrocytes were also seen within the right dorsal funiculus of T11, occupying the area between the pial surface and the corticospinal tract.

4.3.3 DiI+ve Cells Rostral to Cell Graft

Tissue sections from the rostral, middle and caudal thirds of each spinal cord segment rostral to T12 (i.e. T11, T10, T9 and T8) were randomly chosen for fluorescent microscopy. Four sections from each third of the segment (twelve per segment) were

FIGURE 4.5: Microglial and astrocytic reactivity in the dorsal funiculus of T11 in Group 1 rats.

OX-42+ve (a, b) and OX-18+ve (c, d) microglia, and GFAP+ve (e, f) and vimentin+ve (g, h) astrocytes in the dorsal funiculus of T11 at 4 (a, c, e, g) and 8 weeks (b, d, f, h) after grafting OECs at T12 in Group 1 rats. Minimal microglial and astrocytic reactivity is seen at T11 when the only experimental manipulation was grafting OECs one spinal cord segment caudally. Bar = 400 μ m



used for cell counts. DiI+ve OECs were present within the dorsal funiculus of each spinal cord segment from T11 to T8 (the most rostral segment examined in this study). The average number of these cells within the dorsal funiculus of T11 varied from 17.1 per 40 μ m tissue section at 2 weeks survival (Fig. 4.6a), to 30.9 after 4 weeks (Fig. 4.6b) and to 7.6 per tissue section at 8 weeks after grafting (Fig. 4.6c). In more rostral spinal cord segments (i.e. T8-T10) the numbers of DiI+ve OECs in the Group 1 rats always averaged less than 8 per 40 μ m tissue section regardless of survival time, except for the average cell counts at T10 at the 4 week time point where an average of 12.1 cells per tissue section were counted (Fig. 4.7 - 4.9). The small numbers of migrating DiI+ve OECs that were seen in each spinal cord segment rostral to the cell graft were found on the pial surface and along the septi that lay between the fiber tracts comprising the dorsal funiculi (Fig. 4.10a & b).

4.4 Inducing OECs To Migrate Towards A Focal Demyelination

To address specific aim #1, a gliotoxin-induced demyelination of axons was accomplished by stereotactically injecting 1 μ l of a 0.03% solution of EtBr into the right dorsal funiculus of either the T10 (Group 2 rats) or T8 (Group 3 rats) spinal cord segment of separate groups of adult rats (see Table 4.1). Three days after the EtBr injection, DiI+ve OECs were grafted into the right dorsal funiculus of the T12 spinal cord segment of each rat (Fig. 4.4c & d).

4.4.1 The Cell Graft

In Groups 2 and 3 rats the cell grafts had a similar appearance to that described in Section 4.3.1 for the control group (i.e. Group 1 rats) that had received only a cell graft (data not shown). This cell graft consisted of a closely packed DiI+ve collection of OECs. Although there were more than 200 DiI+ve OECs present in each coronal spinal cord section through the midpoint of the rostral-caudal extent of the graft, no attempt was made to obtain an accurate quantitation since it was impossible to distinguish individual cells throughout much of the compact mass. DiI+ve OECs were also found scattered within the plane of the section in the neural tissue outside of the main body of the graft.

FIGURE 4.6: Number of DiI+ve OECs in the dorsal funiculus of T11 in Groups 1-3 rats.

Histograms depicting the mean number (+/- SEM) of DiI+ve Cells (Y-axis) in the dorsal funiculus of the T11 spinal cord segment of control rats (OECs@T12; Group 1) or rats injected with EtBr at either T8 (EtBr@T8/OECs@T12; Group 3) or T10 (EtBr@T10/OECs@T12; Group 2) at 2 (a), 4 (b) or 8 weeks (c) after grafting of OECs at T12 (n = 3 for each Group at each survival time). A one-way analysis of variance was applied to the data with the data for each survival time being analyzed separately for statistical significance. The resulting value of F (6.56) was statistically significant only at the 8 weeks (c) time point (p < 0.05). Multiple post hoc comparisons between Groups 1-3 at the 8 week time point were performed with Bonferonni's multiple comparison post-test revealing a significant difference between Groups 1 and 2 (p < 0.05; 1 asterisk).



FIGURE 4.7: Number of DiI+ve OECs in the dorsal funiculus of T10 in Groups 1-3 rats.

Histograms depicting the mean number (+/- SEM) of DiI+ve Cells (Y-axis) in the dorsal funiculus of the T10 spinal cord segment of control rats (OECs@T12; Group 1) or rats injected with EtBr at either T8 (EtBr@T8/OECs@T12; Group 3) or T10 (EtBr@T10/OECs@T12; Group 2) at 2 (a), 4 (b) or 8 weeks (c) after grafting of OECs at T12 (n = 3 for each Group at each survival time). A one-way analysis of variance was applied to the data with the data for each survival time being analyzed separately for statistical significance. The resulting values of F were not significant at any time point (p > 0.05).



FIGURE 4.8: Number of DiI+ve OECs in the dorsal funiculus of T9 in Groups 1-3 rats.

Histograms depicting the mean number (+/- SEM) of DiI+ve Cells (Y-axis) in the dorsal funiculus of the T9 spinal cord segment of control rats (OECs@T12; Group 1) or rats injected with EtBr at either T8 (EtBr@T8/OECs@T12; Group 3) or T10 (EtBr@T10/OECs@T12; Group 2) at 2 (a), 4 (b) or 8 weeks (c) after grafting of OECs at T12 (n = 3 for each Group at each survival time). A one-way analysis of variance was applied to the data with the data for each survival time being analyzed separately for statistical significance. The resulting values of F (8.182) were statistically significant only at the 8 weeks time points (p < 0.02). Multiple post hoc comparisons between Groups 1-3 were performed with Bonferonni's multiple comparison post-test revealing a significant difference between Groups 1 and 2 at 8 weeks survival (p < 0.05; 1 asterisk).



FIGURE 4.9: Number of DiI+ve OECs in the dorsal funiculus of T8 in Groups 1-3 rats.

Histograms depicting the mean number (+/- SEM) of DiI+ve Cells (Y-axis) in the dorsal funiculus of the T8 spinal cord segment of control rats (OECs@T12; Group 1) or rats injected with EtBr at either T8 (EtBr@T8/OECs@T12; Group 3) or T10 (EtBr@T10/OECs@T12; Group 2) at 2 (a), 4 (b) or 8 weeks (c) after grafting of OECs at T12 (n = 3 for each Group at each survival time). A one-way analysis of variance was applied to the data with the data for each survival time being analyzed separately for statistical significance. The resulting values of F (18.43) were statistically significant only at the 8 week time point (p < 0.003). Multiple post hoc comparisons between Groups 1-3 were performed with Bonferonni's multiple comparison post-test revealing a significant difference between Groups 1 and 2 (p < 0.05; 1 asterisk) and between Groups 1 and 3 (p < 0.01; 2 asterisks) at 8 weeks survival. However, the numerical differences were therapeutically of any interest.



FIGURE 4.10: Dil+ve cells in the dorsal funiculus of T11 and T9 in Groups 1-3 rats.

Fluorescent images of the dorsal funiculus of the T11 and T9 spinal cord segments of control rats (Group 1) and of rats injected with EtBr at either T10 (Group 2) or T8 (Group 3). All rats survived for 8 weeks after grafting. a & b) In control rats, DiI+ve OECs (white arrow) were seen along the septa at T11 (a) and T9 (b). c & d) In rats injected with EtBr at T10, DiI+ve OECs (white arrows) were seen not only along the septa but also within the neuropil at both T11 (c) and T9 (d). e & f) In rats injected with EtBr at T8, DiI+ve OECs (white arrow) were also found along the septa and in neuropil at both T11 (e) and T9 (f), although they were much fewer in number than those in Group 2 rats. Bar = 100 μ m.



4.4.2 The Ethidium Bromide Lesion.

After injecting EtBr into the right dorsal funiculus, the area of the injection acquired yellow discoloration due to host tissue reaction to needle insertion and injection of the EtBr. Macroscopically, the EtBr lesion was identified by the yellow discoloration (see Section 4.2.1) of the dorsal funiculus immediately to the right of the dorsal spinal artery in tissue sections cut through the middle of either the tenth (T10) or the eighth (T8) spinal cord segments in Groups 2 and 3 rats, respectively. In the rats used in this study, the EtBr lesion was fusiform in shape and about 1 mm long, with the yellow discoloration being most intense at 2 weeks of survival. At 4 and 8 weeks after cell grafting, the intensity of yellow discoloration was less intense and harder to identify. The spinal cord segments with an EtBr injection were identified by comparing the location of the EtBr injection with the location of the laminectomy at the time of spinal cord dissection and spinal rootlet count prior spinal cord segment sectioning. For an entire spinal cord segment (e.g. T10), up to twenty 40 μ m coronal tissue sections contained the demyelinated lesion, which was confined to the right dorsal funiculus without involvement of the corticospinal tract (Fig. 4.2e). By 8 weeks of survival, the right dorsal funiculus of T10 (Group 2 rats) and T8 (Group 3 rats) was close to being fully remyelinated, at least in terms of the histological appearance as seen in tissue sections stained with Luxol Fast Blue (Fig. 4.2f).

4.4.3 Host Glial Cell Reactivity.

Similar to what was seen in the T12 segment of the spinal cord in Group 1 rats, reactive microglia and reactive astrocytes were present in and around the cell graft in the right dorsal funiculus. Microglial and astrocytic reactivity was also observed within the ipsilateral dorsal funiculus of the T11 spinal cord segment and at the site of EtBr injection at T8 (Group 3 rats) and T10 (Group 2 rats). In the rats injected with EtBr at T10 and killed 4 weeks after receiving an OEC graft, the intensity of OX-42 (Fig. 4.11a) and OX-18 (Fig. 4.11c) expressed by microglia within the right dorsal funiculus or adjacent right dorsal horn. At the same survival time, the level of GFAP expression by astrocytes within the right dorsal funiculus was also close to that of the

FIGURE 4.11: Microglial and astrocytic reactivity in the dorsal funiculus of T11 in Group 2 rats.

OX-42+ve (a, b) and OX-18+ve (c, d) microglia, and GFAP+ve (e, f) and vimentin+ve (g, h) astrocytes in the dorsal funiculus of T11 at 4 (a, c, e, g) and 8 weeks (b, d, f, h) after grafting OECs at T12 in Group 2 rats. Minimal microglial and astrocytic reactivity is seen ipsilaterally at T11 by 4 weeks (a, c, e & g) after grafting, but the reactivity as evidenced by OX-18 (d) and vimentin (h) immunostaining is much more intense by 8 weeks. Insets in b, d, f and h show the morphology of reactive microglia (b, d) and astrocytes (f, h). Bar in $h = 400 \mu m$ and applies to a-h. Bar in inset in $h = 20 \mu m$ and applies to all insets.



other parts of the section (Fig. 4.11e). Only vimentin expression by astrocytes localized within the right dorsal funiculus looked more intense in comparison with level of its expression by astrocytes in the left funiculus (Fig. 4.11g). Interestingly, the expression of OX-18 by reactive microglia (Fig. 4.11d) and vimentin by reactive astrocytes (Fig. 4.11h) was noticeably more intense in the rats surviving for 8 weeks to those at shorter survival time. Numerous intensely stained microglia and astrocytes were found throughout much of the extent of the right dorsal funiculus reaching but sparing the right corticospinal tract, left dorsal funiculus and right dorsal horn. Although, microglia and astrocytes expressed OX-42 (Fig. 4.11b) and GFAP (Fig. 4.11f), respectively, at 8 week survival time at more intense, these expressions were not as prominent as that of OX-18 (Fig. 4.11d) and vimentin (Fig. 4.11h).

In the rats that received an EtBr injection at T8, OX-42- (Fig. 4.12a & b) and OX-18-expressing (Fig. 4.12c & d) microglia grouped predominantly within the right dorsal funiculus at both 4 and 8 weeks after grafting cells. GFAP+ve astrocytes (Fig. 4.12e & f) were seen in groups also throughout the extent of the right dorsal funiculus. Single OX-42- and OX-18-positive microglia as well as GFAP-expressing astrocytes lined along the numerous septi of both dorsal funiculi and scattered throughout dorsal horns. However, vimentin+ve astrocytes (Fig. 4.12g & h) were more prominent in the right dorsal funiculus than in adjacent structures at both 4 and 8 week survival period.

4.4.4 DiI+ve Cells Rostral to Cell Graft

DiI+ve OECs were present within the dorsal funiculus of T11, with the numbers of these migrating cells being quite similar in the experimental and control groups at two (Fig. 4.6a) and four (Fig. 4.6b) weeks after grafting. However, by 8 weeks after grafting there were several fold more DiI+ve OECs in the dorsal funiculus of T11 (Fig. 4.6c) and T9 (Fig. 4.8c) in those rats that had received an EtBr injection at T10, in comparison to that seen in either of the other two groups of rats. In contrast to that seen in control rats, the migrating DiI+ve OECs in rats in which EtBr had been injected at T10 were found not only along the septi of both dorsal funiculi but also within the neuropil of both right and left dorsal funiculi of T11 (Fig. 4.10c) and T9 (Fig. 4.10d). Although, the mean numbers of DiI+ve OECs in the dorsal funiculus of T8 at the 8

FIGURE 4.12: Microglial and astrocytic reactivity in the dorsal funiculus of T11 in Group 3 rats.

OX-42+ve (a, b) and OX-18+ve (c, d) microglia, and GFAP+ve (e, f) and vimentin+ve (g, h) astrocytes in the dorsal funiculus of T11 at 4 (a, c, e, g) and 8 weeks (b, d, f, h) after grafting OECs at T12 in Group 3 rats. Minimal microglial reactivity is seen ipsilaterally at T11 regardless of survival time (a-d). There was also no detectable difference in GFAP (e, f) or vimentin (g, h) immunostaining between the 4 and 8 week survival. In contrast to GFAP immunostaining, however, the vimentin immunostaining suggested there was astrocytic reactivity ipsilaterally at T11 at least as early as 4 weeks. Bar = 400 μ m



week time point after grafting in Group 2 and 3 rats were both significantly higher than that of the control group (Fig. 4.9), the numerical difference in cell counts was so small it is questionable whether the difference is therapeutically of any interest.

4.5 Inducing OECs To Migrate Away From A Focal Demyelination

To address specific aim #2, DiI+ve OECs were stereotactically injected into the right dorsal funiculus of the T12 spinal cord segment of 9 rats (Group 4 rats; see Table 4.1) three days after an injection of 1 μ l of a 0.03% solution of EtBr into right dorsal funiculus of T12 (Fig. 4.4e). In order to address specific aim #3, an additional nine rats received injections of 1 μ l of a 0.03% solution of EtBr into right dorsal funiculus of both T10 and T12 (Group 5 rats; see Table 4.1). Three days later, these latter nine rats were also stereotactically injected with DiI+ve OECs into the right dorsal funiculus of the T12 spinal cord segment (Fig. 4.4f).

4.5.1 The Ethidium Bromide Lesions

The EtBr lesion was identified by a yellow-pink discoloration at the middle of the T12 spinal cord segment and on the right side immediately lateral to the dorsal spinal artery. The size of the discoloration was slightly bigger due to additional injection of DiI-labelled OECs but it never exceeded 1-1.2 mm in rostral-caudal extent. Approximately 30 spinal cord sections of the middle portion of T12 contained the lesion and occupied the entire right dorsal funiculus with minimal extension into adjacent structures. The EtBr lesion at T10 of the Group 5 rats was similar in its location and extent to that described in Section 4.4.2.

4.5.2 The Cell Graft.

DiI+ve OECs were grafted into a three day old EtBr lesion within the midportion of the T12 spinal cord segment. Similar to that previously described in Section 4.3.1 with respect to the appearance of the cell graft within undamaged CNS tissue, compact groups of 20 to 30 DiI+ve OECs were seen throughout the right dorsal funiculus, with the total number of grafted cells per tissue section averaging more than 200 at the rostral-caudal midpoint of the graft. However, due to grafting the OECs into an EtBr-induced demyelinated lesion, the OECs were scattered throughout the whole

right dorsal funiculus and within the entire area encompassed by the yellow-pink discoloration of the lesion.

4.5.3 Host Glial Cell Reactivity.

As expected, reactive microglia and reactive astrocytes were both found throughout the whole right dorsal funiculus of the T12 spinal cord segment where OECs had been grafted into an EtBr-induced demyelinated lesion in both Group 4 and 5 rats (data not shown). OX-18-expressing microglia and vimentin+ve astrocytes were also seen occupying the entire right dorsal funiculus of the T10 spinal cord segment in Group 5 rats, which had received a second EtBr injection at T10.

In the T11 spinal cord segment, microglia expressing OX-42 (Fig. 4.13a, c, e & g) and OX-18 (Fig. 4.13b, d, f & h) were single cells with small round cell bodies and elongated, branched processes evenly distributed throughout the dorsal horns and dorsal funiculi. However, OX-42+ve/OX-18+ve microglia with round cell bodies and short or no processes grouped together within the right dorsal funiculus of T11 regardless of the survival time (i.e. 2 or 8 weeks) and presence of one (Group 4 rats) or two (Group 5 rats) EtBr lesions. Astrocytes expressing GFAP (Fig. 4.14a, c, e &g) and vimentin (Fig. 4.14b, d, f & h) were less abundant than microglia in T11 spinal cord sections, but when present they were located mostly in the white matter adjacent to the pial surface. Although GFAP+ve astrocytes grouped together within the right dorsal funiculus of the T11 spinal cord segment of Group 4 and 5 rats after 2 and 8 weeks, the level of GFAP expression was more prominent in rats of both groups that survived 8 weeks. The level of vimentin expression was always high in astrocytes and it accumulated mostly along the septa/blood vessels within the T11 right dorsal funiculus of Group 4 and 5 rats.

4.5.4 DiI+ve Cells Rostral to Cell Graft.

In the rats with only a single EtBr lesion (Group 4) DiI+ve OECs were present in the dorsal funiculus of T11 (Fig. 4.15a, c & e), T9 (data not shown) and as

FIGURE 4.13: Microglial reactivity in the dorsal funiculus of T11 in Group 4 and 5 rats.

OX-42+ve (a, c, e & g) and OX-18+ve (b, d, f & h) microglia in the dorsal funiculus of T11 at 2 (a-d) and 8 weeks (e-h) after grafting OECs at T12 in Group 4 (a, b, e & f) and 5 (c, d, g, & h) rats. Microglial reactivity in the ipsilateral dorsal funiculus of T11 as evidenced by OX-42 immunostaining was noticeably greater in the double-lesioned Group 5 rats at both 2 (c) and 8 (g) weeks after OEC grafting as compared to that seen in Group 4 rats (a, e). The OX-18 immunostaining suggested a greater microglial reactivity at 8 weeks (f, h) in both groups as compared to the 2 week survival (b, d). Bar = 400 μ m



FIGURE 4.14: Astrocytic reactivity in the dorsal funiculus of T11 in Group 4 and 5 rats.

GFAP+ve (a, c, e & g) and vimentin+ve (b, d, f & h) astrocytes in the dorsal funiculus of T11 at 2 (a-d) and 8 weeks (e-h) after grafting OECs at T12 in Group 4 (a, b, e & f) and 5 (c, d, g, & h) rats. Astrocytic reactivity in the ipsilateral dorsal funiculus of T11 as evidenced by GFAP immunostaining was noticeably greater at 8 weeks survival in both Group 4 (a, e) and 5 (c, g) rats as compared to that seen at 2 weeks. In the doublelesioned Group 5 rats the vimentin immunostaining suggested a more intense astrocytic reactivity at 2 weeks (d) after grafting as compared to 8 weeks (h) and was also more intense than that seen in Group 4 rats (b) at 2 weeks survival. However, the astrocytic reactivity in Group 4 and 5 rats was comparable at 8 weeks survival as evidenced by vimentin immunostaining (f, h). Bar = 400 μ m



FIGURE 4.15: DiI+ve cells in the dorsal funiculus of T11 in Group 4 and 5 rats.

Fluorescent images of the dorsal funiculus of the T11 spinal cord segment of rats that were injected with EtBr at either T12 (Group 4) or at both T10 and T12 (Group 5) and three days later received an OEC graft into the lesion at T12. The rats survived for 2 (a, b), 4 (c, d) and 8 (e, f) weeks after grafting. Consistently at each survival time there were several fold more DiI+ve cells (white arrows) in the dorsal funiculus of T11 in Group 5 rats (b, d, &f) as compared to that seen in Group 4 rats (a, c & e). Bar = 100 μ m.



far rostrally as T8 (data not shown), which was the most rostral spinal cord segment examined. In the single lesioned rats (Group 4), however, the DiI+ve OECs had migrated singly or at most as pairs of cells. The number of such cells at the levels of T11 and T10 was consistently lower at two (Fig. 4.16a & 4.17a) and four (Fig. 4.16b & 4.17b) weeks postoperatively than had been observed in Groups 2 and 3 rats at equivalent survival times (1.7 - 6.8 fold fewer cells) and was similar to that seen in the latter two groups of rats at the T9 (Fig. 4.18) and T8 (Fig. 4.19) spinal cord segments at two and four weeks after grafting. Even by 8 weeks postoperatively, the average number of DiI+ve OECs in the dorsal funiculus of T11 (Fig. 4.16) to T8 (Fig. 4.19) remained relatively low.

However, the numbers of DiI+ve OECs were significantly higher in the spinal cord segments of rats with two EtBr lesions (i.e., at both T12 and T10) (Fig. 4.15b, d & f). The average number of DiI+ve OECs in the T11 spinal cord segment of Group 5 rats was 71.8 cells per tissue section as soon as 2 weeks after grafting (Fig. 4.16a). Mean numbers of DiI+ve OECs remained high in Group 5 rats even up to 8 weeks (50.4 cells) after grafting (Fig. 4.16c). These numbers were significantly higher than those rats that had received only a single EtBr injection at T12 (i.e. Group 4 rats) regardless of survival time (2, 4, or 8 weeks) and in comparison to those in control rats at both 2 and 8 weeks survival. Proceeding rostrally, there were also significantly more DiI+ve OECs in the dorsal funiculus of T10 at 4 (39.7 DiI+ve OECs) and 8 (20.9 DiI+ve cells) weeks after grafting in Group 5 rats as compared to Group 4 rats (Fig. 4.17). The average number of DiI+ve cells at T9 (5.5 - 12.8; Fig. 4.18) and at T8 (4.9 - 6.4; Fig. 4.19) in Group 5 rats was not significantly different from that of the other groups regardless of survival time, except for the cell counts at T8 for the 2 week time point (Fig. 4.19). However, the actual numerical difference between the cell counts for Groups 4 and 5 at T8 was so small it is questionable whether they are of any therapeutic interest. Either singly or in small groups the DiI+ve OECs were found along the septi of both dorsal funiculi and also within the neuropil of the right dorsal funiculus of T11-T10 in both Groups 4 and 5 rats.

FIGURE 4.16: Number of DiI+ve OECs in the dorsal funiculus of T11 in Groups 1, 4 and 5 rats.

Histograms depicting the mean number (+/- SEM) of DiI+ve Cells (Y-axis) in the dorsal funiculus of the T11 spinal cord segment of control rats (OECs@T12; Group 1) or rats injected with EtBr at either T12 (EtBr@T12/OECs@T12; Group 4) or T10 and T12 (EtBr@T10, T12/OECs@T12; Group 5) 2 (a), 4 (b) or 8 weeks (c) after grafting of OECs at T12 (n = 3 for each Group at each survival time). A one-way analysis of variance was applied to the data with the data for each survival time being analyzed separately for statistical significance. The resulting values of F were statistically significant at the 2 (F = 12.51; p < 0.0075), 4 (F = 11.98; 0.0085) and 8 (F = 14.17; p < 0.006) weeks time points. Multiple post hoc comparisons between Groups 1, 4 and 5 at each time point were performed with Bonferonni's multiple comparison post-test revealing a significant difference between Groups 4 and 5 at all survival times and Groups 1 and 5 at 2 and 8 weeks survival times. 1 asterisk = p < 0.05; 2 asterisks = p < 0.01.


FIGURE 4.17: Number of DiI+ve OECs in the dorsal funiculus of T10 in Groups 1, 4 and 5 rats.

Histograms depicting the mean number (+/- SEM) of DiI+ve Cells (Y-axis) in the dorsal funiculus of the T10 spinal cord segment of control rats (OECs@T12; Group 1) or rats injected with EtBr at either T12 (EtBr@T12/OECs@T12; Group 4) or T10 and T12 (EtBr@T10, T12/OECs@T12; Group 5) 2 (a), 4 (b) or 8 weeks (c) after grafting of OECs at T12 (n = 3 for each Group at each survival time). A one-way analysis of variance was applied to the data with the data for each survival time being analyzed separately for statistical significance. The resulting values of F were statistically significant at the 4 (F = 6.025; p < 0.04) and 8 (F = 22.02; p < 0.002) weeks time points. Multiple post hoc comparisons between Groups 1, 4 and 5 were performed with Bonferonni's multiple comparison post-test revealing a significant difference between Groups 4 and 5 at 4 and 8 weeks survival times, as well as between Groups 1 and 5 at only the 2 week time point. 1 asterisk = p < 0.05; 2 asterisks = p < 0.01.



FIGURE 4.18: Number of DiI+ve OECs in the dorsal funiculus of T9 in Groups 1, 4 and 5 rats.

Histograms depicting the mean number (+/- SEM) of DiI+ve Cells (Y-axis) in the dorsal funiculus of the T9 spinal cord segment of control rats (OECs@T12; Group 1) or rats injected with EtBr at either T12 (EtBr@T12/OECs@T12; Group 4) or T10 and T12 (EtBr@T10, T12/OECs@T12; Group 5) 2 (a), 4 (b) or 8 weeks (c) after grafting of OECs at T12 (n = 3 for each Group at each survival time). A one-way analysis of variance was applied to the data with the data for each survival time being analyzed separately for statistical significance. The resulting values of F were not significantly different at any survival time.



FIGURE 4.19: Number of DiI+ve OECs in the dorsal funiculus of T8 in Groups 1, 4 and 5 rats.

Histograms depicting the mean number (+/- SEM) of DiI+ve Cells (Y-axis) in the dorsal funiculus of the T8 spinal cord segment of control rats (OECs@T12; Group 1) or rats injected with EtBr at either T12 (EtBr@T12/OECs@T12; Group 4) or T10 and T12 (EtBr@T10, T12/OECs@T12; Group 5) 2 (a), 4 (b) or 8 weeks (c) after grafting of OECs at T12 (n = 3 for each Group at each survival time). A one-way analysis of variance was applied to the data with the data for each survival time being analyzed separately for statistical significance. The resulting values of F (8.429) were statistically significant (p < 0.02) only at the 2 week time point. Multiple post hoc comparisons between Groups 1, 4 and 5 at the 2 week time point were performed with Bonferonni's multiple comparison post-test revealing a significant difference between Groups 4 and 5 (p < 0.05; 1 asterisk). However, the numerical differences were therapeutically of any interest.



4.6 Microglial Reactivity Contributes To The Migratory Signal

To address specific aim #4, systemically administered minocycline was used to reduce the activation of microglia both at the demyelinating lesion(s) and along the white matter extending throughout the ipsilateral dorsal funiculus of T11. Six rats (Group 6; see Table 4.1) received an injection of 1 µl of 0.03% EtBr in the right dorsal funiculus of the T10 spinal cord, followed three days later by an injection of DiIlabelled OECs into the right dorsal funiculus of the T12 spinal cord segment (Fig. 4.4g). An additional six rats (Group 7; see Table 4.1) received an injection of 1 µl of 0.03% EtBr in the dorsal funiculus of both the T10 and the T12 spinal cord segments, followed three days later by an injection of DiI-+ve OECs into the right dorsal funiculus of the T12 spinal cord segment (Fig. 4.4h). Half of the rats in each group received daily intraperitoneal injections of minocycline while the remainder rats received vehicle (see Section 2.4.7) for the first four weeks after grafting the cells. This four week treatment schedule was chosen to ensure the anti-inflammatory actions of minocycline would be active for at least the first four weeks since it was during this time that the first signs of lesion-induced migration of OECs was seen in the experiment of Section 4.4. All rats in both Groups 6 and 7 were killed 8 weeks after receiving an OEC graft.

4.6.1 Ethidium Bromide Lesion at T10 (Group 6 rats)

The spinal cord tissues of Group 6 rats were stained with antibodies that identified reactive microglia (OX42 and OX18 monoclonal antibodies) and reactive astrocytes (anti-GFAP and anti-vimentin antisera) to confirm that the minocycline treatment had been effective in preventing the activation of microglia and in assessing the treatment's effect on astrogliosis. In the saline-treated, OX-42 (Fig 4.20a) and OX-18-expressing (Fig. 4.20c) microglia were located in groups occupying the entire right dorsal funiculus of T11. Visibly fewer OX-42+ve (Fig. 4.20b) and small OX-18+ve (Fig. 4.20d) microglia with short processes were identified within the T11 right dorsal funiculus of Group 6 rats treated with minocycline. No cell counts of microglia or other tests to determine the level of OX-42 and OX-18 expression were performed on spinal cord tissue of Group 6 rats treated with either saline or

FIGURE 4.20: Microglial and astrocytic reactivity in the dorsal funiculus of T11 in Group 6 rats.

OX-42+ve (a, b) and OX-18+ve (c, d) microglia, and vimentin+ve (e, f) astrocytes in the dorsal funiculus of T11 at 8 weeks after grafting OECs at T12 in Group 6 rats receiving minocycline (b, d & f) or vehicle (a, c & e) for the first 4 weeks after EtBr injection at T10. The minocycline treatment reduced the microglial reactivity as evidenced by both OX-42 (b) and OX-18 (d) immunostaining as compared to vehicle controls (a, c). There was no noticeable difference in the density of vimentin+ve astrocytes between the vehicle and minocycline treated groups, although the mediallateral extent of the astrocytic reactivity was smaller after minocycline treatment. Bar = $400 \mu m$



minocycline. Vimentin+ve astrocytes were primarily identified in the spinal cord tissue adjacent to the pial surface and within the right dorsal funiculus of the Group 6 rats regardless of treatment with saline (Fig. 4.20e) or minocycline (Fig. 4.20f).

Tissue sections from the T11 spinal cord segment of each Group 6 rat were selected from the rostral, middle and caudal thirds of the segment and examined to count the number of DiI+ve cells. In Group 6 rats treated with saline, the numbers of DiI+ve OECs found within the T11 dorsal funiculus averaged 49.3 per 40 µm tissue section (Fig. 4.21a), values that were similar to those of Group 2 rats that survived for 8 weeks. However, the average number of DiI+ve OECs found in the same location in Group 6 rats treated with minocycline for the first 4 weeks after EtBr injection was only 6.8 per 40 µm tissue section, which was more than 7 fold less than that observed in the vehicle control rats. Mostly single DiI+ve OECs were localized to septi/blood vessels of the right dorsal funiculus of Group 2 rats, single and small groups of DiI+ve OECs were found throughout the neuropil as well as along the septi predominantly of the right dorsal funiculus of the saline-treated rats of Group 6 (Fig. 4.21c).

4.6.2 Ethidium Bromide Lesions at T10 and T12 (Group 7 rats)

Tissue sections from the rostral, middle and caudal thirds of T11 in Group 7 rats were examined for the presence and the location of reactive microglia and reactive astrocytes. Similar to that seen in Group 2 and 6 rats after 8 week survival, OX-42-expressing microglia scattered throughout the grey and white matter as single cells with small round cell body and short, branched processes (Fig. 4.22a & b). Under light microscopy, there were visibly more OX-42 (Fig. 4.22a) and OX-18+ve (Fig. 4.22c) microglia present within the right dorsal funiculus of Group 7 rats treated with saline than that observed in corresponding spinal cord area of Group 7 rats treated with minocycline (Fig. 4.22b &d). Neither cell counts of microglia nor other tests to determine the level of OX-42 and OX-18 expression were performed on spinal cord tissue of Group 7 rats treated with either saline or minocycline. GFAP+ve astrocytes were distributed throughout the grey and white matter of T11 spinal cord sections with

FIGURE 4.21: Dil+ve OECs in the dorsal funiculus of T11 in Group 6 rats.

a) A histogram depicting the mean number (+/- SEM) of DiI+ve cells (Y-axis) in the dorsal funiculus of the T11 spinal cord segment of rats injected with EtBr at T10 (EtBr@T10/OECs@T12; Group 6) and OECs at T12 and treated with either saline or with minocycline for the first 4 weeks after EtBr injection (n = 3 for each Group). The Mann-Whitney U test was applied to the data. The resulting value of U (9.0) was statistically significant (p < 0.05; 1 asterisk) thus supporting the hypothesis of the importance of the microglial reactivity in contributing to the strength of the migratory signal(s). b & c) Fluorescent images of the dorsal funiculus of the T11 spinal cord segment of Group 6 rats. All rats survived for 8 weeks after grafting at which time the saline treated rats (c) had several fold more DiI+ve cells (white arrows) in the dorsal funiculus of T11 as compared to that of minocycline treated rats (b). Bar = 100 μ m and applies to both 'b' and 'c'.



FIGURE 4.22: Microglial and astrocytic reactivity in the dorsal funiculus of T11 in Group 7 rats.

OX-42+ve (a, b) and OX-18+ve (c, d) microglia, and GFAP+ve (e, f) and vimentin+ve (g, h) astrocytes in the dorsal funiculus of T11 at 8 weeks after grafting OECs at T12 in Group 7 rats. The minocycline treatment dramatically reduced the microglial reactivity as evidenced by both OX-42 (b) and OX-18 (d) immunostaining as compared to vehicle controls (a, c). There was no significant difference in the density of GFAP+ve (e, f) or vimentin+ve (g, h) astrocytes between the vehicle and minocycline treated groups. Bar = 400 μ m



some accumulating in groups within the right dorsal funiculus of Group 7 rats treated with either saline (Fig. 4.22e) or minocycline (Fig. 4.22f). There was an accumulation of vimentin-expressing astrocytes in the right dorsal funiculus and at the pial surface of T11 of all Group 7 rats regardless of treatment (Fig. 4.22g & h). In contrast to observed differences in microglia distribution, there was no such effect of minocycline on either GFAP- or vimentin-expressing astrocytes in comparison to that after saline treatment.

The location and the numbers of DiI+ve OECs were also determined in the rostral, middle and caudal thirds of the T11 spinal cord segment in Group 7 rats. In the three saline-treated rats of this group, the number of DiI+ve OECs averaged 60.4 per 40 µm tissue section, which was similar to that observed in Group 5 rats at 8 weeks survival (Fig. 4.23a). However, the average numbers of DiI+ve OECs found in the right dorsal funiculus of T11 in minocycline-treated Group 7 rats was significantly lower, being only 19.9 DiI+ve OECs per 40 µm tissue section, which was 3 fold less than that observed in saline-treated rats. Single DiI+ve OECs were found migrating along septi of T11 dorsal funiculi in Group 7 rats treated with minocycline (Fig. 4.23b). However, migrating DiI+ve OECs were found throughout the neuropil and along septi of the T11 spinal cord segment of the Group 7 rats injected with saline instead (Fig. 4.23c).

4.7 Grafting Retroviral-Infected OECs

To address specific aim #5, OECs were infected with a replication-deficient retrovirus (RV) containing the Lac-Z gene and a neomycin-resistance gene (Fig. 4.24). Geneticin-selected LacZ-expressing OECs were either injected into the right dorsal funiculus of the T12 spinal cord segment or grafted onto the pial surface at the level of the T12 spinal cord segment.

4.7.1 Retroviral-Infected OECs In Vitro

OEC cultures were obtained from the olfactory bulbs of E18 rat embryos as described in Section 2.1.1. After 10 days in mG5 medium (Fig. 4.24a), the cells were exposed to the RV (Fig. 4.24b). One control flask from each set of cultures was fed with DMEM/F12/10%FBS medium containing no RV (Fig. 4.24c). After the two week

FIGURE 4.23: Dil+ve OECs in the dorsal funiculus of T11 in Group 7 rats.

a) A histogram depicting the mean number (+/- SEM) of DiI+ve cells (Y-axis) in the dorsal funiculus of the T11 spinal cord segment of rats injected with EtBr at both T10 and T12 (EtBr@T10, T12/OECs@T12; Group 7) and OECs at T12 and treated with either saline or with minocycline for the first 4 weeks after EtBr injections (n = 3 for each Group). The Mann-Whitney U test was applied to the data. The resulting value of U (9.0) was statistically significant (p < 0.05; 1 asterisk) thus supporting the hypothesis of the importance of the microglial reactivity in contributing to the strength of the migratory signal(s). b & c) Fluorescent images of the dorsal funiculus of the T11 spinal cord segment of Group 7 rats. All rats survived for 8 weeks after grafting at which time the saline treated rats (c) had several fold more DiI+ve cells (white arrows) in the dorsal funiculus of T11 as compared to that of minocycline treated rats (b). Bar = 100 μ m and applies to both 'b' and 'c'.



FIGURE 4.24: Schematic description of retroviral infection of OEC cultures.

(a) OEC cultures in serum-free medium (mG5). b) RV-infection starts when the OEC cultures are switched from mG5 to serum-containing medium. c) As a control, selected flasks are fed with medium containing no RV supernatant. d & e) Both control and RV-infected flasks are fed with medium containing geneticin for the duration of time *in vitro*. Cells in the control flasks were killed by the geneticin treatment within 4-5 days *in vitro*. g) RV-infected and geneticin-selected OECs are frozen for long term storage in the vapor phase of liquid nitrogen. f & h) x-Gal staining of representative OECs from an RV-infected cell culture at the time they were harvested for freezing (f) and in a cell culture initiated from a cell suspension of OECs that had been grafted onto the pial surface of the spinal cord (h). Bar = 200 μ m



period, both infected (Fig. 4.24d) and uninfected control (Fig. 4.24e) cells were exposed to geneticin. Close monitoring of control cultures determined that treatment with geneticin for 4-5 days killed 99% cells in uninfected cultures. At this time, 50-70% of the cells in cultures infected with RV survived the geneticin treatment. Then, OEC cultures were expanded with the DMEM/F12/10%FBS medium with no geneticin. Cell cultures approached confluency within one month. In order to eliminate persisting uninfected cells if present, OEC cultures were exposed to geneticin for 2 additional days at this time. After the second treatment with geneticin, OEC cultures from each set were harvested to assay the cells for β -galactosidase expression. Every single cell in all OEC cultures with a fusiform cell body and flat processes contained blue particles, the protein products of β -galactosidase activity, which densely packed the entire cell cytoplasm but not cell nuclei (Fig. 4.24f). The remaining OEC cultures were then harvested for storage at -70°C (Fig. 4.24g). RV-infected OECs were used later for implantation in experiments described in Sections 4.7.2 and 4.7.3. After grafting, the remaining cell suspensions were plated onto glass coverslips and grown in vitro for 48 hours to assay the Lac-Z-expression in OECs after freezing and thawing. As prior to freezing, all OECs expressed β -galactosidase as seen by the presence of blue particles throughout the cell cytoplasm but not in cell nuclei (Fig. 4.24h).

4.7.2 OEC Grafts Into Dorsal Funiculus

Prior to addressing the Specific Aim #5, RV-infected OECs were grafted into the lower thoracic spinal segment in order to confirm that β -Gal-expressing OECs could be detected *in vivo*. RV-infected OECs were stereotactically injected into the right dorsal funiculus of the T12 spinal cord segment of one rat. This rat did not receive any other experimental manipulations and was sacrificed 4 days after grafting. The rat was perfused with 4% paraformaldehyde and free-floating 40 µm thick coronal sections were cut through the spinal cord. The tissue sections were stained with x-Gal (to identify β -galactosidase+ve cells) and counterstained with Fast Red (to identify all cells in the tissue section) (see Section 2.5.8). The grafted OECs occupied the entire right dorsal funiculus at the grafting site and were easily identified by virtue of their blue, x-Gal-staining (Fig. 4.25a). Within the graft, OECs were closely packed in groups of 20-

FIGURE 4.25: β-Gal+ve OECs in the spinal cord, brainstem and cerebellum

a & b) These micrographs are of the T12 segment of the spinal cord in a rat that had received an injection of OECs expressing β -Gal into the dorsal funiculus of this segment 4 days previously. Numerous β -Gal+ve cells (blue staining) can be seen (arrows in 'b'). c-f) After being grafted onto the pial surface at the level of T12, β -Gal+ve cells could be seen in close association with blood vessels in rostral spinal cord segments (e.g. the dorsal horn of T9 as shown in 'c'). These β -Gal+ve were also found along the pial surface of the cerebellum (d), which presumably represents one possible route by which the cells could gain access to the neuropil of the CNS such as the molecular layer of the cerebellum (e). The β -Gal+ve cells were also seen associated with blood vessels in the tegmentum of the brainstem (f). All of these sections were counter stained with Neutral fast Red. Bar = 400 µm (a), 400 µm (b, d), and 200 µm (c, e & f).



30. X-Gal stained OECs had round cell bodies with short processes (Fig. 4.25b).

4.7.3 OEC Grafts Onto Pial Surface

RV-infected OECs were also grafted onto the pial surface of the right dorsal funiculus of the T12 spinal cord segment of three rats, with one rat being killed at each of 11 days, 1 month and 2 months after grafting. These rats received no other experimental manipulation aside from the OEC graft. In the spinal cord tissue obtained from the cervical and thoracic levels of the rat killed at 11 days after grafting, β -Gal+ve cells were distributed across the coronal plane of the spinal cord sections with no particular pattern. These cells were lying along the septi/blood vessels of the dorsal, lateral and anterior funiculi, but were never seen lying close to the pial surface of the spinal cord. The β -Gal+ve cells were found singly or in small groups of 2-4 cells. Within the white matter, β -Gal+ve cells were sometimes found within the perivascular space in close apposition to the endothelial lining of blood vessels. Some β -Gal+ve cells were also found in the grey matter, including the dorsal and anterior horns and Clark's nucleus. Within the grey matter the β -Gal+ve cells were found in close apposition to blood vessels whereas others were closely apposed to neuronal cell bodies. The RV-infected OECs very likely had floated into the CSF shortly after the graft had been placed on the pial surface, since no cells were found on the pial surface of the spinal cord of any tissue section examined. The number of migrating cells varied from 40 to 50 cells per section at the rostral part of T12 to 20-30 per section at the midportion of T8 to 5-10 per section in the lower medulla. Cell counts for sections of rostral part of T12 and for sections from the middle part of T8 are presented in Table 4.2. The medulla was the most rostral part of the CNS examined in this rat. No cell count was performed on tissue sections from cervical and lumbar segments of this rat.

In the remaining rats, one killed one month after grafting and the second rat at two months, β -Gal+ve cells were found within the white and grey matter of the cervical, thoracic and lumbar spinal cord segments in a similar distribution to that described above for the rat killed 11 days after receiving a pial surface graft. In addition, the brain stem, cerebellum, cerebral hemispheres and olfactory bulbs were also examined to determine whether they contained any β -Gal+ve cells. β -Gal+ve OECs were found in

the white and grey matter of the brain stem where they were found along septi/blood vessels or located close to neuronal cell bodies. A few single β -Gal+ve cells were also found within the olfactory bulbs and on the pial surface of the cerebellum as well as within the Purkinje cell layer (Fig. 4.25d). β -Gal+ve OECs were not seen in the cortex or white matter of the frontal, parietal and temporal lobes of either rat brain in coronal sections collected from the anterior pole of the frontal lobe back to the level of the optic chiasm. β -Gal+ve OECs were also absent from the basal ganglia and diencephalon.

Spinal Cord Segment	T12, rostral part			T8, middle part		
Section:	1 st	2 nd	3 rd	1 st	2 nd	3 rd
WHITE MATTER						
Anterior funiculus	4	3	5	3	3	3
Right lateral funiculus	3	0	1	3	2	1
Left lateral funiculus	2	0	1	2	1	3
Dorsal funiculi	3	0	1	5	5	0
GREY MATTER						
Right anterior horn	9	7	4	4	6	7
Left anterior horn	8	7	8	6	4	6
Right dorsal horn including	5	8	8	4	3	2
Clark's nucleus						
Left dorsal horn including	8	3	5	2	1	3
Clark's nucleus						

Table 4.2Number of β-Gal-expressing OECs found in the T12 and T8 spinal
cord segments after a pial surface graft

5 **DISCUSSION.**

5.1 Animal Models of CNS Demyelination for Studying the *In Vivo* Migratory Ability of Myelinating Glia

Several animal models have been developed to study the morphological, cellular and molecular mechanisms regulating demyelination and remyelination of CNS axons. Each model is designed to mimic some, but not all, of the pathological and neurological sequelae of MS. The most commonly used animal models (see Section 1.6) are: a) toxin-induced demyelination, such as with EtBr (Yajima and Suzuki, 1979), lysolecithin (Hall, 1972) or cuprizone (Blakemore, 1972, 1973; Matsushima and Morell, 2001); b) autoimmune-induced demyelination, the most common being experimental autoimmune encephalomyelitis (EAE) (Lyman et al., 1985; Traugott et al., 1985); and c) viralinduced demyelination [e.g., hepatitis virus (Herndon et al., 1977) or murine viral encephalomyelitis (Lang et al., 1984)]. Although each of these models are useful for addressing some of the questions concerning the mechanisms driving the demyelination and remyelination of CNS axons, most are of limited use in identifying factors important for inducing migration of the cells needed to assist in neural repair. Cell migration towards the site of injury is an important area of concern as part of any therapeutic approach to treating hypomyelinating or demyelinating disorders that involve glial cell transplantation. The grafted cells must be able to respond to migratory signals directing them to migrate towards focal areas of demyelination, such as the isolated plaques in the spinal cord of an MS patient. For this research project, the EtBr model of CNS demyelination was chosen because with the concentration (0.03%) and volume $(1 \mu l)$ of injected neurotoxicant chosen both the location and the size of the lesion could be controlled.

The EtBr model used in this research project was modified in two major ways from that used by other investigators (Yajima and Suzuki, 1979; Blakemore, 1984;

Iwashita and Blakemore, 2000). Firstly, the neurotoxicant was injected at a concentration [i.e., 0.03% as opposed to 0.1% (Yajima and Suzuki, 1979; Franklin et al., 1996; Iwashita and Blakemore, 2000) or 0.3% (Honmou et al., 1996)] that demyelinated only one half of the dorsal funiculus over a rostral-caudal length of less than 1 mm. Secondly, x-irradiation was not used thus permitting host glial cells to respond properly to the lesion and initiate neural repair. Previous investigators have made much larger EtBr-demyelinated lesions (7-8 mm in length) in the spinal cord of adult rats in combination with x-irradiation (Akiyama et al., 2001, 2002a, 2002b; Blakemore and Crang, 1985; Franklin et al., 1996; Honmou et al., 1996) to study the migratory ability of oligodendrocyte progenitor cells (Duncan et al., 1992; Franklin and Blakemore, 1997; Groves et al., 1993; Nait-Oumesmar et al., 1999; O'Leary and Blakemore, 1997; Reynolds et al., 2001; Warrington et al., 1993) and Schwann cells (Baron-Van Evercooren et al., 1992; Blakemore and Crang, 1985, 1989; Blakemore et al., 1995; Franklin et al., 1997; Gilson and Blakemore, 2002). In spite of the larger size of these lesions, there was no detectable Schwann cell migration nor did the oligodendrocyte progenitor cells migrate more than a couple of mm towards the lesion (Baron-Van Evercooren et al., 1992; Blakemore et al., 2002; Chari et al., 2003; Franklin et al., 1996, 1997; Franklin and Blakemore, 1997). Other than these few studies, there has been no systematic examination of the factors inducing or guiding the migration of myelinating glia using an *in vivo* model in the CNS of adult mammals. The EtBr model used in the present research project can be used to study the migratory ability of myelinating glia in response to single or multiple small, focal (<1 mm long) gliotoxin-induced demyelinated lesions in the dorsal funiculus (unilaterally) of the adult rat spinal cord. Due to the small size of the lesions, the EtBr can even be injected at variable distances from the site of the graft. With this EtBr model it will be possible to identify factors that induce or guide the migration of myelinating glia, as was done in the present project where microglial reactivity was identified as one potential source of a migratory signal inducing OECs to migrate over distances of up to 20 mm towards a more rostrally located focal demyelination.

This research project focused on the migratory abilities of OECs grafted into the adult rat spinal cord in either the presence or absence of a focal demyelinated lesion(s). One µl of a 0.03% EtBr solution was injected at different levels of the lower thoracic spinal cord (i.e., at T8, T10 or T12, or at both T10 and T12) in separate groups of rats. Three days after the EtBr injection, OECs were grafted into the right dorsal funiculus of the T12 spinal cord segment. The experimental design gave control over both the location and the size of the demyelinated lesion as well as the distance of the lesion from the OEC graft. This EtBr model was used to study: a) the migration of OECs towards a single demyelinated lesion located either two or four spinal cord segments away from the cell graft; b) the migration of OECs away from a focal EtBr-demyelinated area of CNS white matter in the presence and absence of a second EtBr lesion located two spinal cord segments away; c) the contribution of microglial reactivity to the migratory signal(s) inducing OECs to migrate when EtBr was used to make a focal demyelinated lesion in CNS white matter.

In order to study the migratory ability of any type of myelinating glia grafted into an animal's CNS, it is important to have a model in which the number, location and size of the demyelinated lesions can be reproducibly controlled. The EtBr model used in this research project has several advantages over the models used by other investigators to study demyelination and remyelination of CNS axons. First, the precise location of the EtBr injection can be controlled as well as the size of the demyelinated area. Second, by eliminating x-irradiation from the experimental design, host glia can contribute to neural repair at the site of the EtBr injection thus contributing to the generation of a more natural response by cells of the host animal. By allowing the host glial cells to contribute to neural repair, the dorsal funiculus at the site of EtBr injection appeared to be almost completely repaired by 8 weeks, as judged by Luxol Fast Blue staining (Fig. 4.2f). With these modifications included in the EtBr model used in this research project, reactive microglia were identified as one potential source of a migratory signal inducing OECs to migrate towards a focal EtBr-induced demyelination. In the autoimmune- or viral-induced *in vivo* animal models of CNS demyelination the lesions are typically induced by injections of MBP or MOG (i.e., the EAE model; Raine *et al.*, 1974, 1977, Raine and Traugott, 1983, 1984; Steinman, 1999) or by infecting the animal with Theiler's virus (murine encephalomyelitis model; Dal Canto *et al.*, 1996; Lipton and Jelachich, 1997; Monteyne *et al.*, 1997; Oleszak *et al.*, 1995; Tsunoda and Fujinami, 1996), respectively. The demyelinated lesions in these animal models are found scattered throughout the CNS white matter with no consistent location or size. It would be extremely difficult to identify factors regulating the migration of any of the myelinating glia when both the location and the size of the demyelinated lesion cannot be consistently controlled. Therefore, a direct injection of a chemical agent (e.g., EtBr) into the CNS white matter is much more appropriate for creating a focal demyelination in each animal that is in a precise location and is of a predetermined size. If these criteria are met, then the myelinating glia can be injected at variable distances from the focal demyelination to study their migratory abilities and the signal(s) regulating their migration.

5.2 Migration of Myelinating Glia in the CNS

Remyelination of CNS axons remains an important goal of researchers who are working towards developing therapeutic approaches to treating neurological diseases such as MS and spinal cord injury. With respect to demyelinating diseases, any therapeutic approach to treating hypomyelinating or demyelinating disorders that involve the migration of myelinating glia must take into account factors responsible for inducing or guiding not only the migration of grafted cells but also that of the host myelinating glia (i.e. oligodendrocyte progenitors and Schwann cells) whose remyelinating ability is known to decrease as one gets older (Ando *et al.*, 2003; Berlet and Volk, 1980; Chari *et al.*, 2003; Franklin *et al.*, 2002c). Any therapeutic approach must also take into account that in demyelinating diseases such as MS the endogenous migratory signals, which direct the migration of host and grafted myelinating glia towards focal areas of demyelination, are actually arising from numerous spatially separated lesions, such as for example the isolated plaques in the spinal cord of an MS

patient. It is important to have a better understanding of these migratory signals and/or be able to induce them so appropriate therapeutic approaches can be developed to enhance the migration of myelinating glia without impeding the process of neural repair.

5.2.1 Host Myelinating Glia

In order for spontaneous remyelination to occur either adjacent oligodendrocytes have to remyelinate the axons or oligodendrocyte progenitor cells must migrate into the lesion. Within the oligodendrocyte lineage, oligodendrocyte progenitors have the highest migratory abilities (Gansmuller et al., 1991; Small et al., 1987). Endogenous oligodendrocyte progenitor cells located in the subventricular zone respond to a lysolecithin-induced demyelination in the corpus callosum by migrating towards the lesion and expressing PSA-NCAM all the way to the lesion (Nait-Oumesmar *et al.*, 1999). The migration of the oligodendrocyte progenitors can be facilitated by FGF-2 and PDGF (Lachapelle et al., 2002) and neutrin-1 (Spassky et al., 2002). On the other hand, oligodendrocyte progenitor migration can be inhibited by the chemokine CXCL1 (Tsai et al., 2002), by apotransferrin (Paez et al., 2002) and by class 3 semaphorins (Spassky et al., 2002). Several studies indicate that oligodendrocyte progenitors are present in the demyelinating lesions of an MS patient's CNS (Chang et al. 2002; Wolswijk, 2002) indicating that in spite of the slower recruitment of progenitors that occurs as one gets older this is not the sole obstacle impeding remyelination in the plaques in MS.

However, there is an age-related decline in the rate of recruitment and differentiation of oligodendrocyte progenitor cells in the repair of a demyelinated lesion (Ando *et al.*, 2003; Berlet and Volk, 1980; Chari *et al.*, 2003; Franklin *et al.*, 2002), which is of obvious relevance to MS patients who may suffer for several decades. Franklin *et al.* (2002) and Sim *et al.* (2002) suggested that impairment both of the recruitment of oligodendrocyte progenitors cells into the lesion as well as the differentiation of these cells into myelinating glia recently observed in the CNS of aged animals collectively are primarily responsible for the age-associated decline in remyelination efficiency. The expression of growth factors, including PDGF, IGF and

TGF- β , has been shown to be important for the development, migration and maturation of oligodendrocytes (Milner *et al.*, 1997). A temporal delay in the expression profiles of these growth factors was observed in old animals in comparison to that observed in younger animals (Franklin *et al.*, 2002; Hinks and Franklin, 2000). The decreased expression of myelin genes (i.e. MBP and PLP) was also shown in old rats (>12 months of age; Ibanez *et al.*, 2003). Therefore, the recruitment and differentiation of oligodendrocyte progenitors in older people likely occurs more slowly than in younger individuals.

A number of developmental and pathological studies have shown the ability of endogenous Schwann cells to migrate into the CNS toward a chemically induced (EtBr, lysolecithin) demyelinated lesion and to remyelinate CNS axons (Blakemore, 1975, 1976, 1979; Graca and Blakemore, 1986). The exposure of the neonatal spinal cord to x-irradiation results in prevention not only of the generation of myelin-forming oligodendrocytes just prior to myelination (Gnatack et al., 1973) but also compromises the generation of astrocytes needed to populate the developmentally increasing size of the spinal cord. In an environment where astrocytes fail to maintain the CNS-PNS border, Schwann cells now can enter the CNS, passing through the incomplete glia limitans to myelinate axons (Blakemore and Patterson, 1978; Sims and Gilmore, 1983; Sims et al., 1985). In myelin-deficient rats myelinating oligodendrocytes degenerate leading to a failure in myelination (Duncan et al., 1987). However, astrocytes in myelin-deficient rats are not compromised and thus maintain an intact glia limitans throughout development and into adulthood. In contrast to what is seen with xirradiation of the neonatal spinal cord of wild-type rats, no Schwann cell invasion is detected in myelin-deficient rats. However, Schwann cells will migrate into the CNS and myelinate axons if the spinal cord of neonatal myelin-deficient rats is x-irradiated neonatally. This illustrates the importance of an intact glia limitans, which is formed by astrocytes, in preventing Schwann cell invasion into the CNS (Duncan et al., 1988).

In a different animal model, where gliotoxins like EtBr (see Section 1.6.1) or lysolecithin (see Section 1.6.2) are injected into white matter tracts of the spinal cord, Schwann cells can be observed migrating into the CNS and contributing to remyelination of the axons (Blakemore, 1975, 1976, 1979; Graca and Blakemore, 1986). EtBr kills oligodendrocytes, microglia and astrocytes resulting in demyelination as well as breaking down the glia limitans that had been formed by the astrocytes, whereas lysolecithin is more selectively toxic to oligodendrocytes, thus largely sparing astrocytes. Therefore, Schwann cells extensively repopulate EtBr-induced demyelinated areas (e.g. caudal cerebellar peduncle) and remyelinate CNS axons in these lesions (Woodruff and Franklin, 1999), but in lysolecithin demyelinated lesions Schwann cell invasion is restricted only to the central, astrocyte-poor core of the area of demyelination (Blakemore, 1976). Itoyama *et al.* (1983) showed that Schwann cells could be identified inside the astrocyte-poor central area of MS plaques. Thus, Schwann cell migration into the CNS and their remyelination of CNS axons are related to the presence or absence of an intact astrocytic environment.

Schwann cell migration *in vivo* is likely to involve several cellular interactions acting in concert, even though each interaction alone may promote sufficient migration in vitro. For example, molecules such as laminin, fibronectin and N-cadherin (Itoyama et al., 1983; Iwashita et al., 1983; Jasmin et al., 2000), the CD9 glycoprotein (Anton et al., 1995), the NG2 proteoglycan (Schneider et al., 2001), and the p75NGFR (Anton et al., 1994) have all been implicated in the migratory ability of these glial cells. Anton et al. (1994) demonstrated that the CD9 glycoprotein enhances Schwann cell migration both on rat pup DRG neurites and on rat pup sciatic nerve pieces in vitro. The CD9induced increase in migratory ability of Schwann cells was correlated with elevation of intracellular calcium level and tyrosine phosphorylation (Anton et al., 1995). Studies on the development of peripheral nerves (Heumann et al., 1987; Yan and Johnson, 1988; Di Stefano and Johnson, 1988; Rush, 1984) and on peripheral nerve injury (Fields et al., 1989; Williams et al., 1983) have demonstrated an increased NGF and p75NGFR expression by migrating Schwann cells. Although the mechanisms by which NGF binding to p75NGFR activates Schwann cell motility are not known, NGF was reported to increase the expression of the L1 cell adhesion molecule by Schwann cells in vitro and it was suggested that this neurotrophin may control cell migration (Seilheimer and Schacher, 1987). Which interaction or combination of interactions regulates Schwann

cell migration *in vivo* may depend on the immediate environment and the molecular characteristics of the migrating cells. For example, in contrast to oligodendrocyte progenitor cells, the intense inflammation and glial reactivity that accompanies infection with Theiler's virus in highly susceptible strains of mice appears to prevent Schwann cells from migrating into the CNS (Ozden *et al.*, 1993).

Although not the focus of this study, there was a contribution of host myelinating glia (i.e. oligodendrocytes and Schwann cells) to the repair of the EtBr lesion induced in the model used in this research project. At 2 weeks after injection of EtBr into the right dorsal funiculus of the adult rat spinal cord, demyelination was still evident at the lesion site (Fig. 2e) but by 8 weeks this region of white matter was almost entirely remyelinated (Fig. 2f). Based on studies by Yajima and Suzuki (1979), Blakemore (1984) and Felts *et al.* (1997), this remyelination would be by both oligodendrocyte progenitors and Schwann cells that had migrated into the EtBr-induced lesion. Whether the grafted OECs also contributed to this neural repair, since we now know they will migrate to an EtBr-induced lesion, is a topic for future investigations; the focus of the present study was on the migration of the OECs, not what the cells did upon reaching the lesion.

5.2.2 Donor-Derived Myelinating Glia

The transplantation of myelin forming cells including oligodendrocytes and their progenitors, Schwann cells, OECs, neuronal precursor cells or bone marrow cells has shown therapeutic potential in promoting CNS repair (see Section 1.7). The transplanted cells have been examined for their ability to remyelinate CNS axons, to survive within damaged CNS environment, and to migrate in the presence or absence of reactive astrocytes and microglia. The ability of donor-derived myelinating glial cells to respond to migratory signals arising from focal areas of demyelination has to be carefully investigated in order to identify the ideal cell candidate for therapeutic transplantation.

In the EtBr model used in this study, OECs were grafted at T12 and the EtBr was injected either at T8, T10, or T12 to induce demyelination at various distances (0 mm to > 15 mm) from the cell graft. With an average length of each spinal cord

segment in the lower thoracic area of 5 mm, OECs were able to migrate through the dorsal funiculus at least as far as the T8 spinal cord segment (most rostral segment examined), which was located as far as 15 - 20 mm from the grafting site. When the OECs were injected directly into an EtBr-induced demyelinated lesion at T12 there was limited cell migration. Perhaps in the latter instance the majority of OECs stayed within the lesion vicinity to help in neural repair, maybe even being "instructed" to provide such assistance by activated microglia. OECs can, however, be induced to migrate away from such a demyelinated lesion provided a second lesion is made a short distance away from the first.

This is not the first study to find that OECs can migrate after grafting into the CNS of an adult rat. Using other models, OECs were shown to migrate from the dorsal root entry zone of the spinal cord into dorsal horn and were thought to promote axonal regeneration through laminae 1 to 5 of the grey matter (Ramon-Cueto and Nieto-Sampedro, 1994). The farthest distance to which OECs are thought to have migrated after being grafted at T9 in the proximal stump of the transected spinal cord of an adult rat is as far as the caudal cervical level (i.e. C7; Ramon-Cueto *et al.*, 1998). However, a recent report suggested that OECs, derived from either the lamina propria of the mouse olfactory mucosa (Ramer *et al.* (2004) or neonatal rat olfactory bulb (Riddell et al., 2004), grafted directly into the DRG failed to migrate across the DREZ and also failed to promote the regeneration of dorsal root axons across the DREZ.

Other myelinating glial cells (e.g. Schwann cells and oligodendrocyte progenitor cells) have been shown to migrate for only limited distances after being grafted into the CNS (Baron-Van Evercooren *et al.*, 1992; Blakemore *et al.*, 1986; Fawcett and Asher, 1999). Transplanted into the cerebral hemisphere of newborn shiverer mice, Schwann cells stayed within the vicinity of the graft and eventually disappeared from the CNS parenchyma. The migration and survival of CG4 cells (an oligodendrocyte progenitor cell line) was found to be limited to areas of the injection site after transplantation into the normal rat spinal cord (Franklin *et al.*, 1996). Host oligodendrocyte progenitors were also shown to be unable to migrate over distances greater than 2 mm into the areas of demyelination (Franklin *et al.*, 1997). Reactive astrocytes appear to impede the

migration of both oligodendrocyte progenitor cells and Schwann cells (Fawcett and Asher, 1999; Lakatos *et al.*, 2003; Shields *et al.*, 2000). In contrast, OECs in our experiments were able to migrate through areas containing reactive astrocytes provided reactive microglia were also present along the migratory pathway.

Our data indicate that reactive microglia enhance OEC migration. The distance over which migratory signals were distributed in most of the experiments performed using this *in vivo* model was only one spinal cord segment (~ 5 mm). This distance was still 2-3 times further than oligodendrocyte progenitors (~ 2 mm; Nait-Oumesmar *et al.*, 1995; Franklin and Blakemore, 1997) or Schwann cells (around the implantation site; Baron-Van Evercooren et al., 1992; Blakemore et al., 1986) migrated in response to a demyelinated lesion even in the presence of activated host glia in the intervening CNS tissue. Thus, not only do oligodendrocyte progenitor cells and Schwann cells very likely differ in their ability to respond to migratory signals emanating from an area of tissue injury, but the migratory ability of both of these cell types within the injured CNS of adult mammals is also far less that of OECs.

5.3 Contribution of Reactive Microglial Cells to Migration of Myelinating Glia

Microglia are extraordinarily sensitive to any changes within the CNS microenvironment (Barron, 1995; Gabicke-Haerter *et al.*, 1996; Kreuzberg, 1996; Moore and Thanos, 1996). In response to CNS injury (i.e., trauma, inflammation, etc.) microglial reactivity is sometimes detrimental (Piani *et al.*, 1991; Chao *et al.*, 1992; Giulian *et al.*, 1993) and sometimes beneficial (Mallat *et al.*, 1989; Elkabes *et al.*, 1996; Rabchewsky and Streit, 1998) in terms of contributing to the repair of neural tissue. They play a major role in orchestrating the contribution of multiple cell types to immune reactivity, inflammation, and demyelination within the CNS (Chao *et al.*, 1995; Giulian and Lachman, 1985; Giulian *et al.*, 1986; Lee *et al.*, 1993; Righi *et al.*, 1989; Selmaj *et al.*, 1990), and thus in many instances exacerbate the inflammatory response associated with CNS injury (Benveniste, 1997). Microglia have been shown to release cytokines, including IL-1, TNF- α , IL-6 (Dickson *et al.*, 1993; Hetier *et al.*, 1988; Lee *et al.*, 1993), excitatory amino acids (i.e. glutamate), oxidative radicals, and nitric oxide
(reviewed in Banati *et al.*, 1994; Gordon, 1995; Gehrmann *et al.*, 1995), all of which contribute to the inflammatory reaction in the damaged CNS. The proinflammatory cytokines (i.e. IL-1, TNF- α , IL-6) are secreted early in the glial cell response to brain injury (Eddleston and Mucke, 1993; Hebert *et al.*, 2003; Ridet *et al.*, 1997) and in addition to promoting the inflammatory response also stimulate astrogliosis (Tetzlaff *et al.*, 1988). The inflammatory environment of the damaged CNS, where activated microglia and astrocytes are present, has been suggested to limit the migration of myelinating glia including oligodendrocyte progenitors (Fawcett and Asher, 1999) and Schwann cells (Baron-Van Evercooren *et al.*, 1992; Fawcett and Asher, 1999; Shields *et al.*, 2000).

Activated microglia in MS also express proteinases such as MMPs (Colton et al., 1993; Gehrmann et al., 1995; Gottschall et al., 1995). MMPs represent a family of 23 zinc-containing endopeptidases characterized in mammals (reviewed in Yong et al., 2001), which are known to degrade proteins of the extracellular matrix during embryogenesis and wound healing (Basbaum and Web, 1996; Werb, 1997). Increased expression of MMPs has been implicated in several neurological diseases such as MS and ischemic stroke, in which inflammation occurs (reviewed in Yong et al., 2001). Pro-inflammatory cytokines (i.e. IL-1, TNF- α ,) can up-regulate MMP expression, in part through NFkB transcription activity (Lee *et al.*, 2001). MMPs, in turn, can proteolytically cleave precursor molecules such as pre-IL-6, pre-TNF- α , pre-TGF- α (Black et al., 1997; Yu and Stamenkovic, 2000), and thus significantly promote inflammation in the CNS. In further contribution to the pathogenesis of MS, MMPs cause the disruption of blood-brain barrier and facilitate leukocyte invasion into the CNS parenchyma (Yong et al., 1998), where activated leukocytes make their contribution to the inflammatory response by secreting proinflammatory cytokines (i.e. INF-γ, IL-6, TNF-α, TGF-α; Merrill and Benveniste, 1996; Selmaj and Raine, 1988). Gelatinase B (MMP9), in particular, was shown to degrade MBP in the myelin sheath in vitro (Gijbelts et al., 1993; Chandler et al., 1995) and this action of MMP9 could be an important contributor to demyelination in neuroinflammatory disease. Breakage of cellextracellular matrix contacts by MMPs leads to cell apoptosis as a result of cell

detachment and loss of integrin signaling (Ruoslanti and Reed, 1994; Chen and Strichland, 1997). Production of these cytokines and MMPs by host glia and infiltrating inflammatory cells plays therefore a detrimental role in MS.

How increased secretion of proinflammatory cytokines and MMPs might influence the migration of myelinating glia remains to be elucidated. Tourbah et al. (1997) noted that migrating CG4 cells were often associated with reactive astrocytes. However, the migration of these cells was limited possibly due to an inhibitory effect of the CCL3 chemokine (i.e. macrophage inflammatory protein-1 α ; Nguyen *et al.*, 2003) via CCR1, which is the the CCL3 chemokine receptor that is expressed by CG4 cells (Nguyen and Stangel, 2001; Stangel et al., 1999). Filipovic et al. (2003) observed the expression of one chemokine, growth-related oncogene-alpha (GRO-alpha) and its receptor CXCR2 by oligodendrocyte progenitors in the highly proliferative subventricular zone of the telencephalon of human fetuses. However, neither GROalpha nor CXCR2 were present in oligodendrocyte progenitors found on autopsy in the brain sections of MS patients. Instead, activated microglia localized on the border of demyelinated plaques expressed both GRO-alpha and CXCR2. Thus, the change in the chemokine receptor profile of oligodendrocyte progenitors, for example downregulating the expression of CXCR2 and possibly up-regulating the expression of CCR1, may partially explain the limited migration of oligodendrocyte progenitors within areas of inflammation in the CNS and why remyelination is not more efficient in MS.

There is very little data on how and whether microglial reactivity can influence the migration of oligodendrocytes and Schwann cells during the remyelination phase of neural repair. Among the cells of the oligodendrocyte lineage, O4+ve/GalC-ve oligodendrocyte progenitor cells possess the best migratory ability (Gansmuller *et al.*, 1991; Small *et al.*, 1987). The presence of EAE was reported to enhance CG4 cell migration (Tourbah *et al.*, 1997). Although Tourbah *et al.* (1997) did not mention the association of migrating CG4 cells with reactive microglia, one could assume that microglia would be an important player within the inflammatory environment of the CNS in EAE. Microglial reactivity in response to trauma can also be beneficial to neural repair and facilitate axonal regeneration through areas of injured CNS tissue (Prewitt *et al.*, 1997; Rabchevsky and Streit, 1997; Lazarov-Spiegler *et al.*, 1996; Stichel *et al.*, 1999). Schwann cells migrate into such areas of injury, which may be under both the facilitatory control of microglia (Zeev-Brann *et al.*, 1998; Dezawa *et al.*, 1999) and the inhibitory influence of astrocytes (Fawcett and Asher, 1999). On the other hand, the depletion of microglia and macrophages by clodronate in young adult female rats that have a lysolecithin-induced demyelinated lesion in their spinal cord had no effect on Schwann cell remyelination but resulted in a decreased remyelination by oligodendrocytes (Kotter *et al.*, 2001). The specific response of microglia to CNS damage may well depend on the type of injury and, depending on how the demyelination was induced, could potentially change the profile of migratory signals expressed by these glial cells and to which myelinating glia respond.

The contribution of reactive microglia to the generation of migratory signals for cells of the blood has been known for some time now (Babcock *et al.*, 2003; Gehrmann *et al.*, 1993), but only recently has it been suggested that these chemotactic signals likely also induce the migration of other cell types (Aarum *et al.*, 2003; Bettinger et al. 2002). Aaron *et al.* (2003) have shown that EGF released from mouse microglial cells influence the directional migration of neural precursor cells of the CNS and hypothesized that activated microglia play a role in directing the replacement of cells damaged or lost from injury to the brain or spinal cord. Bettinger *et al.* (2002) reported a similar chemotactic effect on GL261 mouse glioma cells and suggested microglial reactivity may promote the invasive phenotype of astrocytoma cells.

The findings reported in this thesis also implicate microglial cells in contributing to the generation of a migratory signal for cells other than lymphocytes, monocytes, neural precursor cells or glioma cells. The numbers of OECs that migrated into the T11 segment of the spinal cord after grafting into the dorsal funiculus of T12 varied depending on the presence, location, and number of small, focal, EtBr-induced demyelinated lesions. It was also observed that as increased numbers of OECs entered the T11 segment they were found within the area of astrocytic and microglial reactivity in the ipsilateral dorsal funiculus. In contrast to other studies concentrating on the association between reactive astrocytes and migrating oligodendrocyte progenitors

(Armstrong *et al.*, 1990; Fruttiger *et al.*, 1999; Milner *et al.*, 1997; Osterhout *et al.*, 1999; Tourbah *et al.*, 1998) or Schwann cells (Fawcett and Asher, 1999; Franklin and Blakemore, 1993; Harrison, 1985; Itoyama *et al.*, 1983), the migration of OECs in the animal model used in this research project seemed to be dependent on the presence of reactive microglia since it was significantly diminished when the microglial response was dampened by a 4 week exposure to daily injections of minocycline. The minocycline had no apparent effect on the astrocytic reactivity, at least as evidenced by staining with vimentin and GFAP antisera, thus implicating reactive microglia as one potential source, either directly or indirectly, of the migratory signal(s) driving OEC migration in this animal model.

Microglial activity observed at the EtBr-induced lesion (i.e. at T10) and along the migratory path (i.e. T11) may well differentially contribute to the generation of a migratory signal for myelinating glia. Similar to that reported in previous studies (Yajima and Suzuki, 1979), an EtBr injection in our experiments resulted in killing all glial cells including oligodendrocytes, microglia, and astrocytes in the vicinity of the injection site with the demyelination of the right dorsal funiculus still being evident at 2 weeks survival (Fig. 4.2e). At the lesion site, dying cells release various signals including growth factors, cytokines, etc., that activate microglia, macrophages, and astrocytes to begin the process of neural repair. These cells, as well as oligodendrocyte progenitor cells and Schwann cells, migrated into the lesion and contributed to the injury repair that was observed by 8 weeks survival (Fig. 4.2f). Microglia and astrocytes of the adjacent spinal cord segments also become activated under signals (e.g. cytokines, MMPs, etc.) released from the cells of the damaged area and in turn also secrete cytokines and MMPs. Along with those factors secreted at the site of injury, these additional induced signals may facilitate not only the migration of host myelinating glia but also that of grafted OECs. The specific signals generated by this glial reactivity and that contributed to OEC migration in the present study require additional investigation. It is possible OECs express cytokine receptors that can bind microglial-derived cytokines, which in turn could facilitate their migration chemotactically through areas of neuropil containing reactive microglia. OECs are

known to express the IL-6 receptor (Nan *et al.*, 2001), but it is not known which additional cytokine receptors they might also express. In addition, MMPs released by microglia in response to injury may have also contributed to inducing migration of OECs along the path of glial reactivity by reconstructing the extracellular matrix, breaking cell-matrix contacts and activating cytokine precursors (e.g., pre-IL-6), all of which may contribute to generating a chemotactic gradient for the migration of OECs.

Why does it take more than 4 weeks for a significant number of OECs to migrate into T11 when a single lesion is made at T10 even though microglia in the T11 segment react much sooner than this? The glial reactivity detected at T11 appeared to be more intense in rats surviving for 8 weeks in comparison to those at shorter survival times (fig. 4). Whatever the reason is, it was found in this study that a 4 week treatment with minocycline diminished microglia activity and resulted in decreased OEC migration, thus implicating the microglial response during the first 4 weeks after injury in contributing to the generation of a migratory signal. In contrast, precocious (at 2) weeks) migration of OECs away from a demyelinated lesion was noted in rats in which a second lesion had been made 2 segments away. This may indicate that two EtBrinduced lesions result in the migratory signal(s) being generated faster and/or more intensely when there are 2 lesions rather than one, even though the OX-42 and OX-18 immunostaining could not detect a difference in the microglial response. To further complicate matters, OECs did not migrate away when injected directly into the EtBr lesion of a single lesioned rat even though the intensity of the microglial reactivity detected by staining with anti-OX-18 antibody in the adjacent segment (i.e. T11) was similar to those of rats that had received an EtBr injection at T10 and an OEC graft at T12. Clearly, these experimental manipulations reveal variations in the generation of a migratory signal(s) that may be of use in identifying the molecules involved.

5.4 Relevance of Migratory Ability of OECs to Neural Repair

MS pathology is characterized by the presence of multiple plaques of demyelination scattered throughout the CNS. In order to contribute to neural repair in these demyelinated lesions, myelinating glia transplanted into the CNS of an MS patient as part of a therapeutic approach to repairing the damage would have to migrate through both normal and inflammed areas of neuropil to reach the numerous isolated plaques. Upon reaching each demyelinated lesion the myelinating glia will then have to contribute to remyelination of the axons in the presence of myelin debris and of reactive astrocytes and microglia. In addition to remyelination, the ability to promote the regeneration of transected axons that are often found in MS plaques (Bitsch *et al.*, 2000; Davie *et al.*, 1995; De Stefano *et al.*, 1998; Ferguson *et al.*, 1997; Kornek *et al.*, 2000; Trapp *et al.*, 1998) would be an additional benefit of myelinating glia following transplantation. OECs have a greater likelihood of being able to perform the migratory, remyelination, and axonal growth promoting tasks than either oligodendrocyte progenitors or Schwann cells, which is why they are being considered as potential candidates as part of a therapeutic approach to treating neurological disorders including MS and spinal cord injury (Barnett and Chang, 2004; Barnett and Riddell, 2004; Boyd *et al.*, 2003, 2004; Doucette, 1995, 2001; Franklin and Barnett, 2000; Li *et al.*, 1997, 1998; Nieto-Sampedro, 2003; Ramon-Cueto *et al.*, 1998, 2000).

The beneficial contribution of OECs to neural repair include their abilities: 1) to facilitate remyelination of CNS axons *in vivo* in animal models with demyelination and spinal cord injury, as well as to remyelinate DRG axons *in vitro* (see Section 1.2.5); 2) to promote the regeneration of not only olfactory axons but also of other PNS and CNS axons, with the axon regeneration leading to functional recovery (see Section 1.2.5); 3) to integrate into the neuropil of the CNS, coming into close contact with astrocytes with no intervening basal lamina (see Section 1.2.5); 4) to secrete growth factors such as NGF, express cell adhesion molecules such as L1, N-CAM, PSA-N-CAM, and express extracellular matrix molecules such as laminin and fibronectin, which collectively play key roles in neural repair (see Section 1.2); 5) to migrate through CNS neuropil containing reactive astrocytes and microglia (see Section 1.8.3); and 6) to respond to migratory signal(s) arising as a result of a focal demyelination and move towards the demyelinated lesion, as described in this research project.

The main goal of transplanting myelinating glia as part of a therapeutic approach to treating neurological disorders such as MS and spinal cord injury is to facilitate host tissue repair processes including remyelination and axonal regeneration. There is a great possibility that implanted OECs will facilitate the remyelination of CNS axons after they have migrated some distance from the grafting site. OEC remyelination of CNS axons *in vivo* has been reported in animals with EtBr-induced demyelination (Akiyama *et al.*, 2004; Franklin *et al.*, 1996; Imaizumi *et al.*, 1998, 2000; Kato *et al.*, 2000; Smith *et al.*, 2002) and after spinal cord injury when the OECs were grafted into the lesion (Li *et al.*, 1997, 1998). The *in vitro* remyelination of DRG neurites by OECs was also previously shown (Devon and Doucette, 1992; 1995). In addition to their remyelinating ability, OECs have been shown to enhance axonal conduction through previously demyelinated areas of the spinal cord (Imaizumi *et al.*, 1998, 2000). However, the ability to remyelinate either CNS (Boyd *et al.*, 2004; Takami *et al.*, 2002) or PNS (Plant *et al.*, 2002) axons by OECs has been questioned.

Although Boyd et al. (2004) recently reported that Lac-Z expressing OECs implanted into a cystic cavity of the adult rat spinal cord facilitated axonal regeneration and remyelination, these OECs did not associate directly with the remyelinated CNS axons. Peripheral type remyelination of the regenerating axons was observed, however, and was found to be due to Schwann cell migration into the lesion in the presence of transplanted OECs. This indicates that OECs may facilitate axonal regeneration in the damaged spinal cord by providing a microenvironment conducive to Schwann cell migration into the site of injury, which in turn provides support for axonal growth and results in remyelination of the axons. When grafted into the cystic cavity of the adult rat spinal cord, OECs may not only be "instructed" by reactive glia, and perhaps also by other cells comprising the inflammatory response, to stay in the vicinity of the lesion but they themselves may also be playing an "instructive" role in attracting host Schwann cells to migrate into the lesion. In addition, NGF secreted in part by OECs (Boruch et al., 2001; Ramon-Cueto et al., 1993) may bind to the p75NGFR on Schwann cells, possibly facilitating their migration chemotactically as has been shown *in vitro* (Anton et al., 1994). The OECs in the lesion cavity may also shelter the Schwann cells that migrate into the lesion, effectively allowing them to function as axonal growth promoting and remyelinating glia in an area of extensive astrocytic and microglial

reactivity. Such sheltering may especially protect these Schwann cells from activated astrocytes previously shown to inhibit Schwann cell migration into the CNS (Fawcett and Asher, 1999).

The very first clinical signs of disability experienced by MS patients are due to axonal injury that occurs inside the earliest forming plaques (Bitsch et al., 2000; Davie et al., 1995; De Stefano et al., 1998; Ferguson et al., 1997; Kornek et al., 2000; Trapp et al., 1998). Therefore, transplantation of myelinating glia for the treatment of demyelinating diseases should also address the regeneration of these transected axons to restore function and reverse the disability. OECs have been shown to promote the regeneration of not only olfactory axons (Graziadei and Monti Graziadei, 1978, 1979; Barber and Lindsay, 1982; Doucette et al., 1983; Doucette, 1990) but also of other PNS (Guntinas-Lichius et al., 2001, 2002; Ramon-Cueto and Nieto Sampedro, 1994) and CNS (Li et al., 1997, 1998; Nash et al., 2002; Ramon-Cueto et al., 1998; Smale et al., 1996) axons. Transplantation of OECs into animal models of spinal cord injury enhances functional recovery of spinal reflexes (Navarro et al., 1999), locomotion (Keyvan-Fouladi et al., 2003; Li et al., 2003; Lu et al., 2001, 2002; Plant et al., 2003; Ramon-Cueto et al., 2000), forepaw reaching (Nash et al., 2002), and breathing (Li et al., 2003). However, recent reports have tempered the excitement surrounding the therapeutic potential of OECs in promoting the regeneration (Gomez et al., 2003; Riddell et al., 2004; Ramer et al., 2004; Takami et al., 2002) and remyelination (Takami et al., 2002; Boyd et al., 2004) of damaged CNS axons. Possible failure of OECs to survive after being grafted into the cystic cavity of a compression injury in the adult rat spinal cord in a study by Takami et al. (2002) was associated with far less axonal regeneration and locomotor recovery in comparison to that of grafted Schwann cells OECs derived from either the lamina propria of the mouse olfactory mucosa (Ramer et al., 2004) or neonatal rat olfactory bulb (Riddell et al., 2004) also failed to promote the regeneration of dorsal root axons across the DREZ. The type of injury and the complexity of the reparative processes due to the host cell response to tissue damage, as well as the origin of OECs used for grafting, may all potentially influence

the survival and subsequent response of the grafted OECs to neural repair of the CNS tissue.

In order for myelinating glia to perform the migratory, remyelination, and axonal growth promoting tasks discussed above, these cells must integrate into the CNS neuropil thus coming into close contact with host glial cells, especially reactive astrocytes. Reactive astrocytes impede the migration of both oligodendrocyte progenitor cells and Schwann cells (Fawcett and Asher, 1999; Lakatos et al., 2003; Shields et al., 2000). In contrast, OECs in the present research project migrated through areas containing both reactive astrocytes and reactive microglia. Grafting of OECs into the spinal cord of Group 2-7 rats (Table 4.1) resulted in activation of microglia and astrocytes at the site of the graft as well as in adjacent spinal cord segments, as detected with OX-18 and vimentin immunostaining, respectively. The migration of OECs into the neuropil of the lower thoracic spinal cord segments rostral to the site of injection (i.e., T11-T8) at all survival times (2-8 weeks) in these experimental groups shows that OECs can coexist with reactive astrocytes and microglia within the CNS. Lakatos et al. (2003) reported lessening of the response of host astrocytes and of the expression of chondroitin sulphate proteoglycans at the site of injection after grafting neonatallyderived OECs into the non-lesioned dorsal funiculus of the adult rat spinal cord in comparison to that seen after grafting Schwann cells. In addition, OECs can intermingle with astrocytes in co-culture without causing astrocytes to undergo hypertrophy and to increase their expression of chondroitin sulphate proteoglycan in comparison to that of Schwann cells and astrocytes (Lakatos et al., 2000). Collectively, these data suggest the OECs have little difficulty integrating into the neuropil of damaged CNS tissue primarily because they can co-exist, unlike Schwann cells, with astrocytes without being rejected, much as occurs normally in the CNS portion of the primary olfactory pathway (see Section 1.2.3).

Originally identified as critical mediators of neuronal survival and nerve fiber outgrowth during development, neurotrophic factors have been shown to protect neurons and facilitate repair in areas of CNS injury in adult animals (Campenot *et al.*, 2004; D'Ambrosi *et al.*, 2004; Jones *et al.*, 2001; Pierchala et al., 2004; Tuszynski *et al.*, 1999). The infusion of neurotrophic factors such as BDNF and NT-3 into Schwann cell-containing guidance channels connecting the proximal and distal stumps of the transected spinal cord promoted regeneration of axons in descending pathways from the brainstem that grew into these guidance channels (Xu et al., 1995). OECs have been shown to produce PDGF (Kott et al., 1994), NGF (Boruch et al., 2001; Guthrie and Gall, 1991; Ramon-Cueto et al., 1993), neuropeptide Y (Ubnik et al., 1994), and S100 (Barnett et al., 1993; Cummings and Brunjes, 1995; Doucette and Devon, 1995; Franceschini and Barnett, 1996; Gong et al., 1994). Furthermore, Ramon-Cueto and Avila (1994) suggested that OECs might secrete additional growth factors including BDNF, neurotrophin-3 and neurotrophin-4 based on evidence that these cells express the mRNAs for these growth factors. The action of secreted neurotrophins on neurons and Schwann cells might be facilitated by the OEC expression of the p75NGFR with which OECs might use to present the specific neurotrophins, such as NGF, to these other cells types (Barnett et al., 1993; Franceschini and Barnett, 1996; Goodman et al., 1993; Ramon-Cueto and Nieto Sampedro, 1992; Turner and Perez-Polo, 1993, 1994; Vickland et al., 1991). This close contact putative chemotactic effect of the OEC expression of the p75NGFR and NGF secretion would be in addition to a possible longer range migratory signal(s) that the OECs may provide to induce p75NGFR+ve Schwann cells to migrate into the lesion. Migration of Schwann cells in vitro has been shown to be influenced by levels of NGF (Anton et al., 1994). OECs also express several adhesion molecules including N-CAM (Key and Akeson, 1990), PSA-N-CAM (Franceschini and Barnett, 1996; Miragall et al., 1988), and L1 (Doucette, 1995; Miragall et al., 1988; Ramon-Cueto and Avila, 1994), as well as extracellular matrix molecules such as laminin (Doucette, 1996; Kafitz and Greer, 1997; Liesi, 1985), and other growth promoting molecules such as glia-derived nexin (Reinhardt *et al.*, 1988). All of these molecules are known to facilitate axonal growth both in vivo and in vitro (Bixby et al., 1988; Boyd and Gordon, 2003; Fu and Gordon, 1997; Kleitman et al., 1988).

Transplantation of myelinating glia and the administration of growth factors to areas of CNS injury have helped to overcome the normally abortive regeneration of

CNS axons and to facilitate their remyelination within the injured CNS. Researchers have used various molecular genetic techniques to transfer selected growth factor genes into cells that will then be used to deliver growth factors constitutively to all cells at the site of injury. Both retroviral and adenoviral vectors have been utilized to introduce the genes of interest (Owens, 1991). Menei *et al.* (1998) transfected Schwann cells *in vitro* with a replication deficient Moloney murine leukaemia retrovirus carrying the human preproBDNF cDNA. These BDNF-secreting Schwann cells were transplanted into the transected adult rat spinal cord and were found to enhance axonal regeneration across the injury in comparison to wild-type Schwann cells. OECs transfected with the gene for glial cell line-derived neurotrophic factor (GDNF) were also found to promote CNS repair after being grafted into the transected adult rat spinal cord (Cao *et al.*, 2004). The expression of either BDNF by Schwann cells or GDNF by OECs enhanced the growth-promoting properties of these cells within the microenvironment of the lesion site during recovery period that followed the spinal cord injury.

The migration of OECs towards a focal demyelination after being grafted into normal or demyelinated areas of the adult rat spinal cord, as found in this thesis, is a very important ability of these cells. Several previous studies have reported circumstantial evidence that OECs can migrate within the microenvironment of the injured spinal cord when Hoechst prelabelled OECs were grafted very close (1 mm) to the site of injury (Navarro *et al.*, 1999; Ramon-Cueto and Nieto-Sampedro, 1994; Ramon-Cueto *et al.*, 1998, 2000), or when unlabelled OECs are injected directly into a very small lesion (Li *et al.*, 1997, 1998). An important finding arising from the present study is the ability of OECs to respond to migratory signals arising as a result of making a small, focal demyelination two to four spinal segments rostral to the site of grafting. This migratory ability of OECs, coupled with their potential ability to respond to endogenous migratory signal(s) arising from the numerous isolated plaques in the spinal cord of an MS patient, are invaluable qualities that make OECs very attractive as part of a therapeutic approach to treating the pathology of MS.

An additional important finding of this research project is that the migration of OECs within the spinal cord is in response to a signal(s) associated with the activation

of microglia, as opposed to activation of astrocytes, at the lesion site and/or along the migratory path. The injection of EtBr into the spinal cord at different distances from the site of an OEC graft in Group 2, 3 and 5 rats (see Table 4.1) resulted in an activation of astrocytes and microglia in the right dorsal funiculus at the site of the EtBr injection as well as along the migratory path between the cell graft and the lesion. In these rats, an increased number of OECs migrated preferentially through the neuropil of the ipsilateral dorsal funiculus, which was highly populated by reactive astrocytes and microglia. In addition to their being microglial-derived migratory signal(s) generated in the chemically induced demyelinated area, one has to also consider the possibility that there are concentration gradients of the migratory signal(s) along the migratory path between the cell graft and the lesion. In contrast to the migration of OECs when they were grafted at a distance from the EtBr lesion in Group 2 and 3 rats, when they were grafted directly into a small, focal demyelinated lesion (i.e., Group 4 rats) far fewer OECs migrated away from the grafting site even though reactive microglia were present in the adjacent spinal cord segments, as has been also reported by other neuroscientists (Boyd et al., 2004; Plant et al., 2003; Takami et al., 2002). However, when two EtBr lesions were made 2 spinal cord segments apart (i.e., Group 5 rats) and the OECs were grafted directly into one of these lesions the OECs were found to migrate precociously towards the second lesion. Activated microglia at the lesion site in the Group 4 rats might have "instructed" grafted OECs to contribute to the repair process of demyelinated lesion, perhaps by creating a microenvironment conducive to Schwann cell migration.

The minocycline data reported in this thesis implicate microglial reactivity as a major contributor to the generation of the migratory signal(s) driving the migration of OECs in the EtBr model used in this study (see Section 5.3). Specifically, decreasing the reactivity of microglia by a four week daily exposure to minocycline resulted in a corresponding decrease in the number of OECs that migrated into the T11 segment of the spinal cord. There was no indication, as judged by GFAP and vimentin immunostaining, that the minocycline treatment reduced the astrocytic reactivity. Thus, this research project did not identify astrocytes as contributors of any potential signals

inducing OECs to migrate away from the site of a cell graft towards a small, focal demyelinated lesion. Although, the minocycline experiments identified the microglial reactivity as contributing to the EtBr lesion-induced OEC migration, it is possible the astrocytic reactivity may also have played a role in generating a migratory signal(s) but that the generation and secretion of such a signal(s) might be dependent on the presence of a microglial response. Therefore, even if the reactive astrocytes in the T11 segment of Group 2, 3 and 5 rats may have contributed to the migration of OECs away from the grafting site, the signal obviously was not powerful enough in rats in which the microglial reactivity was impeded by minocycline treatment (i.e., Group 6 and 7 rats). It is also possible that the presence of an astrocyte-derived signal(s) is totally dependent on the presence of reactive microglia.

The present study focused on the ability of OECs to respond to migratory signal(s) arising as a result of a small, focal demyelination in the spinal cord of adult rats and on the ability of these grafted cells to migrate through areas of neuropil containing reactive microglia and astrocytes. Both reactive astrocytes and microglia are important players in the complex repair processes that must occur to repair the demyelinated lesions in MS as well as during the regeneration and remyelination of damaged axons following spinal cord trauma. Because OECs are not rejected from the neuropil of the CNS, this enables them to migrate amongst reactive microglia and reactive astrocytes without obstructing the ability of these host glial cells to do the job of neural repair. Whether the OECs that migrate towards the demyelinated lesion contribute to neural repair upon reaching the lesion is an area for future study. Based on what is known about the growth promoting ability of OECs (see Section 1.2.5), however, it can be hypothesized that upon reaching the lesion the OECs will create a more favorable microenvironment for axonal regeneration and remyelination by host-derived myelinating glia, including even host Schwann cells.

5.5 Prelabelling Cells Prior to Grafting

Transplantation of myelinating glia for the purpose of neural repair is a promising approach in the treatment of demyelinating diseases and spinal cord injury. It

is critically important to determine the efficacy of grafting the different types of myelinating glia using animal models before the approach is translated to humans. A proper evaluation of the contribution of grafted oligodendrocyte progenitor cells, Schwann cells or OECs to neural repair cannot be made unless the cells are prelabelled prior to grafting. Otherwise there is no way to tell whether the oligodendrocyte remyelination is due to host oligodendrocytes or ones that arose from the grafted cells. Likewise with OECs and Schwann cells, which both remyelinate axons with peripheral type myelin (see Section 1.7), it is essential for the experimenter to determine the source of the remyelinating cells; i.e. are they host or donor in origin. For example, following intraspinal implantation, OECs are believed to transform into Schwann cell-like cells that are thought to be the ones facilitating the axonal regrowth and responsible also for remyelinating the axons (see Section 1.2.5). One fact often overlooked in these studies is that Schwann cells, which share several phenotypic features with OECs (see Section 1.2), migrate into the spinal cord after injury (Beattie et al., 1997; Brook et al., 1998; Namiki et al., 2000; West et al., 2001; Takami et al., 2002). Given that the invasion of Schwann cells likely coincides with the retention of grafted OECs at sites of spinal cord injury, new questions are raised as to which glial cell type is responsible for supporting the regeneration of central axons. The study by Boyd et al. (2004) was the first ultrastructural study to use prelabelled OECs in a compression injury model and provided direct evidence that OECs do not associate with axons nor do they form myelin or basal lamina after implantation into the injured spinal cord of adult rats. None of the previous studies provided ultrastructural evidence of the fate of prelabelled OECs after intraspinal implantation into or adjacent to the site of injury, even though they attempted to describe in detail the interactions of these grafted cells with central axons and other neural elements (e.g., invading Schwann cells) (Li et al., 1998, 2003; Plant et al., 2003; Takami et al., 2002). Instead, the researchers assumed the Schwann cell-like cells seen at the site of injury arose from the implanted OECs. Because Boyd et al. (2004) used prelabelled OECs in their grafting experiments, they could determine that grafts of OECs are indeed beneficial, but not necessarily for the reasons proposed previously in the literature.

5.5.1 Fluorescent dyes

Fluorescent dyes such as DiI and Hoechst, as well as transfections with a viral vector containing the reporter gene for green fluorescent protein (discussed in Section 5.5.2), have all been widely used to label cells before grafting into host CNS tissue and to visualize the transplanted cells using fluorescent microscopy (Baron-Van Evercooren *et al.*, 1996; Gansmuller *et al.*, 1992; Ramon-Cueto and Nieto-Sampedro, 1994; Ramon-Cueto *et al.*, 2000; Verdu *et al.*, 2001). Prelabeling cells before grafting into the CNS of experimental animals is necessary to determine whether the cells survive grafting into an area of CNS injury, to track their migration (or lack thereof), their migratory routes, and the cells with which they come into intimate contact.

DiI, a carbocianone, is a lipophilic dye that has been utilized in some studies to prelabel cells (e.g. Schwann cells, oligodendrocyte progenitors, astrocytes) prior to grafting (Baron-Van Evercooren *et al.*, 1996; Gansmuller *et al.*, 1992). Soluble in 100% ethanol but not in aqueous solution (Honig and Hume, 1986), DiI is inserted into the plasma membrane and diffuses within the plane of the membrane to label the entire cell surface. As noted by Gansmuller *et al.* (1992), DiI labeling does not affect the survival, migratory ability of astrocytes, oligodendrocyte progenitors and the Schwann cell line MSC80, transplanted into the newborn Shiverer and normal mouse brain. Baron-Van Evercooren *et al.* (1996) also showed that DiI-labeling did not impinge on the migration of oligodendrocyte progenitors and Schwann cells in response to migratory signals arising from a lysolecithin induced demyelinated lesion in the adult wild type and Shiverer mouse spinal cord.

In this research project, DiI-labelled OECs were frozen and stored at -70°C until needed for grafting in one of the cell migration experiments (see Section 2.4.4). Similar to that of unlabelled OECs, the viability of the cells after thawing was approximately 75%. Furthermore, some animals included in Group 1 to 5 rats and all animals in Group 6 and 7 rats survived for 8 weeks, at which time they were perfused, their spinal cords were dissected and then processed for fluorescent microscopy. After all this time, DiI-labelled OECs were easily detected within the rat's spinal cord rostral to the cell graft. Therefore, DiI is a reliable marker, which can help identify grafted cells within host animal tissue, without compromising the viability of the grafted cells.

One disadvantage to the use of DiI for prelabelling cells is the possibility that the dye may leak out of the cell. If leakage does occur, then cells of the host animal may incorporate the dye into their plasma membrane and be mistakenly identified as grafted cells. In the present study, DiI+ve particles were found intra- and extracellularly in spinal cord sections obtained from experimental animals. Spinal cord tissue in our experiments was additionally stained with Hoechst dye in order to label cell nuclei. Each DiI+ve/Hoechst+ve cell had an elongated cell body and unbranched cytoplasmic processes and these were the cells counted for statistical analysis (Fig. 4.2). Several spinal cord sections chosen for staining with the microglial markers OX-18 and OX-42 were also studied using fluorescent microscopy and in none of these sections were microglia seen to be DiI+ve, nor did any of the DiI+ve cells have a multipolar morphology characteristic of microglia or astrocytes. Furthermore, leakage of DiI from the grafting site into the adjacent spinal cord segment is extremely unlikely since there was no staining of axons in the dorsal funiculi of any of the rats in this study.

Hoechst (2'-(4-Ethoxyphenyl)-5-(4-methyl-1-piperazinyl)-2,5'-bi-1*H*benzimidazole trihydrochloride, bisBenzimide Hoechst 33342) is membrane-permeable fluorescent dye, which intercalates in A-T regions of DNA and stains nuclei with low cytotoxicity. Labeling with Hoechst was used to identify cells within host tissue after transplantation of either Schwann cells in animal models with lysolecithin–induced demyelination (Baron-Van Evercooren *et al.*, 1992, 1993) or OECs in animals with spinal cord injury (Ramon-Cueto and Nieto-Sampedro, 1994; Ramon-Cueto *et al.*, 2000; Verdu *et al.*, 2001). Both Schwann cells and OECs were visualized by virtue of their blue stained nuclei. Baron-Van Evercooren *et al.* (1992) also demonstrated that the Hoechst-labelled Schwann cells, transplanted into or at a short distance (2-8 mm) from a demyelinated lesion in Shiverer mouse spinal cord, were able to migrate into the lesion and remyelinate CNS axons. Limited migration of both Hoechst-labelled Schwann cells and a Schwann cell line MSC80 was observed in wild type (2-3 mm) or shiverer (8 mm) mouse spinal cord towards a focal demyelination, with some of these cells contributing to myelin repair upon reaching the lesion and expressing the peripheral myelin protein, P0. Hoechst labeling did not impede the expression of O4, Gal-C or MBP by oligodendrocyte progenitors and their progeny after being transplanted into the newborn shiverer mouse brain. However, tissue from animals that received Hoechst-labelled cells has to be processed very carefully as Ramon-Cueto and Nieto-Sampedro (1994) reported that a long paraformaldehyde incubation results in the loss of Hoechst flourescence. In addition, it has been suggested that phagocytosis of cellular debris by microglia and/or macrophages at the site of implantation can result in an overestimation of the survival and migration of transplanted cells because the phagocytosed Hoechst dye and the nuclear staining do not reveal the morphology of the cells and therefore it is impossible to exclude reactive glia or macrophages from the quantification (Ruitenberg *et al.*, 2002).

Labelling cells with fluorescent dyes (e.g. DiI, Hoechst) have advantages over other prelabelling methods. Firstly, this method consumes less time and secondly, it is more cost effective in comparison to other methods such as viral infection. Prelabelling of cells with DiI prior to grafting does not impede the phenotype or the function of the cells after transplantation. In the present study the DiI could be detected for at least 8 weeks after grafting, although there was a tendency for the fluorescence to fade after a 2-3 minute exposure to light while performing the cell counts on the fluorescent microscope. However, DiI is easiest to use when fluorescent microscopy is being used to detect the presence of prelabelled cells.

5.5.2 Viral Infection

Advances in molecular biology allow scientists to transfer a foreign gene(s) into different cells. Transfer of such a foreign gene into a cell results in the production of the foreign gene's protein, which in turn serves as a "reporter" protein or marker of transfected cells. Viral vectors such as RV (Langford and Owens, 1990; Homnou *et al.*, 1996), the Meloney Murine Leukemia Virus packaging cell line PT67 (Mosahebi *et al.*, 2001), adenovirus (Cot *et al.*, 1998; Ruitenberg *et al.*, 2002; Shy *et al.*, 1995), and HIV-1-derived lentivirus (Kosaka *et al.*, 2004; Ruitenberg *et al.*, 2002) have been utilized to transfect cells designated for transplantation. Genes encoding chloramphenicol acetyltransferase, luciferase, and β -galactosidase have been widely as reporter genes in viral vectors (reviewed by Kosaras and Snyder, 2002). For example, the *Lac-Z* gene encodes the "reporter" enzyme protein, β -galactosidase. RV vectors harboring the *Lac-Z* gene have been utilized for infection of Schwann cells (Langford and Owens, 1990; Homnou *et al.*, 1996) and OECs (Boyd *et al.*, 2004; Ruitenberg *et al.*, 2002) before transplanting these cells into the adult rat spinal cord.

In the experiments reported in this thesis, OEC cultures were exposed to the BAG replication-deficient RV harboring Lac-Z gene. Lac-Z-expression by OECs did not alter the morphology of these cells either in vitro or in vivo. After grafting onto the pial surface of the lower thoracic segment of the adult rat spinal cord, Lac-Z expressing OECs were detected within the cerebellum, brainstem and spinal cord 4 weeks after transplantation in Group 8 rats. The morphology of these cells was similar to that of the DiI-labelled OECs found in the spinal cord of Group 1-7 rats, being elongated cells with unbranched cytoplasmic processes. The expression of β -galactosidase by OECs transplanted into a one-week-old cystic cavity in the adult rat spinal cord was also detected three weeks after transplantation (Boyd *et al.*, 2004). The expression of *Lac-Z* gene did not appear to affect the phenotype, survival or migratory ability of the OECs in either the present study or the study of Boyd et al. (2004). Moreover, Arroyo et al. (1998) reported that Lac-Z expression does not hinder the ability of Schwann cells to produce myelin or associate with unmyelinated axons in vivo. Remyelination of spinal cord axons was also observed by either Schwann cells or OECs derived from a transgenic rat expressing human placental alkaline phosphatase (Akiyama et al., 2004).

A fluorescent marker, GFP (Clontech, Palo alto, CA), can be also incorporated into a cell by transfecting it with adenovirus (Li *et al.*, 2003). GFP expressing mice bred in the Jackson Laboratory (Bar Harbor, ME, USA) have been used as a source for obtaining various GFP expressing cells, e.g. bone marrow cells (Akiyama *et al.*, 2002a). In addition, GFP can be delivered into cells (e.g. OECs) *in vitro* by means of infection with either adenovirus or lentivirus harboring the GFP gene. GFP-expression by OECs, which were transplanted into an adult rat spinal cord, did not alter the phenotype of grafted cells, which displayed a bipolar morphology similar to that seen in the present study with DiI+ve cells and β -Gal+ve cells, and also expressed the p75NGFR (Ruitenberg *et al.*, 2002). In contrast to DiI, GFP does not leak out of a cell. However, Ruitenberg *et al.* (2002) found that only approximately 50 % of the OECs expressed GFP *in vitro* after the infection of cultures with either adenovirus or lentivirus. In addition, the expression of GFP by OECs was detected for only up to 12-14 days after cell transplantation into an adult rat spinal cord (Ruitenberg *et al.*, 2002), in contrast to being able to see DiI+ve cells and β -Gal+ve cells as long as 2 months after grafting into the dorsal funiculus or onto the pial surface of the spinal cord, respectively (present study).

It is still unclear whether viral transfection has any long-term consequences on the phenotype of infected cells. Based on the data presented in this research project and that reported by others it would appear that transfection with a viral vector has no cytopathic effect on the cells (Akiyama *et al.*, 2004; Boyd *et al.*, 2004; Homnou *et al.*, 1996; Mosahebi *et al.*, 2001; Shy *et al.*, 1995). Genetic labeling is therefore a stable and reliable method for prelabeling cells prior to grafting (Mosahebi, *et al.*, 2001) provided the viral vector has a selectable marker gene for purifying the cell culture.

5.6 Methods of Delivery of Cells to Injured CNS

Manipulation of the cellular environment within and adjacent to injured CNS tissue through the use of cell transplantation is being studied in animal models of demyelinating disease (e.g., EtBr or lysolecithin) and of spinal cord injury (e.g., spinal compression). These manipulations have included the transplantation of Schwann cells (Neuberger *et al.*, 1992; Ramon-Cueto *et al.*, 1998; Xu *et al.*, 1997), OECs (Imaizumi *et al.*, 1998; Li *et al.*, 1997; Navarro *et al.*, 1999; Ramon-Cueto *et al.*, 2000; Smale *et al.*, 1996), immature astrocytes (Silver, 1998; Smith *et al.*, 1986), oligodendrocyte progenitor cells (Franklin, 1996; Lachapelle et al., 1994; Milward et al., 2000; O'Leary and Blakemore, 1997), and bone marrow progenitor cells (Akiyama et al., 2002a; Sasaki *et al.*, 2001). Cells that have been genetically modified to secrete large amounts of specific growth factors have also been used to make the microenvironment more conducive to axonal growth (Cunningham *et al.*, 1991; Kawaja *et al.*, 1991; Kawaja

al., 1992). Although the usual method of delivery of such cells to the damaged neural tissue is by injection, a procedure that by its very nature is traumatic to the brain and spinal cord, there are other less invasive options that have also been explored.

5.6.1 Grafting Cells Directly into CNS Tissue

Bony structures form a protective shield around the fragile brain and spinal cord. Therefore, in order for the cells to be grafted directly into the CNS tissue the investigator must inflict various degrees of destruction on either the bones of the skull or on the vertebrae, unless perhaps these bones have already been broken as the initial injury was induced. In most of the animal models, however, the cells have been injected into the neuropil of the spinal cord or brain only after mechanical removal of part of the bony protection. Directly injecting cells into an EtBr-induced (Akiyama et al., 2001, 2002a, 2004; Barnett et al., 2000; Blakemore and Crang, 1989; Franklin et al., 1996, 1997; Jeffery et al., 1999; Sasaki et al., 2001) or lysolecithin-induced (Baron-Van Evercooren et al., 1992, 1996; Duncan et al., 1981; Vignais et al., 1993) demyelinated lesion, into the cystic cavity that forms after spinal injury (Boyd et al., 2003; Lu et al., 2002; Resnick et al., 2003; Takami et al., 2002), or into the proximal and/or distal stumps of the transected spinal cord (Lu et al., 2001; Ramon-Cueto et al., 1998, 2000), has been one of the favoured approaches for delivering cells of interest to injured neural tissue. In many of these cell grafting experiments a stereotaxic apparatus was used to deliver the cells to a precise location, which was especially important when the migratory abilities of the transplanted cells were being studied. The cell suspensions were mostly injected into the neural tissue using either 5 or 10 µl Hamilton syringes with an attached glass or metal (30-32 G) needle. Compared to cell injection done by hand, a syringe pump like the one used in the present experiments makes cell grafting far more accurate by being able to control both the flow rate and the volume injected. There is no way to avoid some mechanical damage to the neural tissue when a needle, no matter how small, is inserted into the brain or spinal cord. Moreover, it would be inappropriate, as part of a therapeutic approach to neural repair in humans, to inflict yet additional damage on the patient's nervous tissue during the delivery of cells meant to facilitate the tissue repair.

Most of the experimental cell therapies being developed in animal models for delivering myelinating glia to the site of demyelination have used a cell grafting approach (see Section 1.7). Although this approach shows promise in terms of neural repair, facilitating functional recovery, and restoring axonal conduction, the limited migration of oligodendrocyte progenitor cells and Schwann cells through the CNS parenchyma restricts the beneficial effect of transplanting these two cell types to the immediate vicinity of the site of injury (Baron-Van Evercooren *et al.*, 1992; Blakemore *et al.*, 1986; Fawcett and Asher, 1999; Franklin and Blakemore, 1993, 1997). The therapeutic usefulness of focal cell grafts, therefore, is of limited benefit in the treatment of diseases such as MS in which multifocal demyelination is the main pathological feature. Thus, in an MS patient there would have to be multiple such injections of myelinating glia, even taking into account the ability of OECs to respond to a migratory signal(s) arising as a result of the multiple lesions of focal demyelination.

5.6.2 The Vascular Route

The intravenous route of administration is widely used in modern medicine to deliver various drugs to target tissues and has also been used to deliver cells to an EtBrinduced and x-irradiated demyelinated lesion (Akiyama *et al.*, 2002; Inoue *et al.*, 2003) or to an ischemic area of the cerebral hemisphere induced by occlusion of the middle cerebral artery (Chen *et al.*, 2001). Bone marrow cells (Akiyama *et al.*, 2002; Li *et al.*, 2001; Inoue *et al.*, 2003), Schwann cells and OECs (Akiyama *et al.*, 2002) are the cell types that have been delivered via this route. Intravenous administration of bone marrow cells in animals with demyelinated lesions resulted in varying degrees of remyelination, with the myelin formed being of both peripheral (i.e., Schwann cell-like) and central (i.e., oligodendrocyte-like) type myelin, as well as improving the conduction velocity of the remyelinated axons (Akiyama *et al.*, 2002; Inoue *et al.*, 2003). An improvement on the adhesive-removal behavioral test and the modified neurologic severity scores in adult rats with cerebral ischemia were also reported subsequent to an intravenous infusion of bone marrow cells (Li *et al.*, 2001).

Adverse effects, if any, on non-neural organs such as the liver, spleen, etc through administration of cells derived from bone marrow via the intravenous route has

yet to be determined, and especially whether there are effects that may arise in animals after long survival times. Inoue *et al.* (2003) reported that they had to inject 100 times more bone marrow cells via the femoral vein to achieve a comparable density of remyelination relative to that achieved after direct intraspinal injection of these cells. However, although a relatively large number of axons were remyelinated in such a lesion subsequent to transplantation of Lac-Z expressing bone marrow cells directly into the demyelinated lesion, Akiyama *et al.* (2002) reported that only a small percentage of the myelinating cells were β -galactosidase positive. Further studies are needed to determine whether intravenously injected bone marrow cells do actually give rise to myelinating glia or whether the tissue repair is due to the facilitation of an endogenous repair mechanism. Furthermore, in contrast to that seen with the bone marrow cells, no remyelination was seen in areas of x-irradiation/EtBr-induced demyelination after intravenous delivery of either Schwann cells or OECs (Akiyama *et al.*, 2002).

Akiyama et al. (2002) suggested that intravenously delivered cells (e.g., bone marrow cells) are unlikely to migrate across the blood brain barrier into normal spinal cord or brain tissue, because this barrier would continue to function as it normally does in restricting access to the nervous system parenchyma. In support of this suggestion they reported that no intravenously delivered bone marrow cells were observed in CNS areas other than in the lesion. The focused entry of prelabeled bone marrow cells into sites of CNS injury in experiments where they were delivered intravenously is likely due to the ability of the cells to respond to chemotactic factors (e.g., monocyte chemoattractant protein 1), the expression of which is increased in damaged CNS tissue (Kim, 1996). In addition, the expression of cell adhesion molecules, such as intercellular adhesion molecule (Zhang *et al.*, 1995), vascular adhesion molecule 1, and E-selectin (Blann et al., 1990; Haraldsen et al., 1996) may also contribute to the migration of the prelabeled bone marrow cells across the blood brain barrier and into the damaged neuropil. Therefore, the vascular route is an approach that may prove benefical in the treatment of neurological diseases like MS and spinal cord injury where complex neurosurgical interventions can and should be avoided.

5.6.3 Injection of Cells Into CSF

There have been a few studies using an animal model of CNS demyelination in which progenitor cells have been delivered intracisternally (Einstein et al., 2003; Learish et al., 1999; Pluchino et al., 2003). For example, neuronal progenitor cells prelabelled with bromodeoxyuridine have been injected into the lateral ventricle of C57BL/6 mice that are suffering from EAE (Einstein et al., 2003). Transplantation of these neuronal progenitor cells resulted in attenuation of clinical severity of EAE and brain inflammation of animals, indicated by reduction in perivascular infiltrates. Pluchino et al. (2003) injected neuronal progenitor cells infected with lentivirus harboring a Lac-Z gene into the lateral ventricle of mice with EAE and observed an increased number of oligodendrocyte progenitors and remyelinated axons within demyelinated areas. Learish et al. (1999) injected Lac-Z-expressing oligodendrocyte progenitor cells into the cerebral ventricles of embryonic myelin-deficient mice. The donor progenitor cells migrated into the CNS parenchyma including the corpus callosum, inferior hippocampus, thalamus, and inferior colliculus and facilitated neural repair. Learish et al. (1999) suggested that an intraventricular injection is a feasible approach to cell delivery in patients with demyelinating disorders, particularly for the demyelinated lesions located in periventricular white mater. However, an intraventricular injection of cells still involves some degree of mechanical injury to the cerebral hemisphere because the injection needle must pass through normal brain tissue before the tip enters the ventricular system.

In clinical practice and in experimental studies that involve the use of large animals, lumbar puncture is a widely used technique for collecting CSF or for delivery of drugs. An alternative approach for cell delivery is an intrathecal injection into the subarachnoid space, which is located between the arachnoid and pia mater and is filled with CSF. Since the pia mater intimately adheres to the surface of the brain and spinal cord, the subarachnoid space widens where the brain surface exhibits a deep sulcus, for example between the cerebellum and medulla, thus forming a cistern (e.g., cisterna magna). The lumbar cistern is formed as a pouch by the extension of the arachnoid mater caudally into the lumbar and sacral portions of the vertebral column leaving behind the pia mater covering the conus medularis and fillum terminalis, as well as the lumbar and sacral rootlets. In adult humans, the spinal cord ends at the level of the first lumbar vertebra although it is much lower in the vertebral column in children. A needle insertion into the lumbar cistern is called a lumbar puncture and is considered a relatively safe procedure in comparison to needle insertion into the cisterna magna.

Only one study so far has reported on an intrathecal injection of neuronal progenitor cells in an EAE animal model (Ben-Hur et al., 2003). Injected into the CSF, neuronal precursor cells were found exclusively within demyelinated areas of brain and spinal cord white matter. Recently, De la Calle and Paino (2002) described a detailed procedure for direct lumbar puncture in rats, which was modified from that described previously by Hylden and Wilcox *et al.* (1980). Under isofluorane anesthesia, a 2 cm longitudinal percutaneous incision was made with a scalpel at the lower lumbar level and a 25G neonatal lumbar puncture needle was inserted through the intervertebral space between the L5 and L6 vertebrae. De la Calle *et al.* (2002) used this lumbar puncture technique to inject a cell suspension of Hoechst-prelabelled NB69 neuroblastoma cells, which are known to synthesize and secrete monoamines into the subarachnoid space of the rat with chronic contusion injury. Intrathecally grafted NB69 neuroblastoma cells survived and maintained their ability to secrete monoamines, thus abolishing chronic contusion injury-induced heat hyperalgesia.

The use of a direct lumbar puncture rather than injecting the cells into the cisterna magna or through brain tissue into the lateral ventricles is of particular clinical relevance. A lumbar puncture can be done relatively quickly, on the order of 10 minutes or less, and in comparison to needle insertion into the cisterna magna or the ventricles is far less traumatic for the patient. An additional advantage of injecting the cells directly into the CSF is that instead of the grafted cells immediately encountering the aggressive, inflammatory environment of the lesion epicenter, and thus possibly compromising the survival of the grafted cells, the myelinating glia could potentially benefit from the CSF in terms of being a source of nutrients. Admittedly, there is also a concern with possible damage to the spinal cord and/or the rootlets at the level of the lumbar puncture. Other possible complications that might arise after lumbar puncture

include headache, CNS infection (especially in immunocompromised patients), and intraspinal hemorrhage in patients receiving anticoagulation therapy. However, these risks are relatively low. In their study, De la Calle and Paino (2002) detected no signs of motor or sensory deficit in the rats subjected to lumbar puncture and saw no grossly detectable lesions in or on the lumbar spinal cord on post mortem examination.

5.7 General Conclusions

An *in vivo* model for studying the migratory ability of OECs within the adult rat spinal cord was developed for the present research study. A small focal (< 1 mm long) demyelination was induced by injecting 1µl of a 0.03% EtBr solution into the right dorsal funiculus of the spinal cord. This lesion was made at variable distances from the site of an OEC graft. In this model, the strength of the migratory signal(s) inducing OECs to migrate increased as the demyelinated lesion was located closer to the grafting site, with a significant increase in the number of OECs in the dorsal funiculus of the intervening segments being first apparent by 8 weeks after grafting. OECs grafted at T12 migrated as far as four spinal cord segments to reach a focal EtBr-induced demyelinated lesion located at T8. There was minimal migration of OECs when the cells were grafted directly into a demyelinated lesion unless a second demyelinated lesion was made two spinal cord segments away from the first, in which case there was precocious migration of OECs. The OEC migration occurred along a migratory path containing many reactive astrocytes and microglia and this migration of OECs was significantly reduced when the microglial reactivity was dampened using minocycline. OECs also survived grafting into cerebrospinal fluid (i.e. subarachnoid space) and migrated into the neuropil of the brain and spinal cord. The major conclusions are that OECs can respond to migratory signal(s) arising as a result of a focal EtBr-induced demyelination and that microglia are one potential source of these migratory signal(s).

5.8 **Future Directions**

The EtBr-induced demyelination animal model used in the present research project will be extremely useful for determining:

- the time frame of microglial response contributing to the generation of the migratory signal(s) that induces OECs to migrate towards the lesion;
- the molecular mechanisms facilitating and impeding the migration of OECs, for example:
 - a. whether proinflammatory cytokines contribute to a migratory signal inducing OECs to migrate;
 - b. whether anti-inflammatory cytokines attenuate the migratory signal;
 - whether OECs express specific cytokine receptors to facilitate or impede their migration in response to proinflammatory or anti-inflammatory cytokines.
- the contribution of grafted OECs to neural repair (including remyelination) once they have migrated into the neuropil of a focal demyelination lesion;
- the ability of OECs to migrate towards a focal demyelination of spinal cord white matter and contribute to neural repair when the cells are grafted into the CSF via a lumbar puncture.

Before adopting a therapeutic approach that involves glial cell transplantations in treating demyelinated diseases, it is important to get a better understanding of host cell-grafted cell interactions in order to maximize the beneficial consequences of myelinating glial cell grafting. In MS multiple demyelinated lesions are not only scattered throughout the CNS neuropil, but the pathogenesis of each plaque develops with its own time frame with each new relapse adding to the pathology and creating new lesions. Thus, at any point during the pathogenesis of the disease in the same MS patient plaques can be found at different stages of progression. The host glial cell response within the demyelinated area and within the surrounding tissue depends on the stage of plaque progression with the highest level of glial reactivity being seen at the early, acute stages and lower reactivity at later, chronic stages. As presented in this research project, microglia are implicated as one potential source, either directly or indirectly, of the migratory signal(s) driving OEC migration in an EtBr-induced demyelination animal model. Why is it important to identify the source of these migratory signals? Cell migration would be a far more efficient way of delivering myelinating glia to multiple areas of CNS injury, such as the isolated MS plaques, than trying to deliver the cells directly to each injured area.

Although the distances over which migratory signals are working in the present study is only one spinal cord segment, this is still 2-3 times further than either oligodendrocyte progenitors or Schwann cells will migrate in response to a demyelinated lesion. OEC migration occurred along a migratory path containing many reactive astrocytes and microglia and this migration was significantly reduced when the microglial reactivity was dampened using a 4 week minocycline treatment. This indicates that the first 4 week period after cell grafting and EtBr injection is the time period when the microglia contributed to the generation of a migratory signal(s). By varying the duration and when in the first 4 week period the minocycline treatment is started, it will be possible to determine whether it is the early (i.e. 3-7 days) or late (1-4 weeks) microglial response to injury that is likely to be most important in contributing to the generation of a migratory signal(s). Knowing this time frame is important for the design of experiments to identify the molecular mechanisms. One can hypothesize that the microglial reactivity during the first two weeks after EtBr injection is the most critical for generating a migratory signal(s) and for creating a migratory pathway conducive to OEC migration because this would be the time interval during which the microglial cells would be the most reactive. For example, the early treatment with clodronate of young adult female rats that have a lysolecithin-induced demyelinated lesion in their spinal cord resulted in the depletion of microglia and macrophages and in a decreased remyelination by oligodendrocytes, whereas a late clodronate administration had no effect on the repair outcome even though still depleting numbers of activated macrophages (Kotter et al., 2001). In MS the microglial response to a progression of the pathogenesis in each plaque is reflected in the profile of released cytokines (e.g., the type and level of cytokine), excitatory amino acids, oxidative radicals, and nitric oxide. By identifying the time frame of the microglial response to

plaque progression, we can predict whether early (e.g., IL-1b) vs. late (e.g., IL-12) proinflammatory and/or anti-inflammatory (e.g., IL-10) cytokines contribute to the migratory signal(s) inducing OECs to migrate.

Transplantation of myelinating glia will be an essential part of potential therapeutic approaches to neural repair in MS and spinal cord injury. In addition to finding the best cell candidate, the method of cell delivery is also an important issue of any therapeutic approach involving cell transplantation. Although various degrees of success in CNS tissue regeneration and remyelination have been achieved after cell transplantation, several aspects specific to MS remain neglected in these animal models. The demyelinated lesions in MS are disseminated throughout the brain and spinal cord and, thus, it is certainly not possible to transplant cells into each and every lesion. Moreover, some demyelinating lesions can be located within vitally important areas such as the brain stem or located in CNS areas that are surgically inaccessible without causing undue mechanical damage to surrounding tissues (e.g., the optic nerve). Therefore, delivery of cells into the CSF by means of a lumbar puncture has many advantages over direct injection into the neuropil of the CNS. Without inflicting unnecessary additional damage to the CNS tissue, the lumbar puncture route could use the CSF to distribute the grafted cells to those parts of the CNS that were demyelinated. As reported in this thesis, OECs grafted into either normal or demyelinated areas of the CNS neuropil can be induced to migrate towards a focal area(s) of demyelination. When the OECs were grafted into the CSF, some were found attached to the pia mater at some distance from the site of grafting as well as being found within the neuropil of the same brain areas. One can hypothesize, therefore, that once OECs are grafted into the subarachnoid space they will float in the CSF, attach to the pia matter at various levels of the neuraxis, and then migrate across the damaged blood-brain barrier into the demyelinated lesions where they will contribute to neural repair.

6. **REFERENCES**

- Aarum, J., K. Sandberg, S. L. Haeberlein and M. A. Persson (2003). "Migration and differentiation of neural precursor cells can be directed by microglia." <u>Proc Natl</u> <u>Acad Sci U S A</u> 100(26): 15983-8.
- Aboul-Enein, F., H. Rauschka, B. Kornek, C. Stadelmann, A. Stefferl, W. Bruck, C. Lucchinetti, M. Schmidbauer, K. Jellinger and H. Lassmann (2003).
 "Preferential loss of myelin-associated glycoprotein reflects hypoxia-like white matter damage in stroke and inflammatory brain diseases." J Neuropathol Exp Neurol 62(1): 25-33.
- Adamson, P., S. Etienne, P. O. Couraud, V. Calder and J. Greenwood (1999).
 "Lymphocyte migration through brain endothelial cell monolayers involves signaling through endothelial ICAM-1 via a rho-dependent pathway." J
 <u>Immunol</u> 162(5): 2964-73.
- Agranoff, B. W. and D. Goldberg (1974). "Diet and the geographical distribution of multiple sclerosis." Lancet 2(7888): 1061-6.
- Aguayo,A.J. (1985). Axonal regeneration from injured neurons in the adult mammalian central nervous system. In: Synaptic Plasticity. C.W. Cotman, ed. Guilford Press, New York, pp.457-483.
- Akassoglou, K., L. Probert, G. Kontogeorgos and G. Kollias (1997). "Astrocyte-specific but not neuron-specific transmembrane TNF triggers inflammation and degeneration in the central nervous system of transgenic mice." <u>J Immunol</u> 158(1): 438-45.
- Akiyama, Y., O. Honmou, T. Kato, T. Uede, K. Hashi and J. D. Kocsis (2001).
 "Transplantation of clonal neural precursor cells derived from adult human brain establishes functional peripheral myelin in the rat spinal cord." <u>Exp Neurol</u> 167(1): 27-39.
- Akiyama Y., K. Lankford, C. Radtke, CA. Greer, and JD. Kocsis (2004).
 "Remyelination of spinal cord axons by olfactory ensheathing cells and Schwann cells derived from a transgenic rat expressing alkaline phosphatase marker gene Neuron." <u>Glia Biology</u> 1: 47–55.
- Akiyama, Y., C. Radtke, O. Honmou and J. D. Kocsis (2002a). "Remyelination of the spinal cord following intravenous delivery of bone marrow cells." <u>Glia</u> 39(3): 229-36.
- Akiyama, Y., C. Radtke and J. D. Kocsis (2002b). "Remyelination of the rat spinal cord by transplantation of identified bone marrow stromal cells." <u>J Neurosci</u> 22(15): 6623-30.
- Al-Omaishi, J., R. Bashir and H. E. Gendelman (1999). "The cellular immunology of multiple sclerosis." J Leukoc Biol 65(4): 444-52.

- Alonso, G. and A. Privat (1993a). "Neuropeptide Y-producing neurons of the arcuate nucleus regenerate axons after surgical deafferentation of the mediobasal hypothalamus." J Neurosci Res 34(5): 510-22.
- Alonso, G. and A. Privat (1993b). "Reactive astrocytes involved in the formation of lesional scars differ in the mediobasal hypothalamus and in other forebrain regions." J Neurosci Res 34(5): 523-38.
- Anderson, T. J., P. Montague, N. Nadon, K. A. Nave and I. R. Griffiths (1997).
 "Modification of Schwann cell phenotype with Plp transgenes: evidence that the PLP and DM20 isoproteins are targeted to different cellular domains." J Neurosci Res 50(1): 13-22.
- Ando, S., Y. Tanaka, Y. Toyoda and K. Kon (2003). "Turnover of myelin lipids in aging brain." <u>Neurochem Res</u> 28(1): 5-13.
- Anton, E. S., M. Hadjiargyrou, P. H. Patterson and W. D. Matthew (1995). "CD9 plays a role in Schwann cell migration in vitro." J Neurosci 15(1 Pt 2): 584-95.
- Anton, E. S., G. Weskamp, L. F. Reichardt and W. D. Matthew (1994). "Nerve growth factor and its low-affinity receptor promote Schwann cell migration." <u>Proc Natl Acad Sci U S A</u> 91(7): 2795-9.
- Archer, D. R., P. A. Cuddon, D. Lipsitz and I. D. Duncan (1997). "Myelination of the canine central nervous system by glial cell transplantation: a model for repair of human myelin disease." <u>Nat Med</u> 3(1): 54-9.
- Armstrong, R. C., L. Harvath and M. E. Dubois-Dalcq (1990). "Type 1 astrocytes and oligodendrocyte-type 2 astrocyte glial progenitors migrate toward distinct molecules." <u>J Neurosci Res</u> 27(3): 400-7.
- Armstrong, R. C., T. Q. Le, E. E. Frost, R. C. Borke and A. C. Vana (2002). "Absence of fibroblast growth factor 2 promotes oligodendroglial repopulation of demyelinated white matter." J Neurosci 22(19): 8574-85.
- Arnett, H. A., J. Mason, M. Marino, K. Suzuki, G. K. Matsushima and J. P. Ting (2001). "TNF alpha promotes proliferation of oligodendrocyte progenitors and remyelination." <u>Nat Neurosci</u> 4(11): 1116-22.
- Arnon, R. (1996). "The development of Cop 1 (Copaxone), an innovative drug for the treatment of multiple sclerosis: personal reflections." <u>Immunol Lett</u> 50(1-2): 1-15.
- Arroyo, E. J., J. R. Bermingham, Jr., M. G. Rosenfeld and S. S. Scherer (1998).
 "Promyelinating Schwann cells express Tst-1/SCIP/Oct-6." J Neurosci 18(19): 7891-902.
- Asensio, V. C. and I. L. Campbell (1999). "Chemokines in the CNS: plurifunctional mediators in diverse states." <u>Trends Neurosci</u> 22(11): 504-12.
- Azizi, S. A., D. Stokes, B. J. Augelli, C. DiGirolamo and D. J. Prockop (1998).
 "Engraftment and migration of human bone marrow stromal cells implanted in the brains of albino rats--similarities to astrocyte grafts." <u>Proc Natl Acad Sci U S A</u> 95(7): 3908-13.
- Babbe, H., A. Roers, A. Waisman, H. Lassmann, N. Goebels, R. Hohlfeld, M. Friese, R. Schroder, M. Deckert, S. Schmidt, R. Ravid and K. Rajewsky (2000). "Clonal expansions of CD8(+) T cells dominate the T cell infiltrate in active multiple sclerosis lesions as shown by micromanipulation and single cell polymerase chain reaction." J Exp Med 192(3): 393-404.

- Babcock, A. A., W. A. Kuziel, S. Rivest and T. Owens (2003). "Chemokine expression by glial cells directs leukocytes to sites of axonal injury in the CNS." J Neurosci 23(21): 7922-30.
- Bach, J. F. (2002). "Current concepts of autoimmunity." <u>Rev Neurol (Paris)</u> 158(10 Pt 1): 881-6.
- Badie, B., J. Schartner, J. Klaver and J. Vorpahl (1999). "In vitro modulation of microglia motility by glioma cells is mediated by hepatocyte growth factor/scatter factor." <u>Neurosurgery</u> 44(5): 1077-82; discussion 1082-3.
- Baehr, M. and R. P. Bunge (1989). "Functional status influences the ability of Schwann cells to support adult rat retinal ganglion cell survival and axonal regrowth." <u>Exp</u> <u>Neurol</u> 106(1): 27-40.
- Bailey, M. S. and M. T. Shipley (1993). "Astrocyte subtypes in the rat olfactory bulb morphological heterogeneity and differential laminar distribution." <u>J Comp</u> <u>Neurol</u> 328(4): 501-526.
- Baker, H. and A. I. Farbman (1993). "Olfactory afferent regulation of the dopamine phenotype in the fetal rat olfactory system." <u>Neuroscience</u> 52(1): 115-134.
- Bakker, D. A. and S. K. Ludwin (1987). "Blood-brain barrier permeability during Cuprizone-induced demyelination. Implications for the pathogenesis of immunemediated demyelinating diseases." J Neurol Sci 78(2): 125-37.
- Balasingam, V., T. Tejada-Berges, E. Wright, R. Bouckova and V. W. Yong (1994).
 "Reactive astrogliosis in the neonatal mouse brain and its modulation by cytokines." J Neurosci 14(2): 846-56.
- Banati, R. B. and M. B. Graeber (1994). "Surveillance, intervention and cytotoxicity: is there a protective role of microglia?" <u>Dev Neurosci</u> 16(3-4): 114-27.
- Bansal, R., A. E. Warrington, A. L. Gard, B. Ranscht and S. E. Pfeiffer (1989).
 "Multiple and novel specificities of monoclonal antibodies O1, O4, and R-mAb used in the analysis of oligodendrocyte development." <u>J Neurosci Res</u> 24(4): 548-57.
- Barber, P. C. (1981). "Regeneration of vomeronasal nerves into the main olfactory bulb in the mouse." <u>Brain Res</u> 216(2): 239-51.
- Barber, P. C. (1982). "Neurogenesis and regeneration in the primary olfactory pathway of mammals." <u>Bibl Anat(23)</u>: 12-25.
- Barber, P. C. and R. M. Lindsay (1982). "Schwann cells of the olfactory nerves contain glial fibrillary acidic protein and resemble astrocytes." <u>Neurosci.</u> 7: 3077-3090.
- Barber, P. C. and G. Raisman (1978). "Inhibition by puromycin of transneuronal transport in the mouse accessory olfactory bulb." Brain Res 152(2): 303-11.
- Barnett, S. C. and L. Chang (2004). "Olfactory ensheathing cells and CNS repair: going solo or in need of a friend?" <u>Trends Neurosci</u> 27(1): 54-60.
- Barnett, S. C. and J. S. Riddell (2004). "Olfactory ensheathing cells (OECs) and the treatment of CNS injury: advantages and possible caveats." J Anat 204(1): 57-67.
- Barnett, M. H. and J. W. Prineas (2004). "Relapsing and remitting multiple sclerosis: pathology of the newly forming lesion." <u>Ann Neurol</u> 55(4): 458-68.
- Barnett, S. C., C. L. Alexander, Y. Iwashita, J. M. Gilson, J. Crowther, L. Clark, L. T. Dunn, V. Papanastassiou, P. G. Kennedy and R. J. Franklin (2000).

"Identification of a human olfactory ensheathing cell that can effect transplantmediated remyelination of demyelinated CNS axons." <u>Brain</u> 123 (Pt 8): 1581-8.

- Barnett, S. C., R. J. Franklin and W. F. Blakemore (1993). "In vitro and in vivo analysis of a rat bipotential O-2A progenitor cell line containing the temperaturesensitive mutant gene of the SV40 large T antigen." <u>Eur J Neurosci</u> 5(10): 1247-60.
- Barnett, S. C., A. M. Hutchins and M. Noble (1993). "Purification of olfactory nerve ensheathing cells from the olfactory bulb." <u>Dev Biol</u> 155(2): 337-50.
- Baron-Van Evercooren, A., V. Avellana-Adalid, A. Ben Younes-Chennoufi, A. Gansmuller, B. Nait-Oumesmar and L. Vignais (1996). "Cell-cell interactions during the migration of myelin-forming cells transplanted in the demyelinated spinal cord." <u>Glia</u> 16(2): 147-64.
- Baron-Van Evercooren, A., E. Clerin-Duhamel, P. Lapie, A. Gansmuuler, F. Lachapelle and M. Gumpel (1992). "The fate of Schwann cells transplanted in the brain during development." <u>Devel. Neurosci.</u> 14: 73-84.
- Baron-Van Evercooren, A., E. Duhamelclerin, J. M. Boutry, J. J. Hauw and M. Gumpel (1993). "Pathways of migration of transplanted Schwann cells in the demyelinated mouse spinal cord." <u>Journal of Neuroscience Research</u> 35(4): 428-438.
- Barres, B. A., I. K. Hart, H. S. Coles, J. F. Burne, J. T. Voyvodic, W. D. Richardson and M. C. Raff (1992). "Cell death and control of cell survival in the oligodendrocyte lineage." <u>Cell</u> 70(1): 31-46.
- Barron, K. D. (1995). "The microglial cell. A historical review." <u>J Neurol Sci</u> 134 Suppl: 57-68.
- Bartholdi, D. and M. E. Schwab (1997). "Expression of pro-inflammatory cytokine and chemokine mRNA upon experimental spinal cord injury in mouse: an in situ hybridization study." <u>Eur J Neurosci</u> 9(7): 1422-38.
- Bartnik, B. L., B. H. Juurlink and R. M. Devon (2000). "Macrophages: their myelinotrophic or neurotoxic actions depend upon tissue oxidative stress." <u>Mult</u> <u>Scler</u> 6(1): 37-42.
- Basbaum, C. B. and Z. Werb (1996). "Focalized proteolysis: spatial and temporal regulation of extracellular matrix degradation at the cell surface." <u>Curr Opin Cell</u> <u>Biol</u> 8(5): 731-8.
- Baumann, N. and D. Pham-Dinh (2001). "Biology of oligodendrocyte and myelin in the mammalian central nervous system." <u>Physiol Rev</u> 81(2): 871-927.
- Beattie, M. S., J. C. Bresnahan, J. Komon, C. A. Tovar, M. Van Meter, D. K. Anderson, A. I. Faden, C. Y. Hsu, L. J. Noble, S. Salzman and W. Young (1997).
 "Endogenous repair after spinal cord contusion injuries in the rat." <u>Exp Neurol</u> 148(2): 453-63.
- Beck, R. W. (1992). "Corticosteroid treatment of optic neuritis: a need to change treatment practices. The Optic Neuritis Study Group." <u>Neurology</u> 42(6): 1133-5.
- Beck, R. W. (1995). "The optic neuritis treatment trial: three-year follow-up results." <u>Arch Ophthalmol</u> 113(2): 136-7.
- Begolka, W. S., C. L. Vanderlugt, S. M. Rahbe and S. D. Miller (1998). "Differential expression of inflammatory cytokines parallels progression of central nervous

system pathology in two clinically distinct models of multiple sclerosis." J Immunol 161(8): 4437-46.

- Ben-Hur, T., O. Einstein, R. Mizrachi-Kol, O. Ben-Menachem, E. Reinhartz, D. Karussis and O. Abramsky (2003). "Transplanted multipotential neural precursor cells migrate into the inflamed white matter in response to experimental autoimmune encephalomyelitis." <u>Glia</u> 41(1): 73-80.
- Benveniste, E. N. (1997). "Role of macrophages/microglia in multiple sclerosis and experimental allergic encephalomyelitis." J Mol Med 75(3): 165-73.
- Berger, B. (1971). "Etude ultrastructurale de la degenerescence Wallerienne experimentale d'un nerf entierement amyelinique: II. Reactions cellulairs." <u>J.</u> <u>Ultrastruct. Res.</u> 37: 479-494.
- Berlet, H. H. and B. Volk (1980). "Studies of human myelin proteins during old age." <u>Mech Ageing Dev</u> 14(1-2): 211-22.
- Bermingham, J. R., Jr., S. S. Scherer, S. O'Connell, E. Arroyo, K. A. Kalla, F. L. Powell and M. G. Rosenfeld (1996). "Tst-1/Oct-6/SCIP regulates a unique step in peripheral myelination and is required for normal respiration." <u>Genes Dev</u> 10(14): 1751-62.
- Berry, M., S. Hall, L. Rees, J. Carlile and J. P. Wyse (1992). "Regeneration of axons in the optic nerve of the adult Browman-Wyse (BW) mutant rat." <u>J Neurocytol</u> 21(6): 426-48.
- Berthold, C. H. and T. Carlstedt (1977). "Observations on the morphology at the transition between the peripheral and the central nervous system in the cat. II. General organization of the transitional region in S1 dorsal rootlets." <u>Acta Physiol Scand Suppl</u> 446: 23-42.
- Bertolotto, A., M. Capobianco, S. Malucchi, E. Manzardo, L. Audano, M. Bergui, G. B. Bradac and R. Mutani (1999). "Transforming growth factor beta1 (TGFbeta1) mRNA level correlates with magnetic resonance imaging disease activity in multiple sclerosis patients." <u>Neurosci Lett</u> 263(1): 21-4.
- Bettinger, I., S. Thanos and W. Paulus (2002). "Microglia promote glioma migration." <u>Acta Neuropathol (Berl)</u> 103(4): 351-5.
- Bhattacharyya, A., R. W. Oppenheim, D. Prevette, B. W. Moore, R. Brackenbury and N. Ratner (1992). "S100 is present in developing chicken neurons and Schwann cells and promotes motor neuron survival in vivo." J Neurobiol 23(4): 451-66.
- Bielekova, B., A. Lincoln, H. McFarland and R. Martin (2000). "Therapeutic potential of phosphodiesterase-4 and -3 inhibitors in Th1-mediated autoimmune diseases." J Immunol 164(2): 1117-24.
- Bignami, A., L. F. Eng, D. Dahl and C. T. Uyeda (1972). "Localization of the glial fibrillary acidic protein in astrocytes by immunofluorescence." <u>Brain Res</u> 43(2): 429-35.
- Birkedal-Hansen, H., W. G. Moore, M. K. Bodden, L. J. Windsor, B. Birkedal-Hansen, A. DeCarlo and J. A. Engler (1993). "Matrix metalloproteinases: a review." <u>Crit</u> <u>Rev Oral Biol Med</u> 4(2): 197-250.
- Birling, M. C., G. Roussel, F. Nussbaum and J. L. Nussbaum (1993). "Biochemical and immunohistochemical studies with specific polyclonal antibodies directed against bovine myelin/oligodendrocyte glycoprotein." <u>Neurochem Res</u> 18(8): 937-45.

- Bitsch, A., C. da Costa, S. Bunkowski, F. Weber, P. Rieckmann and W. Bruck (1998).
 "Identification of macrophage populations expressing tumor necrosis factoralpha mRNA in acute multiple sclerosis." <u>Acta Neuropathol (Berl)</u> 95(4): 373-7.
- Bitsch, A., J. Schuchardt, S. Bunkowski, T. Kuhlmann and W. Bruck (2000). "Acute axonal injury in multiple sclerosis. Correlation with demyelination and inflammation." <u>Brain</u> 123 (Pt 6): 1174-83.
- Bixby, J. L., J. Lilien and L. F. Reichardt (1988). "Identification of the major proteins that promote neuronal process outgrowth on Schwann cells in vitro." <u>J Cell Biol</u> 107(1): 353-61.
- Bjartmar, C., R. P. Kinkel, G. Kidd, R. A. Rudick and B. D. Trapp (2001). "Axonal loss in normal-appearing white matter in a patient with acute MS." <u>Neurology</u> 57(7): 1248-52.
- Bjorklund, L. M., R. Sanchez-Pernaute, S. Chung, T. Andersson, I. Y. Chen, K. S. McNaught, A. L. Brownell, B. G. Jenkins, C. Wahlestedt, K. S. Kim and O. Isacson (2002). "Embryonic stem cells develop into functional dopaminergic neurons after transplantation in a Parkinson rat model." <u>Proc Natl Acad Sci U S</u> A 99(4): 2344-9.
- Black, J. A. and S. G. Waxman (1988). "The perinodal astrocyte." Glia 1(3): 169-83.
- Blakemore, W. F. (1972). "Observations on oligodendrocyte degeneration, the resolution of status spongiosus and remyelination in cuprizone intoxication in mice." <u>J Neurocytol</u> 1(4): 413-26.
- Blakemore, W. F. (1973). "Demyelination of the superior cerebellar peduncle in the mouse induced by cuprizone." J Neurol Sci 20(1): 63-72.
- Blakemore, W. F. (1974). "Remyelination of the superior cerebellar peduncle in old mice following demyelination induced by cuprizone." J Neurol Sci 22(1): 121-6.
- Blakemore, W. F. (1975). "Remyelination by Schwann cells of axons demyelinated by intraspinal injection of 6-aminonicotinamide in the rat." J Neurocytol 4(6): 745-57.
- Blakemore, W. F. (1978). "Observations on remyelination in the rabbit spinal cord following demyelination induced by lysolecithin." <u>Neuropathol Appl Neurobiol</u> 4(1): 47-59.
- Blakemore W. F. (1979). "Remyelination in the CNS. In Progress in Neurological Research." (eds Behan P. 0. and Clifford Rose F.). pp. 12-25. Pittman. Bath.
- Blakemore, W. F. (1984). "Limited remyelination of CNS axons by Schwann cells transplanted into the sub-arachnoid space." J Neurol Sci 64(3): 265-76.
- Blakemore, W. F. (1992). "Transplanted cultured type-1 astrocytes can be used to reconstitute the glia limitans of the CNS: the structure which prevents Schwann cells from myelinating CNS axons." <u>Neuropathol Appl Neurobiol</u> 18(5): 460-6.
- Blakemore, W. F., D. M. Chari, J. M. Gilson and A. J. Crang (2002). "Modelling large areas of demyelination in the rat reveals the potential and possible limitations of transplanted glial cells for remyelination in the CNS." <u>Glia</u> 38(2): 155-68.
- Blakemore, W. F. and A. J. Crang (1985). "The use of cultured autologous Schwann cells to remyelinate areas of persistent demyelination in the central nervous system." J Neurol Sci 70(2): 207-23.
- Blakemore, W. F. and A. J. Crang (1989). "The relationship between type-1 astrocytes, Schwann cells and oligodendrocytes following transplantation of glial cell

cultures into demyelinating lesions in the adult rat spinal cord." <u>J Neurocytol</u> 18(4): 519-28.

- Blakemore, W. F., A. J. Crang and R. Curtis (1986). "The interaction of Schwann cells with CNS axons in regions containing normal astrocytes." <u>Acta Neuropathol</u> (Berl) 71(3-4): 295-300.
- Blakemore, W. F., R. A. Eames, K. J. Smith and W. I. McDonald (1977).
 "Remyelination in the spinal cord of the cat following intraspinal injections of lysolecithin." J Neurol Sci 33(1-2): 31-43.
- Blakemore, W. F., J. M. Gilson and A. J. Crang (2003). "The presence of astrocytes in areas of demyelination influences remyelination following transplantation of oligodendrocyte progenitors." <u>Exp Neurol</u> 184(2): 955-63.
- Blakemore, W. F., N. J. Olby and R. J. Franklin (1995). "The use of transplanted glial cells to reconstruct glial environments in the CNS." <u>Brain Pathol</u> 5(4): 443-50.
- Blakemore, W. F. and R. C. Patterson (1978). "Suppression of remyelination in the CNS by X-irradiation." <u>Acta Neuropathol (Berl)</u> 42(2): 105-13.
- Blanchard, A. D., A. Sinanan, E. Parmantier, R. Zwart, L. Broos, D. Meijer, C. Meier, K. R. Jessen and R. Mirsky (1996). "Oct-6 (SCIP/Tst-1) is expressed in Schwann cell precursors, embryonic Schwann cells, and postnatal myelinating Schwann cells: comparison with Oct-1, Krox-20, and Pax-3." J Neurosci Res 46(5): 630-40.
- Blinzinger, K. and G. Kreutzberg (1968). "Displacement of synaptic terminals from regenerating motoneurons by microglial cells." <u>Z Zellforsch Mikrosk Anat</u> 85(2): 145-57.
- Bo, L., S. Mork, P. A. Kong, H. Nyland, C. A. Pardo and B. D. Trapp (1994).
 "Detection of MHC class II-antigens on macrophages and microglia, but not on astrocytes and endothelia in active multiple sclerosis lesions." <u>J Neuroimmunol</u> 51(2): 135-46.
- Bogumil, T., P. Rieckmann, B. Kubuschok, K. Felgenhauer and W. Bruck (1998).
 "Serum levels of macrophage-derived protein MRP-8/14 are elevated in active multiple sclerosis." <u>Neurosci Lett</u> 247(2-3): 195-7.
- Boison, D., H. Bussow, D. D'Urso, H. W. Muller and W. Stoffel (1995). "Adhesive properties of proteolipid protein are responsible for the compaction of CNS myelin sheaths." J Neurosci 15(8): 5502-13.
- Bonetti, B. and C. S. Raine (1997). "Multiple sclerosis: oligodendrocytes display cell death-related molecules in situ but do not undergo apoptosis." <u>Ann Neurol</u> 42(1): 74-84.
- Bornstein, M. B., A. Miller, S. Slagle, M. Weitzman, H. Crystal, E. Drexler, M. Keilson, A. Merriam, S. Wassertheil-Smoller, V. Spada and et al. (1987). "A pilot trial of Cop 1 in exacerbating-remitting multiple sclerosis." <u>N Engl J Med</u> 317(7): 408-14.
- Boruch, A. V., J. J. Conners, M. Pipitone, G. Deadwyler, P. D. Storer, G. H. Devries and K. J. Jones (2001). "Neurotrophic and migratory properties of an olfactory ensheathing cell line." <u>Glia</u> 33(3): 225-9.
- Bot, J. C., F. Barkhof, C. H. Polman, G. J. Lycklama a Nijeholt, V. de Groot, E. Bergers, H. J. Ader and J. A. Castelijns (2004). "Spinal cord abnormalities in

recently diagnosed MS patients: added value of spinal MRI examination." <u>Neurology</u> 62(2): 226-33.

- Boyd, J. G. and T. Gordon (2003). "Neurotrophic factors and their receptors in axonal regeneration and functional recovery after peripheral nerve injury." <u>Mol</u> <u>Neurobiol</u> 27(3): 277-324.
- Boyd, J. G., J. Lee, V. Skihar, R. Doucette and M. D. Kawaja (2004). "LacZ-expressing olfactory ensheathing cells do not associate with myelinated axons after implantation into the compressed spinal cord." <u>Proc Natl Acad Sci U S A</u> 101(7): 2162-6.
- Boyd, J. G., V. Skihar, M. Kawaja and R. Doucette (2003). "Olfactory ensheathing cells: historical perspective and therapeutic potential." <u>Anat Rec</u> 271B(1): 49-60.
- Brazelton, T. R., F. M. Rossi, G. I. Keshet and H. M. Blau (2000). "From marrow to brain: expression of neuronal phenotypes in adult mice." <u>Science</u> 290(5497): 1775-9.
- Brehmer, A., F. Schrodl and W. Neuhuber (1999). "Morphological classifications of enteric neurons--100 years after Dogiel." <u>Anat Embryol (Berl)</u> 200(2): 125-35.
- Brodkey, J. A., E. D. Laywell, T. F. O'Brien, A. Faissner, K. Stefansson, H. U. Dorries, M. Schachner and D. A. Steindler (1995). "Focal brain injury and upregulation of a developmentally regulated extracellular matrix protein." <u>J Neurosurg</u> 82(1): 106-12.
- Brook, G. A., J. M. Lawrence and G. Raisman (1993). "Morphology and migration of cultured Schwann cells transplanted into the fimbria and hippocampus in adult rats." <u>Glia</u> 9(4): 292-304.
- Brook, G. A., D. Plate, R. Franzen, D. Martin, G. Moonen, J. Schoenen, A. B. Schmitt, J. Noth and W. Nacimiento (1998). "Spontaneous longitudinally orientated axonal regeneration is associated with the Schwann cell framework within the lesion site following spinal cord compression injury of the rat." J Neurosci Res 53(1): 51-65.
- Brosnan, C. F. and C. S. Raine (1996). "Mechanisms of immune injury in multiple sclerosis." <u>Brain Pathol</u> 6(3): 243-57.
- Brown, A. M. and D. E. McFarlin (1981). "Relapsing experimental allergic encephalomyelitis in the SJL/J mouse." Lab Invest 45(3): 278-84.
- Brunden, K. R. and D. T. Brown (1990). "P0 mRNA expression in cultures of Schwann cells and neurons that lack basal lamina and myelin." <u>J Neurosci Res</u> 27(2): 159-68.
- Brunden, K. R., A. J. Windebank and J. F. Poduslo (1990). "Role of axons in the regulation of P0 biosynthesis by Schwann cells." J Neurosci Res 26(2): 135-43.
- Brunner, C., H. Lassmann, T. V. Waehneldt, J. M. Matthieu and C. Linington (1989).
 "Differential ultrastructural localization of myelin basic protein, myelin/oligodendroglial glycoprotein, and 2',3'-cyclic nucleotide 3'phosphodiesterase in the CNS of adult rats." J Neurochem 52(1): 296-304.
- Brustle, O., U. Maskos and R. D. McKay (1995). "Host-guided migration allows targeted introduction of neurons into the embryonic brain." <u>Neuron</u> 15(6): 1275-85.
- Buiakova, O. I., H. Baker, J. W. Scott, A. Farbman, R. Kream, M. Grillo, L. Franzen, M. Richman, L. M. Davis, S. Abbondanzo, C. L. Stewart and F. L. Margolis
(1996). "Olfactory marker protein (OMP) gene deletion causes altered physiological activity of olfactory sensory neurons." <u>Proc Natl Acad Sci U S A</u> 93(18): 9858-63.

- Bunge, M. B., R. P. Bunge, N. Kleitman and A. C. Dean (1989). "Role of peripheral nerve extracellular matrix in Schwann cell function and in neurite regeneration." <u>Dev Neurosci</u> 11(4-5): 348-60.
- Bunge, M. B., R. P. Bunge and G. D. Pappas (1962). "Electron microscopic demonstration of connections between glia and myelin sheaths in the developing mammalian central nervous system." J Cell Biol 12: 448-53.
- Bunge, M. B., R. P. Bunge and H. Ris (1961). "Ultrastructural study of remyelination in an experimental lesion in adult cat spinal cord." <u>J Biophys Biochem Cytol</u> 10: 67-94.
- Bunge, R. P. (1968). "Glial cells and the central myelin sheath." <u>Physiol Rev</u> 48(1): 197-251.
- Bunge, R. P., M. B. Bunge and C. F. Eldridge (1986). "Linkage between axonal ensheathment and basal lamina production by Schwann cells." <u>Annu Rev</u> <u>Neurosci</u> 9: 305-28.
- Butt, A. M., M. Ibrahim, F. M. Ruge and M. Berry (1995). "Biochemical subtypes of oligodendrocyte in the anterior medullary velum of the rat as revealed by the monoclonal antibody Rip." <u>Glia</u> 14(3): 185-97.
- Butt, A. M., A. Duncan, M. F. Hornby, S. L. Kirvell, A. Hunter, J. M. Levine, M. Berry (1999). "Cells expressing the NG2 antigen contact nodes of Ranvier in adult CNS white matter." Glia 26(1): 84-91.
- Butt, A. M. and B. R. Ransom (1989). "Visualization of oligodendrocytes and astrocytes in the intact rat optic nerve by intracellular injection of lucifer yellow and horseradish peroxidase." <u>Glia</u> 2(6): 470-5.
- Buttini, M., A. Sauter and H. W. Boddeke (1994). "Induction of interleukin-1 beta mRNA after focal cerebral ischaemia in the rat." <u>Brain Res Mol Brain Res</u> 23(1-2): 126-34.
- Byrd, C. A. and G. D. Burd (1991). "Development of the olfactory bulb in the clawed frog, Xenopus laevis: a morphological and quantitative analysis." <u>J Comp</u> <u>Neurol</u> 314(1): 79-90.
- Byrd, C. A. and G. D. Burd (1993). "Morphological and quantitative evaluation of olfactory bulb development in xenopus after olfactory placode transplantation." Journal of Comparative Neurology 331(4): 551-563.
- Caggiano, M., J. S. Kauer and D. D. Hunter (1994). "Globose basal cells are neuronal progenitors in the olfactory epithelium: a lineage analysis using a replication-incompetent retrovirus." <u>Neuron</u> 13(2): 339-52.
- Calof, A. L., A. Bonnin, C. Crocker, S. Kawauchi, R. C. Murray, J. Shou and H. H. Wu (2002). "Progenitor cells of the olfactory receptor neuron lineage." <u>Microsc Res</u> <u>Tech</u> 58(3): 176-88.
- Calof, A. L. and D. M. Chikaraishi (1989). "Analysis of neurogenesis in a mammalian neuroepithelium: proliferation and differentiation of an olfactory neuron precursor in vitro." <u>Neuron</u> 3(1): 115-27.
- Calof, A. L., N. Hagiwara, J. D. Holcomb, J. S. Mumm and J. Shou (1996). "Neurogenesis and cell death in olfactory epithelium." <u>J Neurobiol</u> 30(1): 67-81.

- Calof, A. L., J. S. Mumm, P. C. Rim and J. Shou (1998). "The neuronal stem cell of the olfactory epithelium." J Neurobiol 36(2): 190-205.
- Cammer, W. and H. Zhang (1993). "Atypical localization of the oligodendrocytic isoform (PI) of glutathione-S-transferase in astrocytes during cuprizone intoxication." J Neurosci Res 36(2): 183-90.
- Campenot, R. B. and B. L. MacInnis (2004). "Retrograde transport of neurotrophins: fact and function." J Neurobiol 58(2): 217-29.
- Cancalon, P. (1985). "Influence of temperature on various mechanisms associated with neuronal growth and nerve regeneration." <u>Prog Neurobiol</u> 25(1): 27-92.
- Cannella, B. and C. S. Raine (1995). "The adhesion molecule and cytokine profile of multiple sclerosis lesions." <u>Ann Neurol</u> 37(4): 424-35.
- Cao, L., L. Liu, Z. Y. Chen, L. M. Wang, J. L. Ye, H. Y. Qiu, C. L. Lu and C. He (2004). "Olfactory ensheathing cells genetically modified to secrete GDNF to promote spinal cord repair." <u>Brain</u> 127(Pt 3): 535-49.
- Caplan, A. I. (1991). "Mesenchymal stem cells." J Orthop Res 9(5): 641-50.
- Carenini, S., D. Montag, H. Cremer, M. Schachner and R. Martini (1997). "Absence of the myelin-associated glycoprotein (MAG) and the neural cell adhesion molecule (N-CAM) interferes with the maintenance, but not with the formation of peripheral myelin." <u>Cell Tissue Res</u> 287(1): 3-9.
- Carey, E. M. and N. M. Freeman (1983). "Biochemical changes in Cuprizone-induced spongiform encephalopathy. I. Changes in the activities of 2',3'-cyclic nucleotide 3'-phosphohydrolase, oligodendroglial ceramide galactosyl transferase, and the hydrolysis of the alkenyl group of alkenyl, acyl-glycerophospholipids by plasmalogenase in different regions of the brain." <u>Neurochem Res</u> 8(8): 1029-44.
- Carr, V. M. and A. I. Farbman (1993). "The dynamics of cell death in the olfactory epithelium." <u>Exp Neurol</u> 124(2): 308-14.
- Carrieri, P. B., V. Provitera, T. De Rosa, G. Tartaglia, F. Gorga and O. Perrella (1998).
 "Profile of cerebrospinal fluid and serum cytokines in patients with relapsing-remitting multiple sclerosis: a correlation with clinical activity." Immunopharmacol Immunotoxicol 20(3): 373-82.
- Cattaneo, E. and R. McKay (1990). "Proliferation and differentiation of neuronal stem cells regulated by nerve growth factor." <u>Nature</u> 347(6295): 762-5.
- Chabas, D., S. E. Baranzini, D. Mitchell, C. C. Bernard, S. R. Rittling, D. T. Denhardt, R. A. Sobel, C. Lock, M. Karpuj, R. Pedotti, R. Heller, J. R. Oksenberg and L. Steinman (2001). "The influence of the proinflammatory cytokine, osteopontin, on autoimmune demyelinating disease." <u>Science</u> 294(5547): 1731-5.
- Challoner, P. B., K. T. Smith, J. D. Parker, D. L. MacLeod, S. N. Coulter, T. M. Rose, E. R. Schultz, J. L. Bennett, R. L. Garber, M. Chang and et al. (1995). "Plaqueassociated expression of human herpesvirus 6 in multiple sclerosis." <u>Proc Natl</u> <u>Acad Sci U S A</u> 92(16): 7440-4.
- Chandler, S., R. Coates, A. Gearing, J. Lury, G. Wells and E. Bone (1995). "Matrix metalloproteinases degrade myelin basic protein." <u>Neurosci Lett</u> 201(3): 223-6.
- Chandler, S., K. M. Miller, J. M. Clements, J. Lury, D. Corkill, D. C. Anthony, S. E. Adams and A. J. Gearing (1997). "Matrix metalloproteinases, tumor necrosis factor and multiple sclerosis: an overview." J Neuroimmunol 72(2): 155-61.

- Chang, A., W. W. Tourtellotte, R. Rudick and B. D. Trapp (2002). "Premyelinating oligodendrocytes in chronic lesions of multiple sclerosis." <u>N Engl J Med</u> 346(3): 165-73.
- Chao, C. C., S. Hu, T. W. Molitor, E. G. Shaskan and P. K. Peterson (1992). "Activated microglia mediate neuronal cell injury via a nitric oxide mechanism." <u>J Immunol</u> 149(8): 2736-41.
- Chao, C. C., S. Hu and P. K. Peterson (1995). "Glia, cytokines, and neurotoxicity." <u>Crit</u> <u>Rev Neurobiol</u> 9(2-3): 189-205.
- Chari, D. M., A. J. Crang and W. F. Blakemore (2003). "Decline in rate of colonization of oligodendrocyte progenitor cell (OPC)-depleted tissue by adult OPCs with age." <u>J Neuropathol Exp Neurol</u> 62(9): 908-16.
- Chauvet, N., M. Prieto and G. Alonso (1998). "Tanycytes present in the adult rat mediobasal hypothalamus support the regeneration of monoaminergic axons." <u>Exp Neurol</u> 151(1): 1-13.
- Chen, Z. L. and S. Strickland (1997). "Neuronal death in the hippocampus is promoted by plasmin-catalyzed degradation of laminin." <u>Cell</u> 91(7): 917-25.
- Chen, L. M., D. Bailey and C. Fernandez-Valle (2000). "Association of beta 1 integrin with focal adhesion kinase and paxillin in differentiating Schwann cells." J <u>Neurosci</u> 20(10): 3776-84.
- Chiang, C. S., H. C. Powell, L. H. Gold, A. Samimi and I. L. Campbell (1996).
 "Macrophage/microglial-mediated primary demyelination and motor disease induced by the central nervous system production of interleukin-3 in transgenic mice." J Clin Invest 97(6): 1512-24.
- Choi, D. W. (1993). "NMDA receptors and AMPA/kainate receptors mediate parallel injury in cerebral cortical cultures subjected to oxygen-glucose deprivation." <u>Prog Brain Res</u> 96: 137-43.
- Chuah, M. I. and C. Au (1991). "Olfactory Schwann cells are derived from precursor cells in the olfactory epithelium." <u>J Neurosci Res</u> 29(2): 172-80.
- Chuah, M. I. and C. Au (1993). "Cultures of ensheathing cells from neonatal rat olfactory bulbs." <u>Brain Res</u> 601(1-2): 213-20.
- Chuah, M. I. and C. Au (1994). "Olfactory cell cultures on ensheathing cell monolayers." <u>Chem Senses</u> 19(1): 25-34.
- Chuah, M. I. and R. Teague (1999). "Basic fibroblast growth factor in the primary olfactory pathway: mitogenic effect on ensheathing cells." <u>Neuroscience</u> 88(4): 1043-50.
- Coffey, J. C. and K. W. McDermott (1997). "The regional distribution of myelin oligodendrocyte glycoprotein (MOG) in the developing rat CNS: an in vivo immunohistochemical study." J Neurocytol 26(3): 149-61.
- Colton, C. A., J. E. Keri, W. T. Chen and W. L. Monsky (1993). 'Protease production by cultured microglia: substrate gel analysis and immobilized matrix degradation." <u>J Neurosci Res</u> 35(3): 297-304.
- Compston, A., J. Zajicek, J. Sussman, A. Webb, G. Hall, D. Muir, C. Shaw, A. Wood and N. Scolding (1997). "Glial lineages and myelination in the central nervous system." <u>J Anat</u> 190 (Pt 2): 161-200.
- Compston, D. A., B. P. Morgan, A. K. Campbell, P. Wilkins, G. Cole, N. D. Thomas and B. Jasani (1989). "Immunocytochemical localization of the terminal

complement complex in multiple sclerosis." <u>Neuropathol Appl Neurobiol</u> 15(4): 307-16.

- Conlon, P., J. R. Oksenberg, J. Zhang and L. Steinman (1999). "The immunobiology of multiple sclerosis: an autoimmune disease of the central nervous system." <u>Neurobiol Dis</u> 6(3): 149-66.
- Costanzo, R. M. (1985). "Neural regeneration and functional reconnection following olfactory nerve transection in hamster." <u>Brain Res</u> 361(1-2): 258-66.
- Costanzo, R. M. and P. P. Graziadei (1983). "A quantitative analysis of changes in the olfactory epithelium following bulbectomy in hamster." <u>J Comp Neurol</u> 215(4): 370-81.
- Cot, C., A. Privat and C. Levallois (1998). "Adenoviral-mediated transfection of the Lac-Z gene into rat dissociated embryonic central nervous system cells before and after seeding." Int J Dev Neurosci 16(1): 9-17.
- Cross, A. K. and M. N. Woodroofe (1999). "Chemokines induce migration and changes in actin polymerization in adult rat brain microglia and a human fetal microglial cell line in vitro." J Neurosci Res 55(1): 17-23.
- Cummings, D. M. and P. C. Brunjes (1995). "Migrating luteinizing hormone-releasing hormone (LHRH) neurons and processes are associated with a substrate that expresses S100." <u>Brain Res Dev Brain Res</u> 88(2): 148-57.
- Cuschieri, A. and L. H. Bannister (1975). "The development of the olfactory mucosa in the mouse: light microscopy." J Anat 119(2): 277-86.
- Cuzner, M. L., A. J. Loughlin, K. Mosley and M. N. Woodroofe (1994). "The role of microglia macrophages in the processes of inflammatory demyelination and remyelination." <u>Neuropathol Appl Neurobiol</u> 20(2): 200-1.
- D'Ambrosi, N., B. Murra, F. Vacca and C. Volonte (2004). "Pathways of survival induced by NGF and extracellular ATP after growth factor deprivation." <u>Prog</u> <u>Brain Res</u> 146: 93-100.
- D'Urso, D., P. J. Brophy, S. M. Staugaitis, C. S. Gillespie, A. B. Frey, J. G. Stempak and D. R. Colman (1990). "Protein zero of peripheral nerve myelin: biosynthesis, membrane insertion, and evidence for homotypic interaction." <u>Neuron</u> 4(3): 449-60.
- D'Urso, D., P. Ehrhardt and H. W. Muller (1999). "Peripheral myelin protein 22 and protein zero: a novel association in peripheral nervous system myelin." J Neurosci 19(9): 3396-403.
- Dal Canto, M. C., B. S. Kim, S. D. Miller and R. W. Melvold (1996). "Theiler's Murine Encephalomyelitis Virus (TMEV)-Induced Demyelination: A Model for Human Multiple Sclerosis." Methods 10(3): 453-61.
- Daubas, P., D. Pham-Dinh and A. Dautigny (1994). "Structure and polymorphism of the mouse myelin/oligodendrocyte glycoprotein gene." <u>Genomics</u> 23(1): 36-41.
- Dautigny, A., M. G. Mattei, D. Morello, P. M. Alliel, D. Pham-Dinh, L. Amar, D. Arnaud, D. Simon, J. F. Mattei, J. L. Guenet and et al. (1986). "The structural gene coding for myelin-associated proteolipid protein is mutated in jimpy mice." <u>Nature</u> 321(6073): 867-9.
- David, S., C. Bouchard, O. Tsatas and N. Giftochristos (1990). "Macrophages can modify the nonpermissive nature of the adult mammalian central nervous system." <u>Neuron</u> 5(4): 463-9.

- Davie, C. A., G. J. Barker, S. Webb, P. S. Tofts, A. J. Thompson, A. E. Harding, W. I. McDonald and D. H. Miller (1995). "Persistent functional deficit in multiple sclerosis and autosomal dominant cerebellar ataxia is associated with axon loss." <u>Brain</u> 118 (Pt 6): 1583-92.
- De la Calle, J. L., M. A. Mena, J. R. Gonzalez-Escalada and C. L. Paino (2002). "Intrathecal transplantation of neuroblastoma cells decreases heat hyperalgesia and cold allodynia in a rat model of neuropathic pain." <u>Brain Res Bull</u> 59(3): 205-11.
- De la Calle, J. L. and C. L. Paino (2002). "A procedure for direct lumbar puncture in rats." <u>Brain Res Bull</u> 59(3): 245-50.
- De Lorenzo, A. J. (1957). "Electron microscopic observations of the olfactory mucosa and olfactory nerve." J Biophys Biochem Cytol 3(6): 839-50.
- De Stefano, N., P. M. Matthews, L. Fu, S. Narayanan, J. Stanley, G. S. Francis, J. P. Antel and D. L. Arnold (1998). "Axonal damage correlates with disability in patients with relapsing-remitting multiple sclerosis. Results of a longitudinal magnetic resonance spectroscopy study." <u>Brain</u> 121 (Pt 8): 1469-77.
- Deckert-Schluter, M., D. Schluter, H. Hof and O. D. Wiestler (1995). "Rapid progress in the field of infectious and autoimmune disorders of the nervous system." <u>Neuropathol Appl Neurobiol</u> 21(6): 548-51.
- Derivot, J. H. (1984). "Functional anatomy of the peripheral olfactory system of the African lungfish Protopterus annectens Owen: macroscopic, microscopic, and morphometric aspects." <u>Am J Anat</u> 169(2): 177-92.
- Devon, R. and R. Doucette (1992). "Olfactory ensheathing cells myelinate dorsal root ganglion neurites." <u>Brain Res</u> 589(1): 175-9.
- Devon, R. and R. Doucette (1995). "Olfactory ensheathing cells do not require Lascorbic acid in vitro to assemble a basal lamina or to myelinate dorsal root ganglion neurites." <u>Brain Res</u> 688(1-2): 223-9.
- Dezawa, M., K. Kawana, H. Negishi and E. Adachi-Usami (1999). "Glial cells in degenerating and regenerating optic nerve of the adult rat." <u>Brain Res Bull</u> 48(6): 573-9.
- Dickson, D. W., S. C. Lee, L. A. Mattiace, S. H. Yen and C. Brosnan (1993). "Microglia and cytokines in neurological disease, with special reference to AIDS and Alzheimer's disease." <u>Glia</u> 7(1): 75-83.
- Diemel, L. T., C. A. Copelman and M. L. Cuzner (1998). "Macrophages in CNS remyelination: friend or foe?" <u>Neurochem Res</u> 23(3): 341-7.
- Ding, Y. and K. R. Brunden (1994). "The cytoplasmic domain of myelin glycoprotein P0 interacts with negatively charged phospholipid bilayers." J Biol Chem 269(14): 10764-70.
- DiStefano, P. S. and E. M. Johnson, Jr. (1988). "Nerve growth factor receptors on cultured rat Schwann cells." J Neurosci 8(1): 231-41.
- Doble, A. (1999). "The role of excitotoxicity in neurodegenerative disease: implications for therapy." <u>Pharmacol Ther</u> 81(3): 163-221.
- Domaradzka-Pytel, B., B. Ludkiewicz, J. Morys and H. M. Wisniewski (1999). "Expression and distribution of various antigens of developing microglial cells in the rat telencephalon." J Hirnforsch 39(3): 283-91.

- Doucette J. R., J. A. Kiernan and B. A. Flumerfelt. (1983). "The re-innervation of olfactory glomeruli following transection of primary olfactory axons in the central or peripheral nervous system." J Anat.137 (Pt 1): 1-19.
- Doucette, R. (1984). "The glial cells in the nerve fiber layer of the rat olfactory bulb." <u>Anatomical Record</u> 210: 385-391.
- Doucette, R. (1986). Astrocytes in the olfactory bulb. <u>Astrocytes</u>. S. Fedoroff and A. Vernadakis. New York, Academic Press. 1: 293-310.
- Doucette, R. (1989). "Development of the nerve fiber layer in the olfactory bulb of mouse embryos." Journal of Comparative Neurology 285: 514-527.
- Doucette, R. (1990). "Glial influences on axonal growth in the primary olfactory system." <u>Glia</u> 3: 433-449.
- Doucette, R. (1991). "The PNS-CNS transitional zone of the first cranial nerve." Journal of Comparative Neurology 312: 451-466.
- Doucette, R. (1993). "Glial cells in the nerve fiber layer of the main olfactory bulb of embryonic and adult mammals." <u>Microscopy Research and Technique</u> 24: 113-130.
- Doucette, R. (1995). "Olfactory ensheathing cells: potential for glial cell transplantation into areas of CNS injury." <u>Histol Histopathol</u> 10(2): 503-7.
- Doucette, R. (1996). "Immunohistochemical localization of laminin, fibronectin and collagen type IV in the nerve fiber layer of the olfactory bulb." <u>Int J Dev</u> <u>Neurosci</u> 14(7-8): 945-59.
- Doucette R. (2001). "Modulation of axonal growth by glia cells. Inflammatory events in neurodegeneration." Bondy SC and Campbell A. Prominent Press pp1-10.
- Doucette, R. and R. Devon (1994). "Media that support the growth and differentiation of oligodendrocytes do not induce olfactory ensheathing cells to express a myelinating phenotype." <u>Glia</u> 10(4): 296-310.
- Doucette, R. and R. Devon (1995). "Elevated intracellular levels of cAMP induce olfactory ensheathing cells to express GAL-C and GFAP but not MBP." <u>Glia</u> 13(2): 130-40.
- Dowling, P., G. Shang, S. Raval, J. Menonna, S. Cook and W. Husar (1996).
 "Involvement of the CD95 (APO-1/Fas) receptor/ligand system in multiple sclerosis brain." J Exp Med 184(4): 1513-8.
- Drescher, K. M., L. R. Pease and M. Rodriguez (1997). "Antiviral immune responses modulate the nature of central nervous system (CNS) disease in a murine model of multiple sclerosis." <u>Immunol Rev</u> 159: 177-93.
- Duncan, I. D., J. P. Hammang and S. A. Gilmore (1988). "Schwann cell myelination of the myelin deficient rat spinal cord following X-irradiation." <u>Glia</u> 1(3): 233-9.
- Duncan, I. D., J. P. Hammang, S. Goda and R. H. Quarles (1989). "Myelination in the jimpy mouse in the absence of proteolipid protein." <u>Glia</u> 2(3): 148-54.
- Duncan I. D., J. P. Hammang, K. F. Jackson, P. M. Wood, R. P. Bunge, L. Langford (1988). "Transplantation of oligodendrocytes and Schwann cells into the spinal cord of the myelin-deficient rat." <u>J Neurocytol</u> 17(3): 351-360.
- Duncan, I. D., J. P. Hammang and B. D. Trapp (1987). "Abnormal compact myelin in the myelin-deficient rat: absence of proteolipid protein correlates with a defect in the intraperiod line." <u>Proc Natl Acad Sci U S A</u> 84(17): 6287-91.

- Duncan, I. D. and E. A. Milward (1995). "Glial cell transplants: experimental therapies of myelin diseases." <u>Brain Pathol</u> 5(3): 301-10.
- Duncan, I. D., C. Paino, D. R. Archer and P. M. Wood (1992). "Functional capacities of transplanted cell-sorted adult oligodendrocytes." <u>Dev Neurosci</u> 14(2): 114-22.
- Dutta, A. S., J. J. Gormley, M. Coath, L. Hassall, C. F. Hayward, P. R. Gellert, R. S. Kittlety, P. J. Alcock, R. Ferguson, T. Halterman, A. Jamieson, J. A. Moors, J. M. Moores, A. Rees, L. J. Wood, C. F. Reilly and D. Haworth (2000). "Potent cyclic peptide inhibitors of VLA-4 (alpha4beta1 integrin)-mediated cell adhesion. Discovery of compounds like cyclo(MePhe-Leu-Asp-Val-D-Arg-D-Arg) (ZD7349) compatible with depot formulation." J Pept Sci 6(8): 398-412.
- Ebers, G. C., A. D. Sadovnick and N. J. Risch (1995). "A genetic basis for familial aggregation in multiple sclerosis. Canadian Collaborative Study Group." <u>Nature</u> 377(6545): 150-1.
- Eddleston, M. and L. Mucke (1993). "Molecular profile of reactive astrocytes-implications for their role in neurologic disease." <u>Neuroscience</u> 54(1): 15-36.
- Eglitis, M. A., D. Dawson, K. W. Park and M. M. Mouradian (1999). "Targeting of marrow-derived astrocytes to the ischemic brain." <u>Neuroreport</u> 10(6): 1289-92.
- Eglitis, M. A. and E. Mezey (1997). "Hematopoietic cells differentiate into both microglia and macroglia in the brains of adult mice." <u>Proc Natl Acad Sci U S A</u> 94(8): 4080-5.
- Elkabes, S., E. M. DiCicco-Bloom and I. B. Black (1996). "Brain microglia/macrophages express neurotrophins that selectively regulate microglial proliferation and function." J Neurosci 16(8): 2508-21.
- Emerson, M. R., S. Biswas and S. M. LeVine (2001). "Cuprizone and piperonyl butoxide, proposed inhibitors of T-cell function, attenuate experimental allergic encephalomyelitis in SJL mice." J Neuroimmunol 119(2): 205-13.
- Evangelou, N., M. Jackson, D. Beeson and J. Palace (1999). "Association of the APOE epsilon4 allele with disease activity in multiple sclerosis." <u>J Neurol Neurosurg</u> <u>Psychiatry</u> 67(2): 203-5.
- Fagan, A. M. and F. H. Gage (1990). "Cholinergic sprouting in the hippocampus: a proposed role for IL-1." <u>Exp Neurol</u> 110(1): 105-20.
- Fallis, R. J., C. S. Raine and D. E. McFarlin (1989). "Chronic relapsing experimental allergic encephalomyelitis in SJL mice following the adoptive transfer of an epitope-specific T cell line." J Neuroimmunol 22(2): 93-105.
- Farbman, A. I. and F. L. Margolis (1980). "Olfactory marker protein during ontogeny: immunohistochemical localization." <u>Dev Biol</u> 74(1): 205-15.
- Farbman, A. I. and L. M. Squinto (1985). "Early development of olfactory receptor cell axons." <u>Brain Res</u> 351(2): 205-13.
- Fawcett, J. W. (1998). "Spinal cord repair: from experimental models to human application." <u>Spinal Cord</u> 36(12): 811-7.
- Fawcett, J. W. and R. A. Asher (1999). "The glial scar and central nervous system repair." <u>Brain Res Bull</u> 49(6): 377-91.
- Fawcett, J. W., E. Housden, L. Smith-Thomas and R. L. Meyer (1989). "The growth of axons in three-dimensional astrocyte cultures." <u>Dev Biol</u> 135(2): 449-58.
- Fawcett, J. W. and R. J. Keynes (1990). "Peripheral nerve regeneration." <u>Annu Rev</u> <u>Neurosci</u> 13: 43-60.

- Fazakerley, J. K. and M. J. Buchmeier (1993). "Pathogenesis of virus-induced demyelination." Adv Virus Res 42: 249-324.
- Felts P. A and K. J. Smith (1996) "Blood-brain barrier permeability in astrocyte-free regions of the central nervous system remyelinated by Schwann cells." <u>Neuroscience</u> 75(2): 643-55.
- Felts P. A., T. A. Baker and K. J. Smith (1997). "Conduction in segmentally demyelinated mammalian central axons." J Neurosci 17(18): 7267-7277.
- Ferguson, B., M. K. Matyszak, M. M. Esiri and V. H. Perry (1997). "Axonal damage in acute multiple sclerosis lesions." <u>Brain</u> 120 (Pt 3): 393-9.
- Fernandez-Valle, C., D. Gorman, A. M. Gomez and M. B. Bunge (1997). "Actin plays a role in both changes in cell shape and gene-expression associated with Schwann cell myelination." J Neurosci 17(1): 241-50.
- Fernandez-Valle, C., L. Gwynn, P. M. Wood, S. Carbonetto and M. B. Bunge (1994). "Anti-beta 1 integrin antibody inhibits Schwann cell myelination." <u>J Neurobiol</u> 25(10): 1207-26.
- Feron, F., C. Perry, J. J. McGrath and A. Mackay-Sim (1998). "New techniques for biopsy and culture of human olfactory epithelial neurons." <u>Arch Otolaryngol</u> <u>Head Neck Surg</u> 124(8): 861-6.
- Field, P., Y. Li and G. Raisman (2003). "Ensheathment of the olfactory nerves in the adult rat." J Neurocytol 32(3): 317-24.
- Fields, R. D., J. M. Le Beau, F. M. Longo and M. H. Ellisman (1989). "Nerve regeneration through artificial tubular implants." Prog Neurobiol 33(2): 87-134.
- Filbin, M. T., F. S. Walsh, B. D. Trapp, J. A. Pizzey and G. I. Tennekoon (1990). "Role of myelin P0 protein as a homophilic adhesion molecule." <u>Nature</u> 344(6269): 871-2.
- Filipovic, R., I. Jakovcevski, N. Zecevic (2003). "GRO-alpha and CXCR2 in the human fetal brain and multiple sclerosis lesions." <u>Dev Neurosci</u> 25(2-4)" 279-90.
- Folch, J. and M. Lees (1951). "Proteolipides, a new type of tissue lipoproteins; their isolation from brain." J Biol Chem 191(2): 807-17.
- Foran, D. R. and A. C. Peterson (1992). "Myelin acquisition in the central nervous system of the mouse revealed by an MBP-Lac Z transgene." J Neurosci 12(12): 4890-7.
- Fraher, J. P. (1999). "The transitional zone and CNS regeneration." J Anat 194(Pt 2): 161-82.
- Frail, D. E., H. D. Webster and P. E. Braun (1985). "Developmental expression of the myelin-associated glycoprotein in the peripheral nervous system is different from that in the central nervous system." J Neurochem 45(4): 1308-10.
- Franceschini, I. A. and S. C. Barnett (1996). "Low-affinity NGF-receptor and E-N-CAM expression define two types of olfactory nerve ensheathing cells that share a common lineage." <u>Dev Biol</u> 173(1): 327-43.
- Franklin, R. J. (2002). "Obtaining olfactory ensheathing cells from extra-cranial sources a step closer to clinical transplant-mediated repair of the CNS?" <u>Brain</u> 125(Pt 1): 2-3.
- Franklin, R. J. (2002). "Remyelination of the demyelinated CNS: the case for and against transplantation of central, peripheral and olfactory glia." <u>Brain Res Bull</u> 57(6): 827-32.

- Franklin, R. J. and S. C. Barnett (1997). "Do olfactory glia have advantages over Schwann cells for CNS repair?" J Neurosci Res 50(5): 665-72.
- Franklin, R. J. and S. C. Barnett (2000). "Olfactory ensheathing cells and CNS regeneration: the sweet smell of success?" <u>Neuron</u> 28(1): 15-8.
- Franklin, R. J. and W. F. Blakemore (1993). "Requirements for Schwann cell migration within CNS environments: a viewpoint." Int J Dev Neurosci 11(5): 641-9.
- Franklin R. J. and W. F. Blakemore (1997). "Transplanting oligodendrocyte progenitors into the adult CNS." J Anat 190(Pt1):23-33.
- Franklin, R. J., A. J. Crang and W. F. Blakemore (1992). "Type 1 astrocytes fail to inhibit Schwann cell remyelination of CNS axons in the absence of cells of the O-2A lineage." <u>Dev Neurosci</u> 14(2): 85-92.
- Franklin, R. J., J. M. Gilson, I. A. Franceschini and S. C. Barnett (1996). "Schwann cell-like myelination following transplantation of an olfactory bulb-ensheathing cell line into areas of demyelination in the adult CNS." <u>Glia</u> 17(3): 217-24.
- Franklin, R. J., J. M. Gilson and W. F. Blakemore (1997). "Local recruitment of remyelinating cells in the repair of demyelination in the central nervous system." <u>J Neurosci Res</u> 50(2): 337-44.
- Franklin, R. J., C. Zhao and F. J. Sim (2002). "Ageing and CNS remyelination." <u>Neuroreport</u> 13(7): 923-8.
- Fredman, P., J. L. Magnani, M. Nirenberg and V. Ginsburg (1984). "Monoclonal antibody A2B5 reacts with many gangliosides in neuronal tissue." <u>Arch</u> <u>Biochem Biophys</u> 233(2): 661-6.
- Friedman, B., S. Hockfield, J. A. Black, K. A. Woodruff and S. G. Waxman (1989). "In situ demonstration of mature oligodendrocytes and their processes: an immunocytochemical study with a new monoclonal antibody, rip." <u>Glia</u> 2(5): 380-90.
- Friedman, J. E., M. J. Lyons, G. Cu, D. V. Ablashl, J. E. Whitman, M. Edgar, M. Koskiniemi, A. Vaheri and J. B. Zabriskie (1999). "The association of the human herpesvirus-6 and MS." <u>Mult Scler</u> 5(5): 355-62.
- Frisch, D. (1967). "Ultrastructure of mouse olfactory mucosa." <u>Am J Anat</u> 121(1): 87-120.
- Fruttiger, M., L. Karlsson, A. C. Hall, A. Abramsson, A. R. Calver, H. Bostrom, K.
 Willetts, C. H. Bertold, J. K. Heath, C. Betsholtz and W. D. Richardson (1999).
 "Defective oligodendrocyte development and severe hypomyelination in PDGF-A knockout mice." <u>Development</u> 126(3): 457-67.
- Fruttiger, M., D. Montag, M. Schachner and R. Martini (1995). "Crucial role for the myelin-associated glycoprotein in the maintenance of axon-myelin integrity." <u>Eur J Neurosci</u> 7(3): 511-5.
- Fu, S. Y. and T. Gordon (1997). "The cellular and molecular basis of peripheral nerve regeneration." <u>Mol Neurobiol</u> 14(1-2): 67-116.
- Fujita, N., A. Kemper, J. Dupree, H. Nakayasu, U. Bartsch, M. Schachner, N. Maeda, K. Suzuki and B. Popko (1998). "The cytoplasmic domain of the large myelinassociated glycoprotein isoform is needed for proper CNS but not peripheral nervous system myelination." <u>J Neurosci</u> 18(6): 1970-8.
- Gage, F. H., J. Ray and L. J. Fisher (1995). "Isolation, characterization, and use of stem cells from the CNS." <u>Annu Rev Neurosci</u> 18: 159-92.

- Gallo, V. and R. C. Armstrong (1995). "Developmental and growth factor-induced regulation of nestin in oligodendrocyte lineage cells." J Neurosci 15(1 Pt 1): 394-406.
- Gansmuller, A., E. Clerin, F. Kruger, M. Gumpel and F. Lachapelle (1991). "Tracing transplanted oligodendrocytes during migration and maturation in the shiverer mouse brain." <u>Glia</u> 4(6): 580-590.
- Gansmuller, A., F. Kruger, M. Gumpel and A. Baron-Van Evercooren (1992).
 "Photoconverted carbocyanine DiI allows direct visualization of transplanted glial cells at the ultrastructural level." <u>Neurosci Lett</u> 147(2): 151-4.
- Gao, F. B., J. Apperly and M. Raff (1998). "Cell-intrinsic timers and thyroid hormone regulate the probability of cell-cycle withdrawal and differentiation of oligodendrocyte precursor cells." Dev Biol 197(1): 54-66.
- Gebicke-Haerter, P. J., D. Van Calker, W. Norenberg and P. Illes (1996). "Molecular mechanisms of microglial activation. A. Implications for regeneration and neurodegenerative diseases." <u>Neurochem Int</u> 29(1):1-12.
- Gehrmann, J., Y. Matsumoto and G.W. Kreutzberg (1995). "Microglia: intrinsic immuneffector cell of the brain." <u>Brain Res Rev</u> 20:269–287.
- Gehrmann, J., G. Mies, P. Bonnekoh, R. Banati, T. Iijima, G. W. Kreutzberg and K. A. Hossmann (1993). "Microglial reaction in the rat cerebral cortex induced by cortical spreading depression." <u>Brain Pathol</u> 3(1): 11-7.
- Gehrmann, J., S. W. Schoen and G. W. Kreutzberg (1991). "Lesion of the rat entorhinal cortex leads to a rapid microglial reaction in the dentate gyrus. A light and electron microscopical study." <u>Acta Neuropathol (Berl)</u> 82(6): 442-55.
- Gensburger, C., G. Labourdette and M. Sensenbrenner (1987). "Brain basic fibroblast growth factor stimulates the proliferation of rat neuronal precursor cells in vitro." <u>FEBS Lett</u> 217(1): 1-5.
- Getchell, M. L., J. A. Rafols and T. V. Getchell (1984). "Histological and histochemical studies of the secretory components of the salamander olfactory mucosa: effects of isoproterenol and olfactory nerve section." <u>Anat Rec</u> 208(4): 553-65.
- Ghatak, N. R., A. Hirano, Y. Doron and H. M. Zimmerman (1973). "Remyelination in multiple sclerosis with peripheral type myelin." <u>Arch Neurol</u> 29(4): 262-7.
- Giese, K. P., R. Martini, G. Lemke, P. Soriano and M. Schachner (1992). "Mouse P0 gene disruption leads to hypomyelination, abnormal expression of recognition molecules, and degeneration of myelin and axons." <u>Cell</u> 71(4): 565-76.
- Giftochristos, N. and S. David (1988). "Laminin and heparan sulphate proteoglycan in the lesioned adult mammalian central nervous system and their possible relationship to axonal sprouting." <u>J Neurocytol</u> 17(3): 385-97.
- Gijbels, K., R. E. Galardy and L. Steinman (1994). "Reversal of experimental autoimmune encephalomyelitis with a hydroxamate inhibitor of matrix metalloproteases." J Clin Invest 94(6): 2177-82.
- Gijbels, K., P. Proost, S. Masure, H. Carton, A. Billiau and G. Opdenakker (1993).
 "Gelatinase B is present in the cerebrospinal fluid during experimental autoimmune encephalomyelitis and cleaves myelin basic protein." J Neurosci <u>Res</u> 36(4): 432-40.
- Gilden, D. H. (1999). "Chlamydia: a role for multiple sclerosis or more confusion?" <u>Ann Neurol</u> 46(1): 4-5.

- Gilson, J. M. and W. F. Blakemore (2002). "Schwann cell remyelination is not replaced by oligodendrocyte remyelination following ethidium bromide induced demyelination." <u>Neuroreport</u> 13(9): 1205-8.
- Giulian, D., M. Corpuz, S. Chapman, M. Mansouri, and C. Robertson (1993). "Reactive mononuclear phagocytes release neurotoxins after ischemic and traumatic injury to the central nervous system." J Neurosci Res 36: 681-693.
- Giulian, D. and L. B. Lachman (1985). "Interleukin-1 stimulation of astroglial proliferation after brain injury." <u>Science</u> 228(4698): 497-9.
- Gomez, V. M., S. Averill, V. King, Q. Yang, E. D. Perez, S. C. Chacon, R. Ward, M. Nieto-Sampedro, J. Priestley and J. Taylor (2003). "Transplantation of olfactory ensheathing cells fails to promote significant axonal regeneration from dorsal roots into the rat cervical cord." J Neurocytol 32(1): 53-70.
- Gong, Q., M. S. Bailey, S. K. Pixley, M. Ennis, W. Liu and M. T. Shipley (1994).
 "Localization and regulation of low affinity nerve growth factor receptor expression in the rat olfactory system during development and regeneration." J <u>Comp Neurol</u> 344(3): 336-48.
- Gong, Q., W. L. Liu, M. Srodon, T. D. Foster and M. T. Shipley (1996). "Olfactory epithelial organotypic slice cultures: a useful tool for investigating olfactory neural development." <u>Int J Dev Neurosci</u> 14(7-8): 841-52.
- Goodman, A. D., D. J. Mock, J. M. Powers, J. V. Baker and B. M. Blumberg (2003). "Human herpesvirus 6 genome and antigen in acute multiple sclerosis lesions." J Infect Dis 187(9): 1365-76.
- Goodman, M. N., J. Silver and J. W. Jacobberger (1993). "Establishment and neurite outgrowth properties of neonatal and adult rat olfactory bulb glial cell lines." <u>Brain Res</u> 619(1-2): 199-213.
- Gordon S. (1995). "The macrophage." Bioessays 17: 977-986.
- Gottschall, P. E., X. Yu and B. Bing (1995). "Increased production of gelatinase B (matrix metalloproteinase-9) and interleukin-6 by activated rat microglia in culture." J Neurosci Res 42(3): 335-342.
- Gould, R. M. (1990). "Metabolic organization of the myelinating Schwann cell." <u>Ann N</u> <u>Y Acad Sci</u> 605: 44-54.
- Gow, A., V. L. Friedrich, Jr. and R. A. Lazzarini (1992). "Myelin basic protein gene contains separate enhancers for oligodendrocyte and Schwann cell expression." J <u>Cell Biol</u> 119(3): 605-16.
- Graca, D. L. and W. F. Blakemore (1986). "Delayed remyelination in rat spinal cord following ethidium bromide injection." <u>Neuropathol Appl Neurobiol</u> 12(6): 593-605.
- Graeber, M. B. and G. W. Kreutzberg (1988). "Delayed astrocyte reaction following facial nerve axotomy." J Neurocytol 17(2): 209-20.
- Graeber, M. B., W. J. Streit and G. W. Kreutzberg (1988). "Axotomy of the rat facial nerve leads to increased CR3 complement receptor expression by activated microglial cells." J Neurosci Res 21(1): 18-24.
- Gravel, M., J. Peterson, V. W. Yong, V. Kottis, B. Trapp and P. E. Braun (1996).
 "Overexpression of 2',3'-cyclic nucleotide 3'-phosphodiesterase in transgenic mice alters oligodendrocyte development and produces aberrant myelination." <u>Mol Cell Neurosci</u> 7(6): 453-66.

- Graziadei, G. A., R. S. Stanley and P. P. Graziadei (1980a). "The olfactory marker protein in the olfactory system of the mouse during development." <u>Neuroscience</u> 5(7): 1239-52.
- Graziadei, P. P. (1973). "Cell dynamics in the olfactory mucosa." <u>Tissue Cell</u> 5(1): 113-31.
- Graziadei, P. P., M. S. Karlan, G. A. Graziadei and J. J. Bernstein (1980b).
 "Neurogenesis of sensory neurons in the primate olfactory system after section of the fila olfactoria." <u>Brain Res</u> 186(2): 289-300.
- Graziadei, P. P. and J. F. Metcalf (1971). "Autoradiographic and ultrastructural observations on the frog's olfactory mucosa." <u>Z Zellforsch Mikrosk Anat</u> 116(3): 305-18.
- Graziadei, P. P. and A. G. Monti Graziadei (1983). "Regeneration in the olfactory system of vertebrates." <u>Am J Otolaryngol</u> 4(4): 228-33.
- Graziadei, P. P. C. and G. A. Monti Graziadei (1978). The olfactory system: A model for the study of neurogenesis and axon regeneration in mammals. <u>Neuronal</u> <u>Plasticity</u>. C. W. Cotman. New York, Raven Press: 131-153.
- Graziadei, P. P. C. and G. A. Monti Graziadei (1979). "Neurogenesis and neuron regeneration in the olfactory system of mammals. 1. Morphological aspects of differentiation and structural organization of the olfactory sensory neurons." Journal of Neurocytology 8: 1-18.
- Griffin, D. E., B. Levine, W. R. Tyor and D. N. Irani (1992). "The immune response in viral encephalitis." <u>Semin Immunol</u> 4(2): 111-9.
- Griffiths, I., M. Klugmann, T. Anderson, D. Yool, C. Thomson, M. H. Schwab, A.
 Schneider, F. Zimmermann, M. McCulloch, N. Nadon and K. A. Nave (1998).
 "Axonal swellings and degeneration in mice lacking the major proteolipid of myelin." <u>Science</u> 280(5369): 1610-3.
- Grinspan, J. B. and B. Franceschini (1995). "Platelet-derived growth factor is a survival factor for PSA-NCAM+ oligodendrocyte pre-progenitor cells." <u>J Neurosci Res</u> 41(4): 540-51.
- Grinspan, J. B., M. A. Marchionni, M. Reeves, M. Coulaloglou and S. S. Scherer (1996). "Axonal interactions regulate Schwann cell apoptosis in developing peripheral nerve: neuregulin receptors and the role of neuregulins." <u>J Neurosci</u> 16(19): 6107-18.
- Groves, A. K., S. C. Barnett, R. J. Franklin, A. J. Crang, M. Mayer, W. F. Blakemore and M. Noble (1993). "Repair of demyelinated lesions by transplantation of purified O-2A progenitor cells." <u>Nature</u> 362(6419): 453-5.
- Gudino-Cabrera, G. and M. Nieto-Sampedro (1999). "Estrogen receptor immunoreactivity in Schwann-like brain macroglia." J Neurobiol 40(4): 458-70.
- Gudino-Cabrera, G. and M. Nieto-Sampedro (2000). "Schwann-like macroglia in adult rat brain." <u>Glia</u> 30(1): 49-63.
- Gudino-Cabrera, G., A. M. Pastor, R. R. de la Cruz, J. M. Delgado-Garcia and M. Nieto-Sampedro (2000). "Limits to the capacity of transplants of olfactory glia to promote axonal regrowth in the CNS." <u>Neuroreport</u> 11(3): 467-71.
- Guenard, V., L. A. Gwynn and P. M. Wood (1994). "Astrocytes inhibit Schwann cell proliferation and myelination of dorsal root ganglion neurons in vitro." J Neurosci 14(5 Pt 2): 2980-92.

- Guntinas-Lichius, O., D. N. Angelov, T. L. Tomov, J. Dramiga, W. F. Neiss and K. Wewetzer (2001). "Transplantation of olfactory ensheathing cells stimulates the collateral sprouting from axotomized adult rat facial motoneurons." <u>Exp Neurol</u> 172(1): 70-80.
- Guntinas-Lichius, O., F. Martinez-Portillo, J. Lebek, D. N. Angelov, E. Stennert and W. F. Neiss (1997). "Nimodipine maintains in vivo the increase in GFAP and enhances the astroglial ensheathment of surviving motoneurons in the rat following permanent target deprivation." J Neurocytol 26(4): 241-8.
- Guntinas-Lichius, O., K. Wewetzer, T. L. Tomov, N. Azzolin, S. Kazemi, M. Streppel,
 W. F. Neiss and D. N. Angelov (2002). "Transplantation of olfactory mucosa minimizes axonal branching and promotes the recovery of vibrissae motor performance after facial nerve repair in rats." J Neurosci 22(16): 7121-31.
- Gutowski, N. J., J. Newcombe and M. L. Cuzner (1999). "Tenascin-R and C in multiple sclerosis lesions: relevance to extracellular matrix remodelling." <u>Neuropathol</u> <u>Appl Neurobiol</u> 25(3): 207-14.
- Haas, S., J. Brockhaus, A. Verkhratsky and H. Kettenmann (1996). "ATP-induced membrane currents in ameboid microglia acutely isolated from mouse brain slices." <u>Neuroscience</u> 75(1): 257-61.
- Haberly, L. B. and J. L. Price (1977). "The axonal projection patterns of the mitral and tufted cells of the olfactory bulb in the rat." <u>Brain Res</u> 129(1): 152-7.
- Hall, S. M. (1972). "The effect of injections of lysophosphatidyl choline into white matter of the adult mouse spinal cord." <u>J Cell Sci</u> 10(2): 535-46.
- Hall, S. M. and N. A. Gregson (1971). "The in vivo and ultrastructural effects of injection of lysophosphatidyl choline into myelinated peripheral nerve fibres of the adult mouse." J Cell Sci 9(3): 769-89.
- Hamilton, S. P. and L. H. Rome (1994). "Stimulation of in vitro myelin synthesis by microglia." <u>Glia</u> 11(4): 326-35.
- Hammang, J. P., D. R. Archer and I. D. Duncan (1997). "Myelination following transplantation of EGF-responsive neural stem cells into a myelin-deficient environment." <u>Exp Neurol</u> 147(1): 84-95.
- Harding, J., P. P. Graziadei, G. A. Monti Graziadei and F. L. Margolis (1977).
 "Denervation in the primary olfactory pathway of mice. IV. Biochemical and morphological evidence for neuronal replacement following nerve section." Brain Res 132(1): 11-28.
- Hardy, R. and R. Reynolds (1991). "Proliferation and differentiation potential of rat forebrain oligodendroglial progenitors both in vitro and in vivo." <u>Development</u> 111(4): 1061-80.
- Hardy, R. and R. Reynolds (1993). "Neuron-oligodendroglial interactions during central nervous system development." J Neurosci Res 36(2): 121-6.
- Hardy, R. J. and V. L. Friedrich, Jr. (1996). "Progressive remodeling of the oligodendrocyte process arbor during myelinogenesis." <u>Dev Neurosci</u> 18(4): 243-54.
- Harrison, B. (1985). "Schwann cell and oligodendrocyte remyelination in lysolecithininduced lesions in irradiated rat spinal cord." J Neurol Sci 67(2): 143-59.

- Hart, I. K., W. D. Richardson, S. R. Bolsover and M. C. Raff (1989). "PDGF and intracellular signaling in the timing of oligodendrocyte differentiation." <u>J Cell</u> <u>Biol</u> 109(6 Pt 2): 3411-7.
- Hatten, M. E., R. K. Liem, M. L. Shelanski and C. A. Mason (1991). "Astroglia in CNS injury." <u>Glia</u> 4(2): 233-43.
- Hebert, G., J. Arsaut, R. Dantzer and J. Demotes-Mainard (2003). "Time-course of the expression of inflammatory cytokines and matrix metalloproteinases in the striatum and mesencephalon of mice injected with 1-methyl-4-phenyl-1,2,3,6-tetrahydropyridine, a dopaminergic neurotoxin." <u>Neurosci Lett</u> 349(3): 191-5.
- Heimer, L. (1968). "Synaptic distribution of centripetal and centrifugal nerve fibres in the olfactory system of the rat. An experimental anatomical study." J Anat 103(3): 413-32.
- Henion, P. D. and J. A. Weston (1997). "Timing and pattern of cell fate restrictions in the neural crest lineage." <u>Development</u> 124(21): 4351-9.
- Hermanson, M., T. Olsson, B. Westermark and K. Funa (1995). "PDGF and its receptors following facial nerve axotomy in rats: expression in neurons and surrounding glia." <u>Exp Brain Res</u> 102(3): 415-22.
- Herndon, R. M., D. L. Price and L. P. Weiner (1977). "Regeneration of oligodendroglia during recovery from demyelinating disease." <u>Science</u> 195(4279): 693-4.
- Hetier, E., J. Ayala, P. Denefle, A. Bousseau, P. Rouget, M. Mallat and Prochiantz, A. (1988). "Brain macrophages synthesize interleukin-1 and interleukin-1 mRNAs in vitro." <u>J Neurosci Res</u> 21: 391-397.
- Heumann, R., D. Lindholm, C. Bandtlow, M. Meyer, M. J. Radeke, T. P. Misko, E. Shooter and H. Thoenen (1987). "Differential regulation of mRNA encoding nerve growth factor and its receptor in rat sciatic nerve during development, degeneration, and regeneration: role of macrophages." <u>Proc Natl Acad Sci U S A</u> 84(23): 8735-9.
- Hickey, W. F., B. L. Hsu and H. Kimura (1991). "T-lymphocyte entry into the central nervous system." J Neurosci Res 28(2): 254-60.
- Hickey, W. F. and H. Kimura (1988). "Perivascular microglial cells of the CNS are bone marrow-derived and present antigen in vivo." <u>Science</u> 239(4837): 290-2.
- Hickey, W. F., K. Vass and H. Lassmann (1992). "Bone marrow-derived elements in the central nervous system: an immunohistochemical and ultrastructural survey of rat chimeras." J Neuropathol Exp Neurol 51(3): 246-56.
- Hildebrand, J. G., W. Rossler and L. P. Tolbert (1997). "Postembryonic development of the olfactory system in the moth Manduca sexta: primary-afferent control of glomerular development." Semin Cell Dev Biol 8(2): 163-70.
- Hinds, J. W. (1972). "Early neuron differentiation in the mouse of olfactory bulb. I. Light microscopy." J Comp Neurol 146(2): 233-52.
- Hinks, G. L. and R. J. Franklin (2000). "Delayed changes in growth factor gene expression during slow remyelination in the CNS of aged rats." <u>Mol Cell</u> <u>Neurosci</u> 16(5): 542-56.
- Hiremath, M. M., Y. Saito, G. W. Knapp, J. P. Ting, K. Suzuki and G. K. Matsushima (1998). "Microglial/macrophage accumulation during cuprizone-induced demyelination in C57BL/6 mice." <u>J Neuroimmunol</u> 92(1-2): 38-49.

- Hofman, F. M., D. R. Hinton, K. Johnson and J. E. Merrill (1989). "Tumor necrosis factor identified in multiple sclerosis brain." J Exp Med 170(2): 607-12.
- Holbrook, E. H., K. E. Szumowski and J. E. Schwob (1995). "An immunochemical, ultrastructural, and developmental characterization of the horizontal basal cells of rat olfactory epithelium." J Comp Neurol 363(1): 129-46.
- Holtzman, D. A. and M. Halpern (1990). "Embryonic and neonatal development of the vomeronasal and olfactory systems in garter snakes (Thamnophis spp.)." J Morphol 203(2): 123-40.
- Honig, M. G. and R. I. Hume (1986). "Fluorescent carbocyanine dyes allow living neurons of identified origin to be studied in long-term cultures." <u>J Cell Biol</u> 103(1): 171-87.
- Honmou, O., P. A. Felts, S. G. Waxman and J. D. Kocsis (1996). "Restoration of normal conduction properties in demyelinated spinal cord axons in the adult rat by transplantation of exogenous Schwann cells." <u>J Neurosci</u> 16(10): 3199-208.
- Horner, P. J. and T. D. Palmer (2003). "New roles for astrocytes: the nightlife of an 'astrocyte'. La vida loca!" <u>Trends Neurosci</u> 26(11): 597-603.
- Huseby, E. S., D. Liggitt, T. Brabb, B. Schnabel, C. Ohlen and J. Goverman (2001). "A pathogenic role for myelin-specific CD8(+) T cells in a model for multiple sclerosis." J Exp Med 194(5): 669-76.
- Hylden, J. L. and G. L. Wilcox (1980). "Intrathecal morphine in mice: a new technique." <u>Eur J Pharmacol</u> 67(2-3): 313-6.
- Ibanez, C., S. A. Shields, M. El-Etr, E. Leonelli, V. Magnaghi, W. W. Li, F. J. Sim, E. E. Baulieu, R. C. Melcangi, M. Schumacher and R. J. Franklin (2003). "Steroids and the reversal of age-associated changes in myelination and remyelination." <u>Prog Neurobiol</u> 71(1): 49-56.
- The IFNB Multiple Sclerosis Study Group and The University of British Columbia MS/MRI Analysis Group." (1995). "Interferon beta-1b in the treatment of multiple sclerosis: final outcome of the randomized controlled trial. <u>Neurology</u> 45(7): 1277-85.
- Ikenaka, K., T. Kagawa and K. Mikoshiba (1992). "Selective expression of DM-20, an alternatively spliced myelin proteolipid protein gene product, in developing nervous system and in nonglial cells." J Neurochem 58(6): 2248-53.
- Illes, P., W. Norenberg and P. J. Gebicke-Haerter (1996). "Molecular mechanisms of microglial activation. B. Voltage- and purinoceptor-operated channels in microglia." <u>Neurochem Int</u> 29(1): 13-24.
- Imaizumi, T., K. L. Lankford and J. D. Kocsis (2000). "Transplantation of olfactory ensheathing cells or Schwann cells restores rapid and secure conduction across the transected spinal cord." <u>Brain Res</u> 854(1-2): 70-8.
- Imaizumi, T., K. L. Lankford, S. G. Waxman, C. A. Greer and J. D. Kocsis (1998). "Transplanted olfactory ensheathing cells remyelinate and enhance axonal conduction in the demyelinated dorsal columns of the rat spinal cord." <u>J</u> <u>Neurosci</u> 18(16): 6176-85.
- Inouye, H., A. L. Ganser and D. A. Kirschner (1985). "Shiverer and normal peripheral myelin compared: basic protein localization, membrane interactions, and lipid composition." J Neurochem 45(6): 1911-22.

- Itoyama, Y., A. Ohnishi, J. Tateishi, Y. Kuroiwa and H. D. Webster (1985). "Spinal cord multiple sclerosis lesions in Japanese patients: Schwann cell remyelination occurs in areas that lack glial fibrillary acidic protein (GFAP)." <u>Acta</u> <u>Neuropathol (Berl)</u> 65(3-4): 217-23.
- Itoyama, Y., H. D. Webster, E. P. Richardson, Jr. and B. D. Trapp (1983). "Schwann cell remyelination of demyelinated axons in spinal cord multiple sclerosis lesions." <u>Ann Neurol</u> 14(3): 339-46.
- Iwashita, Y. and W. F. Blakemore (2000). "Areas of demyelination do not attract significant numbers of schwann cells transplanted into normal white matter." <u>Glia</u> 31(3): 232-40.
- Iwashita, Y., J. W. Fawcett, A. J. Crang, R. J. Franklin and W. F. Blakemore (2000). "Schwann cells transplanted into normal and X-irradiated adult white matter do not migrate extensively and show poor long-term survival." <u>Exp Neurol</u> 164(2): 292-302.
- Jacobs, L. D., R. W. Beck, J. H. Simon, R. P. Kinkel, C. M. Brownscheidle, T. J. Murray, N. A. Simonian, P. J. Slasor and A. W. Sandrock (2000).
 "Intramuscular interferon beta-1a therapy initiated during a first demyelinating event in multiple sclerosis. CHAMPS Study Group." <u>N Engl J Med</u> 343(13): 898-904.
- Jacobs, L. D., D. L. Cookfair, R. A. Rudick, R. M. Herndon, J. R. Richert, A. M. Salazar, J. S. Fischer, D. E. Goodkin, C. V. Granger, J. H. Simon, J. J. Alam, D. M. Bartoszak, D. N. Bourdette, J. Braiman, C. M. Brownscheidle, M. E. Coats, S. L. Cohan, D. S. Dougherty, R. P. Kinkel, M. K. Mass, F. E. Munschauer, 3rd, R. L. Priore, P. M. Pullicino, B. J. Scherokman, R. H. Whitham and et al. (1996). "Intramuscular interferon beta-1a for disease progression in relapsing multiple sclerosis. The Multiple Sclerosis Collaborative Research Group (MSCRG)." Ann Neurol 39(3): 285-94.
- Jafek, B. W., B. Murrow, R. Michaels, D. Restrepo and M. Linschoten (2002). "Biopsies of human olfactory epithelium." <u>Chem Senses</u> 27(7): 623-8.
- Janzer, R. C. and M. C. Raff (1987). "Astrocytes induce blood-brain barrier properties in endothelial cells." <u>Nature</u> 325(6101): 253-7.
- Jasmin, L., G. Janni, T. M. Moallem, D. A. Lappi and P. T. Ohara (2000). "Schwann cells are removed from the spinal cord after effecting recovery from paraplegia." <u>J Neurosci</u> 20(24): 9215-23.
- Jeffery, D. R., J. Absher, F. E. Pfeiffer and H. Jackson (2000). "Cortical deficits in multiple sclerosis on the basis of subcortical lesions." <u>Mult Scler</u> 6(1): 50-5.
- Jeffery, N. D., A. J. Crang, T. O'Leary M, S. J. Hodge and W. F. Blakemore (1999). "Behavioural consequences of oligodendrocyte progenitor cell transplantation into experimental demyelinating lesions in the rat spinal cord." <u>Eur J Neurosci</u> 11(5): 1508-14.
- Jeserich, G. and T. V. Waehneldt (1986). "Bony fish myelin: evidence for common major structural glycoproteins in central and peripheral myelin of trout." J Neurochem 46(2): 525-33.
- Jessen, K. R. and R. Mirsky (1992). "Schwann cells: early lineage, regulation of proliferation and control of myelin formation." <u>Curr Opin Neurobiol</u> 2(5): 575-81.

- Jessen, K. R. and R. Mirsky (1997). "Embryonic Schwann cell development: the biology of Schwann cell precursors and early Schwann cells." J Anat 191 (Pt 4): 501-5.
- Jessen, K. R. and R. Mirsky (1998). "Origin and early development of Schwann cells." <u>Microsc Res Tech</u> 41(5): 393-402.
- Jessen, K. R., L. Morgan, H. J. Stewart and R. Mirsky (1990). "Three markers of adult non-myelin-forming Schwann cells, 217c(Ran-1), A5E3 and GFAP: development and regulation by neuron-Schwann cell interactions." <u>Development</u> 109(1): 91-103.
- Jiang, S., M. I. Khan, J. Wang, P. J. Middlemiss, E. S. Werstiuk, R. Wickson and M. P. Rathbone (2003). "Enteric glia promote functional recovery of CTM reflex after dorsal root transection." <u>Neuroreport</u> 14(10): 1301-4.
- Johe, K. K., T. G. Hazel, T. Muller, M. M. Dugich-Djordjevic and R. D. McKay (1996). "Single factors direct the differentiation of stem cells from the fetal and adult central nervous system." <u>Genes Dev</u> 10(24): 3129-40.
- Johnson, K. P., B. R. Brooks, J. A. Cohen, C. C. Ford, J. Goldstein, R. P. Lisak, L. W. Myers, H. S. Panitch, J. W. Rose and R. B. Schiffer (1995). "Copolymer 1 reduces relapse rate and improves disability in relapsing-remitting multiple sclerosis: results of a phase III multicenter, double-blind placebo-controlled trial. The Copolymer 1 Multiple Sclerosis Study Group." <u>Neurology</u> 45(7): 1268-76.
- Johnson, K. P., B. R. Brooks, J. A. Cohen, C. C. Ford, J. Goldstein, R. P. Lisak, L. W. Myers, H. S. Panitch, J. W. Rose, R. B. Schiffer, T. Vollmer, L. P. Weiner and J. S. Wolinsky (1998). "Extended use of glatiramer acetate (Copaxone) is well tolerated and maintains its clinical effect on multiple sclerosis relapse rate and degree of disability. Copolymer 1 Multiple Sclerosis Study Group." <u>Neurology</u> 50(3): 701-8.
- Jones, L. L., M. Oudega, M. B. Bunge and M. H. Tuszynski (2001). "Neurotrophic factors, cellular bridges and gene therapy for spinal cord injury." <u>J Physiol</u> 533(Pt 1): 83-9.
- Joy, J. E. and Richard B. Johnston (2001). Multiple Sclerosis: Current Status and Strategies for the Future." National Academy Press Washington, D.C.
- Kafitz, K. W. and C. A. Greer (1997). "Role of laminin in axonal extension from olfactory receptor cells." J Neurobiol 32(3): 298-310.
- Kafitz, K. W. and C. A. Greer (1998). "The influence of ensheathing cells on olfactory receptor cell neurite outgrowth in vitro." <u>Ann N Y Acad Sci</u> 855: 266-9.
- Kalyani, A., K. Hobson and M. S. Rao (1997). "Neuroepithelial stem cells from the embryonic spinal cord: isolation, characterization, and clonal analysis." <u>Dev</u> <u>Biol</u> 186(2): 202-23.
- Kappos, L., G. Comi, H. Panitch, J. Oger, J. Antel, P. Conlon and L. Steinman (2000).
 "Induction of a non-encephalitogenic type 2 T helper-cell autoimmune response in multiple sclerosis after administration of an altered peptide ligand in a placebo-controlled, randomized phase II trial. The Altered Peptide Ligand in Relapsing MS Study Group." <u>Nat Med</u> 6(10): 1176-82.

- Kato, T., O. Honmou, T. Uede, K. Hashi and J. D. Kocsis (2000). "Transplantation of human olfactory ensheathing cells elicits remyelination of demyelinated rat spinal cord." <u>Glia</u> 30(3): 209-18.
- Kaur, C. and E. A. Ling (1992). "Activation and re-expression of surface antigen in microglia following an epidural application of kainic acid in the rat brain." J <u>Anat</u> 180 (Pt 2): 333-42.
- Kazmierski, R., M. Wender, P. Guzik and D. Zielonka (2004). "Association of influenza incidence with multiple sclerosis onset." Folia Neuropathol 42(1): 19-23.
- Kerlero de Rosbo, N., R. Milo, M. B. Lees, D. Burger, C. C. Bernard and A. Ben-Nun (1993). "Reactivity to myelin antigens in multiple sclerosis. Peripheral blood lymphocytes respond predominantly to myelin oligodendrocyte glycoprotein." J Clin Invest 92(6): 2602-8.
- Kettenmann, H., R. Banati and W. Walz (1993). "Electrophysiological behavior of microglia." <u>Glia</u> 7(1): 93-101.
- Key, B. and R. A. Akeson (1990). "Olfactory neurons express a unique glycosylated form of the neural cell adhesion molecule (N-CAM)." <u>J Cell Biol</u> 110(5): 1729-43.
- Keyvan-Fouladi, N., G. Raisman and Y. Li (2003). "Functional repair of the corticospinal tract by delayed transplantation of olfactory ensheathing cells in adult rats." J Neurosci 23(28): 9428-34.
- Kidd, G. J., P. E. Hauer and B. D. Trapp (1990). "Axons modulate myelin protein messenger RNA levels during central nervous system myelination in vivo." <u>J</u> <u>Neurosci Res</u> 26(4): 409-18.
- Kiernan, J. A. (1979). "Hypotheses concerned with axonal regeneration in the mammalian nervous system." <u>Biol Rev Camb Philos Soc</u> 54(2): 155-97.
- Kies, M. W., R. E. Martenson and G. E. Deibler (1972). "Myelin basic proteins." <u>Adv</u> <u>Exp Med Biol</u> 32(0): 201-14.
- Kirschner, D. A. and A. L. Ganser (1980). "Compact myelin exists in the absence of basic protein in the shiverer mutant mouse." <u>Nature</u> 283(5743): 207-10.
- Kleitman, N., D. K. Simon, M. Schachner and R. P. Bunge (1988). "Growth of embryonic retinal neurites elicited by contact with Schwann cell surfaces is blocked by antibodies to L1." Exp Neurol 102(3): 298-306.
- Kloss, C. U., G. W. Kreutzberg and G. Raivich (1997). "Proliferation of ramified microglia on an astrocyte monolayer: characterization of stimulatory and inhibitory cytokines." J Neurosci Res 49(2): 248-54.
- Knox, K. K., J. H. Brewer, J. M. Henry, D. J. Harrington and D. R. Carrigan (2000).
 "Human herpesvirus 6 and multiple sclerosis: systemic active infections in patients with early disease." <u>Clin Infect Dis</u> 31(4): 894-903.
- Kohama, I., K. L. Lankford, J. Preiningerova, F. A. White, T. L. Vollmer and J. D. Kocsis (2001). "Transplantation of cryopreserved adult human Schwann cells enhances axonal conduction in demyelinated spinal cord." <u>J Neurosci</u> 21(3): 944-50.
- Kolodny, E. H. (1993). "Dysmyelinating and demyelinating conditions in infancy." <u>Curr</u> <u>Opin Neurol Neurosurg</u> 6(3): 379-86.

- Komoly, S., M. D. Jeyasingham, O. E. Pratt and P. L. Lantos (1987). "Decrease in oligodendrocyte carbonic anhydrase activity preceding myelin degeneration in cuprizone induced demyelination." J Neurol Sci 79(1-2): 141-8.
- Kondo, A., T. Nakano and K. Suzuki (1987). "Blood-brain barrier permeability to horseradish peroxidase in twitcher and cuprizone-intoxicated mice." <u>Brain Res</u> 425(1): 186-90.
- Kornek, B. and H. Lassmann (2003). "Neuropathology of multiple sclerosis-new concepts." <u>Brain Res Bull</u> 61(3): 321-6.
- Kornek, B., M. K. Storch, R. Weissert, E. Wallstroem, A. Stefferl, T. Olsson, C. Linington, M. Schmidbauer and H. Lassmann (2000). "Multiple sclerosis and chronic autoimmune encephalomyelitis: a comparative quantitative study of axonal injury in active, inactive, and remyelinated lesions." <u>Am J Pathol</u> 157(1): 267-76.
- Kosaka, Y., N. Kobayashi, T. Fukazawa, T. Totsugawa, M. Maruyama, C. Yong, T. Arata, H. Ikeda, K. Kobayashi, T. Ueda, Y. Kurabayashi and N. Tanaka (2004).
 "Lentivirus-based gene delivery in mouse embryonic stem cells." <u>Artif Organs</u> 28(3): 271-7.
- Kosaras, B. and E. Snyder (2002). "Electron microscopy and lac-Z labeling." <u>Methods</u> <u>Mol Biol</u> 198: 157-75.
- Koshinaga, M. and S. R. Whittemore (1995). "The temporal and spatial activation of microglia in fiber tracts undergoing anterograde and retrograde degeneration following spinal cord lesion." J Neurotrauma 12(2): 209-22.
- Kott, J. N., L. E. Westrum, E. W. Raines, M. Sasahara and R. Ross (1994). "Olfactory ensheathing glia and platelet-derived growth factor B-chain reactivity in the transplanted rat olfactory bulb." <u>Int J Dev Neurosci</u> 12(4): 315-23.
- Kotter, M. R., A. Setzu, F. J. Sim, N. Van Rooijen and R. J. Franklin (2001).
 "Macrophage depletion impairs oligodendrocyte remyelination following lysolecithin-induced demyelination." <u>Glia</u> 35(3): 204-12.
- Kraus, J., B. Engelhardt, N. Chatzimanolis, R. Bauer, J. Tofighi, B. S. Kuehne, C. Laske, E. Stolz, P. Frielinghaus, C. Schaefer, F. Blaes, H. Traupe, M. Kaps and P. Oschmann (2002). "Cell surface bound and soluble adhesion molecules in CSF and blood in multiple sclerosis: correlation with MRI-measures of subclinical disease severity and activity." J Neuroimmunol 122(1-2): 175-85.
- Kreutzberg, G.W. (1995). "Reaction of the neuronal cell body to axonal damage." in: S.G. Waxman, J.D. Kocsis, P.K. Stys Eds., The Axon, Oxford Univ. Press, pp. 355–374.
- Kreutzberg, G. W. (1996). "Microglia: a sensor for pathological events in the CNS." <u>Trends Neurosci</u> 19(8): 312-8.
- Kuhlbrodt, K., B. Herbarth, E. Sock, I. Hermans-Borgmeyer and M. Wegner (1998).
 "Sox10, a novel transcriptional modulator in glial cells." J Neurosci 18(1): 237-50.
- Kukekov, V. G., E. D. Laywell, O. Suslov, K. Davies, B. Scheffler, L. B. Thomas, T. F. O'Brien, M. Kusakabe and D. A. Steindler (1999). "Multipotent stem/progenitor cells with similar properties arise from two neurogenic regions of adult human brain." <u>Exp Neurol</u> 156(2): 333-44.

- Kurtzke, J. F. (1980). "Epidemiologic contributions to multiple sclerosis: an overview." <u>Neurology</u> 30(7 Pt 2): 61-79.
- Kurz, H. and B. Christ (1998). "Embryonic CNS macrophages and microglia do not stem from circulating, but from extravascular precursors." <u>Glia</u> 22(1): 98-102.
- Lachapelle, F., V. Avellana-Adalid, B. Nait-Oumesmar and A. Baron-Van Evercooren (2002). "Fibroblast growth factor-2 (FGF-2) and platelet-derived growth factor AB (PDGF AB) promote adult SVZ-derived oligodendrogenesis in vivo." <u>Mol</u> <u>Cell Neurosci</u> 20(3): 390-403
- Lakatos, A., S. C. Barnett and R. J. Franklin (2003). "Olfactory ensheathing cells induce less host astrocyte response and chondroitin sulphate proteoglycan expression than Schwann cells following transplantation into adult CNS white matter." <u>Exp</u> <u>Neurol</u> 184(1): 237-46.
- Lakatos, A., R. J. Franklin and S. C. Barnett (2000). "Olfactory ensheathing cells and Schwann cells differ in their in vitro interactions with astrocytes." <u>Glia</u> 32(3): 214-25.
- Lang, W., M. Rodriguez, V. A. Lennon and P. W. Lampert (1984). "Demyelination and remyelination in murine viral encephalomyelitis." <u>Ann N Y Acad Sci</u> 436: 98-102.
- Langford, L. A. and G. C. Owens (1990). "Resolution of the pathway taken by implanted Schwann cells to a spinal cord lesion by prior infection with a retrovirus encoding beta-galactosidase." Acta Neuropathol (Berl) 80(5): 514-20.
- Lanza, D. C., D. A. Deems, R. L. Doty, D. Moran, D. Crawford, J. C. Rowley, 3rd, A. Sajjadian and D. W. Kennedy (1994). "The effect of human olfactory biopsy on olfaction: a preliminary report." <u>Laryngoscope</u> 104(7): 837-40.
- Lanza, D. C., D. T. Moran, R. L. Doty, J. Q. Trojanowski, J. H. Lee, J. C. Rowley, 3rd, D. Crawford and D. W. Kennedy (1993). "Endoscopic human olfactory biopsy technique: a preliminary report." <u>Laryngoscope</u> 103(7): 815-9.
- Lassmann, H. (2003). "Hypoxia-like tissue injury as a component of multiple sclerosis lesions." J Neurol Sci 206(2): 187-91.
- Lassmann, H., W. Bruck and C. Lucchinetti (2001). "Heterogeneity of multiple sclerosis pathogenesis: implications for diagnosis and therapy." <u>Trends Mol Med</u> 7(3): 115-21.
- Lassmann, H., C. Brunner, M. Bradl and C. Linington (1988). "Experimental allergic encephalomyelitis: the balance between encephalitogenic T lymphocytes and demyelinating antibodies determines size and structure of demyelinated lesions." <u>Acta Neuropathol (Berl)</u> 75(6): 566-76.
- Lassmann, H., C. S. Raine, J. Antel and J. W. Prineas (1998). "Immunopathology of multiple sclerosis: report on an international meeting held at the Institute of Neurology of the University of Vienna." J Neuroimmunol 86(2): 213-7.
- Lauer, K. (1997). "Ecologic studies of multiple sclerosis." <u>Neurology</u> 49(2 Suppl 2): S18-26.
- Laywell, E. D., U. Dorries, U. Bartsch, A. Faissner, M. Schachner and D. A. Steindler (1992). "Enhanced expression of the developmentally regulated extracellular matrix molecule tenascin following adult brain injury." <u>Proc Natl Acad Sci U S</u> <u>A</u> 89(7): 2634-8.

- Lazarov-Spiegler, O., A.S. Solomon, A.B. Zeev-Brann, D.L. Hirschberg, V. Lavie and M. Schwartz (1996). "Transplantation of activated macrophages overcomes central nervous system regrowth failure." <u>FASEB J</u> 10: 1296–1302
- Le Douarin, N., C. Dulac, E. Dupin and P. Cameron-Curry (1991). "Glial cell lineages in the neural crest." <u>Glia</u> 4(2): 175-84.
- Learish, R. D., O. Brustle, S. C. Zhang and I. D. Duncan (1999). "Intraventricular transplantation of oligodendrocyte progenitors into a fetal myelin mutant results in widespread formation of myelin." <u>Ann Neurol</u> 46(5): 716-22.
- Lebar, R., J. M. Boutry, C. Vincent, R. Robineaux and G. A. Voisin (1976). "Studies on autoimmune encephalomyelitis in the guinea pig. II. An in vitro investigation on the nature, properties, and specificity of the serum-demyelinating factor." J Immunol 116(5): 1439-46.
- Lebar, R., C. Lubetzki, C. Vincent, P. Lombrail and J. M. Boutry (1986). "The M2 autoantigen of central nervous system myelin, a glycoprotein present in oligodendrocyte membrane." Clin Exp Immunol 66(2): 423-34.
- Lee, S. C., W. Liu, D. W. Dickson, C. F. Brosnan, and J. W. Berman (1993). "Cytokine production by human fetal microglia and astrocytes. Differential induction by lipopolysaccharide and IL-1 beta." J Immunol 150(7): 2659-67.
- Lee, W. H., S. H. Kim, Y. Lee, B. B. Lee, B. Kwon, H. Song, B. S. Kwon and J. E. Park (2001). "Tumor necrosis factor receptor superfamily 14 is involved in atherogenesis by inducing proinflammatory cytokines and matrix metalloproteinases." <u>Arterioscler Thromb Vasc Biol</u> 21: 2004–2010.
- Lemke, G. (1993). "The molecular genetics of myelination: an update." <u>Glia</u> 7(4): 263-71.
- Lemke, G. and R. Axel (1985). "Isolation and sequence of a cDNA encoding the major structural protein of peripheral myelin." <u>Cell</u> 40(3): 501-8.
- Lemke, G. and M. Chao (1988). "Axons regulate Schwann cell expression of the major myelin and NGF receptor genes." <u>Development</u> 102(3): 499-504.
- Lemke, G., E. Lamar and J. Patterson (1988). "Isolation and analysis of the gene encoding peripheral myelin protein zero." <u>Neuron</u> 1(1): 73-83.
- Li, C., M. B. Tropak, R. Gerlai, S. Clapoff, W. Abramow-Newerly, B. Trapp, A. Peterson and J. Roder (1994). "Myelination in the absence of myelin-associated glycoprotein." <u>Nature</u> 369(6483): 747-50.
- Li, Y., P. Decherchi and G. Raisman (2003). "Transplantation of olfactory ensheathing cells into spinal cord lesions restores breathing and climbing." <u>J Neurosci</u> 23(3): 727-31.
- Li, Y., P. M. Field and G. Raisman (1997). "Repair of adult rat corticospinal tract by transplants of olfactory ensheathing cells." <u>Science</u> 277(5334): 2000-2.
- Li, Y., P. M. Field and G. Raisman (1998). "Regeneration of adult rat corticospinal axons induced by transplanted olfactory ensheathing cells." <u>J Neurosci</u> 18(24): 10514-24.
- Liesi, P. (1985). "Laminin-immunoreactive glia distinguish regenerative adult CNS systems from non-regenerative ones." <u>Embo J</u> 4(10): 2505-11.
- Lindholm, D., R. Heumann, M. Meyer and H. Thoenen (1987). "Interleukin-1 regulates synthesis of nerve growth factor in non-neuronal cells of rat sciatic nerve." <u>Nature</u> 330(6149): 658-9.

- Ling, E. A. and W. C. Wong (1993). "The origin and nature of ramified and amoeboid microglia: a historical review and current concepts." <u>Glia</u> 7(1): 9-18.
- Linington, C., M. Bradl, H. Lassmann, C. Brunner and K. Vass (1988). "Augmentation of demyelination in rat acute allergic encephalomyelitis by circulating mouse monoclonal antibodies directed against a myelin/oligodendrocyte glycoprotein." <u>Am J Pathol</u> 130(3): 443-54.
- Lips, K., C. C. Stichel and H. W. Muller (1995). "Restricted appearance of tenascin and chondroitin sulphate proteoglycans after transection and sprouting of adult rat postcommissural fornix." J Neurocytol 24(6): 449-64.
- Lipton, H. L. (1975). "Theiler's virus infection in mice: an unusual biphasic disease process leading to demyelination." <u>Infect Immun</u> 11(5): 1147-55.
- Lipton, H. L. and M. L. Jelachich (1997). "Molecular pathogenesis of Theiler's murine encephalomyelitis virus-induced demyelinating disease in mice." <u>Intervirology</u> 40(2-3): 143-52.
- Liu, J., M. W. Marino, G. Wong, D. Grail, A. Dunn, J. Bettadapura, A. J. Slavin, L. Old and C. C. Bernard (1998). "TNF is a potent anti-inflammatory cytokine in autoimmune-mediated demyelination." <u>Nat Med</u> 4(1): 78-83.
- Lois, C. and A. Alvarez-Buylla (1993). "Proliferating subventricular zone cells in the adult mammalian forebrain can differentiate into neurons and glia." <u>Proc Natl</u> <u>Acad Sci U S A</u> 90(5): 2074-7.
- Lopes-Cardozo, M., J. E. Sykes, R. H. Van der Pal and L. M. van Golde (1989). "Development of oligodendrocytes. Studies of rat glial cells cultured in chemically-defined medium." <u>J Dev Physiol</u> 12(3): 117-27.
- Lorch, Y., A. Friedmann, H. L. Lipton and M. Kotler (1981). "Theiler's murine encephalomyelitis virus group includes two distinct genetic subgroups that differ pathologically and biologically." J Virol 40(2): 560-7.
- Losseff, N. A., L. Wang, H. M. Lai, D. S. Yoo, M. L. Gawne-Cain, W. I. McDonald, D. H. Miller and A. J. Thompson (1996). "Progressive cerebral atrophy in multiple sclerosis. A serial MRI study." <u>Brain</u> 119 (Pt 6): 2009-19.
- Losseff, N. A., S. L. Webb, J. I. O'Riordan, R. Page, L. Wang, G. J. Barker, P. S. Tofts, W. I. McDonald, D. H. Miller and A. J. Thompson (1996). "Spinal cord atrophy and disability in multiple sclerosis. A new reproducible and sensitive MRI method with potential to monitor disease progression." <u>Brain</u> 119 (Pt 3): 701-8.
- Love, S. (1988)." Cuprizone neurotoxicity in the rat: morphologic observations." J Neurol Sci 84(2-3): 223-37.
- Loughlin, A. J., C. A. Copelman, A. Hall, T. Armer, B. C. Young, D. N. Landon and M. L. Cuzner (1997). "Myelination and remyelination of aggregate rat brain cell cultures enriched with macrophages." J Neurosci Res 47(4): 384-92.
- Lu, J., F. Feron, S. M. Ho, A. Mackay-Sim and P. M. Waite (2001). "Transplantation of nasal olfactory tissue promotes partial recovery in paraplegic adult rats." <u>Brain</u> <u>Res</u> 889(1-2): 344-57.
- Lu, J., F. Feron, A. Mackay-Sim and P. M. Waite (2002). "Olfactory ensheathing cells promote locomotor recovery after delayed transplantation into transected spinal cord." <u>Brain</u> 125(Pt 1): 14-21.
- Lublin, F. D., P. H. Maurer, R. G. Berry and D. Tippett (1981). "Delayed, relapsing experimental allergic encephalomyelitis in mice." J Immunol 126(3): 819-22.

- Lucchinetti, C., W. Bruck, J. Parisi, B. Scheithauer, M. Rodriguez and H. Lassmann (1999). "A quantitative analysis of oligodendrocytes in multiple sclerosis lesions. A study of 113 cases." <u>Brain</u> 122 (Pt 12): 2279-95.
- Lucchinetti, C., W. Bruck, J. Parisi, B. Scheithauer, M. Rodriguez and H. Lassmann (2000). "Heterogeneity of multiple sclerosis lesions: implications for the pathogenesis of demyelination." Ann Neurol 47(6): 707-17.
- Lucchinetti, C. F., W. Bruck, M. Rodriguez and H. Lassmann (1996). "Distinct patterns of multiple sclerosis pathology indicates heterogeneity on pathogenesis." <u>Brain</u> <u>Pathol</u> 6(3): 259-74.
- Lucius, R., H. P. Young, S. Tidow and J. Sievers (1996). "Growth stimulation and chemotropic attraction of rat retinal ganglion cell axons in vitro by co-cultured optic nerves, astrocytes and astrocyte conditioned medium." Int J Dev Neurosci 14(4): 387-98.
- Ludwin, S. K. (1978). "Central nervous system demyelination and remyelination in the mouse: an ultrastructural study of cuprizone toxicity." <u>Lab Invest</u> 39(6): 597-612.
- Ludwin, S. K. (1997). "The pathobiology of the oligodendrocyte." <u>J Neuropathol Exp</u> <u>Neurol</u> 56(2): 111-24.
- Lundberg, C. and A. Bjorklund (1996). "Host regulation of glial markers in intrastriatal grafts of conditionally immortalized neural stem cell lines." <u>Neuroreport</u> 7(4): 847-52.
- Lustgarten, J. H., M. Proctor, R. I. Haroun, A. M. Avellino, A. A. Pindzola and M. Kliot (1991). "Semipermeable polymer tubes provide a microenvironment for in vivo analysis of dorsal root regeneration." J Biomech Eng 113(2): 184-8.
- Lyman, W. D., C. F. Brosnan and C.S. Raine (1985). "Chronic-relapsing experimental autoimmune encephalomyelitis. Myelin basic protein induces suppression of blastogenesis during remissions but not during exacerbations." <u>J Neuroimmunol</u> 7(5-6): 345-353.
- Mackay-Sim, A. and P. Kittel (1991). "Cell dynamics in the adult mouse olfactory epithelium: a quantitative autoradiographic study." J Neurosci 11(4): 979-84.
- Macklin, W. B., C. L. Weill and P. L. Deininger (1986). "Expression of myelin proteolipid and basic protein mRNAs in cultured cells." <u>J Neurosci Res</u> 16(1): 203-17.
- Mahanthappa, N. K., D. N. Cooper, S. H. Barondes and G. A. Schwarting (1994). "Rat olfactory neurons can utilize the endogenous lectin, L-14, in a novel adhesion mechanism." <u>Development</u> 120(6): 1373-84.
- Mahanthappa, N. K. and G. A. Schwarting (1993). "Peptide growth factor control of olfactory neurogenesis and neuron survival in vitro: roles of EGF and TGF-beta s." <u>Neuron</u> 10(2): 293-305.
- Mallat, M., R. Houlgatte, P. Brachet and A. Prochiantz (1989). "Lipopolysaccharidestimulated rat brain macrophages release NGF in vitro." <u>Dev Biol</u> 133(1): 309-11.
- Margolis F.L. (1985). "Olfactory marker protein: from PAGE band to cDNA." <u>Trends</u> <u>in Neuroscience</u> 8: 542-546.

- Marin-Padilla, M. and M. R. Amieva (1989). "Early neurogenesis of the mouse olfactory nerve: Golgi and electron microscopic studies." <u>J Comp Neurol</u> 288(2): 339-52.
- Martini, R., M. H. Mohajeri, S. Kasper, K. P. Giese and M. Schachner (1995). "Mice doubly deficient in the genes for P0 and myelin basic protein show that both proteins contribute to the formation of the major dense line in peripheral nerve myelin." J Neurosci 15(6): 4488-95.
- Martini, R., J. Zielasek, K. V. Toyka, K. P. Giese and M. Schachner (1995). "Protein zero (P0)-deficient mice show myelin degeneration in peripheral nerves characteristic of inherited human neuropathies." <u>Nat Genet</u> 11(3): 281-6.
- Mason, J. L., J. J. Jones, M. Taniike, P. Morell, K. Suzuki and G. K. Matsushima (2000). "Mature oligodendrocyte apoptosis precedes IGF-1 production and oligodendrocyte progenitor accumulation and differentiation during demyelination/remyelination." J Neurosci Res 61(3): 251-62.
- Mason, J. L., C. Langaman, P. Morell, K. Suzuki and G. K. Matsushima (2001).
 "Episodic demyelination and subsequent remyelination within the murine central nervous system: changes in axonal calibre." <u>Neuropathol Appl Neurobiol</u> 27(1): 50-8.
- Mason, J. L., A. Toews, J. D. Hostettler, P. Morell, K. Suzuki, J. E. Goldman and G. K. Matsushima (2004). "Oligodendrocytes and progenitors become progressively depleted within chronically demyelinated lesions." <u>Am J Pathol</u> 164(5): 1673-82.
- Matsuda, Y., H. Koito and H. Yamamoto (1997). "Induction of myelin-associated glycoprotein expression through neuron-oligodendrocyte contact." <u>Brain Res</u> <u>Dev Brain Res</u> 100(1): 110-6.
- Matsushima, G.K. and P. Morell (2001). "The neurotoxicant, cuprizone, as a model to study demyelination and remyelination in the central nervous system." <u>Brain</u> <u>Pathol</u> 11(1): 107-16.
- Mattei, M. G., P. M. Alliel, A. Dautigny, E. Passage, D. Pham-Dinh, J. F. Mattei and P. Jolles (1986). "The gene encoding for the major brain proteolipid (PLP) maps on the q-22 band of the human X chromosome." <u>Hum Genet</u> 72(4): 352-3.
- Maxwell, D. S., L. Kruger and A. Pineda (1969). "The trigeminal nerve root with special reference to the central-peripheral transition zone: an electron microscopic study in the macaque." <u>Anat Rec</u> 164(1): 113-25.
- McCallum, K., M. M. Esiri, W. W. Tourtellotte and J. Booss (1987). "T cell subsets in multiple sclerosis. Gradients at plaque borders and differences in nonplaque regions." <u>Brain</u> 110 (Pt 5): 1297-308.
- McCarthy, J. B., S. L. Palm and L. T. Furcht (1983). "Migration by haptotaxis of a Schwann cell tumor line to the basement membrane glycoprotein laminin." J Cell Biol 97(3): 772-7.
- McFarland, H. F., J. A. Frank, P. S. Albert, M. E. Smith, R. Martin, J. O. Harris, N. Patronas, H. Maloni and D. E. McFarlin (1992). "Using gadolinium-enhanced magnetic resonance imaging lesions to monitor disease activity in multiple sclerosis." <u>Ann Neurol</u> 32(6): 758-66.
- McKeon, R. J., R. C. Schreiber, J. S. Rudge and J. Silver (1991). "Reduction of neurite outgrowth in a model of glial scarring following CNS injury is correlated with

the expression of inhibitory molecules on reactive astrocytes." <u>J Neurosci</u> 11(11): 3398-411.

- McMahon, E. J., K. Suzuki and G. K. Matsushima (2002). "Peripheral macrophage recruitment in cuprizone-induced CNS demyelination despite an intact blood-brain barrier." J Neuroimmunol 130(1-2): 32-45.
- McMorris, F. A., T. M. Smith, S. DeSalvo and R. W. Furlanetto (1986). "Insulin-like growth factor I/somatomedin C: a potent inducer of oligodendrocyte development." <u>Proc Natl Acad Sci U S A</u> 83(3): 822-6.
- Medana, I., M. A. Martinic, H. Wekerle and H. Neumann (2001). "Transection of major histocompatibility complex class I-induced neurites by cytotoxic T lymphocytes." <u>Am J Pathol</u> 159(3): 809-15.
- Meier, C., E. Parmantier, A. Brennan, R. Mirsky and K. R. Jessen (1999). "Developing Schwann cells acquire the ability to survive without axons by establishing an autocrine circuit involving insulin-like growth factor, neurotrophin-3, and platelet-derived growth factor-BB." J Neurosci 19(10): 3847-59.
- Menei, P., C. Montero-Menei, S. R. Whittemore, R. P. Bunge and M. B. Bunge (1998).
 "Schwann cells genetically modified to secrete human BDNF promote enhanced axonal regrowth across transected adult rat spinal cord." <u>Eur J Neurosci</u> 10(2): 607-21.
- Merrill, J. E. and E. N. Benveniste (1996). "Cytokines in inflammatory brain lesions: helpful and harmful." <u>Trends Neurosci</u> 19: 331-338.
- Merrill, J. E., L. J. Ignarro, M. P. Sherman, J. Melinek and T. E. Lane (1993).
 "Microglial cell cytotoxicity of oligodendrocytes is mediated through nitric oxide." J Immunol 151(4): 2132-41.
- Meyer-Franke, A. and B. Barres (1994). "Axon myelination. Myelination without myelin-associated glycoprotein." <u>Curr Biol</u> 4(9): 847-50.
- Mezey, E. and K. J. Chandross (2000). "Bone marrow: a possible alternative source of cells in the adult nervous system." <u>Eur J Pharmacol</u> 405(1-3): 297-302.
- Miescher, G. C., R. Lutzelschwab, B. Erne, F. Ferracin, S. Huber and A. J. Steck (1997). "Reciprocal expression of myelin-associated glycoprotein splice variants in the adult human peripheral and central nervous systems." <u>Brain Res Mol</u> Brain Res 52(2): 299-306.
- Miller, A., S. Shapiro, R. Gershtein, A. Kinarty, H. Rawashdeh, S. Honigman and N. Lahat (1998). "Treatment of multiple sclerosis with copolymer-1 (Copaxone): implicating mechanisms of Th1 to Th2/Th3 immune-deviation." J <u>Neuroimmunol</u> 92(1-2): 113-21.
- Milner, R. (1997). "Understanding the molecular basis of cell migration; implications for clinical therapy in multiple sclerosis." <u>Clin Sci (Lond)</u> 92(2): 113-22.
- Milner, R., H. J. Anderson, R. F. Rippon, J. S. McKay, R. J. Franklin, M. A. Marchionni, R. Reynolds and C. Ffrench-Constant (1997). "Contrasting effects of mitogenic growth factors on oligodendrocyte precursor cell migration." <u>Glia</u> 19(1): 85-90.
- Milner, R. and C. Ffrench-Constant (1994). "A developmental analysis of oligodendroglial integrins in primary cells: changes in alpha v-associated beta subunits during differentiation." <u>Development</u> 120(12): 3497-506.

- Milner, R., M. Wilby, S. Nishimura, K. Boylen, G. Edwards, J. Fawcett, C. Streuli, R. Pytela and C. ffrench-Constant (1997). "Division of labor of Schwann cell integrins during migration on peripheral nerve extracellular matrix ligands." <u>Dev</u> <u>Biol</u> 185(2): 215-28.
- Milward, E. A., C. G. Lundberg, B. Ge, D. Lipsitz, M. Zhao and I. D. Duncan (1997).
 "Isolation and transplantation of multipotential populations of epidermal growth factor-responsive, neural progenitor cells from the canine brain." J Neurosci Res 50(5): 862-71.
- Miragall, F., G. Kadmon, M. Husmann and M. Schachner (1988). "Expression of cell adhesion molecules in the olfactory system of the adult mouse: presence of the embryonic form of N-CAM." <u>Dev Biol</u> 129(2): 516-31.
- Miragall, F., G. Kadmon and M. Schachner (1989). "Expression of L1 and N-CAM cell adhesion molecules during development of the mouse olfactory system." <u>Dev</u> <u>Biol</u> 135(2): 272-86.
- Mirsky, R. and K. R. Jessen (1996). "Schwann cell development, differentiation and myelination." <u>Curr Opin Neurobiol</u> 6(1): 89-96.
- Mirsky, R., D. B. Parkinson, Z. Dong, C. Meier, E. Calle, A. Brennan, P. Topilko, B. S. Harris, H. J. Stewart and K. R. Jessen (2001). "Regulation of genes involved in Schwann cell development and differentiation." <u>Prog Brain Res</u> 132: 3-11.
- Miwa, T., T. Moriizumi, H. Sakashita, Y. Kimura, T. Donjo and M. Furukawa (1993).
 "Transection of the olfactory nerves induces expression of nerve growth factor receptor in mouse olfactory epithelium." <u>Neurosci Lett</u> 155(1): 96-8.
- Montag, D., K. P. Giese, U. Bartsch, R. Martini, Y. Lang, H. Bluthmann, J. Karthigasan, D. A. Kirschner, E. S. Wintergerst, K. A. Nave and et al. (1994).
 "Mice deficient for the myelin-associated glycoprotein show subtle abnormalities in myelin." <u>Neuron</u> 13(1): 229-46.
- Monteyne, P., J. F. Bureau and M. Brahic (1997). "The infection of mouse by Theiler's virus: from genetics to immunology." <u>Immunol Rev</u> 159: 163-76.
- Monti-Graziadei, G. A., F. L. Margolis, J. W. Harding and P. P. Graziadei (1977). "Immunocytochemistry of the olfactory marker protein." <u>J Histochem Cytochem</u> 25(12): 1311-6.
- Monti Graziadei, G. A., R.S. Stanley and P.P.C. Graziadei (1980). "The olfactory marker protein in the olfactory system of the mouse during development." <u>Neuroscience</u> 5: 1239–1252.
- Moore, S. and S. Thanos (1996). "The concept of microglia in relation to central nervous system disease and regeneration" <u>Prog Neurobiol</u> 48(4-5): 441-460.
- Morell, P., C. V. Barrett, J. L. Mason, A. D. Toews, J. D. Hostettler, G. W. Knapp and G. K. Matsushima (1998). "Gene expression in brain during cuprizone-induced demyelination and remyelination." <u>Mol Cell Neurosci</u> 12(4-5): 220-7.
- Morell, P., R. H. Quarles, and W. T. Norton (1994). "Myelin formation, structure and biochemistry." In: <u>Basic Neurochemistry</u>, edited by G. J. Siegel, B. W. Agranoff, R. W. Albers, and P. B. Molinoff. New York: Raven, p. 117–143.
- Morgan, L., K. R. Jessen and R. Mirsky (1991). "The effects of cAMP on differentiation of cultured Schwann cells: progression from an early phenotype (04+) to a myelin phenotype (P0+, GFAP-, N-CAM-, NGF-receptor-) depends on growth inhibition." J Cell Biol 112(3): 457-67.

- Morioka, T., A. N. Kalehua and W. J. Streit (1991). "The microglial reaction in the rat dorsal hippocampus following transient forebrain ischemia." <u>J Cereb Blood</u> <u>Flow Metab</u> 11(6): 966-73.
- Morrison, E. E. and R. M. Costanzo (1992). "Morphology of olfactory epithelium in humans and other vertebrates." <u>Microsc Res Tech</u> 23(1): 49-61.
- Mosahebi, A., B. Woodward, M. Wiberg, R. Martin and G. Terenghi (2001). "Retroviral labeling of Schwann cells: in vitro characterization and in vivo transplantation to improve peripheral nerve regeneration." <u>Glia</u> 34(1): 8-17.
- Moulton, D. G., G. Celebi, and R. P. Fink (1970). "Olfaction in mammals: proliferation of cell in the olfactory epithelium and sensitivity to odors." In <u>Taste and Smell in Vertebrates.</u> CIBA Foundation Symposium. (edited by G. E. W. Wolstenholme and J. Knight) pp. 227-246. London. Churchill.
- Mozell, R. L. and F. A. McMorris (1991). "Insulin-like growth factor I stimulates oligodendrocyte development and myelination in rat brain aggregate cultures." J Neurosci Res 30(2): 382-90.
- Murray, R. C., D. Navi, J. Fesenko, A. D. Lander and A. L. Calof (2003). "Widespread defects in the primary olfactory pathway caused by loss of Mash1 function." J Neurosci 23(5): 1769-80.
- Myhr, K. M., G. Raknes, H. Nyland and C. Vedeler (1999). "Immunoglobulin G Fcreceptor (FcgammaR) IIA and IIIB polymorphisms related to disability in MS." <u>Neurology</u> 52(9): 1771-6.
- Nadim, W., P. N. Anderson and M. Turmaine (1990). "The role of Schwann cells and basal lamina tubes in the regeneration of axons through long lengths of freeze-killed nerve grafts." <u>Neuropathol Appl Neurobiol</u> 16(5): 411-21.
- Naef, R. and U. Suter (1998). "Many facets of the peripheral myelin protein PMP22 in myelination and disease." <u>Microsc Res Tech</u> 41(5): 359-71.
- Nait-Oumesmar, B., L. Decker, F. Lachapelle, V. Avellana-Adalid, C. Bachelin and A. B. Van Evercooren (1999). "Progenitor cells of the adult mouse subventricular zone proliferate, migrate and differentiate into oligodendrocytes after demyelination." <u>Eur J Neurosci</u> 11(12): 4357-66.
- Nait-Oumesmar, B., L. Vignais, E. Duhamel-Clerin, V. Avellana-Adalid, G. Rougon and A. Baron-Van Evercooren (1995). "Expression of the highly polysialated neural cell adhesion molecule during postnatal myelination of the adult spinal cord." <u>Eur J Neurosci</u> 7: 480-491.
- Namiki, J., A. Kojima and C. H. Tator (2000). "Effect of brain-derived neurotrophic factor, nerve growth factor, and neurotrophin-3 on functional recovery and regeneration after spinal cord injury in adult rats." J Neurotrauma 17(12): 1219-31.
- Nan, B., M. L. Getchell, J. V. Partin and T. V. Getchell (2001). "Leukemia inhibitory factor, interleukin-6, and their receptors are expressed transiently in the olfactory mucosa after target ablation." <u>J Comp Neurol</u> 435(1): 60-77.
- Nash, H. H., R. C. Borke and J. J. Anders (2002). "Ensheathing cells and methylprednisolone promote axonal regeneration and functional recovery in the lesioned adult rat spinal cord." <u>J Neurosci</u> 22(16): 7111-20.
- Navarro, X., A. Valero, G. Gudino, J. Fores, F. J. Rodriguez, E. Verdu, R. Pascual, J. Cuadras and M. Nieto-Sampedro (1999). "Ensheathing glia transplants promote

dorsal root regeneration and spinal reflex restitution after multiple lumbar rhizotomy." <u>Ann Neurol</u> 45(2): 207-15.

- Nave, K. A. and O. BOESPFLUG-TANGUY (1996). "X-linked developmental defects of myelination: from mouse mutants to human genetic diseases." <u>Neuroscientist</u> 2: 33–43.
- Nguyen, D., M. Hopfner, F. Zobel, U. Henke, H. Scherubl and M. Stangel (2003). "Rat oligodendroglial cell lines express a functional receptor for the chemokine CCL3 (macrophage inflammatory protein-1alpha)." <u>Neurosci Lett</u> 351(2): 71-74.
- Nguyen, D. and M. Stangel (2001). "Expression of the chemokine receptors CXCR1 and CXCR2 in at oligodendroglial cells." <u>Dev Brain Res</u> 128: 77–81.
- Niehaus, A., J. Shi, M. Grzenkowski, M. Diers-Fenger, J. Archelos, H. P. Hartung, K. Toyka, W. Bruck and J. Trotter (2000). "Patients with active relapsing-remitting multiple sclerosis synthesize antibodies recognizing oligodendrocyte progenitor cell surface protein: implications for remyelination." <u>Ann Neurol</u> 48(3): 362-71.
- Niehaus, A., J. Stegmuller, M. Diers-Fenger and J. Trotter (1999). "Cell-surface glycoprotein of oligodendrocyte progenitors involved in migration." J Neurosci 19(12): 4948-61.
- Nieto-Sampedro, M. (2003). "Central nervous system lesions that can and those that cannot be repaired with the help of olfactory bulb ensheathing cell transplants." <u>Neurochem Res</u> 28(11): 1659-76.
- Nieto-Sampedro, M., M. Manthrope, G. Barbin, S. Varon and C. W. Cotman (1983). "Injury-induced neuronotrophic activity in adult rat brain: correlation with survival of delayed implants in the wound cavity." J Neurosci 3(11): 2219-29.
- Nishiyama, A., A. Chang and B. D. Trapp (1999). "NG2+ glial cells: a novel glial cell population in the adult brain." J Neuropathol Exp Neurol 58(11): 1113-24.
- Nishiyama, A., X. H. Lin, N. Giese, C. H. Heldin and W. B. Stallcup (1996). "Colocalization of NG2 proteoglycan and PDGF alpha-receptor on O2A progenitor cells in the developing rat brain." <u>J Neurosci Res</u> 43(3): 299-314.
- Norenberg, M. D. (1994). "Astrocyte responses to CNS injury." <u>J Neuropathol Exp</u> <u>Neurol</u> 53(3): 213-20.
- Norton, W. T. (1981). "Biochemistry of myelin." Adv Neurol 31: 93-121.
- Nowacki, P., A. Potemkowski, T. Korwin-Piotrowska and D. Nocon (2000). "Morphometric analysis of axons in the minute multiple sclerosis lesions and shadow plaques in patients with multiple sclerosis." <u>Folia Neuropathol</u> 38(3): 104-110.
- Oksenberg, J. R., M. A. Panzara and L. Steinman (1993). "Multiple sclerosis: from immunogenetics to immunotherapy." J Neurol Sci 115 Suppl: S29-37.
- O'Leary, M. T. and W. F. Blakemore (1997). "Oligodendrocyte precursors survive poorly and do not migrate following transplantation into the normal adult central nervous system." J Neurosci Res 48(2): 159-167.
- Oleszak, E. L., J. Kuzmak, R. A. Good and C. D. Platsoucas (1995). "Immunology of Theiler's murine encephalomyelitis virus infection." <u>Immunol Res</u> 14(1): 13-33.
- Oley, N., R. S. DeHan, D. Tucker, J. C. Smith and P. P. Graziadei (1975). "Recovery of structure and function following transection of the primary olfactory nerves in pigeons." J Comp Physiol Psychol 88(2): 477-95.

- Opdenakker, G. (1997). "On the roles of extracellular matrix remodeling by gelatinase B." <u>Verh K Acad Geneeskd Belg</u> 59(6): 489-514.
- Orentas, D. M. and R. H. Miller (1996). "A novel form of migration of glial precursors." <u>Glia</u> 16(1): 27-39.
- Osterhout, D. J., A. Wolven, R. M. Wolf, M. D. Resh and M. V. Chao (1999). "Morphological differentiation of oligodendrocytes requires activation of Fyn tyrosine kinase." J Cell Biol 145(6): 1209-18.
- Owen, M. (1988). "Marrow stromal stem cells." J Cell Sci Suppl 10: 63-76.
- Owens, G. C. (1991). "A molecular approach to myelination using recombinant retroviruses." <u>Ann N Y Acad Sci</u> 633: 543-5.
- Ozden, S., C. Aubert, J. F. Bureau, D. Gonzalez-Dunia and M. Brahic (1993). "Analysis of proteolipid protein and P0 transcripts in mice infected with Theiler's virus." <u>Microb Pathog</u> 14(2): 123-31.
- Paez, P. M., C. B. Marta, M. B. Moreno, E. F. Soto and J. M. Pasquini (2002).
 "Apotransferrin decreases migration and enhances differentiation of oligodendroglial progenitor cells in an in vitro system" <u>Dev Neurosci</u> 24(1) 47-58.
- Palmer, T. D., J. Takahashi and F. H. Gage (1997). "The adult rat hippocampus contains primordial neural stem cells." <u>Mol Cell Neurosci</u> 8(6): 389-404.
- Pascual, J. I., G. Gudino-Cabrera, R. Insausti and M. Nieto-Sampedro (2002). "Spinal implants of olfactory ensheathing cells promote axon regeneration and bladder activity after bilateral lumbosacral dorsal rhizotomy in the adult rat." J Urol 167(3): 1522-6.
- Patel, P. I., B. B. Roa, A. A. Welcher, R. Schoener-Scott, B. J. Trask, L. Pentao, G. J. Snipes, C. A. Garcia, U. Francke, E. M. Shooter and et al. (1992). "The gene for the peripheral myelin protein PMP-22 is a candidate for Charcot-Marie-Tooth disease type 1A." <u>Nat Genet</u> 1(3): 159-65.
- Paty, D. W. and D. K. Li (1993). "Interferon beta-1b is effective in relapsing-remitting multiple sclerosis. II. MRI analysis results of a multicenter, randomized, doubleblind, placebo-controlled trial. UBC MS/MRI Study Group and the IFNB Multiple Sclerosis Study Group." <u>Neurology</u> 43(4): 662-7.
- Pekny, M., K. A. Stanness, C. Eliasson, C. Betsholtz and D. Janigro (1998). "Impaired induction of blood-brain barrier properties in aortic endothelial cells by astrocytes from GFAP-deficient mice." <u>Glia</u> 22(4): 390-400.
- Penderis, J., S. A. Shields and R. J. Franklin (2003). "Impaired remyelination and depletion of oligodendrocyte progenitors does not occur following repeated episodes of focal demyelination in the rat central nervous system." <u>Brain</u> 126(Pt 6): 1382-91.
- Perez-Bouza, A., C. B. Wigley, W. Nacimiento, J. Noth and G. A. Brook (1998).
 "Spontaneous orientation of transplanted olfactory glia influences axonal regeneration." <u>Neuroreport</u> 9(13): 2971-5.
- Perry, V. H., M. D. Bell, H. C. Brown and M. K. Matyszak (1995). "Inflammation in the nervous system." <u>Curr Opin Neurobiol</u> 5(5): 636-41.
- Peters, A., L. S. Palay and H. deF. Webster (1976). "The fine structure of the nervous system : the neurons and supporting cells." W. B. Saunders Company, Philadelphia.

- Peters, A., L. S. Palay and H. deF. Webster (1991). "The Fine Structure of the Nervous System: the Neuron and the Supporting Cells." Oxford, UK: Oxford Univ. Press.
- Peterson, J. W., L. Bo, S. Mork, A. Chang and B. D. Trapp (2001). "Transected neurites, apoptotic neurons, and reduced inflammation in cortical multiple sclerosis lesions." <u>Ann Neurol</u> 50(3): 389-400.
- Peyron, F., S. Timsit, J. L. Thomas, T. Kagawa, K. Ikenaka and B. Zalc (1997). "In situ expression of PLP/DM-20, MBP, and CNP during embryonic and postnatal development of the jimpy mutant and of transgenic mice overexpressing PLP." J <u>Neurosci Res</u> 50(2): 190-201.
- Pfeiffer, S. E., A. E. Warrington and R. Bansal (1993). "The oligodendrocyte and its many cellular processes." <u>Trends Cell Biol</u> 3(6): 191-7.
- Pham-Dinh, D., M. C. Birling, G. Roussel, A. Dautigny and J. L. Nussbaum (1991). "Proteolipid DM-20 predominates over PLP in peripheral nervous system." <u>Neuroreport</u> 2(2): 89-92.
- Pham-Dinh, D., M. G. Mattei, J. L. Nussbaum, G. Roussel, P. Pontarotti, N. Roeckel, I. H. Mather, K. Artzt, K. F. Lindahl and A. Dautigny (1993).
 "Myelin/oligodendrocyte glycoprotein is a member of a subset of the immunoglobulin superfamily encoded within the major histocompatibility complex." <u>Proc Natl Acad Sci U S A</u> 90(17): 7990-4.
- Phinney, D. G., G. Kopen, R. L. Isaacson and D. J. Prockop (1999). "Plastic adherent stromal cells from the bone marrow of commonly used strains of inbred mice: variations in yield, growth, and differentiation." J Cell Biochem 72(4): 570-85.
- Piani, D., K. Frei, K. Q. Do, M. Cuenod and A. Fontana (1991). "Murine brain macrophages induce NMDA receptor mediated neurotoxicity in vitro by secreting glutamate." <u>Neurosci Lett</u> 133: 159-162.
- Piddlesden, S. J. and B. P. Morgan (1993). "Killing of rat glial cells by complement: deficiency of the rat analogue of CD59 is the cause of oligodendrocyte susceptibility to lysis." J Neuroimmunol 48(2): 169-75.
- Pierchala, B. A., R. C. Ahrens, A. J. Paden and E. M. Johnson, Jr. (2004). "Nerve growth factor promotes the survival of sympathetic neurons through the cooperative function of the protein kinase C and phosphatidylinositol 3-kinase pathways." J Biol Chem 279(27): 27986-93.
- Pinching, A. J. and T. P. Powell (1971). "The neuropil of the periglomerular region of the olfactory bulb." J Cell Sci 9(2): 379-409.
- Pitt, D., P. Werner and C. S. Raine (2000). "Glutamate excitotoxicity in a model of multiple sclerosis." <u>Nat Med</u> 6(1): 67-70.
- Pixley, S. K. (1992). "The olfactory nerve contains two populations of glia, identified both in vivo and in vitro." <u>Glia</u> 5(4): 269-84.
- Pixley, S. K. and J. de Vellis (1984). "Transition between immature radial glia and mature astrocytes studied with a monoclonal antibody to vimentin." <u>Brain Res</u> 317(2): 201-9.
- Pixley, S. K. and R. Y. Pun (1990). "Cultured rat olfactory neurons are excitable and respond to odors." <u>Brain Res Dev Brain Res</u> 53(1): 125-30.
- Plant, G. W., C. L. Christensen, M. Oudega and M. B. Bunge (2003). "Delayed transplantation of olfactory ensheathing glia promotes sparing/regeneration of

supraspinal axons in the contused adult rat spinal cord." <u>J Neurotrauma</u> 20(1): 1-16.

- Plant, G. W., P. F. Currier, E. P. Cuervo, M. L. Bates, Y. Pressman, M. B. Bunge and P. M. Wood (2002). "Purified adult ensheathing glia fail to myelinate axons under culture conditions that enable Schwann cells to form myelin." <u>J Neurosci</u> 22(14): 6083-91.
- Pleasure, D., B. Kreider, S. Shuman and G. Sobue (1985). "Tissue culture studies of Schwann cell proliferation and differentiation." Dev Neurosci 7(5-6): 364-73.
- Pluchino, S., A. Quattrini, E. Brambilla, A. Gritti, G. Salani, G. Dina, R. Galli, U. Del Carro, S. Amadio, A. Bergami, R. Furlan, G. Comi, A. L. Vescovi and G. Martino (2003). "Injection of adult neurospheres induces recovery in a chronic model of multiple sclerosis." Nature 422(6933): 688-94.
- Poduslo, S. E., C. H. Pak and K. Miller (1990). "Hydrocortisone induction during oligodendroglial differentiation." <u>Neurosci Lett</u> 113(1): 84-8.
- Prewitt, C. M. F., I. R. Niesman, C. J. M. Kane and J. D. Houle (1997). "Activated macrophage/microglial cells can promote the regeneration of sensory axons into the injured spinal cord." <u>Exp Neurol</u> 148: 433–443.
- Price, J. L. and T. P. Powell (1971). "Certain observations on the olfactory pathway." J Anat 110(1): 105-26.
- Price, J., D. Turner, and C. Cepko (1987). "Lineage analysis in the vertebrate nervous system by retrovirus-mediated gene transfer." <u>PNAS</u> 84: 156-160.
- Prieto, M., N. Chauvet and G. Alonso (2000). "Tanycytes transplanted into the adult rat spinal cord support the regeneration of lesioned axons." <u>Exp Neurol</u> 161(1): 27-37.
- Prineas, J. W., E. E. Kwon, E. S. Cho, L. R. Sharer, M. H. Barnett, E. L. Oleszak, B. Hoffman and B. P. Morgan (2001). "Immunopathology of secondaryprogressive multiple sclerosis." <u>Ann Neurol</u> 50(5): 646-57.
- Pringle, N. P. and W. D. Richardson (1993). "A singularity of PDGF alpha-receptor expression in the dorsoventral axis of the neural tube may define the origin of the oligodendrocyte lineage." <u>Development</u> 117(2): 525-33.
- Privat, A., C. Jacque, J. M. Bourre, P. Dupouey and N. Baumann (1979). "Absence of the major dense line in myelin of the mutant mouse "shiverer"." <u>Neurosci Lett</u> 12(1): 107-12.
- Probert, L. and K. Akassoglou (2001). "Glial expression of tumor necrosis factor in transgenic animals: how do these models reflect the "normal situation"?" <u>Glia</u> 36(2): 212-9.
- Probert, L., K. Akassoglou, M. Pasparakis, G. Kontogeorgos and G. Kollias (1995). "Spontaneous inflammatory demyelinating disease in transgenic mice showing central nervous system-specific expression of tumor necrosis factor alpha." <u>Proc Natl Acad Sci U S A</u> 92(24): 11294-8.
- Prockop, D. J. (1997). "Marrow stromal cells as stem cells for nonhematopoietic tissues." <u>Science</u> 276(5309): 71-4.
- Quarles, R. H. (1997). "Glycoproteins of myelin sheaths." J Mol Neurosci 8(1): 1-12.
- Rabchevsky, A.G. and W.J. Streit (1997). 'Grafting of cultured microglial cells into the lesioned spinal cord of adult rats enhances neurite out-growth." J Neurosci Res 47: 34–48.

- Rabchevsky, A. G. and W. J. Streit (1998). "Role of microglia in post-injury repair and regeneration of the CNS." <u>Ment Retard and Dev Disabil Res Rev</u> 4:187-192.
- Radtke, C., Y. Akiyama, J. Brokaw, K. L. Lankford, K. Wewetzer, W. L. Fodor and J. D. Kocsis (2004). "Remyelination of the nonhuman primate spinal cord by transplantation of H-transferase transgenic adult pig olfactory ensheathing cells." <u>Faseb J</u> 18(2): 335-7.
- Raff, M.C., K. L. Fields, S. Hakamori, R. Mirsky, R.M. Pruss, and J. Winter (1979). "Cell-type-specific markers for distinguishing and studying neurons and the major classes of glial cells in culture." <u>Brain Res</u> 174: 283–308.
- Raff, M. C., R. Mirsky, K. L. Fields, R. P. Lisak, S. H. Dorfman, D. H. Silberberg, N. A. Gregson, S. Leibowitz and M. C. Kennedy (1978). "Galactocerebroside is a specific cell-surface antigenic marker for oligodendrocytes in culture." <u>Nature</u> 274(5673): 813-6.
- Rafols, J. A. and T. V. Getchell (1983). "Morphological relations between the receptor neurons, sustentacular cells and Schwann cells in the olfactory mucosa of the salamander." <u>Anat Rec</u> 206(1): 87-101.
- Raine, C. S. (1984). "Biology of disease. Analysis of autoimmune demyelination: its impact upon multiple sclerosis." Lab Invest 50(6): 608-35.
- Raine, C. S., D. H. Snyder, M. P. Valsamis and S. H. Stone (1974). "Chronic experimental allergic encephalomyelitis in inbred guinea pigs. An ultrastructural study." <u>Lab Invest</u> 31(4): 369-80.
- Raine, C. S. and S. H. Stone (1977). "Animal model for multiple sclerosis. Chronic experimental allergic encephalomyelitis in inbred guinea pigs." <u>N Y State J Med</u> 77(11): 1693-6.
- Raine, C. S. and U. Traugott (1983). "Chronic relapsing experimental autoimmune encephalomyelitis. Ultrastructure of the central nervous system of animals treated with combinations of myelin components." Lab Invest 48(3): 275-84.
- Raine, C. S. and U. Traugott (1984). "Experimental autoimmune demyelination. Chronic relapsing models and their therapeutic implications for multiple sclerosis." <u>Ann N Y Acad Sci</u> 436: 33-51.
- Raisman, G. (1985). "Specialized neuroglial arrangement may explain the capacity of vomeronasal axons to reinnervate central neurons." <u>Neuroscience</u> 14(1): 237-54.
- Raisman, G. (2001). "Olfactory ensheathing cells another miracle cure for spinal cord injury?" <u>Nat Rev Neurosci</u> 2(5): 369-75.
- Raisman, G., J. M. Lawrence and G. A. Brook (1993). "Schwann cells transplanted into the CNS." Int J Dev Neurosci 11(5): 651-69.
- Raivich, G., H. Bluethmann and G. W. Kreutzberg (1996). "Signaling molecules and neuroglial activation in the injured central nervous system." <u>Keio J Med</u> 45(3): 239-47.
- Raivich, G., M. Reddington, C. A. Haas and G. W. Kreutzberg (1995). "Peptides in motoneurons." Prog Brain Res 104: 3-20.
- Ramer, L. M., M. W. Richter, A. J. Roskams, W. Tetzlaff and M. S. Ramer (2004). "Peripherally-derived olfactory ensheathing cells do not promote primary afferent regeneration following dorsal root injury." Glia 47(2): 189-206.
- Ramon-Cueto, A. and J. Avila (1998). "Olfactory ensheathing glia: properties and function." <u>Brain Res Bull</u> 46(3): 175-87.

- Ramon-Cueto, A., M. I. Cordero, F. F. Santos-Benito and J. Avila (2000). "Functional recovery of paraplegic rats and motor axon regeneration in their spinal cords by olfactory ensheathing glia." <u>Neuron</u> 25(2): 425-35.
- Ramon-Cueto, A. and M. Nieto-Sampedro (1992). "Glial cells from adult rat olfactory bulb: immunocytochemical properties of pure cultures of ensheathing cells." <u>Neuroscience</u> 47(1): 213-20.
- Ramon-Cueto, A. and M. Nieto-Sampedro (1994). "Regeneration into the spinal cord of transected dorsal root axons is promoted by ensheathing glia transplants." <u>Exp</u> <u>Neurol</u> 127(2): 232-44.
- Ramon-Cueto, A., G. W. Plant, J. Avila and M. B. Bunge (1998). "Long-distance axonal regeneration in the transected adult rat spinal cord is promoted by olfactory ensheathing glia transplants." J Neurosci 18(10): 3803-15.
- Ramon-Cueto, A., J. Perez and M. Nieto-Sampedro (1993). "In vitro enfolding of olfactory neurites by p75 NGF receptor positive ensheathing cells from adult rat olfactory bulb." <u>Eur J Neurosci</u> 5(9): 1172-80.
- Ramon-Cueto, A. and F. Valverde (1995). "Olfactory bulb ensheathing glia: a unique cell type with axonal growth-promoting properties." <u>Glia</u> 14(3): 163-73.
- Rebiere, A. and J. Dainat (1981). "Quantitative study of synapse formation in the duck olfactory bulb." J Comp Neurol 203(1): 103-20.
- Reier, P. J., B. S. Bregman and J. R. Wujek (1986). "Intraspinal transplantation of embryonic spinal cord tissue in neonatal and adult rats." <u>J Comp Neurol</u> 247(3): 275-96.
- Reier,P.J., L.F. Eng, and L. Jakeman (1989). 'Reactive astrocyte and axonal outgrowth in the injured CNS: Is gliosis really an impediment to regeneration?" In: <u>Neural</u> <u>Regeneration and Transplantation</u>. F.J. Seil, ed. Alan R. Liss, New York, pp.183-209.
- Reier, P. J. and J. D. Houle (1988). "The glial scar: its bearing on axonal elongation and transplantation approaches to CNS repair." <u>Adv Neurol</u> 47: 87-138.
- Reindl, M., C. Linington, U. Brehm, R. Egg, E. Dilitz, F. Deisenhammer, W. Poewe and T. Berger (1999). "Antibodies against the myelin oligodendrocyte glycoprotein and the myelin basic protein in multiple sclerosis and other neurological diseases: a comparative study." <u>Brain</u> 122 (Pt 11): 2047-56.
- Reinhard, E., R. Meier, W. Halfter, G. Rovelli and D. Monard (1988). "Detection of glia-derived nexin in the olfactory system of the rat." <u>Neuron</u> 1(5): 387-94.
- Reynolds, R., I. Cenci di Bello, M. Dawson and J. Levine (2001). "The response of adult oligodendrocyte progenitors to demyelination in EAE." <u>Prog Brain Res</u> 132: 165-174.
- Reynolds, B. A. and S. Weiss (1992). "Generation of neurons and astrocytes from isolated cells of the adult mammalian central nervous system." <u>Science</u> 255(5052): 1707-10.
- Richardson, W. D., H. K. Smith, T. Sun, N. P. Pringle, A. Hall and R. Woodruff (2000). "Oligodendrocyte lineage and the motor neuron connection." <u>Glia</u> 29(2): 136-42.
- Riddell, J. S., M. Enriquez-Denton, A. Toft, R. Fairless and S. C. Barnett (2004).
 "Olfactory ensheathing cell grafts have minimal influence on regeneration at the dorsal root entry zone following rhizotomy." <u>Glia</u> 47(2): 150-67.

- Ridet, J. L., S. K. Malhotra, A. Privat and F. H. Gage (1997). "Reactive astrocytes: cellular and molecular cues to biological function." <u>Trends Neurosci</u> 20(12): 570-7.
- Rieckmann, P., B. Altenhofen, A. Riegel, B. Kallmann and K. Felgenhauer (1998).
 "Correlation of soluble adhesion molecules in blood and cerebrospinal fluid with magnetic resonance imaging activity in patients with multiple sclerosis." <u>Mult</u> Scler 4(3): 178-82.
- Righi, M., L. Mori, G. De Libero, M. Sironi, A. Biondi, A. Mantovani, S. D. Donini and P. Ricciardi-Castagnoli (1989). "Monokine production by microglial cell clones." <u>Eur J Immunol</u> 19: 1443–1448.
- Ritchie, J. M. (1984). "Physiological basis of cinduction in myelinated nerve fibers." In P. Morell (ed.), Myelin, 2nd ed. New York: Plenum, pp. 117-141.
- Roa, B. B., P. J. Dyck, H. G. Marks, P. F. Chance and J. R. Lupski (1993). "Dejerine-Sottas syndrome associated with point mutation in the peripheral myelin protein 22 (PMP22) gene." <u>Nat Genet</u> 5(3): 269-73.
- Roach, A., N. Takahashi, D. Pravtcheva, F. Ruddle and L. Hood (1985). "Chromosomal mapping of mouse myelin basic protein gene and structure and transcription of the partially deleted gene in shiverer mutant mice." <u>Cell</u> 42(1): 149-55.
- Rodriguez, M., J. J. Kenny, R. L. Thiemann and G. E. Woloschak (1990). "Theiler's virus-induced demyelination in mice immunosuppressed with anti-IgM and in mice expressing the xid gene." <u>Microb Pathog</u> 8(1): 23-35.
- Rodriguez, M., C. F. Lucchinetti, R. J. Clark, T. L. Yakash, H. Markowitz and V. A. Lennon (1988). "Immunoglobulins and complement in demyelination induced in mice by Theiler's virus." <u>J Immunol</u> 140(3): 800-6.
- Rosen, C. L., R. P. Bunge, M. D. Ard and P. M. Wood (1989). "Type 1 astrocytes inhibit myelination by adult rat oligodendrocytes in vitro." J Neurosci 9(10): 3371-9.
- Rudick, R. A., D. E. Goodkin, L. D. Jacobs, D. L. Cookfair, R. M. Herndon, J. R.
 Richert, A. M. Salazar, J. S. Fischer, C. V. Granger, J. H. Simon, J. J. Alam, N.
 A. Simonian, M. K. Campion, D. M. Bartoszak, D. N. Bourdette, J. Braiman, C.
 M. Brownscheidle, M. E. Coats, S. L. Cohan, D. S. Dougherty, R. P. Kinkel, M.
 K. Mass, F. E. Munschauer, R. L. Priore, R. H. Whitham and et al. (1997).
 "Impact of interferon beta-1a on neurologic disability in relapsing multiple sclerosis. The Multiple Sclerosis Collaborative Research Group (MSCRG)."
 Neurology 49(2): 358-63.
- Rudick, R. A., R. M. Ransohoff, R. Peppler, S. VanderBrug Medendorp, P. Lehmann and J. Alam (1996). "Interferon beta induces interleukin-10 expression: relevance to multiple sclerosis." <u>Ann Neurol</u> 40(4): 618-27.
- Ruitenberg, M. J., G. W. Plant, C. L. Christensen, B. Blits, S. P. Niclou, A. R. Harvey, G. J. Boer and J. Verhaagen (2002). "Viral vector-mediated gene expression in olfactory ensheathing glia implants in the lesioned rat spinal cord." <u>Gene Ther</u> 9(2): 135-46.
- Ruitenberg, M. J., G. W. Plant, F. P. Hamers, J. Wortel, B. Blits, P. A. Dijkhuizen, W. H. Gispen, G. J. Boer and J. Verhaagen (2003). "Ex vivo adenoviral vector-mediated neurotrophin gene transfer to olfactory ensheathing glia: effects on

rubrospinal tract regeneration, lesion size, and functional recovery after implantation in the injured rat spinal cord." <u>J Neurosci</u> 23(18): 7045-58.

- Ruoslahti, E. and J. C. Reed (1994). "Anchorage dependence, integrins, and apoptosis." <u>Cell</u> 77: 477–478.
- Rush, R. A. (1984). "Immunohistochemical localization of endogenous nerve growth factor." <u>Nature</u> 312(5992): 364-7.
- Rutkowski, J. L., C. J. Kirk, M. A. Lerner and G. I. Tennekoon (1995). "Purification and expansion of human Schwann cells in vitro." <u>Nat Med</u> 1(1): 80-3.
- Ruuls, S. R., M. C. de Labie, K. S. Weber, C. A. Botman, R. J. Groenestein, C. D. Dijkstra, T. Olsson and P. H. van der Meide (1996). "The length of treatment determines whether IFN-beta prevents or aggravates experimental autoimmune encephalomyelitis in Lewis rats." J Immunol 157(12): 5721-31.
- Sadovnick, A. D., H. Armstrong, G. P. Rice, D. Bulman, L. Hashimoto, D. W. Paty, S. A. Hashimoto, S. Warren, W. Hader, T. J. Murray and et al. (1993). "A population-based study of multiple sclerosis in twins: update." <u>Ann Neurol</u> 33(3): 281-5.
- Sailer, M., B. Fischl, D. Salat, C. Tempelmann, M. A. Schonfeld, E. Busa, N. Bodammer, H. J. Heinze and A. Dale (2003). "Focal thinning of the cerebral cortex in multiple sclerosis." <u>Brain</u> 126(Pt 8): 1734-44.
- Salzer, J. L., W. P. Holmes and D. R. Colman (1987). "The amino acid sequences of the myelin-associated glycoproteins: homology to the immunoglobulin gene superfamily." J Cell Biol 104(4): 957-65.
- Sarlieve, L. L., M. Fabre, J. Susz and J. M. Matthieu (1983). "Investigations on myelination in vitro: IV. "Myelin-like" or premyelin structures in cultures of dissociated brain cells from 14--15-day-old embryonic mice." <u>J Neurosci Res</u> 10(2): 191-210.
- Sasaki, M., O. Honmou, Y. Akiyama, T. Uede, K. Hashi and J. D. Kocsis (2001).
 "Transplantation of an acutely isolated bone marrow fraction repairs demyelinated adult rat spinal cord axons." <u>Glia</u> 35(1): 26-34.
- Satoh, M. and M. Takeuchi (1995). "Induction of NCAM expression in mouse olfactory keratin-positive basal cells in vitro." <u>Brain Res Dev Brain Res</u> 87(2): 111-9.
- Saxe, D. F., N. Takahashi, L. Hood and M. I. Simon (1985). "Localization of the human myelin basic protein gene (MBP) to region 18q22----qter by in situ hybridization." <u>Cytogenet Cell Genet</u> 39(4): 246-9.
- Scherer, S. S. (1997). "The biology and pathobiology of Schwann cells." <u>Curr Opin</u> <u>Neurol</u> 10(5): 386-97.
- Scherer, S. S. (1997). "Molecular genetics of demyelination: new wrinkles on an old membrane." <u>Neuron</u> 18(1): 13-6.
- Scherer, S. S., S. M. Deschenes, Y. T. Xu, J. B. Grinspan, K. H. Fischbeck and D. L. Paul (1995). "Connexin32 is a myelin-related protein in the PNS and CNS." <u>J</u> <u>Neurosci</u> 15(12): 8281-94.
- Scherer, S. S., Y. T. Xu, D. Roling, L. Wrabetz, M. L. Feltri and J. Kamholz (1994).
 "Expression of growth-associated protein-43 kD in Schwann cells is regulated by axon-Schwann cell interactions and cAMP." J Neurosci Res 38(5): 575-89.

- Schiffer, D., M. T. Giordana, A. Mauro and A. Migheli (1988). "Reactive astrocytes in the morphologic composition of peripheral areas of gliomas." <u>Tumori</u> 74(4): 411-20.
- Schmidtmayer, J., C. Jacobsen, G. Miksch and J. Sievers (1994). "Blood monocytes and spleen macrophages differentiate into microglia-like cells on monolayers of astrocytes: membrane currents." <u>Glia</u> 12(4): 259-67.
- Schneider, S., F. Bosse, D. D'Urso, H. Muller, M. W. Sereda, K. Nave, A. Niehaus, T. Kempf, M. Schnolzer and J. Trotter (2001). "The AN2 protein is a novel marker for the Schwann cell lineage expressed by immature and nonmyelinating Schwann cells." J Neurosci 21(3): 920-33.
- Schneider-Schaulies, J., A. von Brunn and M. Schachner (1990). "Recombinant peripheral myelin protein P0 confers both adhesion and neurite outgrowth-promoting properties." J Neurosci Res 27(3): 286-97.
- Schnitzer, J., W. W. Franke and M. Schachner (1981). "Immunocytochemical demonstration of vimentin in astrocytes and ependymal cells of developing and adult mouse nervous system." J Cell Biol 90(2): 435-47.
- Schoenfeld, T. A., L. McKerracher, R. Obar and R. B. Vallee (1989). "MAP 1A and MAP 1B are structurally related microtubule associated proteins with distinct developmental patterns in the CNS." J Neurosci 9(5): 1712-30.
- Schrijver, H. M., J. B. Crusius, B. M. Uitdehaag, M. A. Garcia Gonzalez, P. J. Kostense, C. H. Polman and A. S. Pena (1999). "Association of interleukinlbeta and interleukin-1 receptor antagonist genes with disease severity in MS." <u>Neurology</u> 52(3): 595-9.
- Schwab, M. E. (1990). "Myelin-associated inhibitors of neurite growth and regeneration in the CNS." <u>Trends Neurosci</u> 13(11): 452-6.
- Schwob, J. E. (2002). "Neural regeneration and the peripheral olfactory system." <u>Anat</u> <u>Rec</u> 269(1): 33-49.
- Schwob, J. E., N. B. Farber and D. I. Gottlieb (1986). "Neurons of the olfactory epithelium in adult rats contain vimentin." J Neurosci 6(1): 208-17.
- Schwob, J. E., J. M. Huard, M. B. Luskin and S. L. Youngentob (1994). "Retroviral lineage studies of the rat olfactory epithelium." Chem Senses 19(6): 671-82.
- Scotti, A. L., M. C. Hoffmann and C. Nitsch (1994). "The neurite growth promoting protease nexin 1 in glial cells of the olfactory bulb of the gerbil: an ultrastructural study." <u>Cell Tissue Res</u> 278(2): 409-13.
- Seilheimer, B. and M. Schachner (1987). "Regulation of neural cell adhesion molecule expression on cultured mouse Schwann cells by nerve growth factor." <u>Embo J</u> 6(6): 1611-6.
- Seitelberger, F. (1995). "Neuropathology and genetics of Pelizaeus-Merzbacher disease." <u>Brain Pathol</u> 5(3): 267-73.
- Selmaj, K. W. and C. S. Raine (1988). "Tumor necrosis factor mediates myelin and oligodendrocyte damage in vitro." <u>Ann Neurol</u> 23: 339–346.
- Selmaj, K., C. S. Raine, B. Cannella and C. F. Brosnan (1991). "Identification of lymphotoxin and tumor necrosis factor in multiple sclerosis lesions." <u>J Clin</u> <u>Invest</u> 87(3): 949-54.
- Selmaj, K. W., M. Farooq, W. T. Norton, C. S. Raine and C. F. Brosnan (1990).
 "Proliferation of astrocytes in vitro in response to cytokines. A primary role for tumor necrosis factor." J Immunol 144(1): 129-35.
- Shah, N. M., A. K. Groves and D. J. Anderson (1996). "Alternative neural crest cell fates are instructively promoted by TGFbeta superfamily members." <u>Cell</u> 85(3): 331-43.
- Shah, N. M., M. A. Marchionni, I. Isaacs, P. Stroobant and D. J. Anderson (1994).
 "Glial growth factor restricts mammalian neural crest stem cells to a glial fate." <u>Cell</u> 77(3): 349-60.
- Shi, R. and A. R. Blight (1996). "Compression injury of mammalian spinal cord in vitro and the dynamics of action potential conduction failure." J Neurophysiol 76(3): 1572-80.
- Shields, S. A., W. F. Blakemore and R. J. Franklin (2000). "Schwann cell remyelination is restricted to astrocyte-deficient areas after transplantation into demyelinated adult rat brain." <u>J Neurosci Res</u> 60(5): 571-578.
- Shy, M. E., G. Hobson, M. Jain, O. Boespflug-Tanguy, J. Garbern, K. Sperle, W. Li, A. Gow, D. Rodriguez, E. Bertini, P. Mancias, K. Krajewski, R. Lewis and J. Kamholz (2003). "Schwann cell expression of PLP1 but not DM20 is necessary to prevent neuropathy." <u>Ann Neurol</u> 53(3): 354-65.
- Shy, M. E., M. Tani, Y. J. Shi, S. A. Whyatt, T. Chbihi, S. S. Scherer and J. Kamholz (1995). "An adenoviral vector can transfer lacZ expression into Schwann cells in culture and in sciatic nerve." <u>Ann Neurol</u> 38(3): 429-36.
- Sim, F. J., C. Zhao, J. Penderis and R. J. Franklin (2002). "The age-related decrease in CNS remyelination efficiency is attributable to an impairment of both oligodendrocyte progenitor recruitment and differentiation." <u>J Neurosci</u> 22(7): 2451-9.
- Sims, T. J. and S. A. Gilmore (1983). "Interactions between intraspinal Schwann cells and the cellular constituents normally occurring in the spinal cord: an ultrastructural study in the irradiated rat." <u>Brain Res</u> 276(1): 17-30.
- Sims, T. J., S. A. Gilmore, S. G. Waxman and E. Klinge (1985). "Dorsal-ventral differences in the glia limitans of the spinal cord: an ultrastructural study in developing normal and irradiated rats." <u>J Neuropathol Exp Neurol</u> 44(4)415-429.
- Simmons, P. A. and T. V. Getchell (1981). "Neurogenesis in olfactory epithelium: loss and recovery of transpithelial voltage transients following olfactory nerve section." J Neurophysiol 45(3): 516-28.
- Simmons, P. A., J. A. Rafols and T. V. Getchell (1981). "Ultrastructural changes in olfactory receptor neurons following olfactory nerve section." <u>J Comp Neurol</u> 197(2): 237-57.
- Simon, J. H., L. D. Jacobs, M. Campion, K. Wende, N. Simonian, D. L. Cookfair, R. A. Rudick, R. M. Herndon, J. R. Richert, A. M. Salazar, J. J. Alam, J. S. Fischer, D. E. Goodkin, C. V. Granger, M. Lajaunie, A. L. Martens-Davidson, M. Meyer, J. Sheeder, K. Choi, A. L. Scherzinger, D. M. Bartoszak, D. N. Bourdette, J. Braiman, C. M. Brownscheidle, R. H. Whitham and et al. (1998). "Magnetic resonance studies of intramuscular interferon beta-1a for relapsing multiple

sclerosis. The Multiple Sclerosis Collaborative Research Group." <u>Ann Neurol</u> 43(1): 79-87.

- Smale, K. A., R. Doucette and M. D. Kawaja (1996). "Implantation of olfactory ensheathing cells in the adult rat brain following fimbria-fornix transection." <u>Exp Neurol</u> 137(2): 225-33.
- Small, R. K., P. Riddle and M. Noble (1987). "Evidence for migration of oligodendrocyte--type-2 astrocyte progenitor cells into the developing rat optic nerve." <u>Nature</u> 328(6126): 155-157.
- Smith, K. J., R. Kapoor, S. M. Hall and M. Davies (2001). "Electrically active axons degenerate when exposed to nitric oxide." <u>Ann Neurol</u> 49(4): 470-6.
- Smith, P. M., A. Lakatos, S. C. Barnett, N. D. Jeffery and R. J. Franklin (2002).
 "Cryopreserved cells isolated from the adult canine olfactory bulb are capable of extensive remyelination following transplantation into the adult rat CNS." <u>Exp</u> <u>Neurol</u> 176(2): 402-6.
- Smith, P. M., F. J. Sim, S. C. Barnett and R. J. Franklin (2001). "SCIP/Oct-6, Krox-20, and desert hedgehog mRNA expression during CNS remyelination by transplanted olfactory ensheathing cells." <u>Glia</u> 36(3): 342-53.
- Smith, R., B. A. Cornell, M. A. Keniry and F. Separovic (1983). "31P nuclear magnetic resonance studies of the association of basic proteins with multilayers of diacyl phosphatidylserine." <u>Biochim Biophys Acta</u> 732(3): 492-8.
- Smith, T., A. Groom, B. Zhu and L. Turski (2000). "Autoimmune encephalomyelitis ameliorated by AMPA antagonists." <u>Nat Med</u> 6(1): 62-6.
- Snipes, G. J., U. Suter, A. A. Welcher and E. M. Shooter (1992). "Characterization of a novel peripheral nervous system myelin protein (PMP-22/SR13)." <u>J Cell Biol</u> 117(1): 225-38.
- Solly, S. K., J. L. Thomas, M. Monge, C. Demerens, C. Lubetzki, M. V. Gardinier, J. M. Matthieu and B. Zalc (1996). "Myelin/oligodendrocyte glycoprotein (MOG) expression is associated with myelin deposition." <u>Glia</u> 18(1): 39-48.
- Southard-Smith, E. M., L. Kos and W. J. Pavan (1998). "Sox10 mutation disrupts neural crest development in Dom Hirschsprung mouse model." <u>Nat Genet</u> 18(1): 60-4.
- Sparkes, R. S., T. Mohandas, C. Heinzmann, H. J. Roth, I. Klisak and A. T. Campagnoni (1987). "Assignment of the myelin basic protein gene to human chromosome 18q22-qter." <u>Hum Genet</u> 75(2): 147-50.
- Spassky, N., F. de Castro, B. Le Bras, K. Heydon, F. Queraud-LeSaux, E. Bloch-Gallego, A. Chedotal, B. Zalc and J. L. Thomas (2002). "Directional guidance of oligodendroglial migration by class 3 semaphorins and netrin-1." <u>J Neurosci</u> 22(14): 5992-6004.
- Spassky, N., C. Olivier, E. Perez-Villegas, C. Goujet-Zalc, S. Martinez, J. Thomas and B. Zalc (2000). "Single or multiple oligodendroglial lineages: a controversy." <u>Glia</u> 29(2): 143-8.
- Sriram, S. and M. Rodriguez (1997). "Indictment of the microglia as the villain in multiple sclerosis." <u>Neurology</u> 48(2): 464-70.
- Sriram, S., C. W. Stratton, S. Yao, A. Tharp, L. Ding, J. D. Bannan and W. M. Mitchell (1999). "Chlamydia pneumoniae infection of the central nervous system in multiple sclerosis." <u>Ann Neurol</u> 46(1): 6-14.

- Stangel, M., A. Compston and N. J. Scolding (1999). "Polyclonal immunoglobulins for intravenous use do not influence the behaviour of cultured oligodendrocytes." <u>J</u> <u>Neuroimmunol 96</u>: 228–233.
- Steer, J. M. (1971). "Some observations on the fine structure of rat dorsal spinal nerve roots." J Anat 109(3): 467-85.
- Steinman, L. (1993). "Connections between the immune system and the nervous system." Proc Natl Acad Sci U S A 90(17): 7912-4.
- Steinman, L. (1995). "Multiple sclerosis. Presenting an odd autoantigen." <u>Nature</u> 375(6534): 739-40.
- Steinman, L. (1996). "Multiple sclerosis: a coordinated immunological attack against myelin in the central nervous system." <u>Cell</u> 85(3): 299-302.
- Steinman, L. (1999). "Assessment of animal models for MS and demyelinating disease in the design of rational therapy." <u>Neuron</u> 24(3): 511-4.
- Steinman, L. (2001a). "Myelin-specific CD8 T cells in the pathogenesis of experimental allergic encephalitis and multiple sclerosis." J Exp Med 194(5): F27-30.
- Steinman, L. (2001b). "Multiple sclerosis: a two-stage disease." <u>Nat Immunol</u> 2(9): 762-764.
- Steinman, L., R. Martin, C. Bernard, P. Conlon and J. R. Oksenberg (2002). "Multiple sclerosis: deeper understanding of its pathogenesis reveals new targets for therapy." <u>Annu Rev Neurosci</u> 25: 491-505.
- Stemple, D. L. and D. J. Anderson (1992). "Isolation of a stem cell for neurons and glia from the mammalian neural crest." <u>Cell</u> 71(6): 973-85.
- Stichel, C. C., S. Hermanns, H. J. Luhmann, F. Lausberg, H. Niermann, D. D'Urso, G. Servos, H. G. Hartwig and H. W. Muller (1999). "Inhibition of collagen IV deposition promotes regeneration of injured CNS axons." <u>Eur J Neurosci</u> 11(2): 632-46.
- Stidworthy, M. F., S. Genoud, U. Suter, N. Mantei and R. J. Franklin (2003).
 "Quantifying the early stages of remyelination following cuprizone-induced demyelination." <u>Brain Pathol</u> 13(3): 329-39.
- Stone, S. H. and E. M. Lerner, 2nd (1965). "Chronic Disseminated Allergic Encephalomyelitis in Guinea Pigs." <u>Ann N Y Acad Sci</u> 122: 227-41.
- Stout, R. P. and P. P. Graziadei (1980). "Influence of the olfactory placode on the development of the brain in Xenopus laevis (Daudin). I. Axonal growth and connections of the transplanted olfactory placode." <u>Neuroscience</u> 5(12): 2175-86.
- Streit, W. J. and G. W. Kreutzberg (1988). "Response of endogenous glial cells to motor neuron degeneration induced by toxic ricin." <u>J Comp Neurol</u> 268(2): 248-63.
- Streit, W. J., S. L. Semple-Rowland, S. D. Hurley, R. Miller, P. G. Popovich, and B. T. Stokes (1998). "Cytokine mRNA profiles in contused spinal cord and axotomized facial nucleus suggest a beneficial role for inflammation and gliosis." <u>Exp Neurol</u> 152: 74-87.
- Streit, W. J., S. A. Walter and N. A. Pennell (1999). "Reactive microgliosis." Prog Neurobiol 57(6): 563-81.
- Strunk, T., S. Bubel, B. Mascher, P. Schlenke, H. Kirchner and K. P. Wandinger (2000). "Increased numbers of CCR5+ interferon-gamma- and tumor necrosis

factor-alpha-secreting T lymphocytes in multiple sclerosis patients." <u>Ann Neurol</u> 47(2): 269-73.

- Suzuki, K. and Y. Kikkawa (1969). "Status spongiosus of CNS and hepatic changes induced by cuprizone (biscyclohexanone oxalyldihydrazone)." <u>Am J Pathol</u> 54(2): 307-25.
- Svendsen, C. N., D. J. Clarke, A. E. Rosser and S. B. Dunnett (1996). "Survival and differentiation of rat and human epidermal growth factor-responsive precursor cells following grafting into the lesioned adult central nervous system." <u>Exp</u> <u>Neurol</u> 137(2): 376-88.
- Takami, T., M. Oudega, M. L. Bates, P. M. Wood, N. Kleitman and M. B. Bunge (2002). "Schwann cell but not olfactory ensheathing glia transplants improve hindlimb locomotor performance in the moderately contused adult rat thoracic spinal cord." J Neurosci 22(15): 6670-81.
- Takamiya, Y., S. Kohsaka, S. Toya, M. Otani and Y. Tsukada (1988).
 "Immunohistochemical studies on the proliferation of reactive astrocytes and the expression of cytoskeletal proteins following brain injury in rats." <u>Brain Res</u> 466(2): 201-10.
- Tao-Cheng, J. H., Z. Nagy and M. W. Brightman (1987). "Tight junctions of brain endothelium in vitro are enhanced by astroglia." <u>J Neurosci</u> 7(10): 3293-9.
- Taupin, V., S. Toulmond, A. Serrano, J. Benavides and F. Zavala (1993). "Increase in IL-6, IL-1 and TNF levels in rat brain following traumatic lesion. Influence of pre- and post-traumatic treatment with Ro5 4864, a peripheral-type (p site) benzodiazepine ligand." J Neuroimmunol 42(2): 177-85.
- Temple, S. and M. C. Raff (1986). "Clonal analysis of oligodendrocyte development in culture: evidence for a developmental clock that counts cell divisions." <u>Cell</u> 44(5): 773-9.
- Tetzlaff, W., M. A. Bisby and G. W. Kreutzberg (1988). "Changes in cytoskeletal proteins in the rat facial nucleus following axotomy." <u>J Neurosci</u> 8(9): 3181-9.
- Theiler, K. (1972). "The House Mouse." Berlin: Springer-Verlag.
- Timmerman, V., P. De Jonghe, C. Ceuterick, E. De Vriendt, A. Lofgren, E. Nelis, L. E. Warner, J. R. Lupski, J. J. Martin and C. Van Broeckhoven (1999). "Novel missense mutation in the early growth response 2 gene associated with Dejerine-Sottas syndrome phenotype." <u>Neurology</u> 52(9): 1827-32.
- Timmerman, V., E. Nelis, W. Van Hul, B. W. Nieuwenhuijsen, K. L. Chen, S. Wang, K. Ben Othman, B. Cullen, R. J. Leach, C. O. Hanemann and et al. (1992). "The peripheral myelin protein gene PMP-22 is contained within the Charcot-Marie-Tooth disease type 1A duplication." <u>Nat Genet</u> 1(3): 171-5.
- Timsit, S., S. Martinez, B. Allinquant, F. Peyron, L. Puelles and B. Zalc (1995).
 "Oligodendrocytes originate in a restricted zone of the embryonic ventral neural tube defined by DM-20 mRNA expression." J Neurosci 15(2): 1012-24.
- Timsit, S. G., L. Bally-Cuif, D. R. Colman and B. Zalc (1992). "DM-20 mRNA is expressed during the embryonic development of the nervous system of the mouse." <u>J Neurochem</u> 58(3): 1172-5.
- Tontsch, U., D. R. Archer, M. Dubois-Dalcq and I. D. Duncan (1994). "Transplantation of an oligodendrocyte cell line leading to extensive myelination." <u>Proc Natl</u> <u>Acad Sci U S A</u> 91(24): 11616-20.

- Topilko, P., G. Levi, G. Merlo, S. Mantero, C. Desmarquet, G. Mancardi and P. Charnay (1997). "Differential regulation of the zinc finger genes Krox-20 and Krox-24 (Egr-1) suggests antagonistic roles in Schwann cells." <u>J Neurosci Res</u> 50(5): 702-12.
- Topilko, P., S. Schneider-Maunoury, G. Levi, A. Baron-Van Evercooren, A. B. Chennoufi, T. Seitanidou, C. Babinet and P. Charnay (1994). "Krox-20 controls myelination in the peripheral nervous system." <u>Nature</u> 371(6500): 796-9.
- Tourbah, A., A. Clapin, O. Gout, B. Fontaine, R. Liblau, F. Batteux, J. L. Stievenart, B. Weill, C. Lubetzki and O. Lyon-Caen (1998). "Systemic autoimmune features and multiple sclerosis: a 5-year follow-up study." <u>Arch Neurol</u> 55(4): 517-21.
- Tourbah, A., C. Linnington, C. Bachelin, V. Avellana-Adalid, Wekerle, and A. Baron-Van Evercooren (1997). "Inflammation promotes survival and migration of the CG4 oligodendrocyte progenitors transplanted in the spinal cord of both inflammatory and demyelinated EAE rats." J Neurosci Res 50(5): 853-61.
- Trachtenberg, J. T. and W. J. Thompson (1996). "Schwann cell apoptosis at developing neuromuscular junctions is regulated by glial growth factor." <u>Nature</u> 379(6561): 174-7.
- Trapp, B. D., S. B. Andrews, C. Cootauco and R. Quarles (1989). "The myelinassociated glycoprotein is enriched in multivesicular bodies and periaxonal membranes of actively myelinating oligodendrocytes." <u>J Cell Biol</u> 109(5): 2417-26.
- Trapp, B. D., L. Bernier, S. B. Andrews and D. R. Colman (1988). "Cellular and subcellular distribution of 2',3'-cyclic nucleotide 3'-phosphodiesterase and its mRNA in the rat central nervous system." <u>J Neurochem</u> 51(3): 859-68.
- Trapp, B. D., L. Bo, S. Mork and A. Chang (1999). "Pathogenesis of tissue injury in MS lesions." <u>J Neuroimmunol</u> 98(1): 49-56.
- Trapp, B. D., A. Nishiyama, D. Cheng and W. Macklin (1997). "Differentiation and death of premyelinating oligodendrocytes in developing rodent brain." <u>J Cell</u> <u>Biol</u> 137(2): 459-68.
- Trapp, B. D., J. Peterson, R. M. Ransohoff, R. Rudick, S. Mork and L. Bo (1998).
 "Axonal transection in the lesions of multiple sclerosis." <u>N Engl J Med</u> 338(5): 278-85.
- Trapp, B. D. and R. H. Quarles (1982). "Presence of the myelin-associated glycoprotein correlates with alterations in the periodicity of peripheral myelin." <u>J Cell Biol</u> 92(3): 877-82.
- Traugott, U., C. S. Raine and D. E. McFarlin (1985). "Acute experimental allergic encephalomyelitis in the mouse: immunopathology of the developing lesion." <u>Cell Immunology</u> 91(1): 240-254.
- Traugott, U., S. H. Stone and C. S. Raine (1982). "Chronic relapsing experimental autoimmune encephalomyelitis. treatment with combinations of myelin components promotes clinical and structural recovery." J Neurol Sci 56(1): 65-73.
- Treloar, H. B., A. L. Purcell and C. A. Greer (1999). "Glomerular formation in the developing rat olfactory bulb." J Comp Neurol 413(2): 289-304.
- Tsacopoulos, M. and P. J. Magistretti (1996). "Metabolic coupling between glia and neurons." J Neurosci 16(3): 877-85.

- Tsai, H. H., E. Frost, V. To, S. Robinson, C. Ffrench-Constant, R. Geertman, R. M. Ransohoff and R. H. Miller (2002). "The chemokine receptor CXCR2 controls positioning of oligodendrocyte precursors in developing spinal cord by arresting their migration." <u>Cell</u> 110(3): 373-383.
- Tsunoda, I. and R. S. Fujinami (1996). "Two models for multiple sclerosis: experimental allergic encephalomyelitis and Theiler's murine encephalomyelitis virus." <u>J Neuropathol Exp Neurol</u> 55(6): 673-86.
- Tucker, R. P., C. C. Garner and A. Matus (1989). "In situ localization of microtubuleassociated protein mRNA in the developing and adult rat brain." <u>Neuron</u> 2(3): 1245-56.
- Turner, C. P. and J. R. Perez-Polo (1993). "Expression of p75NGFR in the olfactory system following peripheral deafferentation." <u>Neuroreport</u> 4(8): 1023-6.
- Turner, C. P. and J. R. Perez-Polo (1994). "Changes in expression of the low affinity receptor for neurotrophins, p75NGFR, in the regenerating olfactory system." <u>Int</u> <u>J Dev Neurosci</u> 12(8): 767-73.
- Ubink, R., N. Halasz, X. Zhang, A. Dagerlind and T. Hokfelt (1994). "Neuropeptide tyrosine is expressed in ensheathing cells around the olfactory nerves in the rat olfactory bulb." <u>Neuroscience</u> 60(3): 709-26.
- Ubink, R. and T. Hokfelt (2000). "Expression of neuropeptide Y in olfactory ensheathing cells during prenatal development." <u>J Comp Neurol</u> 423(1): 13-25.
- Umemori, H., Y. Kadowaki, K. Hirosawa, Y. Yoshida, K. Hironaka, H. Okano and T. Yamamoto (1999). "Stimulation of myelin basic protein gene transcription by Fyn tyrosine kinase for myelination." J Neurosci 19(4): 1393-7.
- Utzschneider, D. A., D. R. Archer, J. D. Kocsis, S. G. Waxman and I. D. Duncan (1994). "Transplantation of glial cells enhances action potential conduction of amyelinated spinal cord axons in the myelin-deficient rat." <u>Proc Natl Acad Sci U</u> <u>S A</u> 91(1): 53-7.
- Uyemura, K. (1993). "Functional glycoproteins expressed in Schwann cell membrane." <u>Neurosci Res</u> 16(1): 9-13.
- Uyemura, K., H. Asou and Y. Takeda (1995). "Structure and function of peripheral nerve myelin proteins." <u>Prog Brain Res</u> 105: 311-8.
- Valverde, F. and L. Lopez-Mascaraque (1991). "Neuroglial arrangements in the olfactory glomeruli of the hedgehog." J Comp Neurol 307(4): 658-74.
- Valverde, F., M. Santacana and M. Heredia (1992). "Formation of an olfactory glomerulus: morphological aspects of development and organization." <u>Neuroscience</u> 49(2): 255-75.
- Van den Berg, L. H., P. R. Bar, P. Sodaar, I. Mollee, J. J. Wokke and T. Logtenberg (1995). "Selective expansion and long-term culture of human Schwann cells from sural nerve biopsies." <u>Ann Neurol</u> 38(4): 674-8.
- Van Eldik, L. J., B. Christie-Pope, L. M. Bolin, E. M. Shooter and W. O. Whetsell, Jr. (1991). "Neurotrophic activity of S-100 beta in cultures of dorsal root ganglia from embryonic chick and fetal rat." <u>Brain Res</u> 542(2): 280-5.
- van Oosten, B. W., F. Barkhof, P. E. Scholten, B. M. von Blomberg, H. J. Ader and C. H. Polman (1998). "Increased production of tumor necrosis factor alpha, and not of interferon gamma, preceding disease activity in patients with multiple sclerosis." <u>Arch Neurol</u> 55(6): 793-8.

- Ved, H. S., E. Gustow and R. A. Pieringer (1989). "Effect of hydrocortisone on myelin basic protein in developing primary brain cultures." <u>Neurosci Lett</u> 99(1-2): 203-7.
- Verdu, E., G. Garcia-Alias, J. Fores, G. Gudino-Cabrera, V. C. Muneton, M. Nieto-Sampedro and X. Navarro (2001). "Effects of ensheathing cells transplanted into photochemically damaged spinal cord." <u>Neuroreport</u> 12(11): 2303-9.
- Verhaagen, J., C. A. Greer and F. L. Margolis (1990). "B-50/GAP43 Gene Expression in the Rat Olfactory System During Postnatal Development and Aging." <u>Eur J</u> <u>Neurosci</u> 2(5): 397-407.
- Verhaagen, J., A. B. Oestreicher, W. H. Gispen and F. L. Margolis (1989). "The expression of the growth associated protein B50/GAP43 in the olfactory system of neonatal and adult rats." J Neurosci 9(2): 683-91.
- Vickland, H., L. E. Westrum, J. N. Kott, S. L. Patterson and M. A. Bothwell (1991).
 "Nerve growth factor receptor expression in the young and adult rat olfactory system." <u>Brain Res</u> 565(2): 269-79.
- Viereck, C., R. P. Tucker and A. Matus (1989). "The adult rat olfactory system expresses microtubule-associated proteins found in the developing brain." J <u>Neurosci</u> 9(10): 3547-57.
- Voigt, T. (1989). "Development of glial cells in the cerebral wall of ferrets: direct tracing of their transformation from radial glia into astrocytes." J Comp Neurol 289(1): 74-88.
- Walz, W., S. Ilschner, C. Ohlemeyer, R. Banati and H. Kettenmann (1993).
 "Extracellular ATP activates a cation conductance and a K+ conductance in cultured microglial cells from mouse brain." J Neurosci 13(10): 4403-11.
- Warner, L. E., P. Mancias, I. J. Butler, C. M. McDonald, L. Keppen, K. G. Koob and J. R. Lupski (1998). "Mutations in the early growth response 2 (EGR2) gene are associated with hereditary myelinopathies." <u>Nat Genet</u> 18(4): 382-4.
- Warner, L. E., J. Svaren, J. Milbrandt and J. R. Lupski (1999). "Functional consequences of mutations in the early growth response 2 gene (EGR2) correlate with severity of human myelinopathies." <u>Hum Mol Genet</u> 8(7): 1245-51.
- Warrington, A. E., E. Barbarese and S. E. Pfeiffer (1993). "Differential myelinogenic capacity of specific developmental stages of the oligodendrocyte lineage upon transplantation into hypomyelinating hosts." J Neurosci Res 34(1): 1-13.
- Waubant, E., D. E. Goodkin, L. Gee, P. Bacchetti, R. Sloan, T. Stewart, P. B. Andersson, G. Stabler and K. Miller (1999). "Serum MMP-9 and TIMP-1 levels are related to MRI activity in relapsing multiple sclerosis." <u>Neurology</u> 53(7): 1397-401.
- Waxman, S. G., J. D. Kocsis and K. C. Nitta (1979). "Lysophosphatidyl cholineinduced focal demyelination in the rabbit corpus callosum. Light-microscopic observations." <u>J Neurol Sci</u> 44(1): 45-53.
- Weinshenker, B. G., P. C. O'Brien, T. M. Petterson, J. H. Noseworthy, C. F. Lucchinetti, D. W. Dodick, A. A. Pineda, L. N. Stevens and M. Rodriguez (1999). "A randomized trial of plasma exchange in acute central nervous system inflammatory demyelinating disease." <u>Ann Neurol</u> 46(6): 878-86.

- Weinshenker, B. G., P. Santrach, A. S. Bissonet, S. K. McDonnell, D. Schaid, S. B. Moore and M. Rodriguez (1998). "Major histocompatibility complex class II alleles and the course and outcome of MS: a population-based study." Neurology 51(3): 742-7.
- Wekerle, H. (1993). "Experimental autoimmune encephalomyelitis as a model of immune-mediated CNS disease." <u>Curr Opin Neurobiol</u> 3(5): 779-84.
- Werb, Z (1997). "ECM and cell surface proteolysis: regulating cellular ecology." <u>Cell</u> 91: 439-442.
- Werring, D. J., D. Brassat, A. G. Droogan, C. A. Clark, M. R. Symms, G. J. Barker, D. G. MacManus, A. J. Thompson and D. H. Miller (2000). "The pathogenesis of lesions and normal-appearing white matter changes in multiple sclerosis: a serial diffusion MRI study." Brain 123 (Pt 8): 1667-76.
- West, N. R., V. Leblanc and G. H. Collins (2001). "Support of axonal regrowth by endogenous mechanisms following spinal cord injury in adult rats." <u>Neuropathology</u> 21(3): 188-202.
- White, L. E., Jr. (1965). "A morphologic concept of the limbic lobe." <u>Int Rev Neurobiol</u> 8: 1-34.
- Willard, H. F. and J. R. Riordan (1985). "Assignment of the gene for myelin proteolipid protein to the X chromosome: implications for X-linked myelin disorders." <u>Science</u> 230(4728): 940-2.
- Williams, L. R., F. M. Longo, H. C. Powell, G. Lundborg and S. Varon (1983).
 "Spatial-temporal progress of peripheral nerve regeneration within a silicone chamber: parameters for a bioassay." J Comp Neurol 218(4): 460-70.
- Wilson, K. C. and G. Raisman (1981). "Estimation of numbers of vomeronasal synapses in the glomerular layer of the accessory olfactory bulb of the mouse at different ages." <u>Brain Res</u> 205(2): 245-53.
- Windrem, M. S., M. C. Nunes, W. K. Rashbaum, T. H. Schwartz, R. A. Goodman, G. McKhann, 2nd, N. S. Roy and S. A. Goldman (2004). "Fetal and adult human oligodendrocyte progenitor cell isolates myelinate the congenitally dysmyelinated brain." <u>Nat Med</u> 10(1): 93-7.
- Wingerchuk, D. M., C. F. Lucchinetti and J. H. Noseworthy (2001). "Multiple sclerosis: current pathophysiological concepts." Lab Invest 81(3): 263-81.
- Wolswijk, G. (2002). "Oligodendrocyte precursor cells in the demyelinated multiple sclerosis spinal cord." <u>Brain</u> 125(Pt 2): 338-49.
- Wood, D. D. and M. A. Moscarello (1989). "The isolation, characterization, and lipidaggregating properties of a citrulline containing myelin basic protein." J Biol Chem 264(9): 5121-7.
- Woodbury, D., E. J. Schwarz, D. J. Prockop and I. B. Black (2000). "Adult rat and human bone marrow stromal cells differentiate into neurons." <u>J Neurosci Res</u> 61(4): 364-70.
- Woodroofe, M. N., A. S. Bellamy, M. Feldmann, A. N. Davison and M. L. Cuzner (1986). "Immunocytochemical characterisation of the immune reaction in the central nervous system in multiple sclerosis. Possible role for microglia in lesion growth." <u>J Neurol Sci</u> 74(2-3): 135-52.
- Woodroofe, M. N., G. S. Sarna, M. Wadhwa, G. M. Hayes, A. J. Loughlin, A. Tinker and M. L. Cuzner (1991). "Detection of interleukin-1 and interleukin-6 in adult

rat brain, following mechanical injury, by in vivo microdialysis: evidence of a role for microglia in cytokine production." <u>J Neuroimmunol</u> 33(3): 227-36.

- Woodruff, R. H. and R. J. Franklin (1999). "Demyelination and remyelination of the caudal cerebellar peduncle of adult rats following stereotaxic injections of lysolecithin, ethidium bromide, and complement/anti-galactocerebroside: a comparative study." <u>Glia</u> 25(3): 216-28.
- Wujek, J. R., C. Bjartmar, E. Richer, R. M. Ransohoff, M. Yu, V. K. Tuohy and B. D. Trapp (2002). "Axon loss in the spinal cord determines permanent neurological disability in an animal model of multiple sclerosis." <u>J Neuropathol Exp Neurol</u> 61(1): 23-32.
- Xu, X. M., V. Guenard, N. Kleitman, P. Aebischer and M. B. Bunge (1995). "A combination of BDNF and NT-3 promotes supraspinal axonal regeneration into Schwann cell grafts in adult rat thoracic spinal cord." <u>Exp Neurol</u> 134(2): 261-72.
- Yajima, K. and K. Suzuki (1979). "Demyelination and remyelination in the rat central nervous system following ethidium bromide injection." <u>Lab Invest</u> 41(5): 385-92.
- Yamagami, S., M. Tamura, M. Hayashi, N. Endo, H. Tanabe, Y. Katsuura and K. Komoriya (1999). "Differential production of MCP-1 and cytokine-induced neutrophil chemoattractant in the ischemic brain after transient focal ischemia in rats." J Leukoc Biol 65(6): 744-9.
- Yan, Q. and E. M. Johnson, Jr. (1988). "An immunohistochemical study of the nerve growth factor receptor in developing rats." J Neurosci 8(9): 3481-98.
- Yandava, B. D., L. L. Billinghurst and E. Y. Snyder (1999). ""Global" cell replacement is feasible via neural stem cell transplantation: evidence from the dysmyelinated shiverer mouse brain." <u>Proc Natl Acad Sci U S A</u> 96(12): 7029-34.
- Yang, L. J., C. B. Zeller, N. L. Shaper, M. Kiso, A. Hasegawa, R. E. Shapiro and R. L. Schnaar (1996). "Gangliosides are neuronal ligands for myelin-associated glycoprotein." <u>Proc Natl Acad Sci U S A</u> 93(2): 814-8.
- Yong, V. W., C. A. Krekoski, P. A. Forsyth, R. Bell and D. R. Edwards (1998). "Matrix metalloproteinases and diseases of the CNS." <u>Trends Neurosci</u> 21(2): 75-80.
- Yong V. W., C. Power, P. Forsyth and D. Edwards (2001). "Metalloproteinases in biology and pathology of the nervous system." <u>Nat Reviews Neurosci</u> 2: 502-511.
- Yu, Q. and I. Stamenkovic (2000). "Cell surface-localized matrix metalloproteinase-9 proteolytically activates TGF-beta and promotes tumor invasion and angiogenesis." Genes Dev 14(2): 163-76.
- Zalc, B., M. Monge, P. Dupouey, J. J. Hauw and N. A. Baumann (1981).
 "Immunohistochemical localization of galactosyl and sulfogalactosyl ceramide in the brain of the 30-day-old mouse." <u>Brain Res</u> 211(2): 341-54.
- Zamvil, S. S. and L. Steinman (1990). "The T lymphocyte in experimental allergic encephalomyelitis." <u>Annu Rev Immunol</u> 8: 579-621.
- Zeev-Brann, A. B., O. Lazarov-Spiegler, T. Brenner and M. Schwartz (1998).
 "Differential effects of central and peripheral nerves on macrophages and microglia." <u>Glia</u> 23(3): 181-90.

- Zorick, T. S. and G. Lemke (1996). "Schwann cell differentiation." <u>Curr Opin Cell Biol</u> 8(6): 870-6.
- Zorick, T. S., D. E. Syroid, A. Brown, T. Gridley and G. Lemke (1999). "Krox-20 controls SCIP expression, cell cycle exit and susceptibility to apoptosis in developing myelinating Schwann cells." <u>Development</u> 126(7): 1397-406.